## Approaches to causality and multi-agent paradoxes in non-classical theories

 $\mathbf{B}\mathbf{Y}$ 

VILASINI VENKATESH

DOCTOR OF PHILOSOPHY

UNIVERSITY OF YORK

MATHEMATICS

October 2020

## Dedication

To my mother, Sujatha

## Abstract

Causality and logic are both fundamental to our understanding of the universe, but our intuitions about these are challenged by quantum phenomena. This thesis reports progress in the analysis of causality and multi-agent logical paradoxes in quantum and post-quantum theories. Both these research areas are highly relevant for the development of quantum technologies such as quantum cryptography and computing.

Part I of this thesis focuses on causality. Firstly, we develop techniques for using generalised entropies to analyse distinctions between classical and non-classical causal structures. We derive new properties of classical and quantum Tsallis entropies of systems that follow from the relevant causal structure, and apply these to obtain new necessary constraints for classicality in the Triangle causal structure. Supplementing the method with the post-selection technique, we provide evidence that Shannon and Tsallis entropic constraints are insufficient for detecting non-classicality in Bell scenarios with non-binary outcomes. This points to the need for better methods of characterising correlations in non-classical causal structures. Secondly, we investigate the relationships between causality and space-time by developing a framework for modelling cyclic and fine-tuned influences in non-classical theories. We derive necessary and sufficient conditions for such causal models to be compatible with a space-time structure and for ruling out operationally detectable causal loops. In particular, this provides an operational framework for analysing post-quantum theories admitting jamming non-local correlations.

In Part II of this thesis, we investigate multi-agent logical paradoxes, of which the quantum Frauchiger-Renner paradox has been the only example. We develop a framework for analysing such paradoxes in arbitrary physical theories. Applying this to box world, a post-quantum theory, we derive a stronger paradox that does not rely on post-selection. Our results reveal that reversible, unitary evolution of agents' memories is not necessary for deriving multi-agent logical paradoxes, rather that certain forms of contextuality could be.

## Contents

Al	bstra	$\mathbf{ct}$	ii
Co	onter	nts	iii
Li	st of	Tables	vi
Li	st of	Figures	vii
Ac	cknov	vledgements	ix
De	eclara	ations	x
1	Intr	oduction	1
	1.1	Preface	1
	1.2	Synopsis	3
2	Pre	liminaries	7
	2.1	Overview of notational conventions	7
	2.2	Convex geometry and polyhedral computations	9
	2.3	Information-theoretic entropy measures and their properties	21
	2.4	Generalised probabilistic theories	28

	2.5	Causal structures and causal models	34
	2.6	Modal logic	51
Ι	Aj	oproaches to causality in non-classical theories	56
3	Ove	erview of techniques for analysing causal structures	<b>58</b>
	3.1	Certifying non-classicality in probability space	58
	3.2	Certifying non-classicality in entropy space	62
4	Ent	ropic analysis of causal structures without post-selection	68
	4.1	Summary of contributions: in words and in poetry	69
	4.2	Causal constraints and Tsallis entropy vectors	70
	4.3	New Tsallis entropic inequalities for the Triangle causal structure	75
	4.4	Discussion	80
	4.5	Appendix	81
<b>5</b>	Ent	ropic analysis of causal structures with post-selection	94
	5.1	Summary of contributions: in words and in poetry	95
	5.2	Existing result for the $(2, 2, 2, 2)$ Bell scenario in entropy space $\ldots \ldots \ldots$	96
	5.3	New results for the $(2, 2, 3, 3)$ Bell scenario in probability space	97
	5.4	New results for the $(2, 2, 3, 3)$ Bell scenario in entropy space $\ldots \ldots \ldots$	99
	5.5	Discussion	107
	5.6	Appendix	108
6	Cyc prin	clic and fine-tuned causal models and compatibility with relativistic nciples	117

	6.1	Summary of contributions: in words and in poetry	118
	6.2	Motivation for analysing cyclic and fine-tuned causal models	119
	6.3	The framework, Part 1: Causal models	123
	6.4	The framework, Part 2: Space-time	136
	6.5	Necessary and sufficient conditions	138
	6.6	Theories with superluminal influence without superluminal signalling $\ldots$ .	143
	6.7	Discussion	150
	6.8	Appendix	153
Π	IIMulti-agent paradoxes166		
7	Mu	ti-agent paradoxes in quantum theory	168
	7.1	An entanglement version of the Frauchiger-Renner thought experiment	169
	7.2	Discussion	175
8	Mu	ti-agent paradoxes beyond quantum theory	180
	8.1	Summary of contributions: in words and in poetry	180
	8.2	Generalized reasoning, memories and measurements	181
	8.3	Agents, memory and measurement in box world	189
	8.4	Finding the paradox	193
	8.5	Discussion	195
	8.6	Appendix	200
9	Con	clusions and outlook	209

# List of Tables

2.1	Information-theoretic properties of classical and quantum entropies	27
4.1	Looking for violations of our Tsallis entropic inequalities	79
5.1	Vertex representation of the polytope $\Pi^{(2,2,3,3)}_{CHSH}$ .	110

# List of Figures

2.1	Projections of a polyhedral cone	13
2.2	Intuition behind the Simplex algorithm	18
2.3	Dual of a polytope	21
2.4	Individual and composite systems in box world	30
2.5	Bipartite measurements in box world	32
2.6	The bipartite Bell and Triangle causal structures	36
2.7	Example of fine-tuned correlations compatible with a causal structure	38
2.8	Example illustrating d-separation and conditional independence	39
2.9	Two node causal structures with same set of compatible distributions $\ldots \ldots$	40
2.10	Pre-intervention, augmented and post-intervention causal structures	44
3.1	The bipartite Bell causal structure and its post-selected version	65
4.1	Circuit decomposition of quantum common cause channel	85
5.1	Plot of the Tsallis entropic BC expression as a function of the Tsallis parameter $q$ for the distribution of Equation (5.4)	100
5.2	Regions in the $v$ – $\epsilon$ plane where Tsallis entropic BC inequalities can be violated.	106
6.1	Motivating examples for cyclic and fine-tuned causal models	121

LIST OF FIGURES

6.2	Space-time diagram and causal structure for a tripartite Bell experiment	123
6.3	Relationships between concepts associated with causal models	131
6.4	Some fine-tuned causal structures	133
6.5	Operationally detectable causation without correlation	133
6.6	Some fine-tuned and/or cyclic causal structure	135
6.7	Relationships between concepts associated with space-time structure	143
6.8	A protocol that can lead to superluminal signalling in the jamming scenario $\ . \ .$	145
6.9	Counter-examples to the claims of Ref. [116]	147
6.10	Compatible space-time embedding of a cyclic causal model	149
6.11	A cyclic quantum causal model	157
6.12	Possible causal structures for Theorem 6.6.1.	164
7.1	Entanglement based version of the Frauchiger-Renner thought experiment $\ . \ .$	171
8.1	Measurement and memory update in quantum theory from different perspectives	187
8.2	Outside perspective of a measurement and memory update in GPTs $\ldots \ldots \ldots$	190
8.3	Information-preserving memory update in box world	192
8.4	Circuit decomposition of the memory update map in box world	206

## Acknowledgements

First and foremost, I am grateful to my supervisor, Roger Colbeck for his gentle guidance and deep insights without which this thesis would not be possible. I have greatly benefited both from his expertise and from his encouragement to develop my own research ideas and to pursue external collaborations. His ability to express complex concepts in the most effective verbal and mathematical form have inspired me to strive for perfection in my scientific communication.

I wish to thank the Department of Mathematics for my PhD fellowship and all the timely administrative support. I extend my gratitude to other members of our department, in particular Tony Sudbery and Matt Pusey, for valuable discussions that have enriched my knowledge. I am thankful to Mirjam for several stimulating discussions on causal structures, for sharing snippets of her Mathematica code, and to her and Vicky for their help during my initial days in York. I deeply appreciate the experience I have had with all my peers, in particular Peiyun, Giorgos, Rutvij, Vincenzo, Vasilis, Max and Ali— the enjoyable conversations, moral support and spontaneous music-making sessions have made my PhD experience very memorable. I express my deep regards to my collaborators at ETH Zürich. Firstly to Lídia del Rio for the numerous exciting exchanges about quantum foundations, career and life advice, and much more. Thanks are also due to Nuriya Nurgalieva, especially for the pretty diagrams on our joint paper. I am grateful to Renato Renner, whose profound ideas and clarity of thought have always inspired me, and to him and Christopher Portmann for truly insightful interactions that have undoubtedly benefited my work. Further, I thank Stefan Wolf and Matt Pusey for taking the time to examine this long thesis.

The unwavering support of my family and friends have been vital to the completion of this thesis. My aunt, Indumathi for several enlightening discussions on abstract mathematical concepts since high school days. Meghana and Avni for being ever-prepared to endure my excited monologues about quantum physics, and being the main contributors to my knowledge of fascinating scientific facts outside of physics. Nemanja for his invaluable support with all things tech and for being a patient audience to several practice talks. Lastly but importantly, my mother, Sujatha for inspiring my academic pursuits at a young age by impressing upon me that knowledge should be sought not merely as a means to an end, but as an end in itself.

## Declarations

I declare that this thesis is a presentation of original work and I am the sole author. This work has been carried out under the supervision of Dr. Roger Colbeck and has not previously been presented for an award at this, or any other University. Chapters 4 and 5 are primarily based on publications [1] and [2] listed below which were carried out with Dr. Colbeck, while Chapter 8 is based on the publication [3] which is joint work with collaborators from ETH Zürich, Switzerland. Chapter 6 is based on yet unpublished work [4] carried out with Dr. Colbeck. Further, the Mathematica package [5] was developed for use in publications [1] and [2]. Additional research work carried out or published during the doctoral studies in related topics, but not reported in this thesis are given in [6] and [7]. All other sources are acknowledged and listed in the bibliography.

#### List of publications included in this thesis

- Vilasini, V. and Colbeck, R. Analyzing causal structures using Tsallis entropies. Physical Review A 100, 062108 (2019).
- [2] Vilasini, V. and Colbeck, R. Limitations of entropic inequalities for detecting nonclassicality in the postselected Bell causal structure. Physical Review Research 2, 033096 (2020).
- [3] Vilasini, V., Nurgalieva, N. and del Rio, L. Multi-agent paradoxes beyond quantum theory. New Journal of Physics 21, 113028 (2019).

#### In preparation and code

[4] Vilasini, V. and Colbeck, R. Cyclic and fine-tuned causal models and compatibility with relativistic principles. In preparation (2020).

[5] Colbeck, R. and Vilasini, V. LPAssumptions (A Mathematica package for solving linear programming problems with free parameters, subject to assumptions on these parameters) (2019). https://github.com/rogercolbeck/LPAssumptions.

#### Additional research work not included in this thesis

- [6] Vilasini, V., Portmann, C. and del Rio, L. *Composable security in relativistic quantum cryptography.* New Journal of Physics 21, 043057 (2019).
- [7] Vilasini, V., del Rio, L. and Renner, R. Causality in definite and indefinite space-times. In preparation (2020). Extended abstract at: https://wdi.centralesupelec.fr/users/ valiron/qplmfps/papers/qs01t3.pdf

## CHAPTER 1

### Introduction

## 1.1 Preface

UANTUM theory makes several predictions that fly in the face of common intuition and have puzzled its very founders— energy appearing only in discrete quanta, particles exhibiting wave-like properties such as interference and superposition, entanglement of distant particles, the list goes on. Nevertheless, quantum theory has passed several decades of stringent experimental tests, establishing itself as one of the most successful theories of nature currently available. Why does nature seem to comply with the predictions of quantum theory? What are the physical principles that uniquely define quantum theory? These questions continue to be actively researched, even after a century since the inception of the theory. The process has repeatedly revealed the inter-dependent and mutually reinforcing relationship between technological progress and advancements in fundamental scientific research— an understanding of fundamental principles enables us to harness their potential for technologies, which in turn allow us to probe nature at greater detail. Revolutionary technologies such as the laser, nuclear energy, positron emission tomography (i.e., PET medical imaging) and quantum computing would not exist if not for fundamental explorations into the microscopic regime, fed by human curiosity to comprehend the ways of nature. On the other hand, breakthroughs in high-precision measurements are enabling the observation of quantum effects at newer physical regimes, making it imperative to model the implications of extending quantum theory to larger and more complex systems.

In the modern era of physics, the field of quantum information theory has led to great progress in understanding the foundational aspects of the quantum regime, while also bringing to attention the increased information-processing capabilities of quantum theory. A lot of this progress traces back to a seminal theorem by John Bell in 1964 [22] proving the incompatibility of quantum predictions with certain classical models of the physical world. The remarkable fact is

that this incompatibility can in principle be witnessed at the level of empirical data such the settings of knobs and values of pointers on measurement devices, by testing the strength of correlations between the data obtained by non-communicating parties. This has made possible cryptographic protocols such as key distribution whose security is guaranteed by the laws of physics [79, 145, 17], as opposed to weighing an adversary's computing power against the difficulty of solving a mathematical problem. It has also enabled the generation and certification of true randomness [168, 60, 5], leading to the development of quantum random number generators that have now become commercially available. In the field of computing, Shor's quantum algorithm [191] enables prime factorisation in polynomial time, offering a significant speedup over existing classical algorithms, which has major implications for the security of modern day cryptosystems.<sup>1</sup> This has attracted huge investments from governments as well as companies such as Google, IBM and Microsoft, that are making steady progress towards the development of scalable quantum computers as well as quantum cryptographic primitives. Furthermore, quantum advantages in several communication tasks have also been reported [210, 40], and there are global efforts towards establishing quantum communication in space through satellite networks and also developing the quantum internet.

These technological feats are backed by theoretical research into quantum correlations, quantum networks, relativistic quantum information and quantum circuits, among others. Many of these areas fall in the ambit of the study of causality and causal structures in non-classical theories, of which there are several approaches. For example, reformulating Bell's theorem in terms of the different constraints implied by causal structures on correlations in classical and non-classical theories has revealed possible ways to generalise it to more complex scenarios involving multiple parties and sources of correlations [224, 48]. Analysing causal structures beyond the standard CHSH Bell scenario [56] are important for building quantum networks for communication and distributed computing. Furthermore, in protocols involving multiple agents distributed over space-time, relativistic principles such as the finite speed of signalling, also play a role in restricting the information processing possibilities and consequently, in the security of relativistic cryptographic protocols [59, 209]. The study of causation in quantum theory has also extended the analysis of quantum properties such as superposition and entanglement from the spatial to the temporal domain, and generalisations of quantum circuits have been introduced for modelling spatio-temporal correlations [53, 154, 171]. Such scenarios have not only been physically implemented [146, 173, 184], they have also been shown to offer additional information-theoretic advantages in numerous information-processing tasks [37, 9, 107, 51]. These examples can also be used to simulate thought experiments where the space-time structure itself exhibits quantum properties, arising from quantum gravitational effects [230]. Therefore a more complete understanding of causation and its relation to space-time structure in quantum theory is essential for harnessing its full information-processing potential and is likely to aid the long sought after unification of quantum theory with general relativity.

The ability to create and manipulate quantum superpositions of larger composite systems is an important consequence of the advancements in quantum networks and computing technologies. While we don't observe quantum effects at macroscopic scales, the possibility of such an observation under careful experimental conditions is not forbidden by any known physical principle.

<sup>&</sup>lt;sup>1</sup>The RSA encryption scheme is based on the complexity of this problem.

Several theoretical and experimental research groups around the globe have been invested in probing the quantum to classical transition, whether there is a fundamental scale at which such a transition should occur remains a burning open question. Recent work has revealed that extending quantum theory to more complex systems can lead to a conflict with simple and commonly employed principles of logical reasoning [85, 152]. Continued research into this domain would hence enable a better characterisation of the structure of logic that apply to larger scale quantum systems, such as quantum computers. It would also generate insights into whether the principles of quantum theory are universally applicable, or whether they are merely a very good approximation for the microscopic realm offered by a more fundamental theory that we are yet to discover.

This thesis concerns itself with the two areas of quantum information research enunciated above, namely the study of causality and of multi-agent logical paradoxes that can arise when quantum theory is extended to more complex systems. Our analysis of these topics also extends to post-quantum theories, since it is often the case that stepping beyond quantum theory to a more general setting informs an understanding of what makes quantum theory special among the plethora of other physical theories. We summarise the main contributions of this thesis below, and its relevance for the broader open questions described here.

### 1.2 Synopsis

This thesis divided into two parts and consists of a total of nine chapters. The first part focuses on causality while the second on multi-agent paradoxes in non-classical theories. Chapters 4, 5, 6 and 8 contain the majority of the original contributions of this thesis which are based on published [206, 207, 208] as well as unpublished work<sup>2</sup>. Leaving out the present chapter, of which the current section is the last, we summarise the main contents and/or contributions of the remaining eight chapters below.

**Chapter 2: Preliminaries** Here we introduce the background concepts and mathematical tools that will be important for the rest of this thesis. An expert reader may choose to skip over parts or the whole of this chapter. We first cover the basics of polyhedral geometry and computation, followed by an overview of information-theoretic entropic measures and their properties. Both these topics are central to the results of Chapters 4 and 5. We then discuss a class of post-quantum theories, namely generalised probabilistic theories, before moving on to the preliminary concepts about causal structures in classical, quantum and generalised probabilistic theories. The former will be relevant for both parts of the thesis while the latter will predominantly be employed in Part I. We conclude this chapter with a review of epistemic modal logic, a concept that will feature in Part II of the thesis.

<sup>&</sup>lt;sup>2</sup>Vilasini, V. and Colbeck, R. *Cyclic and fine-tuned causal models and compatibility with relativistic principles.* In preparation (2020).

#### Part I: Approaches to causality in non-classical theories

Causality is a fundamental part of our perception of the universe. A precise understanding of cause-effect relationships is integral to the scientific method, having found applications in drug trials, economic predictions, machine learning as well as biological and physical systems. Quantum mechanics, a theory that has been immensely successful in explaining microscopic physics, strongly challenges our classical intuitions about causation, deeming classical models inadequate for describing quantum phenomena. Apart from this fundamental implication, a plethora of information-theoretic applications arise from the study of causal structures, that benefit from a quantum over classical advantage. Extending this study to more general, postquantum theories facilitates this understanding by highlighting the aspects of causality that might be special to quantum theory. Part I of this thesis is dedicated to the analysis of causal structures, both acyclic and cyclic, and using both entropic and non-entropic techniques.

**Chapter 3: Overview of techniques for analysing causal structures** Here, we provide a more specific overview of the techniques for certifying the non-classicality of causal structures, that will be employed in Chapters 4 and 5. We first discuss the techniques employed when the observed correlations in a causal structure are represented in probability space, focusing on the geometry of the correlation sets in the bipartite Bell causal structure. Following this, we present an overview of methods to certify non-classicality in entropy space. This includes a summary of the entropy vector method and the post-selection technique which is often employed with this method. This chapter is based on the introductory sections of our published papers [206, 207].

**Chapter 4: Entropic analysis of causal structures without post-selection** In this chapter, we develop new techniques for using generalised entropy measures to analyse causal structures, particularly Tsallis entropies. Employing the Shannon entropy for this purpose is known to have certain limitations [214, 216] and the use of generalised entropies have not been previously considered. Our first contribution is a new set of constraints on the Tsallis entropies that are implied by (conditional) independence between classical random variables, which we apply to causal structures.<sup>3</sup> Finding that the standard entropic method becomes computationally intractable even for small causal structures, we propose a way to circumvent this issue and obtain a partial solution, namely a new set of Tsallis entropies have found important applications in other areas of information theory and in the field of non-extensive statistical physics, hence the properties of Tsallis entropies derived here can be useful even beyond the analysis of causal structures. Further, we also find that Rényi entropies pose significant limitations for certifying the non-classicality of causal structures. This chapter is based on our published work [206].

<sup>&</sup>lt;sup>3</sup>Additionally, we also generalise these constraints to quantum Tsallis entropies in certain cases, under a suitable notion of quantum conditional independence. But this is not central to the remaining results.

Chapter 5: Entropic analysis of causal structures with post-selection We investigate whether the entropic method combined with the post-selection technique allows non-classicality to be detected whenever it is present. We focus on the bipartite Bell causal structure, where the answer is known to be positive for the case of binary valued measurement outcomes [46]. Developing an analogue of the technique used in [46] to the scenario where there are three outcomes and two inputs per party, we identify two families of non-classical distributions here, namely those whose non-classicality can/cannot be detected through this technique. We then investigate extensions of the technique by allowing the observed correlations to be post-processed according to a natural class of non-classicality non-generating operations prior to testing them entropically. We provide numerical as well as analytical evidence that indicate that even under a natural class of post-processings, entropic inequalities for either the Shannon or Tsallis entropies are generally not sufficient to detect non-classicality in the Bell causal structure. Our work provides insights into some of the advantages and the limitations of the entropic technique, and is the first to report drawbacks of the technique in post-selected causal structures. This chapter is based almost entirely on published material [207].

Chapter 6: Cyclic and fine-tuned causal models and compatibility with relativistic **principles** In this chapter, we investigate the relationships between causation and space-time structure. We develop an operational framework for modelling cyclic and fine-tuned causal influences in non-classical theories, and characterise their compatibility with an underlying spacetime structure. Distinguishing between causal loops based on the ability to operationally detect them, we derive necessary and sufficient conditions for ruling out operationally detectable loops in these causal models, and for the model to not signal superluminally with respect to a spacetime structure. This provides a mathematical framework for analysing generalisations of the so-called "jamming" scenario [105, 116] whereby a party superluminally influences correlations between other space-like separated parties. We prove, through an explicit protocol that these scenarios can lead to superluminal signalling (without additional assumptions), contrary to the original claim that they are compatible with relativistic principles. Furthermore, we analyse a claim made in [116] that a set of no-signaling conditions are necessary and sufficient for ruling out causal loops in multipartite Bell scenarios. Our results also identify missing assumptions in these claims and generalise them to arbitrary causal structures. Finally, we present ideas for building a more complete framework for causally modelling these general scenarios, which is likely to be of broader relevance. This work is yet unpublished.

#### Part II: Multi-agent paradoxes in non-classical theories

Logical deduction is another rudimentary aspect of our understanding of the world around us, which we begin to employ since early childhood. In recent years, it has been shown that quantum theory, when extended to the level of the observer, challenges some of the commonly employed rules of logical reasoning. More precisely, when observers model each other's memories (where they store their measurement outcomes) as quantum systems and use simple logical principles to reason about each other's knowledge, they can arrive at a deterministic contradiction [85, 152]. This multi-agent logical paradox proposed by Frauchiger and Renner has generated a surge of

scientific interest in understanding the role of observers as quantum systems, and the objectivity of their physical observations. An observer here need not necessarily be a conscious entity, but one capable of implementing quantum measurements and performing simple deductions based on the outcomes— something that a relatively small quantum computer can be programmed to do (in principle). Experimental efforts towards the developing scalable quantum computers could make possible a physical implementation of such scenarios in the near future, making it imperative to develop a better theoretical understanding thereof. Part II of this thesis focuses on multi-agent paradoxes, in and beyond quantum theory, for we must step outside the quantum formalism to better comprehend its peculiarities.

**Chapter 7: Multi-agent paradoxes in quantum theory** In this chapter, we present an entanglement-based version of the Frauchiger-Renner thought experiment [85] and explain its seemingly paradoxical result. We briefly analyse the relationship of this to other results with similar implications for the objectivity of agents' observations. We conclude with a discussion about the implications of this result for the interpretations of quantum theory.

**Chapter 8: Multi-agent paradoxes beyond quantum theory** We generalize assumptions involved of the Frauchiger-Renner result so that they can be applied to arbitrary physical theories. This provides an operational framework for modelling agents as systems of a general physical theory, which we apply for modelling how observers' memories may evolve in box world, a particular post-quantum, probabilistic theory. We use this to find a deterministic contradiction in the case where agents share a PR box, a bipartite box world system. Our version of the paradox in box world is stronger than the quantum one of Frauchiger and Renner, in the sense that it does not rely on post-selection. It also reveals that reversibility of the memory update, akin to quantum unitarity is not necessary for witnessing such paradoxes, suggesting that certain forms of contextuality are to be held responsible. Obtaining an inconsistency in the framework of generalised probabilistic theories broadens the landscape of theories which are affected by the application of classical rules of reasoning to physical agents, and enables a deeper understanding of the features of quantum theory that lead to an incompatibility with classical logical structures. This chapter is based on our published paper [208].

**Chapter 9: Conclusions and oulook** We present concluding remarks on the contributions made in both parts of this thesis, and their potential scope for addressing immediate as well as broader open problems in the foundations of physics and information theory.

## CHAPTER 2

### Preliminaries

In this chapter, we review the necessary background and main mathematical tools that will be employed in this thesis. We begin with an overview of convex geometry, optimization and polyhedral computations in Section 2.2. In Section 2.3, we present the information-theoretic entropy measures that are relevant to this thesis, namely Shannon, von-Neumann, Tsallis and Rényi entropies, and outline some of their useful properties. We then review the framework of generalised probabilistic theories (GPTs) in Section 2.4, mainly following the work of Barrett [16]. Section 2.5 summarises the basics of classical and non-classical causal models which will be primarily based on the classical Bayesian networks approach of Judea Pearl [159] and the framework of generalised causal structures developed by Henson, Lal and Pusey [115] that apply to quantum and GPT causal structures. These concepts form the backbone of Part I of this thesis which concerns causality in classical and non-classical theories. In Section 2.6 we summarise the main structures and axioms underlying the modal logic framework. This, along with the framework for GPTs mentioned above form the main prerequisites for Part II of this thesis, which concerns multi-agent logical paradoxes in quantum and GPT settings. We first present a brief overview of notational conventions used in this thesis.

### 2.1 Overview of notational conventions

Throughout this thesis (unless specified otherwise), we will employ the following notational conventions.

**Random variables, probabilities, entropies:** We will use capital letters (typically from the second half of the English alphabets) to denote random variables or RVs for short e.g., X, Y, Z. We will also use the same label to denote the set of possible values taken by the random

variable, the corresponding small letter to denote a particular value of the random variable and |.| to denote cardinality of sets. For example, the random variable X takes values  $x \in X$ , and there are |X| possible values that X can take. We will only consider finite and discrete random variables throughout this thesis. For a probability distribution over a set of random variables  $\{X_1, ..., X_n\}$ , we will use  $P_{X_1,...,X_n} \in \mathscr{P}_n$ , where  $\mathscr{P}_n$  denotes the set of all probability distributions over n discrete random variables. Whenever it is necessary to mention the specific values of the random variables, we will abbreviate  $P_{X_1,...,X_n}(X_1 = x_1, ..., X_n = x_n)$  to  $P(X_1 = x_1, ..., X_n = x_n)$ or simply  $P(x_1, ..., x_n)$ . When an equation contains only distributions labelled by variables denoted in capital letters, it should be interpreted as being satisfied for every value of those variables (denoted by the corresponding small letter). For example  $P_{XY} = P_X P_Y$  should be read as  $P(xy) = P(x)P(y) \ \forall x \in X, y \in Y$ , where P(xy) is short for  $P_{XY}(X = x, Y = y)$ . We will use the short form XY or sometimes X, Y to denote the union of two variables/sets of variables. Further, we will use H(XY...) (or sometimes, H(X, Y, ...)) to denote joint entropies of sets of random variables X, Y, ....

Quantum formalism: Given a Hilbert space  $\mathscr{H}$ , we will use  $\mathscr{P}(\mathscr{H})$  to represent the set of positive, semi-definite operators on  $\mathscr{H}$ , and  $\mathscr{S}(\mathscr{H})$  to denote the set of positive semidefinite and trace one operators on  $\mathscr{H}$ . Quantum states will, in general be represented by density operators, i.e.,  $\rho \in \mathscr{S}(\mathscr{H})$ . Hermitian conjugate will be denoted using the usual dagger notation,  $\rho^{\dagger}$  and matrix transpose by the subscript T,  $\rho^{T}$ . Rank 1 density operators i.e., pure states will be denoted as elements of the Hilbert space  $\mathscr{H}$ , in the conventional bra-ket notation  $|\psi\rangle \in \mathscr{H}$  and  $|\psi\rangle^{\dagger} = \langle\psi| \in \mathscr{H}^{*}$ , where  $\mathscr{H}^{*}$  is the dual space of  $\mathscr{H}$ . Quantum channels or transformations on quantum states correspond to *completely positive trace-preserving* (CPTP) maps  $\mathscr{E} : \mathscr{S}(\mathscr{H}_A) \mapsto \mathscr{S}(\mathscr{H}_B)$  that map an input state  $\rho_A \in \mathscr{S}(\mathscr{H}_A)$  to an output state  $\rho_B = \mathscr{E}(\rho_A) \in \mathscr{S}(\mathscr{H}_B)$ . Quantum measurements will be described by *positive operator valued measures* (POVMs): a POVM is a set  $\{E^x\}_{x \in X}$  with  $E^x \in \mathscr{P}(\mathscr{H})$ , labelled by the classical values x, and summing to identity  $\sum_{x \in X} E^x = 1$ . We will often abbreviate  $\{E^x\}_{x \in X}$  to  $\{E^X\}_X$ . For the purposes of this thesis, we will only consider a discrete and finite set of measurement outcomes, so X corresponds to a discrete and finite random variable taking values  $x \in X$ .

**Causal structures and space-time diagrams:** Everywhere in the thesis, except for Chapter 6, we will use the regular arrows  $\rightarrow$  to denote causal influence (from cause to effect). In Chapter 6, we will use  $\rightsquigarrow$  for the same purpose because we would like to categorise these into solid arrows  $\rightarrow$  and dashed arrows  $-\rightarrow$  based on how these causal influences are detected operationally. A vast majority of the causal structures appearing in the rest of the thesis correspond to the  $\rightarrow$  case, which justifies this notation. All space-time diagrams will have space along the x-axis and time along the y-axis.

**Logical operators and others:** We will use the standard logical operators i.e.,  $\neg$  for 'not',  $\land$  for 'and',  $\lor$  for 'or',  $\Rightarrow$  for 'if...then', and  $\Leftrightarrow$  for 'equivalent.' Further,  $\oplus$  will be used to denote modulo-2 addition of classical bits i.e., the bitwise OR. Logarithms will use base *e* i.e., we will consider natural logarithms and denote it by ln.

## 2.2 Convex geometry and polyhedral computations

The lecture notes [92] provide a thorough introduction to polyhedral computation, while [31, 104, 142] are good resources for convex geometry and their application in optimisation problems. In this section, we give an overview of the main aspects of these topics that will be relevant to the current thesis and will focus our attention on the vector spaces  $V = \mathbb{R}^n$  over the ordered field of reals  $\mathbb{R}$ .

#### 2.2.1 The geometry of convex polyhedra

The following special types of linear combinations are central to the concepts of polyhedral theory and convex optimization.

**Definition 2.2.1** (Affine, conic and convex combinations). Consider a set of points  $x_1, ..., x_m \in V$  and their linear combination  $\sum_{i=1}^{m} \alpha_i x_i$  with real coefficients  $\alpha_i$ . Then,

- 1.  $\sum_{i=1}^{m} \alpha_i x_i$  is an affine combination if  $\sum_{i=1}^{m} \alpha_i = 1$ .
- 2.  $\sum_{i=1}^{m} \alpha_i x_i$  is a conic combination if  $\alpha_i \ge 0 \ \forall i$ .
- 3.  $\sum_{i=1}^{m} \alpha_i x_i$  is a *convex combination* if it is both affine and conic i.e.,  $\sum_{i=1}^{m} \alpha_i = 1$  and  $\alpha_i \ge 0$  $\forall i$ .

The affine, conic and convex hull of a set of points are then naturally defined with respect to the corresponding type of linear combination.

**Definition 2.2.2** (Affine, conic and convex hull). The affine hull of a set of points  $S \subseteq V$ , is the set of all affine combinations of points in S and is denoted as Aff(S). Similarly the conic and convex hull of S correspond to the sets of all conic and convex combinations of S and are denoted by Cone(S) and Conv(S) respectively.

Analogous to the concept of linear independence, a set of points S is said to be *affinely inde*pendent if and only if none of the points in S can be expressed as an affine combination of the remaining points in S. A set  $S \subseteq V$  that is closed under affine combinations is called an *affine* subspace of  $V = \mathbb{R}^n$ .

We now proceed to discuss the central objects of polyhedral theory namely, convex sets, convex cones, polyhedra and polytopes, based on the concepts defined above.

**Definition 2.2.3** (Convex set). A subset  $S \subseteq V$  is said to be *convex* if for any two points  $x_1, x_2 \in S$ , the line segment  $[x_1, x_2] := \{x : x_1 + (1 - \alpha)x_2, 0 \le \alpha \le 1\}$  is contained in S.

By applying Definition 2.2.3 inductively to the set S, it can be shown that a set is convex if and only if it contains every convex combination of its points. Convex sets of particular interest to us are *convex cones*, and a further subset of those, *polyhedral cones*.

**Definition 2.2.4** (Convex cone). A set  $C \subseteq V$  is called a *cone* if for every  $x \in C$ , and  $\alpha \ge 0$ ,  $\alpha x \in C$ . C is a *convex cone* if it is convex and a cone i.e., for any  $x_1, x_2 \in C$ , and  $\alpha_1, \alpha_2 \ge 0$ ,

$$\alpha_1 x_1 + \alpha_2 x_2 \in \mathcal{C} \tag{2.1}$$

Note that a set is a convex cone if and only if it contains every conic combination of its points.

**Definition 2.2.5** (Polyhedral cone). A set  $\mathcal{P} \subseteq V$  is said to be a *polyhedral cone* if it is a *polyhedron* and a *cone*, where a polyhedron  $\mathcal{P}$  is a subset of V that can be expressed as the solution set of a finite number of linear inequalities,<sup>1</sup>

$$\mathcal{P} = \{ x \in V : Ax \le b \text{ and } Cx = d \}, \quad A \in \mathbb{R}^{m \times n}, b \in \mathbb{R}^m, C \in \mathbb{R}^{k \times n}, d \in \mathbb{R}^k.$$
(2.2)

By construction, a polyhedron corresponds to an intersection of half-spaces  $\{x \in V | Ax \leq b\}$  and hyperplanes  $\{x \in V | Cx = d\}$ , and is convex. This is known as the  $\mathcal{H}$ -representation (where  $\mathcal{H}$ stands for half-space) of a polyhedron, and bounded polyhedra are called *polytopes*.<sup>2</sup> Since a cone, by definition contains the origin cone,  $\mathbf{0} \in V = \mathbb{R}^n$ , polyhedral cones can be expressed in the more concise form  $\mathcal{P} = \{x \in V : Ax \leq 0\}$ . Polyhedra (and hence polytopes) can also be represented through a  $\mathcal{V}$ -representation (where  $\mathcal{V}$  stands for vertex)<sup>3</sup> which follows from an important result in polyhedral theory, the Minkowski-Weyl Theorem.

**Theorem 2.2.1** (Minkowski-Weyl Theorem). For  $\mathcal{P} \subseteq V$ , the following statements are equivalent

- 1.  $\mathcal{P}$  is a polyhedron,
- 2.  $\mathcal{P}$  is finitely generated, i.e., there exist finite sets  $S, T \subset V$  such that

$$\mathcal{P} = \operatorname{Conv}(S) + \operatorname{Cone}(T), \tag{2.3}$$

where addition of sets is defined with respect to the Minkowski sum, i.e.,  $S_1+S_2 \coloneqq \{s_1+s_2 \colon s_1 \in S_1, s_2 \in S_2\}$  for  $S_1, S_2 \subseteq V$ .

The second statement defines the  $\mathcal{V}$ -representation of  $\mathcal{P}$ . In the case of a polyhedral cone given by  $\mathcal{P} = \{x \in V : Ax \leq 0\}$  in the  $\mathcal{H}$ -representation,  $S \subset V$  can be taken to be the empty set in the

<sup>&</sup>lt;sup>1</sup>Note that the equalities Cx = d in Equation (2.2) can be equivalently written in terms of inequalities as  $Cx \ge d$  and  $Cx \le d$ , but we will often write these separately (as is common in the convex optimization literature).

 $<sup>^{2}</sup>$ We will adopt these definitions in the rest of thesis, noting that in the literature, the opposite convention for defining polyhedra and polytopes is sometimes used.

<sup>&</sup>lt;sup>3</sup>Strictly speaking, unbounded polyhedra have extremal rays rather than extremal points or vertices (defined later in the section). Nevertheless, the representations in terms of extremal rays (in the unbounded case) as well as in terms of vertices (bounded case) are known as the  $\mathcal{V}$ -representation.

corresponding  $\mathcal{V}$ -representation. On the other hand, a polytope (in its  $\mathcal{V}$ -representation) can be written as the convex hull of a finite set of points, and T can be taken to be the empty set in this case. A *simplex* is a polytope that can be written as the convex hull of a finite set of *affinely independent* points.

The geometry of a polyhedron  $\mathcal{P} \subseteq V$  is characterised by its *faces* which are defined in terms of valid linear inequalities of  $\mathcal{P}$ . A linear inequality is valid for  $\mathcal{P}$  if it holds for all  $x \in \mathcal{P}$ .

**Definition 2.2.6** (Faces of a polyhedron). For a polyhedron  $\mathcal{P} \subseteq V$ , a subset  $\mathcal{F} \subseteq \mathcal{P}$  is called a is called a face of  $\mathcal{P}$  if it is represented as

$$\mathcal{F} = \mathcal{P} \cap \{ x : c^T x = d \}, \tag{2.4}$$

for some valid inequality  $c^T x \leq d$ .

All faces of a polyhedron are by construction, polyhedrons themselves. Faces of dimensions 0, 1 and  $dim(\mathcal{P})-1$  are called *vertices*, *edges* and *facets* respectively. It can be shown that vertices of a polyhedron are equivalent to its *extreme points* which are points in  $\mathcal{P}$  that cannot be written as a convex combination of other points in  $\mathcal{P}$ . Hence will use these terms interchangeably. Further, when an edge of a polyhedron is unbounded, it can either be a line (unbounded in both directions or half-line (starting from a vertex and unbounded in one direction). In the latter case, the edge is called an *extremal ray.*<sup>4</sup> In the  $\mathcal{V}$ -representation of a polyhedron given by statement 2 of Theorem 2.2.1, it is often convenient to take the sets S and T to correspond to the set of *extremal points* and *extremal rays* respectively.

#### 2.2.2 Projections of polyhedra: Fourier-Motzkin Elimination

Fourier-Motzkin Elimination is a method for projecting higher dimensional polyhedra to lower dimensional ones through variable elimination, and forms an important part of the computational methods employed in this thesis. Here, we describe the mathematical concepts underpinning this algorithm. For this, we start with linear transformations of which the transformations of interest, projections are a subset.

**Definition 2.2.7** (Linear transformation). A *linear transformation* is a map  $f : \mathbb{R}^n \to \mathbb{R}^m$  that acts as  $f : x \mapsto Ax$ , where  $A \in \mathbb{R}^{m \times n}$ .

It can be immediately shown that linear transformations preserve convexity (since convex combinations are particular cases of linear combinations). Additionally, linear transformation also map convex cones to convex cones and polyhedral cones to polyhedral cones. Projections  $\pi$  are linear transformations that are idempotent i.e.,  $\pi.\pi = \pi$ . In particular, we will consider orthogonal projections which are projections on a Hilbert space that satisfy  $\langle \pi(u), v \rangle = \langle u, \pi(v) \rangle$  for

<sup>&</sup>lt;sup>4</sup>Extremal rays can also be defined as a subset R of  $\mathcal{P}$  that cannot be expressed as a (non-trivial) conic combination of points not belonging to R.

the Hilbert space inner-product  $\langle, \rangle$  between any vectors u and v in the space. In our case this Hilbert space is simply  $\mathbb{R}^n$  and  $\langle, \rangle$  is the corresponding dot product, and in the remainder of the thesis, projections must be taken to mean orthogonal projections.

For a polyhedron  $\mathcal{P} \subseteq V$  expressed as  $\mathcal{P} = \{(x, y) \in \mathbb{R}^{n_1} \times \mathbb{R}^{n_2} : Ax + By \leq b\}$ , with  $n_1 + n_2 = n$ , the projection of  $\mathcal{P}$  into the subspace of the y variables  $\pi_y(\mathcal{P}) : V \mapsto V$  is defined as

$$\pi_y(\mathcal{P}) \coloneqq \{ (\mathbf{0}, y) : y \in \mathbb{R}^{n_2}, \text{ and } \exists x \in \mathbb{R}^{n_1} \text{ such that } (x, y) \in \mathcal{P} \}.$$

$$(2.5)$$

 $\pi_y$  is a linear transformation that can be represented by a  $n \times n$  matrix with an  $n_2 \times n_2$  identity matrix as its second block, and zeroes everywhere else.<sup>5</sup> In the  $\mathcal{V}$  representation, a projection  $\pi$  transforms the convex/conic hull of a set of points  $\{x_1, ..., x_m\}$  to the convex/conic hull of the points  $\{\pi(x_1), ..., \pi(x_m)\}$ . Given a polyhedron  $\mathcal{P}$  in the  $\mathcal{H}$ -representation  $\{x \in V : Ax \leq b\}$ as input, the Fourier-Motzkin Elimination (FME) procedure outputs the  $\mathcal{H}$ -representation, or the inequalities describing the projected polyhedron  $\pi_y(\mathcal{P})$ . The projection  $\pi_y$  is implemented through FME by eliminating the first  $n_1$  variables in the system of inequalities defining  $\mathcal{P}$ . Taking x to represent the n-dimensional vector with components  $x_i$ , the following steps detail the Fourier-Motzkin Elimination procedure for eliminating the variable  $x_1$ . This procedure can then be iterated  $n_1$  times to eliminate all required variables.

- 1. The inequalities Ax = b are partitioned into three sets  $I_+$ ,  $I_-$  and  $I_0$ .  $I_+$  denotes the set of all inequalities in  $Ax \leq b$  where the variable  $x_1$  has a strictly positive coefficient, i.e., all inequalities where  $A_{i1} > 0$ . Similarly,  $I_-$  is the set of all inequalities where  $x_1$  has a strictly negative coefficient, and  $I_0$  is the set of all inequalities where  $x_1$  does not appear i.e., its coefficient is 0.
- 2. If  $I_+$  is empty, then the inequalities in  $I_-$  are ignored, and vice-versa. If both  $I_+$  and  $I_-$  are non-empty, then the following steps are undertaken. Every inequality  $A_i x \leq b_i$  in  $I_+$  is rearranged and expressed as

$$x_1 \le \frac{1}{A_{i1}} (b_i - \sum_{k \ne 1} A_{ik} x_k) \coloneqq f_i(x_2, \dots, x_n).$$
(2.6)

There are  $|I_+|$  such inequalities. Similarly, every inequality  $A_j x \leq b_j$  in  $I_-$  is expressed as (noting that  $A_{j1} < 0$  in this case)

$$x_1 \ge \frac{1}{A_{j1}} (b_j - \sum_{l \ne 1} A_{jl} x_l) \coloneqq g_j(x_2, \dots, x_n),$$
(2.7)

and there are  $|I_-|$  inequalities of this type. Equations (2.6) and (2.7) are then combined to give the following set of  $|I_+|.|I_-|$  inequalities that are independent of the variable  $x_1$ ,

$$g_j(x_2,...,x_n) \le f_i(x_2,...,x_n).$$
 (2.8)

<sup>&</sup>lt;sup>5</sup>Here, the projection  $\pi$  is defined as an endomorphism on V i.e., a map from V to itself. However, it can be equivalently seen as a map from V to a lower dimensional space by dropping additional zeroes, i.e., replacing  $(\mathbf{0}, y) \in \mathbb{R}^n$  with  $y \in \mathbb{R}^{n_2}$  in the Equation (2.5).



Figure 2.1: Projections of a polyhedral cone (Example 2.2.1): (a) illustrates a polyhedral cone ( $\mathcal{P}$ ), with 3 faces ( $\mathcal{H}$ -representation) and 3 extremal rays ( $\mathcal{V}$ -representation) as indicated. Projections of  $\mathcal{P}$  on to the xy, yz and xz-planes are also polehedra, as shown in (b), (c) and (d) respectively. In all four subfigures, the corresponding polyhedron (blue region) is unbounded, and extends to infinity along the direction of the extremal rays indicated.

3. The union of the  $|I_+|.|I_-|$  inequalities of Equation (2.8) with those of  $I_0$  characterise the final output set of inequalities corresponding to the projection into the space of the n-1 variables  $x_2, ..., x_n$ .

Note that the final set of inequalities obtained in the FME procedure is often not minimal, in the sense that it can contain several redundancies. This can however be checked efficiently through a linear program, and a characterisation of the projected polyhedron in terms of fewer, non-redundant inequalities can be obtained (as described in Section 2.2.3.1 and Figure 2.2).

**Example 2.2.1.** Consider the polyhedral cone  $\mathcal{P} \in V$  of Figure 2.1a which, in the  $\mathcal{H}$ -representation

is given by

$$\mathcal{P} = \left\{ \begin{pmatrix} x \\ y \\ z \end{pmatrix} \in \mathbb{R}^3 \middle| \begin{pmatrix} -2 & 1 & 1 \\ 0 & -3 & 1 \\ 0 & 0 & -1 \end{pmatrix} \cdot \begin{pmatrix} x \\ y \\ z \end{pmatrix} \le \begin{pmatrix} 0 \\ 0 \\ 0 \end{pmatrix} \right\}$$
(2.9)

The same polyhedron in expressed in the  $\mathcal{V}$ -representation as

$$\mathcal{P} = \operatorname{Cone}\left(\begin{pmatrix}1\\0\\0\end{pmatrix}, \begin{pmatrix}1\\2\\0\end{pmatrix}, \begin{pmatrix}2\\1\\3\end{pmatrix}\right)$$
(2.10)

The projection matrices in the 3 coordinate planes are,

$$\pi_{xy} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad \pi_{yz} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{pmatrix}, \quad \pi_{xz} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 1 \end{pmatrix}, \quad (2.11)$$

Then the projections of  $\mathcal{P}$  onto each of the coordinate planes correspond to the polyhedra  $\mathcal{P}_{xy}$ (Figure 2.1b),  $\mathcal{P}_{yz}$  (Figure 2.1c) and  $\mathcal{P}_{xz}$  (Figure 2.1d) expressed in the  $\mathcal{V}$ -representation as

$$\mathcal{P}_{xy} = \operatorname{Cone}\left(\begin{pmatrix}1\\0\\0\end{pmatrix}, \begin{pmatrix}2\\1\\0\end{pmatrix}, \begin{pmatrix}2\\1\\0\end{pmatrix}\right), \quad \mathcal{P}_{yz} = \operatorname{Cone}\left(\begin{pmatrix}0\\0\\0\end{pmatrix}, \begin{pmatrix}0\\2\\0\end{pmatrix}, \begin{pmatrix}0\\1\\3\end{pmatrix}\right) \quad \mathcal{P}_{xz} = \operatorname{Cone}\left(\begin{pmatrix}1\\0\\0\end{pmatrix}, \begin{pmatrix}1\\0\\0\end{pmatrix}, \begin{pmatrix}2\\0\\3\end{pmatrix}\right)$$
(2.12)

The same projections can be obtained by performing Fourier-Motzkin Elimination on the  $\mathcal{H}$ -representation of  $\mathcal{P}$  (Equation (2.9)) as shown below.

1. For the projections  $\pi_{xy}$ ,  $\pi_{yz}$ ,  $\pi_{xz}$ , we wish to eliminate the variables z, x and y respectively. Denoting the variable to be eliminated by a superscript, we have

$$\begin{split} I^z_+ &= \{-2x + y + z \leq 0, -3y + z \leq 0\}, & I^z_- = \{-z \leq 0\}, & I^z_0 = \varnothing \\ I^x_+ &= \varnothing, & I^x_- = \{-2x + y + z \leq 0\}, & I^x_0 = \{-3y + z \leq 0, -z \leq 0\} \\ I^y_+ &= \{-2x + y + z \leq 0\}, & I^y_- = \{-3y + z \leq 0\}, & I^y_0 = \{-z \leq 0\} \end{split}$$

- 2. Since  $I_{+}^{x}$  is empty, we can ignore  $I_{-}^{x}$  and only need  $I_{0}^{x}$  to describe  $\mathcal{P}_{yz}$ .  $I_{+}^{y}$ ,  $I_{-}^{y}$ ,  $I_{+}^{z}$  and  $I_{-}^{z}$  are all non-empty, so we rearrange the corresponding inequalities as  $I_{+}^{z} = \{z \leq 2x y, z \leq 3y\}$ ,  $I_{-}^{z} = \{z \geq 0\}$  and  $I_{+}^{y} = \{y \leq 2x z\}$ ,  $I_{-}^{y} = \{y \geq \frac{z}{3}\}$ . Combining the inequalities in each case we have  $2x y \geq 0, 3y \geq 0$  for the projection  $\pi_{xy}$ ,  $3y z \geq 0$ ,  $z \geq 0$  for the projection  $\pi_{yz}$  and  $2x 4\frac{z}{3} \geq 0$ ,  $z \geq 0$  for the projection  $\pi_{xz}$ .
- 3. We obtain the  $\mathcal{H}$ -representation of the projected polytopes as

$$\mathcal{P}_{xy} = \left\{ \begin{pmatrix} x \\ y \end{pmatrix} \in \mathbb{R}^2 \middle| \begin{pmatrix} -2 & 1 \\ 0 & -3 \end{pmatrix} \cdot \begin{pmatrix} x \\ y \end{pmatrix} \le \begin{pmatrix} 0 \\ 0 \end{pmatrix} \right\}$$
$$\mathcal{P}_{yz} = \left\{ \begin{pmatrix} y \\ z \end{pmatrix} \in \mathbb{R}^2 \middle| \begin{pmatrix} -3 & 1 \\ 0 & -1 \end{pmatrix} \cdot \begin{pmatrix} y \\ z \end{pmatrix} \le \begin{pmatrix} 0 \\ 0 \end{pmatrix} \right\}$$
$$\mathcal{P}_{xz} = \left\{ \begin{pmatrix} x \\ z \end{pmatrix} \in \mathbb{R}^2 \middle| \begin{pmatrix} -2 & \frac{4}{3} \\ 0 & -1 \end{pmatrix} \cdot \begin{pmatrix} x \\ z \end{pmatrix} \le \begin{pmatrix} 0 \\ 0 \end{pmatrix} \right\}$$
(2.13)

One can then see that the  $\mathcal{V}$ -representation of the polyhedra are indeed those found in Equation (2.12), as illustrated in Figure 2.1.

Algorithmic complexity of FME: The FME algorithm can be computationally costly due to its high algorithmic complexity, which is, in the worst case double exponential. For an initial system of  $n_i$  inequalities, a maximum of  $(\frac{n_i}{2})^2$  inequalities can be obtained after one elimination step.<sup>6</sup> After k elimination steps, the algorithm can produce a maximum of  $4(\frac{n_i}{4})^{2^k}$  inequalities, which scales double-exponentially in the number of iterations k. This makes the algorithm quite inefficient in general. However, the method remains useful in several cases for which the scaling is far from the worst-case scaling. It can also be marginally improved by removing some of the redundancies using the so-called Chernikov rules [49, 50]. For the work presented in this thesis, the algorithm was mainly implemented on the PORTA software [55] that takes into account some of the Chernikov rules.

#### 2.2.3 Convex optimization

#### 2.2.3.1 Linear programs

A linear program (LP) consists of optimising a linear function subject to linear equality/inequality constraints. Every linear program can be expressed as a *primal* and a *dual* problem. For  $x, y \in \mathbb{R}^n$ ,  $A \in \mathbb{R}^{m \times n}$ ,  $c \in \mathbb{R}^n$  and  $b \in \mathbb{R}^m$  we have,

Primal	program	Dual program		
Minimise	$c^T x$	Maximise	$b^T y$	
Subject to	$Ax \ge b, x \ge 0$	Subject to	$A^T y \le c, y \ge 0$	

A vector satisfying all the constraints of the primal/dual program is said to be a *feasible solution* of that program. Hence the sets of feasible solutions for the primal and the dual programs are  $\{x \in V : Ax \ge b, x \ge 0\}$  and  $\{y \in V : A^T y \le c, y \ge 0\}$  respectively. An optimal solution for the primal (/dual) program is a feasible solution of that program that attains the smallest (/largest) value of the objective function  $c^T x$  ( $b^T y$ ). A linear program is called *feasible* if the corresponding feasible set of solutions is non-empty and it is said to be unbounded if the objective function problems. Note that the geometry of the feasible region of a linear program is a polyhedron, and hence convex. In the rest of this thesis, we will refer to this as the standard form of the LP<sup>7</sup>.

**Duality:** Using the definitions of the primal and dual programs, we can see that  $c^T x \ge (A^T y)^T x = y^T A x \ge y^T b = b^T y$ . In other words, for any feasible solutions x and y of the primal (minimization) problem and the dual (maximization) problem,  $c^T x \ge b^T y$  holds. This property is called *weak duality*. Another property which is not immediately evident, but nevertheless

<sup>&</sup>lt;sup>6</sup>This corresponds to the case where  $|I_+| = |I_-| = \frac{n_i}{2}$  and  $|I_0| = 0$  for the variable being eliminated.

<sup>&</sup>lt;sup>7</sup>This might differ from other texts but we choose this convention for convenience, since this is also the format used by Mathematica, the main computational software used in this thesis.

true is that of *strong duality*. It states that if the primal and dual problems are feasible, then there exist a pair of feasible solutions  $x^*$  and  $y^*$  of these problems such that  $c^T x^* = b^T y^*$ . By weak duality, these solutions will be optimal.

**Removing redundant constraints using LPs:** Linear programs can be efficiently used to check whether a given system of linear inequalities contains any redundant inequalities that can be dropped without affecting the solution set. In general, to check whether the  $i^{th}$ inequality in  $Ax \ge b$  is redundant, one can minimise  $A_ix$  subject to the remaining inequalities. If the optimal feasible solution is strictly smaller than  $b_i$ , then the inequality  $A_ix \ge b_i$  is not redundant. Otherwise, the inequality is redundant and can be ignored. An example is illustrated in Figure 2.2. Iterating this procedure for all inequalities in the system, we can obtain a minimal set of non-redundant inequalities that are equivalent to the original system. For the work presented in the current thesis, this method was used to remove redundant inequalities in the output of the Fourier-Motzkin elimination procedure. Reducing inequalities at the output of each iteration also helps in speeding up the computational procedure of FME. This was primarily implemented on Mathematica using the inbuilt LinearProgramming function.

The LinearProgramming function in Mathematica can only handle linear programs where the objective vector c, constraint matrix A and the constraint vector b do not contain unspecified variables. For example, we might wish to optimize  $x_1 + ax_2$ , where a is an unspecified constant in the range  $0 \le a \le 1$ , returning the solution for all values of a in the range. For solving linear programs involving such additional unknown variables (other than those being optimised over), subject to assumptions on them, we developed a Mathematica Package, LPASSUMPTIONS [62] that was used to produce some of the main computational results of Chapters 4 and 5. Our package is based on the two-phase simplex method for solving linear programming problems.

#### 2.2.3.2 The Simplex method for solving linear programs

The simplex algorithm is widely used for solving linear programming problems and was originally proposed by George Dantzig [69]. The algorithm is fairly involved, consisting of several steps. Our Mathematica Package [62] (see Appendix 4.5.2 for details) implements this algorithm, however the full details are not required for understanding the numerical results produced using this package. Hence, we provide only a brief overview of the main mathematical concepts underpinning this algorithm and refer the reader to [70, 142, 104] for further details of the algorithm, and other methods for solving linear programs.

As we have previously noted, the feasible region of a linear program is a polyhedron, and by Theorem 2.2.1, every polyhedron can be represented as a Minkowski sum of two sets: the former a convex hull of a finite set of vertices, and the latter a conic hull of a finite set of extremal rays. An extreme point or vertex of the polyhedron defining the feasible region of a linear program is called a *basic feasible solution* (BFS). The following theorem (Theorem 3.3 of [123]) is at the core of the simplex method.

**Theorem 2.2.2** ([123] Theorem 3.3). Let  $\mathcal{P} \subset V$  be polyhedron with at least one extreme point. Then every linear program with the feasible solution set  $\{x \in \mathcal{P}\}$ , is either unbounded or attains its optimal value at an extreme point of  $\mathcal{P}$ .

Further, if a given extreme point  $x_e$  of  $\mathcal{P}$  is not an optimal solution, then it can be shown that there exists an edge of  $\mathcal{P}$  containing  $x_e$  such that the objective function strictly improves (i.e., increases in case of a maximization LP and decreases in case of a minimization LP) as we move away from  $x_e$  along that edge (Section 3.8 of [123]). Finite edges connect extreme points to extreme points and moving along this direction will lead to a new extremal point  $x'_e$  with a better value of the objective function. If the identified edge emanating from  $x_e$  is unbounded, then the linear program has no finite solution. The simplex method is based on this intuition starting with an initial vertex of the feasible set  $\mathcal{P}$  i.e., an initial basic feasible solution, one moves along edges of  $\mathcal{P}$  to other vertices that improve the objective function value, until an optimal vertex is reached or it is revealed that the problem is unbounded.

**Example 2.2.2** (Intuition behind the Simplex algorithm). Consider the following simple linear program (left), expressed in the standard form (right).

Maximise
$$2x + 3y$$
Minimise $(-2, -3)^T \cdot (x, y)$ Subject to $2x + y \le 6$ ,  
 $x \le 2$ ,  
 $y \le 3$ ,  
 $x, y \ge 0$ Subject to $\begin{pmatrix} -2 & -1 \\ -1 & 0 \\ 0 & -1 \end{pmatrix} \cdot \begin{pmatrix} x \\ y \end{pmatrix} \ge \begin{pmatrix} -6 \\ -2 \\ -3 \end{pmatrix}$ 

As shown in Figure 2.2, the feasible region is a polytope (bounded polyhedron) in 2 dimensions with 5 vertices,  $\mathcal{V} = \{(0,0), (0,3), (\frac{3}{2},3), (2,2), (2,0)\}$  and 5 facets  $\mathcal{F} = \{x = 0, y = 0, y = 3, 2x+y = 6, x = 2\}$ . From the figure, it is evident that the maximum must be attained at a vertex since the objective function f(x,y) = 2x + 3y is linear. Calculating the value of the objective function f at all 5 vertices, we immediately see that the maximum value f = 12 is attained at the vertex  $(\frac{3}{2},3)$ . That this is indeed the maximum value of the constrained optimization problem, and that it is attained at the said point, can also be verified by using the method of Lagrange multipliers, for example.

**Phases:** The simplex algorithm proceeds in two phases. Phase I concerns the problem of finding an initial *basic feasible solution*. This is done by formulating a new linear program  $\overline{LP}$  based on the original LP for which an initial BFS  $\overline{x}$  is readily found. The construction of  $\overline{LP}$  is such that applying the simplex method to  $\overline{x}$  either yields an optimum x that would be an initial BFS for the original problem LP, or reveals that LP is infeasible. If feasible, Phase II uses the initial BFS x obtained in Phase I to find an optimum solution (if a finite solution exists) of the original problem LP through the simplex method.



Figure 2.2: Intuition behind the Simplex algorithm: This figure illustrates the simple linear program of Example 2.2.2, where the feasible region corresponds to a polytope (shaded in blue). The thick blue lines correspond to different level curves f(x,y) = c of the objective function f. The maximum value of c for which (x,y) belongs to the feasible region is our required solution. In the present example, the maximum value of f = 12 is attained at the vertex (1.5,3) (in accordance with Theorem 2.2.2). Starting at any other vertex, there is always an edge (in this 2D case, edges are also the facets) along which the value of the f increased, and following the path specified by such an edge at each subsequent vertex, we can always reach the optimal vertex. This figure also illustrates how redundant inequalities can be removed using linear programming (explained in Section 2.2.3.1). Suppose that we have the 5 inequalities of Example 2.2.2, expressed as in the standard form of an LP, along with a sixth inequality  $-2x - 3y \ge -\frac{27}{2}$ . From the figure, it is evident that the sixth inequality is not required to characterise the blue polytope, and can be dropped. This is checked by minimising -2x-3y subject to the other inequalities, which is the LP of Example 2.2.2. Since the minimum value  $-12 \ge -\frac{27}{2}$ we conclude that this inequality is redundant. On the other hand,  $-2x - 3y \ge -9$  along with the other 5 inequalities would define a smaller polytope and would hence not be redundant.

**Column geometry and efficiency:** It can be shown that a linear programming problem with m constraints and n variables has at most  $\binom{m}{n} = \frac{m!}{n!(m-n)!}$  basic feasible solutions when  $m \ge n$ . Hence a brute-force method for finding the optimum solution would be quite inefficient in general. The merit of the simplex method is derived from its ability to find the optimum much more efficiently (relative to a brute-force method) by exploiting the geometry of convex sets. This consists of first expressing the original LP involving inequality constraints in an equivalent form involving only equality constraints, by introducing certain additional *slack variables*. Once the LP is bought to the desired equivalent form, the geometry defined by the column vectors of the new constraint matrix provide a way to systematically compute a better

BFS at each step or to decide whether the LP is unbounded.<sup>8</sup> The improved BFS in every iteration is found by computing its so-called *reduced cost* that indicates how much the objective function value will be improved by moving to that BFS. Hence the method manages to reach the optimum solution (if it exists) in much fewer iterations on an average, as compared to brute-force and other graph-traversal methods [70, 142]. It was originally found that for many linear programs admitting an optimal solution, the simplex method found it in O(n) iterations [69, 70], which was the main reason for the success of the algorithm. However, over the years, a number of examples have surfaced where the number of iterations of the simplex method scales exponentially in the number of variables. In fact, determining the number of iterations needed for solving a given linear program, is known to be an NP-hard problem. Nevertheless, the simplex method continues to be useful in several practical cases (which includes the work presented in this thesis).

#### 2.2.4 Polyhedral representation conversion

The task of converting from the  $\mathcal{H}$  to the  $\mathcal{V}$  representation of a convex polyhedron is called vertex enumeration while the reverse is called as facet enumeration. Representation conversion problems are in general known to be computationally hard— there is no known vertex/facet enumeration algorithm that is polynomial in the input and output size and in the dimension of a polyhedron, and in the case of unbounded polyhedra, the problem is known to be NP hard [124]. In general, it is difficult to characterise the size of the output of such problems given the size of the input. For example, a hypercube in *n*-dimensions has 2n facets and  $2^n$ vertices. The output of the vertex enumeration in this case would scale exponentially in the input while the output of the facet enumeration would be quite small relative to the input size. The computational complexity of vertex/facet enumeration problems continues to be a subject of current research and several algorithms have been proposed for the same.

The most basic algorithm for facet enumeration involves a direct application of Fourier-Motzkin Elimination, which is not always very efficient (as seen in Section 2.2.2). There are also better methods, such as the double description method, that use similar steps as FME, while additionally exploiting properties of polyhedral representations (such as the Minkowski-Weyl Theorem 2.2.1) to achieve better efficiency. Broadly, there are two main classes of polyhedral representation conversion algorithms, namely incremental and graph traversal algorithms. The double description method mentioned above corresponds to an incremental algorithm. In the case of polytopes, one method for implementing the latter is through reversing the procedure of the simplex method (discussed in Section 2.2.3.2), whereby starting at a vertex that optimises a suitably chosen objective function, all other vertices are systematically enumerated until every vertex has been visited at least once [92].

<sup>&</sup>lt;sup>8</sup>Here, the columns of the constraint matrix provide a basis, an appropriate linear combination of which gives the required BFS. Geometrically, the basis vectors in each iteration correspond to the vertices of a simplex and the associated BFS lies in this simplex i.e., belongs to the convex hull of the vertices represented by those basis vectors. The geometry of the simplex provides a minimal, affinely independent basis for representing the basic solution at each step.

Softwares such as PORTA [55], on which the main computational work of this thesis has been carried out, employ FME-based incremental algorithms. We present an overview of a simple FME-based approach for enumerating facets of polyhedral cones, without going into the details of more efficient incremental or graph-traversal algorithms. The concept of a *polar dual* of a polyhedron, which essentially interchanges the role of the vertices and faces of a polyhedron (see Figure 2.3) is then used such that we also get vertex enumeration for free, given an algorithm for facet enumeration.

**Facet enumeration:** Consider a polyhedron  $\mathcal{P} \subseteq V$  expressed as  $\mathcal{P} = \text{Conv}(x_1, ..., x_k) + \text{Cone}(y_1, ..., y_l)$  in the  $\mathcal{V}$ -representation. To convert to the  $\mathcal{H}$ -representation of  $\mathcal{P}$ ,

1. Express the convex and conic hulls in terms of inequalities as follows

$$x = \alpha_1 x_1 + \dots + \alpha_k x_k + \lambda_1 y_1 + \dots + \lambda_l y_l,$$
  

$$1 = \alpha_1 + \dots + \alpha_k,$$
  

$$\alpha_1, \dots, \alpha_k, \lambda_1, \dots, \lambda_l \ge 0$$
(2.14)

2. The variables in the above system of equations are  $x, \alpha_1, ..., \alpha_k, \lambda_1, ..., \lambda_l$ . Project out the variables  $\alpha_1, ..., \alpha_k, \lambda_1, ..., \lambda_l$  using Fourier-Mozkin Elimination (See Section 2.2.2) to obtain linear inequalities only involving components of x, which corresponds to the required  $\mathcal{H}$ -representation.

In order to convert from the  $\mathcal{H}$  to  $\mathcal{V}$ -representation, we require the concept of the polar or polar dual of a polyhedron.

**Definition 2.2.8** (Polar of a polyhedron). The polar  $\mathcal{P}^{\Delta}$  of a polyhedron  $\mathcal{P} \subseteq V$  is defined as

$$\mathcal{P}^{\Delta} = \{ c \in V : c^T x \le 1, \forall x \in \mathcal{P} \}.$$

$$(2.15)$$

The following theorem (Theorem 9.1 of [187]) elucidates some elegant properties of the polar  $\mathcal{P}^{\Delta}$  and relates its  $\mathcal{H}/\mathcal{V}$  representations to those of the original polytope  $\mathcal{P}$ .

**Theorem 2.2.3** (Properties of the polar). Let  $\mathcal{P} \subseteq V$  be a polyhedron that contains the origin. Then,

- 1.  $\mathcal{P}^{\Delta}$  is a polyhedron and  $\mathcal{P}^{\Delta\Delta} = \mathcal{P}$ ,
- 2. If  $\mathcal{P} = \text{Conv}(x_1, ..., x_k) + \text{Cone}(y_1, ..., y_l) \subseteq V$ , then  $\mathcal{P}^{\Delta} = \{u \in V : x_i^T u \leq 1, \forall i = 1, ..., k \text{ and } y_j^T u \leq 0, \forall j = 1, ..., l\}$ ,
- 3. If  $\mathcal{P} = \{u \in V : x_i^T u \le 1, \forall i = 1, ..., k, and y_j^T u \le 0, \forall j = 1, ..., l\}, then \mathcal{P}^{\Delta} = \text{Conv}(x_1, ..., x_k) + \text{Cone}(y_1, ..., y_l).$



Figure 2.3: A polytope (blue cube) and its dual (red octahedron): The dual of a polyhedron has the elegant property that there is a one-to-one correspondence between vertices of the original polyhedron and facets of the dual, and between facets of the original and vertices of the dual. Using this concept, any algorithm for facet enumeration also yields an algorithm for vertex enumeration. The notion of a polar dual that is required for this inter-conversion is defined formally in the main text. Note that this figure is only for the purpose of illustrating the concept of a dual, and does not exactly correspond to the polar dual.

**Vertex enumeration:** Consider a polyhedron  $\mathcal{P} \subseteq V$  that contains the origin in its interior. If a polyhedron does not contain the origin, then by an appropriate coordinate transformation, it can be transformed into one that does. This allows  $\mathcal{P}$  to be expressed in the  $\mathcal{H}$ -representation as  $\mathcal{P} = \{x \in V : Ax \leq b\}$ , with  $b_i \geq 0 \forall i$ . To convert to the  $\mathcal{V}$ -representation of  $\mathcal{P}$ ,

- 1. For all inequalities  $A_i x \leq b_i$  such that  $b_i > 0$ , divide throughout by  $b_i$  to express in the form  $\tilde{A}_i x \leq 1$ . Without loss of generality, let these be the first k inequalities. Then  $\mathcal{P} = \{x \in V : \tilde{A}_i x \leq 1, \forall i = 1, ..., k \text{ and } A_i x \leq 0, \forall i = k + 1, ..., m\}$ , where m is the total number of inequalities.
- 2. Use the third statement of Theorem 2.2.3 to find the  $\mathcal{V}$ -representation of the polar  $\mathcal{P}^{\Delta}$  as  $\mathcal{P}^{\Delta} = \operatorname{Conv}(\tilde{A}_{1}^{T}, ..., \tilde{A}_{k}^{T}) + \operatorname{Cone}(A_{k+1}^{T}, ..., A_{m}^{T}).$
- 3. Convert  $\mathcal{P}^{\Delta}$  to its  $\mathcal{H}$ -representation using facet enumeration.
- 4. Use the first an third statements of Theorem 2.2.3 to find the  $\mathcal{V}$ -representation of  $\mathcal{P} = (\mathcal{P}^{\Delta})^{\Delta}$  using the  $\mathcal{H}$ -representation of  $\mathcal{P}^{\Delta}$  obtained in the previous step.

## 2.3 Information-theoretic entropy measures and their properties

Entropies have played a crucial role in the study of thermodynamic phenomena and statistical mechanics since the 19th century [58, 29, 97], and in 1932, John von Neumann generalised the concept of entropies to quantum systems [212]. An information-theoretic understanding of entropy on the other hand, was first developed by Claude Shannon in his seminal 1948 paper

[190], where he introduced the Shannon entropy as a measure of the information content of a message and used it to characterise the information capacity of communication channels. Interestingly, the von Neumann entropy, even though chronologically prior, turns out to be a quantum generalisation of the Shannon entropy.<sup>9</sup> Since then, entropies have become an integral part of both classical and quantum information theory. Several other information-theoretic entropy measures have also been proposed (such as Rényi [179], Tsallis [200], min and max entropies [177]), and serve as useful mathematical tools for tackling a plethora of informationtheoretic problems such as data compression [190, 172], randomness extraction [24, 128], key distribution [177, 12] as well as causal structures [46, 89, 213, 216]. Here, we provide a brief overview of some of the classical and quantum information-theoretic entropy measures and their properties that are relevant for the work presented in this thesis.

#### 2.3.1 Shannon and von Neumann entropies

**Definition 2.3.1** (Shannon entropy). Given a random variable X distributed according to the discrete probability distribution  $P_X$ , the Shannon entropy of X is given by<sup>10</sup>

$$H(X) = -\sum_{x} P(x) \ln P(x).$$
 (2.16)

The Shannon entropy H(X) is a continuous and strictly concave function of the probability distribution over X i.e., for two probability distributions P and Q over X and  $\lambda \in [0,1]$ , denoting the dependence on the distribution by a subscript,

$$H(X)_{\lambda P+(1-\lambda)Q} \ge \lambda H(X)_P + (1-\lambda)H(X)_Q.$$

$$(2.17)$$

**Definition 2.3.2** (Conditional Shannon entropy). Given two random variables X and Y, distributed according to  $P_{XY}$ , the *conditional Shannon entropy* is defined by

$$H(X|Y) = -\sum_{x,y} P(xy) \ln \frac{P(xy)}{P(y)}$$
(2.18)

It is also useful to define a quantity called the *mutual information*, which, as the name suggests, quantifies the amount of information that two random variables carry about each other. The conditional version of this quantity quantifies how much information two variables share about each other, given the value of a third variable.

**Definition 2.3.3** (Shannon mutual information). Given 3 random variables, X, Y and Z, the Shannon mutual information and Shannon conditional mutual information between X and Y, and X and Y conditioned on Z are respectively given by

$$I(X:Y) = H(X) - H(X|Y),$$
  

$$I(X:Y|Z) = H(X|Z) - H(X|YZ).$$
(2.19)

<sup>&</sup>lt;sup>9</sup>In fact, Shannon had corresponded with von Neumann, who is said to have suggested that Shannon call his new uncertainty measure "entropy", due to its similarities with prior notions of thermodynamic and statistical entropies.

<sup>&</sup>lt;sup>10</sup>Note that it is common to take logarithms in base 2 and measure entropy in *bits*; here we use base e corresponding to measuring entropy in *nats*.

**Properties of Shannon entropy:** Some useful information-theoretic properties satisfied by the Shannon entropy and associated mutual information are discussed below. Some of these properties can be expressed in different, but equivalent forms. We will distinguish between these because equivalences that apply in the case of the Shannon entropy may not necessarily apply for other entropies. Whenever we have an entropic equation/inequality where no conditional entropies are explicitly involved (such as the left most ones in all of the properties below), we will refer to it as the *unconditional form* of that property, otherwise it will be called the *conditional form*. According to the defining of mutual information (Equation (2.19)) adopted in this thesis, the forms of properties expressed in terms of the mutual informations qualify as conditional forms as well. In the following, R, S and T are three disjoint sets of random variables.

- 1. Non-negativity and upper bound (UB): The Shannon entropy is positive i.e.,  $H(R) \ge 0$  and upper bounded as  $H(R) \le \ln |R|$ , where |.| denotes cardinality. Equality for the upper bound is achieved if and only if  $P_R(r) = 1/|R|$  for all possible values  $r \in R$  (i.e., if the distribution on R is uniform).
- 2. Chain rule (CR): For disjoint sets  $R_1, ..., R_n$ , S of random variables,

$$H(R_1, R_2, \dots, R_n | S) = \sum_{i=1}^n H(R_i | R_{i-1}, \dots, R_1, S), \qquad (2.20)$$

which in particular implies the two useful identities H(R|S) = H(RS) - H(S) and H(R|ST) = H(RS|T) - H(S|T).

3. Additivity (A): If R and S are independent i.e., if  $P_{RS} = P_R P_S$ , then

$$H(RS) = H(R) + H(S) \quad \stackrel{CR}{\longleftrightarrow} \quad H(R|S) = H(R) \tag{2.21}$$

4. Monotonicity (M): From the definition (2.18) of the conditional Shannon entropy, it follows that

$$H(S) \le H(RS) \quad \stackrel{CR}{\Longleftrightarrow} \quad H(R|S) \ge 0$$
 (2.22)

5. Subadditivity (SA):

$$H(R) + H(S) \ge H(RS) \quad \stackrel{CR}{\longleftrightarrow} \quad H(R) \ge H(R|S) \quad \stackrel{(2.19)}{\longleftrightarrow} \quad I(R:S) \ge 0 \tag{2.23}$$

6. Strong subadditivity (SSA):

$$H(RT) + H(ST) \ge H(RST) + H(T) \quad \stackrel{CR}{\longleftrightarrow} \quad H(R|T) \ge H(R|ST) \quad \stackrel{(2.19)}{\longleftrightarrow} \quad I(R:S|T) \ge 0$$

$$(2.24)$$

Note that SSA implies SA when T is the empty set, but SA does not imply SSA. The unconditional form of SSA is often called *submodularity*. The conditional forms of SA and SSA (often referred to as *data-processing* inequalities) tell us that entropy cannot increase when conditioning on more random variables, which is a crucial information-theoretic property of the entropy. The conditional and unconditional forms of the properties given above are equivalent for entropic measures that satisfy the chain rule (CR) in the form given in Equation (2.20). However for entropies that do not satisfy this chain rule, the two forms will be inequivalent and we will explicitly specify which form is being referred to in such cases. In the rest of this thesis, when we say that an entropy satisfies the properties given by Equations (2.20)-(2.24), this should be understood as— the corresponding property holds when the Shannon entropy H() in these equations is replaced by the entropy measure under consideration.

The von Neumann entropy generalises the Shannon entropy to quantum states (represented by density operators), and is defined as follows.

**Definition 2.3.4** (von Neumann entropy). The von Neumann entropy of a density operator  $\rho_A \in \mathscr{S}(\mathscr{H}_A)$  is defined as

$$H(A)_{\rho} \coloneqq -\operatorname{tr}(\rho \ln(\rho)). \tag{2.25}$$

It follows that the von Neumann entropy is zero for pure quantum states.

We will often drop the subscript  $\rho$  in the von Neumann entropy even though this will lead to the same notation as the Shannon case. However, this is justified since classical probability distributions can be equivalently encoded using diagonal density matrices (with the distribution along the diagonal), in which case we indeed recover the Shannon entropy. We will often use X, Y, Z etc. to denote classical random variables and A, B, C etc. to denote labels of quantum subsystems, to distinguish between the two entropies. With this, the conditional von Neumann entropy is simply defined through the chain rule, i.e.,

$$H(A|B) = H(AB) - H(B).$$
 (2.26)

The von Neumann mutual information and its conditional version are defined exactly as in the Shannon case, i.e., through Equation (2.19) with H() denoting the von Neumann entropy instead.

**Properties of von Neumann entropy:** The von Neumann entropy is nonnegative and satisfies additivity (Equation (2.21)), chain rule (Equation (2.20)), subadditivity (Equations (2.23)) and strong subadditivity<sup>11</sup> (Equations (2.24)). Further, it also admits a corresponding upper bound,  $H(A) \leq \ln(d_A)$  where  $d_A$  denotes the dimension of the Hilbert space  $\mathscr{H}_A$ , with equality if an only if  $\rho_A$  is the maximally mixed state  $\frac{1}{d_A}$ . Note however that the von Neumann entropy *does not* satisfy monotonicity (Equation (2.22)) i.e., the conditional entropy is not positive. This can be seen by taking  $\rho_{AB}$  to be an entangled pure state in Equation (2.26), such that H(A), H(B) > 0 even though H(AB) = 0. Instead, the von Neumann entropy satisfies a weaker property (*weak monotonicity*) that states that for any tripartite state  $\rho_{ABC}$ , the following holds.

$$H(A|B)_{\rho} \ge -H(A|C)_{\rho} \tag{2.27}$$

<sup>&</sup>lt;sup>11</sup>The strong subadditivity of the von Neumann entropy is an important theorem in quantum information theory, proven by Lieb and Ruskai [135]. The original proof (which is also the one presented in standard textbooks [151]) is particularly known for its complexity and a number of simpler proofs have been proposed since then (see for example [177]).
## 2.3.2 Tsallis entropies

**Definition 2.3.5** ((Classical) Tsallis entropies). Given a random variable X distributed according to the discrete probability distribution  $P_X$ , the order q Tsallis entropy of X for a non-negative real parameter q is defined as [200]

$$S_q(X) = \begin{cases} -\sum_{\{x \in X: P_X(x) > 0\}} P(x)^q \ln_q P(x) & \text{if } q \neq 1 \\ H(X) & \text{if } q = 1 \end{cases}$$
(2.28)

where  $\ln_q P(x) = \frac{P(x)^{1-q}-1}{1-q}$ .

The q-logarithm function converges to the natural logarithm in the limit  $q \to 1$  so that  $\lim_{q \to 1} S_q(X) = H(X)$  and the function is continuous in q. For brevity, we will henceforth write  $\sum_x$  instead of  $\sum_{\{x:P(x)>0\}}$ , keeping it implicit that probability zero events do not contribute to the sum.<sup>12</sup> An equivalent form of Equation (2.28) is the following.

$$S_q(X) = \begin{cases} \frac{1}{1-q} \left( \sum_{x \in X} P(x)^q - 1 \right) & \text{if } q \neq 1 \\ H(X) & \text{if } q = 1 \end{cases}$$
(2.29)

A number of non-equivalent ways of defining the conditional Tsallis entropy have been proposed in the literature [93, 3]. The definitions of [3] does not satisfy the chain rule that is satisfied by the conditional Shannon entropy (Equation (2.20)), while the definition of [93] does. Hence we will stick to the latter definition throughout this thesis, which is as follows.

**Definition 2.3.6** (Conditional Tsallis entropies). Given two random variables X and Y, distributed according to  $P_{XY}$ , the order q conditional Tsallis entropy for  $q \ge 0$  is defined by

$$S_q(X|Y) \coloneqq \begin{cases} -\sum_{x,y} P(xy)^q \ln_q P(x|y) & \text{if } q \neq 1\\ H(X|Y) & \text{if } q = 1 \end{cases}$$
(2.30)

Note that  $S_q(X|Y)$  converges to the Shannon conditional entropy H(X|Y) in the limit  $q \rightarrow 1$ . The unconditional and conditional Tsallis mutual informations are defined analogously to Equation (2.19) for the Shannon case with I replaced by  $I_q$  and H replaced by  $S_q$ .

$$I_q(X:Y) = S_q(X) - S_q(X|Y),$$
  

$$I_q(X:Y|Z) = S_q(X|Z) - S_q(X|YZ).$$
(2.31)

**Properties of Tsallis entropies:** Tsallis entropies are non-negative and satisfy the unconditional and conditional forms of monotonicity (Equation (2.22)) and the chain rule (Equation (2.20)) for all  $q \ge 0$ . They also satisfy both the forms of subadditivity (Equations (2.23))

<sup>&</sup>lt;sup>12</sup>Note that this means the Tsallis entropy for q < 0 is not robust in the sense that small changes in the probability distribution can lead to large changes in the Tsallis entropy.

and strong subadditivity (Equations (2.24)) for all  $q \ge 1$ . For  $q \ge 0$ , they admit an analogous upper-bound as the Shannon case i.e.,  $S_q(X) \le \ln_q |X|$ , where for q > 0, equality is achieved if and only if  $P_X(x) = 1/|X|$  for all x (i.e., if the distribution on X is uniform). However, Tsallis entropies are not additive (Equation (2.21)) in general and instead satisfy a weaker condition known as *pseudo-additivity* [68]— for two independent random variables X and Y i.e.,  $P_{XY} = P_X P_Y$ , and for all q, the Tsallis entropies satisfy

$$S_q(XY) = S_q(X) + S_q(Y) + (1 - q)S_q(X)S_q(Y).$$
(2.32)

Note that in the Shannon case (q = 1), we recover additivity for independent random variables. For q < 1, strong subadditivity (Equation (2.24)) does not hold in general [93]. The results presented in this thesis rely on this property and hence we will restrict to the  $q \ge 1$  case in the remainder of this thesis. The quantum generalisation of the (classical) Tsallis entropy is defined as follows. Note that in the rest of this thesis, "Tsallis entropy" will stand for the "classical Tsallis entropy" defined in Definition 2.3.5. In the quantum case, we will explicitly use "quantum Tsallis entropy".

**Definition 2.3.7** (Quantum Tsallis entropies). The order q quantum Tsallis entropy of a density operator  $\rho_A \in \mathscr{S}(\mathscr{H}_A)$ , for a non-negative real parameter q is defined as [200]

$$S_q(A) = \begin{cases} \frac{1}{1-q} (\operatorname{tr}(\rho^q) - 1) & \text{if } q \neq 1 \\ H(A) & \text{if } q = 1, \end{cases}$$
(2.33)

where H(A) denotes the von Neumann entropy (Definition 2.3.4) of  $\rho_A$ .

Quantum Tsallis entropies can be seen as a generalisation of the von Neumann entropy since they converge to H(A) in the limit of q going to 1. The conditional quantum Tsallis entropy for a density operator  $\rho_{AB} \in \mathscr{S}(\mathscr{H}_A \otimes \mathscr{H}_B)$  can be simply defined as

$$S_q(A|B) = S_q(AB) - S_q(B)$$

$$(2.34)$$

(analogous to the von Neumann case, Equation (2.26)).

**Properties of quantum Tsallis entropies:** Quantum Tsallis entropies are non-negative and satisfy the chain rule (Equation (2.20)) by definition of the conditional entropy. They also satisfy pseudo-additivity (Equation (2.32))  $\forall q \geq 0$ , both forms of subadditivity (Equation (2.23))  $\forall q \geq 1$  and are upper-bounded as  $S_q(A) \leq \ln_q(d_A)$ , where the bound is saturated for q > 0 if and only if  $\rho_A$  is the maximally mixed state. However, they no not satisfy monotonicity and strong-subadditivity in general. The former is evident since the von Neumann entropy, a special case of quantum Tsallis entropies, also does not satisfy this property. The latter was noted in [162] where sufficient conditions for the strong subadditivity of quantum Tsallis entropies expressed in its unconditional form, were also analysed.

	А	CR	М	SA	SSA
	Eq. $(2.21)$	Eq. $(2.20)$	Eq. $(2.22)$	Eq. $(2.23)$	Eq. $(2.24)$
Shannon	✓ [190]	✓ [190]	✓ [190]	✓ [190]	✓ [190]
Von Neumann	✓ [212]	$\checkmark$ (By definition)	× [43] (Weak Mono. [135])	√[7]	✓ [135]
Tsallis	$\times [200]$ (Pseudo-add.)	√ [93]	✓ [71]	$ \checkmark [93] \\ (q \ge 1) $	$ \sqrt{[93]} \\ (q \ge 1) $
Q. Tsallis	$\times [200]$ (Pseudo-add.)	$\checkmark$ (By definition)	× [43]	$ \sqrt{[13]} \\ (q \ge 1) $	× [162]
Rényi	✓ [179]	×[120] (Alternate CR [148])	√[136]	√ (C. form [120])	√ )(C. form [120])
Min	✓ [177]	× [177]	× [177]	√ (C. form [177])	√ )(C. form [177])
Max	√ [177]	× [177]	× [177]	√ (C. form [177])	√ )(C. form [177])

Table 2.1: Information-theoretic properties of generalised entropies: This table summarises the main properties of the entropy measures discussed in the main text along with those of the min and max entropies. The acronyms A, CR, M, SA, SSA and C. form denote additivity, chain rule, monotonicity, subadditivity, strong subadditivity and conditional form respectively. Rényi, Min and Max entropies satisfy SA and SSA only in the conditional form while the remaining entropies satisfy them (wherever indicated so) in both conditional and unconditional forms. For the von Neumann and quantum Tsallis entropies, the chain rule follows by definition (Equations (2.26) and (2.34)). Monotonicity fails for the von Neumann entropy and hence the quantum Tsallis entropy which is a generalisation of it for  $q \neq 1$ . Weak Monotonicity (2.27) holds in the von Neumann case but an analogue for the more general  $q \ge 0$  quantum Tsallis entropies is not known (to the best of our knowledge). Pseudo-additivity (Equation (2.32)) for quantum Tsallis entropies follows in the same manner as it does for the classical case shown in [200]. The table only lists the properties of the classical Rényi entropy, since the quantum version is not relevant for the work presented in this thesis. Multiple definitions of quantum Rényi entropies have been proposed [199, 148], and these properties may differ in each case. For example SSA in conditional form is only known to hold for  $\alpha \in [0,2]$  for the quantum Rényi entropy of [199]. The Rényi entropies do not satisfy the chain rule of Equation (2.20), but they satisfy a weaker, cardinality-dependent chain rule of Equation (2.38). The min and max entropies (which are originally defined as quantum entropies) are also included in this table for comparison, their definition and further properties can be found in [177].

## 2.3.3 Rényi entropies

Here we define only the classical versions of Rényi entropies, as the quantum version will not be used in the remainder of this thesis.

**Definition 2.3.8** ((Classical) Rényi entropies). Given a random variable X distributed according to the discrete probability distribution  $P_X$ , the order  $\alpha$  Rényi entropy of X for a non-negative real parameter  $\alpha$  is defined as

$$R_{\alpha}(X) = \begin{cases} \frac{1}{1-\alpha} \ln\left(\sum_{x \in X} P(x)^{\alpha}\right) & \text{if } \alpha \neq 1\\ H(X) & \text{if } \alpha = 1 \end{cases}$$
(2.35)

There are several inequivalent definitions of the conditional Rényi entropy proposed in the literature [120, 83], in this thesis, we will consider the following definition.

**Definition 2.3.9** (Conditional Rényi entropy). Given two random variables X and Y, distributed according to  $P_{XY}$ , the order  $\alpha$  conditional Rényi entropy for  $\alpha \ge 0$  is defined by

$$R_{\alpha}(X|Y) = \begin{cases} \frac{1}{1-\alpha} \log(\sum_{xy} P(xy)^{\alpha} P(y)^{1-\alpha}) & \text{if } \alpha \neq 1\\ H(X|Y) & \text{if } \alpha = 1 \end{cases}$$
(2.36)

Both the unconditional and conditional Rényi entropies converge to the corresponding Shannon entropy as  $\alpha$  tends to 1.

**Properties of Rényi entropies:** The Rényi entropies are nonnegative and upper-bounded as  $R_{\alpha}(X) \leq \ln |X|$  for all  $\alpha \geq 0$ , with equality (for  $\alpha > 0$ ) if and only if  $P_X$  is the uniform distribution over X [179]. Rényi entropies do not satisfy the chain rule (Equation (2.20)) in general for  $\alpha \neq 1$ . However, an alternate, dimension dependent chain rule has been proposed for quantum Rényi entropies in [148], which therefore also hold for classical Rényi entropies (where dimension would correspond to the cardinality of the variables). This alternate chain rule is given as

$$H_{\alpha}(X|YZ) \ge H_{\alpha}(XZ|Y) - \ln|Z|. \tag{2.38}$$

Further, Rényi entropies also satisfy monotonicity (Equation (2.22)) in both the conditional and unconditional forms i.e.,  $H_{\alpha}(X) \leq H_{\alpha}(XY)$  and  $H_{\alpha}(X|Y) \geq 0$ , even though these forms are not equivalent due to failure of the chain rule (in contrast to the Shannon case). Classical Rényi entropies have been shown to satisfy subadditivity (Equation (2.23)) as well as strong subadditivity (Equation (2.24)), both only in the conditional forms. They fail to satisfy the (inequivalent) unconditional form of these properties for  $\alpha \neq 1$ .

Table 2.1 summarises some of the important information-theoretic properties of the entropies discussed so far, as well as those of the min and max entropies proposed in [177] for comparison. We will not define or discuss these and other entropy measures here as they are not relevant for the work presented in this thesis.

# 2.4 Generalised probabilistic theories

Quantum theory has worked and continues to work remarkably well to explain and predict empirical observations. However there is no consensus on a set of natural physical principles that single out quantum theory from a plethora of other possible theories that are also compatible with relativistic principles (such as finite signalling speed) but produce strongerthan-quantum correlations [170]. Generalised probabilistic theories (GPTs), of which classical and quantum theories can be seen as particular members, were developed out the motivation to derive the mathematical formalism of quantum theory from fundamental physical principles or axioms, and to probe deeper, theory-independent connections between information processing and physical principles [188, 139, 140, 141, 110, 16, 157]. Several properties such as (Bell) non-locality, contextuality, entanglement, non-unique decomposition of a mixed state into pure states, monogamy of correlations, information-theoretically secure cryptography and no-cloning have been shown to extend beyond quantum theory [170, 110, 17, 144, 16]. However, teleportation and entanglement swapping are not always possible in non-classical GPTs despite the existence of entangled states [192, 16, 193, 103, 218].<sup>13</sup> Further, from a foundational point of view, it is of interest to understand whether interpretational issues are peculiarities of quantum theory. The contextuality/non-locality of GPTs suggests that similar interpretational problems could arise here. We will discuss this aspect in more detail in Chapter 8 where we derive a Wigner's friend type paradox in *box world* (a particular GPT).

In the present chapter, we provide an overview of the framework for information processing in GPTs proposed by Barrett in [16]. In the remainder of this thesis, we will use the more colloquial term, "box world" to denote the set of theories that Barrett originally calls *Generalised no-signaling Theories*. This theory allows arbitrary correlations between measurements on separated systems, as long as they are non-signaling i.e., choice of measurement on one subsystem does not affect the measurement outcome on the other. Here separation between systems is expressed in terms of a tensor product structure (that allows us to identify subsystems), and without reference to space-time locations/space-like separation<sup>14</sup>. Not all GPTs need to have a tensor product structure for representing composite systems, but will focus only on those that do, for the purposes of this thesis. The following is based on the review sections of our paper [208].

## 2.4.1 States and transformations

**Individual states.** The so-called generalised bit or *gbit* is a system completely characterized by two binary measurements which can be performed on it [16]. Such sets of measurements that completely characterise the state of a system are known as *fiducial measurements*. The state of a gbit is thus fully specified by the vector

$$\vec{P}_{gbit} = \begin{pmatrix} P(a=0|X=0) \\ P(a=1|X=0) \\ \hline P(a=0|X=1) \\ P(a=1|X=1) \end{pmatrix},$$
(2.39)

<sup>&</sup>lt;sup>13</sup>The intuition here is that there is a trade-off between the state space and the effect space (the dual of the former). A larger state space implies that the set of effects leading to valid probabilities would be smaller, leading to a smaller effects space. Hence, there exist GPTs with entangled states but no analogue of entangling measurements, both of which are needed for teleportation-like protocols.

<sup>&</sup>lt;sup>14</sup>Of course this leads to systems and correlations that are non-signaling also with respect to a space-time structure once this is introduced into the picture.



(a) G-bit. A gbit is a function with binary input and output, characterized by the probability vector  $\vec{P}_{gbit}$ , also called the state vector.



(b) **PR box.** The PR box has two binary inputs X, Y and two binary outputs a, b, satisfying  $XY = a \oplus b$ , and otherwise uniformly random (state vector on the right). Usually it is applied in the context of two space-like separated agents, each providing one of the inputs and obtaining the respective output. The box is non-signaling, and maximally violates the CHSH inequality [170].

**Figure 2.4:** Boxes in Generalized Probabilistic Theories. The modular objects of GPTs are input/output functions depicted as boxes and characterized by probability vectors. Each function (or box) can be evaluated once, and it may or not correspond to a physical system being probed; even if it is, nothing is assumed about the post-evaluation state of the system (unlike quantum theory, which specifies the post-measurement state of a system given its initial state and the measurement device).<sup>15</sup>

where X = 0 and X = 1 represent the two choices of measurements and  $a \in \{0, 1\}$  are the possible outcomes (Figure 2.4a). Analogously, a classical bit is a system characterized by a single binary fiducial measurement,

$$\vec{P}_{bit} = \begin{pmatrix} P(a=0|X=0) \\ P(a=1|X=0) \end{pmatrix},$$
(2.40)

and, in quantum theory, a qubit is characterized by three fiducial measurements (corresponding, for example, to three directions X, Y and Z in the Bloch sphere),

$$\vec{P}_{qubit} = \begin{pmatrix} P(a=0|X=0) \\ P(a=1|X=0) \\ \hline P(a=0|X=1) \\ P(a=1|X=1) \\ \hline P(a=0|X=2) \\ P(a=1|X=2) \end{pmatrix}.$$
(2.41)

<sup>&</sup>lt;sup>15</sup>This figure is taken from our paper [208] which is joint work with Nuriya Nurgalieva and Lídia del Rio, and was made by Lídia.

For normalized states, we have  $|\vec{P}| = \sum_{i} P(a = i | X = j) = 1, \forall j$ . The set of possible states of a gbit is convex, with the extreme points

$$\vec{P}_{00} = \begin{pmatrix} 1\\ 0\\ 1\\ 0 \end{pmatrix}, \qquad \vec{P}_{01} = \begin{pmatrix} 1\\ 0\\ 0\\ 1 \end{pmatrix}, \qquad \vec{P}_{10} = \begin{pmatrix} 0\\ 1\\ 1\\ 0 \end{pmatrix}, \qquad \vec{P}_{11} = \begin{pmatrix} 0\\ 1\\ 0\\ 1 \end{pmatrix}.$$
(2.42)

These correspond to pure states, and the state space of a gbit is a polytope (due to the finite number of extreme points). In the qubit case, the extreme points correspond to all the (infinitely many) pure states on the surface of the Bloch sphere, some of these are

$$\vec{P}_{|+\rangle} = \begin{pmatrix} 1\\ 0\\ \frac{1}{1/2}\\ \frac{1}{2}\\ \frac{1}{$$

Note that in box world, pure gbits are deterministic for both alternative measurements, whereas in quantum theory at most one fiducial measurement can be deterministic for each pure qubit, as reflected by uncertainty relations. We denote the set of allowed states of a system A by  $\mathscr{S}^A$ .

**Composite states.** The state of a bipartite system AB, denoted by  $\vec{P}^{AB} \in \mathscr{S}^{AB}$  can be written in the form  $\vec{P}^{AB} = \sum_i r_i \vec{P}_i^A \otimes \vec{P}_i^B$  where  $r_i$  are real coefficients<sup>16</sup> and  $\vec{P}_i^A \in \mathscr{S}^A$ ,  $\vec{P}_i^B \in \mathscr{S}^B$  can be taken to be pure and normalised states of the individual systems A and B [16]. Thus, a general 2-gbit state  $\vec{P}_2^{AB}$  can be written as in Figure 2.4b (left), where  $X, Y \in \{0, 1\}$  are the two fiducial measurements on the first and second gbit and  $a, b \in \{0, 1\}$  are the corresponding measurement outcomes. The PR box  $\vec{P}_{PR}$ , on the right, is an example of such a 2 gbit state that is valid in box world, which satisfies the condition  $a \oplus b = xy$  [170].

**State transformations.** Valid operations are represented as matrices that transform valid state vectors to valid state vectors. In addition, we only have access to the (single-shot) input/output behaviour of systems, so in practice all valid operations in box world take the form of classical wirings between boxes, which correspond to pre- and post-processing of input and output values, and convex combinations thereof [16]. For example, bipartite joint measurements on a 2-gbit system can be decomposed into convex combinations of classical "wirings", as shown in Figure 2.5. In contrast, quantum theory allows for a richer structure of bipartite measurements by allowing for entangling measurements (e.g. in the Bell basis), which cannot be decomposed into classical wirings. Bipartite transformations on multi-gbit systems turn out to be classical wirings as well [16]. Reversible operations in particular consist only of trivial wirings: local operations and permutations of systems [103]. One cannot perform entangling operations such as a coherent copy (the quantum CNOT gate) [16, 192, 103].

<sup>&</sup>lt;sup>16</sup>Note that it is not necessary that the coefficients  $r_i$  be positive and sum to one. If this is the case, then the composite state would be separable and hence local, otherwise, the state is entangled [16].



Figure 2.5: Bipartite measurements in box world. Any bipartite measurement on a 2-gbit box world system can be decomposed into a procedure (or convex combinations thereof) of the following form. Alice first performs a measurement X on one of the gbits (labelled A), and forwards the outcome a to Bob. Bob then performs a measurement  $Y = f_1(a)$ , which may depend on a, on the other gbit (labelled B), obtaining the outcome b. The final measurement outcome o of the joint measurement can be computed by Bob as a function  $f_2(a, b)$ . All allowed bipartite measurements are convex combinations of this type of classical wirings [16].

## 2.4.2 Measurements: observing outcomes

In quantum theory, systems are described by states that live in a Hilbert space, measurements and transformations on these states are represented by CPTP maps and the Born rule specifies how to obtain the probabilities of possible measurement outcomes give these states and measurements. In more general theories, there is no reason to assume Hilbert spaces or CPTP maps. In fact such a description of the state space and operations may not even be available, systems may be described as black boxes taking in classical inputs (choice of measurements) and giving classical outputs (measurement outcomes). What we can demand is that the theory provides a way for agents to predict the probabilities of obtaining various outputs based on their input choice and some operational description of the box. We have briefly reviewed states and transformations in Barrett's framework for GPTs [16], we now discuss how measurements and associated probabilities are characterised here.

Consider a GPT,  $\mathbb{T}$ . Denoting the set of all allowed states of a system in  $\mathbb{T}$  by  $\mathscr{S}$ , any valid transformation on a normalised GPT state  $\vec{P} \in \mathscr{S}$  maps it to another normalised GPT state in  $\mathscr{S}$ . The set of allowed normalized states is assumed to be closed and convex<sup>17</sup> and transfor-

<sup>&</sup>lt;sup>17</sup>The motivation for this assumption is that if it is possible to prepare two states  $P_1$  and  $P_2$ , then it should

mations on these states, assumed to be linear. Each valid transformation is then represented by a matrix M such that  $\vec{P} \to M.\vec{P}$  under this transformation and  $M.\vec{P} \in \mathscr{S}$  [16]. Further, operations that result in different possible outcomes can be associated with a set of transformations, one for each outcome. These also give an operational meaning to unnormalised states where  $|\vec{P}| = \sum_i P(a = i|X = j) = c \quad \forall j, c \in [0, 1]$  (i.e., the norm is independent of the value of j). Such an operation M on a normalised initial state  $\vec{P}$  can be associated with a set of matrices  $\{M_i\}$  such that the unnormalised state corresponding to the  $i^{th}$  outcome is  $M_i.\vec{P}$ . Then the probability of obtaining this outcome is simply the norm of this unnormalised state,  $|M_i.\vec{P}|$ and the corresponding normalized final state is  $M_i.\vec{P}/|M_i.\vec{P}|$ . A set  $\{M_i\}$  represents a valid operation if the following hold [16].

$$0 \le |M_i.\vec{P}| \le 1 \qquad \forall i, \vec{P} \in \mathscr{S} \tag{2.44a}$$

$$\sum_{i} |M_{i}.\vec{P}| = 1 \qquad \forall \vec{P} \in \mathscr{S}$$
(2.44b)

$$M_i.\vec{P} \in \mathscr{S} \quad \forall i, \vec{P} \in \mathscr{S}$$
 (2.44c)

This is the analogue of quantum Born rule for GPTs. Box world is a GPT where the state space  $\mathscr{S}$  consists of all normalized states  $\vec{P}$  whose entries are valid probabilities (i.e.,  $\in [0, 1]$ ) and satisfy the *no-signaling* constraints i.e., for a *N*-partite state  $\vec{P}$ , the marginal term

$$\sum_{a_i} P(a_1, ..., a_i, ..., a_N | X_1, ..., X_i, ..., X_N)$$

is independent of the setting  $X_i$  for all  $i \in \{1, ..., N\}^{18}$ 

When the GPT  $\mathbb{T}$  is box world, the conditions of Equations 2.44a-2.44c result in the characterization of measurements and transformations in the theory in terms of classical circuits or *wirings* as shown in [16]. We summarise the results of [16] characterising allowed transformations and measurements in box world and will only consider normalization-preserving transformations.

#### • Transformations:

- *Single system:* All transformations on single box world systems are relabellings of fiducial measurements or outcomes or a convex combination thereof.
- Bipartite system: Let X and Y be fiducial measurements performed on the transformed bipartite system with corresponding outcomes a and b, then all transformations of 2-gbit systems can be decomposed into convex combinations of classical circuits of the following form: A fiducial measurement  $X' = f_1(X, Y)$  is performed on the initial state of the first gbit resulting in the outcome a' followed by a fiducial measurement  $Y' = f_2(X, Y, a')$  on the initial state of the second gbit resulting in the outcome b'. The final outcomes are given as  $(a, b) = f_3(X, Y, a', b')$ , where  $f_1$ ,  $f_2$  and  $f_3$  are arbitrary functions.

be possible to prepare a probabilistic mixture of the two (for example, by tossing a coin).

<sup>&</sup>lt;sup>18</sup>This is in the spirit of relativistic causality since one would certainly expect that the input of one party does not affect the output of others when the are all space-like separated from each other.

#### • Measurements:

- *Single system:* All measurements on single box world systems are either fiducial measurements with outcomes relabelled or convex combinations of such.
- Bipartite system: All bipartite measurements on 2-gbit systems can be decomposed into convex combinations of classical circuits of the following form (Figure 2.5): A fiducial measurement X is performed on the initial state of the first gbit resulting in the outcome a' followed by a fiducial measurement Y = f(a') on the second gbit resulting in the outcome b'. The final outcome is a = f'(a', b'), where f and f' are arbitrary functions.

# 2.5 Causal structures and causal models

Cause-effect relationships between systems constrain the possible correlations that can be observed between them. These relationships are naturally represented using *causal structures* or directed graphs. Given a causal structure, a *causal model* provides a set of mathematical rules to define the possible correlations that are compatible with the causal structure. Additionally, it can also provide rules for how the correlations compatible with a causal structure change under external manipulations of the systems involved. There are different approaches for mathematically formalising these concepts in general physical theories, depending on what the nodes and edges of the causal graph correspond to, the compatibility condition required by the causal model and how external manipulations are described. For classical causal models, a common approach is that of Bayesian networks, pioneered by Judea Pearl [159]. This has been generalised to quantum and more general, post-quantum causal structures in [115]. We will mainly adopt this approach in the thesis, and will briefly discuss other approaches for quantum causal modelling at the end of Section 2.5.2.3.

**General definitions and terminology:** Throughout this thesis, we will only consider discrete and finite random variables, and corresponding causal structures with finite number of systems/nodes. More formally, we have the following definitions.

**Definition 2.5.1** (Causal structure). A causal structure is a directed graph  $\mathcal{G}$  over a set of nodes, some of which are labelled observed and the rest unobserved. Each observed node X corresponds to a classical random variable<sup>19</sup> of the same name, while unobserved nodes can correspond to classical, quantum or post-quantum systems. For two (observed or unobserved) nodes X and Y of  $\mathcal{G}$ , X is said to be a *cause* of a Y if and only if there exists a directed path from X to Y in  $\mathcal{G}$ .

The majority of the literature on causal structures as well as Chapters 4 and 5 concern the case of acyclic causal structures represented by directed acyclic graphs (DAGs). In Chapter 6, we

<sup>&</sup>lt;sup>19</sup>Observed variables may represent classical inputs (such as measurement settings) or outputs (such as measurement outcomes) of an experiment.

attempt to model cyclic causal structures as well, but an understanding of the acyclic case is a pre-requisite for this as well. Hence, we will focus on the acyclic case in this review.

Note that by the above definition of *cause*, X is said to be a cause of itself if and only if there is a directed path from X to itself in  $\mathcal{G}$ , which is not possible if the graph is acyclic. We will use *family relations* to describe causal relationships between nodes of a directed graph— for  $i \neq j$ , if  $X_i$  is a direct or indirect cause of  $X_j$ , then  $X_i$  is said to be an *ancestor* of  $X_j$  or equivalently,  $X_j$  a *descendant* of  $X_i$ . We will use either explicitly as  $\operatorname{anc}(X_i)$  or more compactly as  $X^{\downarrow}$  to denote the set of all ancestors of  $X_i$  in  $\mathcal{G}$  and similarly,  $\operatorname{desc}(X_i)$  or  $X_i^{\uparrow}$  for the set of all its descendants. If  $X_i$  is a direct cause of  $X_j$ , then  $X_i$  is said to be a *parent* of  $X_j$  or equivalently  $X_j$ , a *child* of  $X_i$ . We will use  $\operatorname{par}(X_i)$  or  $X_i^{\downarrow_1}$  and  $\operatorname{child}(X_i)$  or  $X_i^{\uparrow_1}$  to denote the sets of all parents and children respectively, of  $X_i$  in  $\mathcal{G}$ . If a node  $X_i$  has no parents in  $\mathcal{G}$ , then it is said to be *exogenous* with respect to that graph, and is called *endogenous* otherwise.

Figure 2.6 illustrates two causal structures that will be considered extensively Part I of this thesis, particularly in Chapters 4 and 5. These are the biparite Bell and Triangle causal structures, which we will denote by  $\mathcal{G}_B$  and  $\mathcal{G}_T$  respectively. Both the Bell [22, 57, 224, 122, 63, 169, 143, 15, 66, 34, 214] and Triangle [88, 47, 48, 33, 215, 178, 129] causal structures have been studied extensively and continue to be a subject of ongoing research, by virtue of being relatively small causal structures that are known to support non-classical correlations. We first outline the preliminaries of classical causal models [159] required for this thesis, and then discuss the generalisation to quantum and post-quantum causal structures, mainly following the approach of [115]. We now enunciate the following broad definition of a causal model.

**Definition 2.5.2** (General definition of a causal model). A causal model consists of a causal structure  $\mathcal{G}$ , and a joint distribution  $P_{X_1,...,X_n}$  over the set  $\{X_1,...,X_n\}$  of observed nodes of  $\mathcal{G}$  that is compatible with the causal structure according to a *compatibility condition*.

## 2.5.1 Classical causal models

The section is primarily based on the framework of classical causal models developed by Judea Pearl [159].

## 2.5.1.1 Directed graphs and compatible probability distributions

**Definition 2.5.3** ((Classical) causal structures). A causal structure is called classical and denoted  $\mathcal{G}^C$  if all its unobserved nodes correspond to classical random variables (denoted by the same name as the associated node).

Hence for classical causal structures, all nodes (observed and unobserved) correspond to classical random variables and we will use the same label to denote the node of the DAG as well as the



Figure 2.6: The bipartite Bell and Triangle causal structures: Two widely studied causal structures that are known to support a gap between correlations realisable through classical vs non-classical unobserved causes. Here observed nodes are circled while the uncircled nodes are unobserved. (a) The bipartite Bell causal structure: This corresponds to an information processing protocol involving two parties Alice and Bob where the observed nodes A and B represent the random variables corresponding their choice of input (measurement setting) while X and Y represent the random variables corresponding to their outputs (measurement outcome). The outcomes X and Y are obtained by measuring the respective halves of a shared system  $\Lambda$ , which may be classical, quantum or post-quantum. (b) The Triangle causal structure: Here, the three observed nodes X, Y and Z have unobserved, pairwise common causes A, B and C, but no joint common cause. This corresponds to a information processing protocol involving three parties who produce correlated data (X, Y and Z) by pre-sharing information pairwise, and never having interacted as a group.

random variable it represents. The framework of classical causal models [159] then specifies when a joint probability distribution over all nodes is said to be compatible with the causal structure.

**Definition 2.5.4** (Compatibility of distribution with  $\mathcal{G}^C$ ). A distribution  $P_{X_1,\ldots,X_n} \in \mathscr{P}_n$  over the random variables  $\{X_1,\ldots,X_n\}$  is said to be *compatible* with a classical causal structure  $\mathcal{G}^C$  (with these variables as nodes) if it satisfies the *causal Markov condition* i.e., the joint distribution decomposes as

$$P_{X_1,\dots,X_n} = \prod_{i=1}^n P_{X_i | X_i^{\downarrow_1}}, \tag{2.45}$$

where  $X_i^{\downarrow_1}$  denotes the set of all parent nodes of the node  $X_i$  in the DAG  $\mathcal{G}^{\mathbb{C}}$ .

A crucial concept relating to the joint distribution of a causal structure is *conditional independence* defined as follows.

**Definition 2.5.5** (Conditional independence). Given pairwise disjoint subsets of random variables X, Y and Z (where X and Y are non-empty) and a joint distribution  $P_{XYZ}$  over them, X and Y are said to be *conditionally independent* given Z, denoted  $X \perp Y | Z$  if

$$P_{XY|Z} = P_{X|Z} P_{Y|Z}.$$
 (2.46)

The relationship between the *Markov compatibility* condition of Definition 2.5.4 and conditional independence as defined above are given by the following Theorem (Theorem 1.2.7 in [159]).

**Theorem 2.5.1** (Pearl 2009). A distribution  $P \in \mathscr{P}_n$  is compatible with a classical causal structure  $\mathscr{G}^C$  over a node set  $\{X_1, ..., X_n\}$  if and only if every node  $X_i$  is conditionally independent from its non-descendants, denoted  $X_i^{\dagger}$  given its parents  $X_i^{\downarrow_1}$  in  $\mathscr{G}^C$  i.e.,  $\forall i \in \{1, ..., n\}$  $X_i \perp X_i^{\dagger} | X_i^{\downarrow_1}$ , or explicitly

$$P_{X_i X_i^{\dagger} | X_i^{\downarrow_1}} = P_{X_i | X_i^{\downarrow_1}} P_{X_i^{\dagger} | X_i^{\downarrow_1}}$$
(2.47)

The equivalent compatibility condition in terms of conditional independence provided by Theorem 2.5.1 is often referred to as *local Markov* or *parental Markov* condition in the literature. It is possible to read off the conditional independences satisfied by a Markov compatible distribution from the corresponding causal graph, rules for doing so were independently developed by Geiger [96] and Verma and Pearl [205]. The following graph theoretic notions are required to understand how this can be done.

**Definition 2.5.6** (Blocked paths). Let  $\mathcal{G}$  be a DAG in which X and Y are distinct nodes and Z be a set of nodes not containing X or Y. A path from X to Y is said to be *blocked* by Z if it contains either  $A \longrightarrow W \longrightarrow B$  with  $W \in Z$ ,  $A \longleftarrow W \longrightarrow B$  with  $W \in Z$  or  $A \longrightarrow W \longleftarrow B$  such that neither W nor any descendant of W belongs to Z.

In the following, three node subgraphs of the form  $A \longrightarrow W \longleftarrow B$  will be known as *colliders* where W is referred to as the collider node. Other three node subgraphs will be referred to as *non-colliders*, in particular  $A \longleftarrow W \longrightarrow B$  corresponds to the *common cause subgraph* where the node W is a common cause of A and B.

**Definition 2.5.7** (d-separation). Let  $\mathcal{G}$  be a DAG in which X, Y and Z are disjoint sets of nodes. X and Y are *d-separated* by Z in  $\mathcal{G}$ , denoted as  $(X \perp^d Y | Z)_{\mathcal{G}}$  (or simply  $X \perp^d Y | Z$  if  $\mathcal{G}$  is obvious from context) if every path from a variable in X to a variable in Y is *blocked* by Z, otherwise, X is said to be *d-connected* with Y given Z.

The concept of *d*-separation was developed independently in [96] and [205]. It follows from the above definition that d-separation is a symmetric relation in the first two arguments (as is conditional independence) i.e.,  $X \perp^d Y | Z$  is equivalent to  $Y \perp^d X | Z$ . When Z is the empty set  $\emptyset$ , we will simply denote this as  $X \perp^d Y$ , and similarly for conditional independences (which will be called *independences* when the conditioning set Z is empty). For arbitrary disjoint subsets X, Y and Z of the nodes, d-separation can be used to determine whether X and Y are conditionally independent given Z for a distribution compatible with the causal structure (according to Definition 2.5.4). The following result which is Theorem 1.2.5 of [159], and was originally presented in [205] illustrates this.

**Theorem 2.5.2** (Verma and Pearl 1988). Let  $\mathcal{G}^C$  be a classical causal structure and let X, Yand Z be pairwise disjoint subsets of nodes in  $\mathcal{G}^C$ . If a probability distribution P is compatible with  $\mathcal{G}$  (according to Definition 2.5.4), then the d-separation  $X \perp^d Y|Z$  implies the conditional independence  $X \perp Y|Z$ . Conversely, if all distributions P compatible with  $\mathcal{G}^C$  satisfy the conditional independence  $X \perp Y|Z$ , then the corresponding d-separation relation  $X \perp^d Y|Z$  holds.



Figure 2.7: Example of fine-tuned correlations compatible with a causal structure: Even though S is a cause of H, it is possible for the correlations  $p_{SEH}$  to be fine-tuned such that the positive dependence  $S \xrightarrow{+} E \xrightarrow{+} H$  exactly cancels out the negative dependence  $S \xrightarrow{-} H$  resulting in S and H being uncorrelated (see Example 2.5.1

The forward direction of the above Theorem is referred to as the soundness of d-separation and the reverse direction as the completeness of d-separation for determining conditional independences. Note that for the latter, it is important that all compatible distributions satisfy the conditional independence. It is possible for a distribution P that is compatible with a graph  $\mathcal{G}$ to satisfy a conditional independence  $X \perp Y | Z$  even when the d-separation relation  $X \perp^d Y | Z$ does not hold in  $\mathcal{G}$ . A causal model defined by such a distribution P and causal structure  $\mathcal{G}$  is said to be unfaithful or fine-tuned. Therefore for a faithful causal model, the only conditional independence relations are those that can be read off from the d-separation relations in the associated causal graph. For a faithful causal model, it can be shown that all the conditional independences in the distribution are implied by the n conditional independence constraints (one for each node) given by the parental/local Markov condition (Equation (2.47)), and can be derived from these constraints and standard probability calculus based on Bayes' rule. In general, not all of these n constraints may be required. The following example (taken from [186]) illustrates a compatible distribution that is unfaithful with respect to its causal graph.

**Example 2.5.1** (A fine-tuned causal model). Consider the causal structure of Figure 2.7 where + or - on an edge indicates whether the nodes connected by that edge are positively or negatively correlated as a result of that causal influence. Note that the Markov condition of Equation (2.45) does not impose any constraints on joint distributions  $P_{SEH}$  compatible with this causal structure, and neither does d-separation. Any probability distribution  $P_{SEH}$  can be expressed as  $P_{SEH} = P_S P_{E|S} P_{H|ES}$  using Bayes' rules, which is the same factorization obtained by applying the Markov condition to this graph. Further, S influences H through two causal mechanisms—a direct, negative causal influence and an indirect, positive influence. It is therefore possible to have a compatible distribution  $P_{SEH}$  in which these two causal mechanisms exactly cancel each other out such that S and H appear to be uncorrelated i.e.,  $P_{SH} = P_S P_H$ , even though S and H are clearly d-connected in the graph.

Note that such examples are *fine-tuned* in the sense that the causal influences will have to conspire in a very specific way in order to just cancel out, and even slight deviations in the strength of the correlations will destroy the apparent independence. We will consider such examples in further detail (including explicit distributions) in Chapter 6. In the rest of this section, we will focus only on faithful correlations and causal models. We now provide an illustrative example for the concepts of d-separation and conditional independence.

**Example 2.5.2** (d-separation and conditional independence). Consider the causal structure of



Figure 2.8: Causal structure of Example 2.5.2

Figure 2.8, where all nodes are observed. Here, the node T (and the corresponding sub-graph  $S \longrightarrow T \longleftarrow U$ ) is a collider (and all remaining nodes are non-colliders), and V is a common cause of U and Y. We now consider d-separation relations involving different conditioning sets Z. If Z is the empty set,  $X \perp^d Y | Z$  (or simply  $X \perp^d Y$ ) because the path between X and Y is blocked by the collider T that does not belong to  $Z = \emptyset$ . If  $\{T, W\}$  or a subset thereof is contained in Z, then X and Y are not d-separated (or equivalently, they are d-connected) by Z since the path contains the node T that either belongs to Z or has a descendant W that belongs to Z. Further, X and T as well as T and Y are d-connected by the empty set  $Z = \emptyset$ . The former become d-separated when Z contains either one or both of R and S since these block the directed path from Z to T. Similarly, T and Y become d-separated by Z if it contains one or both of U and V. By the soundness of d-separation, this implies the conditional independence relations  $X \perp Y$ ,  $X \perp T \mid \{R, S\}$ ,  $T \perp Y \mid \{U, V\}$ . By finding the set of non-descendants and parents for all 8 nodes, we have the following set of 8 conditional independences corresponding to Equation (2.47). For a faithful causal model over the causal structure 2.8, all conditional independence relations in every compatible distribution (including those mentioned previously) are implied by these 8 relations.

$$X \perp \{U, V, Y\}, R \perp \{U, V, Y\}|X, S \perp \{U, V, Y\}|X, T \perp \{X, R, V, Y\}|\{S, U\}, W \perp \{X, R, S, U, V, Y\}|T, U \perp \{X, R, S, Y\}|V, V \perp \{X, R, S\}, Y \perp \{X, R, S, T, W, U\}|V.$$
(2.48)

**d-separation and entropic causal constraints:** An equivalent statement of Theorem 2.5.2 can be obtained by encoding conditional independences in terms of the entropies (rather than the probability distribution) over the nodes. For a classical causal structure  $\mathcal{G}$  with pairwise disjoint subsets of nodes X, Y and  $Z, X \perp Y | Z$  is equivalent to the entropic constraint I(X : Y|Z) = 0, where I(X : Y|Z) is the Shannon conditional mutual information (Definition 2.3.3). Then, X and Y are d-separated by Z in  $\mathcal{G}$  if and only if I(X : Y|Z) = 0 for all distributions compatible with  $\mathcal{G}$  [159]. The complete set of d-separation conditions give all the conditional independence relations implied by the DAG (by soundness of d-separation). In the case of



Figure 2.9: Two node causal structures with same set of compatible distributions: All of the above causal structures contain two observed nodes X and Y. In all cases, neither the Markov compatibility condition (2.45) nor d-separation imply any constraints on the marginal distribution  $P_{XY}$  over the observed nodes. Therefore any 2-variable distribution  $P_{XY}$  is compatible with all 5 of these graphs, and the causal structure cannot be inferred directly from these correlations. However, if we were given, or could deduce additional information regarding the results of active interventions on X and Y, we would be able to rule out some of these explanations. For example, if forcibly setting X to some value x (without changing anything else), resulted in a change in the distribution over Y, we would say that X is a cause of Y. In this case, the opposite direct cause explanations (b) and (e) as well as the purely common cause explanation (c) would be ruled out, leaving us with (a) and (d) as possible explanations.

Shannon entropy for a DAG with n nodes these are all implied by the n constraints

$$I(X_i:X_i^{\ddagger}|X_i^{\downarrow_1}) = 0 \quad \forall i \in \{1,\dots,n\}.$$

$$(2.49)$$

In other words, a distribution over n variables satisfies Equation (2.47) and hence Equation (2.45) (by Theorem 2.5.1) if and only if it satisfies Equation (2.49).

#### 2.5.1.2 Interventions and counterfactuals

Given a causal structure, we can apply Definition 2.5.4 to obtain the set of all joint distributions compatible distributions with it. Causal inference concerns the reverse problem— given a compatible distribution (and possibly some additional information), can we infer the causal structure that it is compatible with? It is not possible to infer an underlying causal structure from correlations alone: correlations are symmetric while causal relationships are directional. For example, if two variables X and Y are correlated, Reichenbach's principle [176] asserts that either X must be a cause of Y, Y must be a cause of X, X and Y share a common cause or any combination thereof and any distribution  $P_{XY}$  over these variables is compatible with all five of these causal structures illustrated in Figure 2.9. Hence these causal explanations cannot be distinguished immediately using observed correlations alone, and additional information/deductions are required in order to infer the underlying causal structure. Intuitively, we can argue that if actively intervening or "doing" something only to X changed something about Y, then we definitely know that X is a cause of Y. This intuition is formalised in terms of interventions and do-conditionals [159], which in addition to the observed correlations allow for the inference of the underlying causal structure in several scenarios that cannot be identified from the correlations alone. We will follow the augmented graph approach (Section 3.2.2 of [159]) to define interventions and do-conditionals and only consider the case of perfect or ideal interventions in this thesis.

First, consider a causal model associated with a causal structure  $\mathcal{G}$  over a set  $\{X_1, ..., X_n\}$  of nodes, all of which are observed. External intervention on a node  $X_i$  can be described using an augmented graph  $\mathcal{G}_{I_{X_i}}$  which is obtained from the original graph  $\mathcal{G}$  by adding a node  $I_{X_i}$  and an edge  $I_{X_i} \longrightarrow X_i$  (with everything else unchanged). The intervention variable  $I_{X_i}$  can take values in the set  $\{idle, \{do(x_i)\}_{x_i \in X_i}\}$ , where  $I_{X_i} = idle$  corresponds to the case where no intervention is performed (i.e., the situation described by the original causal model) and  $I_{X_i} = do(x_i)$  forces  $X_i$  to take the value  $x_i$  by cutting off its dependence on all other parents. Hence the parents of the node  $X_i$  in the original and augmented graphs are related as  $\operatorname{par}_{\mathcal{G}_{I_{X_i}}}(X_i) = \operatorname{par}_{\mathcal{G}}(X_i) \cup I_{X_i}$ . We will denote  $\operatorname{par}_{\mathcal{G}_{I_{X_i}}}(X_i)$  simply as  $\operatorname{par}_I(X_i)$  and the  $\operatorname{par}_{\mathcal{G}}(X_i)$  as  $\operatorname{par}(X_i)$ , in short-hand. The conditional probability of  $X_i = x_i$  conditioned on its parents  $\operatorname{par}_{\mathcal{G}}(X_i)/\operatorname{par}_I(X_i)$  taking the value  $p_{X_i}/p_{I_{X_i}}$  is given as.

$$P(x_i|p_{I_{X_i}}) \coloneqq \begin{cases} P(x_i|p_{X_i}), & \text{if } I_{X_i} = idle \\ 0, & \text{if } I_{X_i} = do(x'_i) \text{ and } x'_i \neq x_i \\ 1, & \text{if } I_{X_i} = do(x'_i) \text{ and } x'_i = x_i \end{cases}$$
(2.50)

From this, we see that whenever  $I_{X_i} \neq idle$ ,  $X_i$  no longer depends on its original parents par $(X_i)$ . Therefore, conditioned on  $I_{X_i} \neq idle$ , it is illustrative to consider a new graph which we denote by  $\mathcal{G}_{do(X_i)}$  that represents the post-intervention causal structure after a non-trivial intervention has been performed. The causal graph  $\mathcal{G}_{do(X_i)}$  is obtained by cutting off all incoming arrows to  $X_i$ except the one from  $I_{X_i}$  in the causal graph  $\mathcal{G}_{I_{X_i}}$ , with everything else unchanged. An example of the graphs  $\mathcal{G}$ ,  $\mathcal{G}_{I_{X_i}}$  and  $\mathcal{G}_{do(X_i)}$  is given in Figure 2.10. Then the effect of an intervention setting  $X_i = x'_i$ , i.e., performing  $do(x'_i)$  is to transform the original probability distribution  $P(X_1 = x_1, ..., X_n = x_n)$  into a new probability distribution  $P(X_1 = x_1, ..., X_n = x_n | do(X_i = x_i))$ (the *do-conditional*) given as

$$P(x_1, ..., x_n | do(x_i)) \coloneqq P_{\mathcal{G}_{do(X_i)}}(x_1, ..., x_n | I_{X_i} = do(x_i)),$$
(2.51)

where  $P(x_1, ..., x_n, I_{X_i} = do(x_i))$  is compatible with the graph  $\mathcal{G}_{do(X_i)}$  and Equation (2.50), which implies that  $P(x_1, ..., x'_i, ..., x_n | do(x_i)) = 0$  if  $x'_i \neq x_i$ . Applying the Markov compatibility condition (Definition 2.5.4) to the original, pre-intervention graph  $\mathcal{G}$  implies that all distributions compatible with this graph must satisfy  $P_{X_1,...,X_n} = \prod_{j=1}^n P_{X_j | par(X_j)}$ . In the post-intervention graph  $\mathcal{G}_{do(X_i)}$ , the only causal relations that change are those involving the intervened node  $X_i$  where the set of parents of  $X_i$  in  $\mathcal{G}_{do(X_i)}$  is the single element set  $\{I_{X_i}\}$ , and the parent sets of all other nodes remain the same as those of  $\mathcal{G}$ . Then any distribution  $P_{X_1,\ldots,X_n,I_{X_i}}$ with  $I_{X_i}$  taking values in  $\{do(x_i)\}_{x_i \in X_i}$  (i.e., not idle), that is compatible with  $\mathcal{G}_{do(X_i)}$  satisfies  $P_{X_1,\ldots,X_n|I_{X_i}} = \prod_{j=1,j\neq i}^n P_{X_j|\text{par}(X_j)} P_{X_i|I_{X_i}}$ , for any distribution over  $I_{X_i}$ . Noting that  $P_{X_i|I_{X_i}}$  is the deterministic distribution which is 1 whenever  $X_i$  takes the same value set by  $I_{X_i}$  and 0 otherwise, this gives the following transformation between the do-conditional (Equation (2.51)) and the original distribution,

$$P(x_1, ..., x_n | do(x'_i)) = \begin{cases} \frac{P(x_1, ..., x_n)}{P(x_i | \text{par}(x_i))}, & \text{if } x'_i = x_i \\ 0, & \text{if } x'_i \neq x_i. \end{cases}$$
(2.52)

Importantly, this allows us to decide, given two nodes X and Y, whether or not X is a cause of Y— if there exists a value  $x \in X$  such that  $P(y|do(x)) \neq P(y)$  for some  $y \in Y$ , then X is a cause of Y (or equivalently, there is a directed path from X to Y). Otherwise, we can conclude that X is not a cause of Y or equivalently that there is no directed path from X to Y (if the causal model is faithful)<sup>20</sup>. This formalises the intuition explained in Figure 2.9, where checking whether an active intervention  $(I_X \neq idle)$  on X changes the distribution over Y allowed us to rule out 3 of 5 causal explanations. As we will see in the following paragraphs, checking whether the do-conditional P(y|do(x)) differs from the regular conditional P(y|x), can allow us to pick out a unique causal structure in this case.

**Interventions on subsets of nodes:** This procedure naturally extends to interventions on subsets of the nodes. If a simultaneous intervention is performed on a subset X of the nodes  $\{X_1, ..., X_n\}$ , an intervention variable  $I_{X_i}$  will be introduced for each element  $X_i$  of the subset X, along with the corresponding edge  $I_{X_i} \to X_i$ . Then conditioning on each of the  $I_{X_i}$  for  $X_i \in X$  being *idle*, the original graph is obtained and conditioning on  $I_{X'} \neq idle$  for  $X' \subseteq X$  (i.e., a "do" operation has been performed on every node in X'), the corresponding graph  $\mathcal{G}_{do(X')}$  is defined by cutting off all incoming edges (except  $I_{X_i} \to X_i$  for all  $X_i \in X'$ ) in  $\mathcal{G}$ . Then the do conditional  $P(x_1, ..., x_n | do(x'))$  and its relation to the original distribution  $P(x_1, ..., x_n)$  are defined analogously, using Equations (2.51) and (2.52) for the graph  $\mathcal{G}_{do(X')}$ .

The importance of the transformation (2.52) is that it allows for *counter-factual reasoning* by mathematically relating the post-intervention distribution to the pre-intervention distribution. It tells us that if all the nodes in a causal structure are observed, then we can deduce the result of any intervention (possibly on a subset of nodes) just by passively observing the correlations over the variables and using mathematical manipulations, without physically performing any interventions. This ability to deduce the post-intervention distribution only from the observed distribution in a consistent manner is known as *identifiability*. Identifiability of interventional effects is of crucial practical use, especially in situations where an actually intervention may be physically (or ethically) forbidden, for example testing the effect of certain drugs and treatments on humans. In causal structures that include unobserved nodes, this kind of counter-factual

<sup>&</sup>lt;sup>20</sup>Note that in unfaithful causal models, it is possible to have P(y|do(x)) = P(y) for all  $x \in X$  and  $y \in Y$  even when X is a cause of Y, and checking for the violation of this condition only provides a sufficient test of causation but not a necessary one. This point is crucial for the results of Chapter 6, which concern unfaithful causal models.

reasoning may not always be possible i.e., the effect of interventions may not be identifiable from the observed distributions, as explained below.

Interventions in the presence of unobserved nodes: For a causal structure containing unobserved nodes, interventions can only be performed on the observed nodes. For a classical causal structure  $\mathcal{G}^C$ , the unobserved nodes are also random variables. Let  $\{X_1, ..., X_n\}$ and  $\{\Lambda_1, ..., \Lambda_m\}$  denote the set of observed and unobserved nodes of  $\mathcal{G}^C$  respectively. Any observed distribution  $P_{X_1,...,X_n}$  compatible with  $\mathcal{G}$  can be obtained by applying the Markov condition (2.45) to the hypothetical distribution over all nodes and summing over all possible values of the unobserved nodes i.e.,

$$P_{X_1,\dots,X_n} = \sum_{\Lambda_1,\dots,\Lambda_m} \prod_{i=1}^n P_{X_1|\operatorname{par}(X_i)} \prod_{j=1}^m P_{\Lambda_j|\operatorname{par}(\Lambda_j)}.$$
(2.53)

Considering interventions only on observed nodes, Equations (2.51) and (2.52) can be derived analogously using this compatibility condition. However there are two important distinctions to be noted. Firstly, the observed distribution  $P_{X_1,...,X_n}$  is in general not Markov (and hence not compatible) with respect to the sub-graph involving only the observed nodes and edges connecting them. However the overall distribution over observed and unobserved nodes is Markov with respect to the complete DAG involving all these nodes if we assume that graph accounts for all the unobserved common causes and corresponds to a classical causal structure. Secondly, the post-intervention distribution may not be identifiable from the observed, preintervention distribution alone. This is because, an intervened node  $X_i$  may contain unobserved parents in the original causal structure and Equation (2.52) would then require division by a quantity that cannot be obtained from the observed distribution alone. More explicitly, the following example describes a non-identifiable causal structure involving an unobserved node.

**Example 2.5.3** (A non-identifiable causal structure). Consider the causal structure of Figure 2.9d with X, Y observed and  $\Lambda$  unobserved. The augmented and post-intervention causal structures corresponding to an intervention on X are illustrated in Figure 2.10. Observed distributions compatible with the original causal structure are given as  $P_{XY} = \sum_{\Lambda} P_{\Lambda XY} = \sum_{\Lambda} P_{\Lambda} P_{X|\Lambda} P_{Y|X\Lambda}$ . Note that any distribution  $P_{XY}$  can be written in this form using Bayes' rule and marginalising over some variable  $\Lambda$ . The post-intervention distribution  $P_{XY|I_X}$  (with  $I_X \neq idle$ ) is compatible with the graph  $\mathcal{G}_{do(X)}$  and is given as  $P_{XY|I_X} = \sum_{\Lambda} P_{\Lambda} P_{X|I_X} P_{Y|X\Lambda}$ . Equation (2.52) in this case corresponds to

$$P(xy|do(x')) = \begin{cases} \sum_{\lambda \in \Lambda} \frac{P(\lambda xy)}{P(x|\lambda)} = \sum_{\lambda \in \Lambda} P(\lambda) P(y|x\lambda), & \text{if } x' = x\\ 0, & \text{if } x' \neq x. \end{cases}$$
(2.54)

Note that the transformation involves division by a value corresponding to the distribution  $P_{X|\Lambda}$  which cannot be determined from the observed distribution  $P_{XY}$  alone. Hence the effect of the intervention is not identifiable for this causal structure. We will discuss identifiability of causal structures in further detail in Chapter 6.

In the above example, the do-conditional P(y|do(x)) is simply the non-zero expression in the above equation i.e.,  $P(y|do(x)) = \sum_{\lambda \in \Lambda} P(\lambda)P(y|x\lambda)$ . The regular conditional P(y|x) is obtained using the Markov compatibility condition as  $P(y|x) = \frac{\sum_{\lambda \in \Lambda} P(\lambda)P(y|x\lambda)}{P(x)}$ . As long as



Figure 2.10: Pre-intervention, augmented and post-intervention causal structures: Taking the original, pre-intervention causal structure to be that of Figure 2.9d (repeated here in (a) for completeness), parts (b) and (c) of this figure illustrate the augmented and post-intervention causal structures corresponding to an intervention on X in the causal structure of (a). In the former, the variable  $I_X$  can take values in the set  $\{idle, \{do(x)\}_{x \in X}\}$  while in the latter, it can only take the values  $\{do(x)\}_{x \in X}$  corresponding to an active intervention. Conditioned on  $I_X = idle$ , we effectively obtain the original causal structure (a) which corresponds to no intervention being performed, as specified by Equation (2.51).

 $P(x|\lambda) \neq P(x)$  for some values  $x \in X, \lambda \in \Lambda$ , the two conditionals will be different. Note that this must indeed hold i.e., X and  $\Lambda$  cannot be independent if the distribution is faithful, since they are d-connected in the causal structure. Therefore, for faithful causal models over the causal structure 2.9d, there exist values  $x \in X, y \in Y$  such that  $P(y|do(x)) \neq P(y|x)$ . On the other hand, one can easily check that for the causal structure 2.9a, P(y|do(x)) = P(y|x) always holds— the intuition being that, in this case X is a parentless (i.e., exogenous) node, hence the intervention does not change any of the causal relations that were present in the original graph. In general, interventions on an exogenous node X does not change the distribution, and whether or not X is a cause of Y in this case can simply be decided by checking whether or not X and Y are correlated i.e., whether  $P_{Y|X} = P_Y$ . In conclusion, while the do-conditional P(y|do(x))for the causal structure 2.9d is not identifiable from the original observed distribution P(xy), it can be determined experimentally by physically performing the intervention. Comparing this experimentally obtained estimate of the do-conditional with the original distribution will allow us to uniquely identify one among the 5 causal structures of Figure 2.9, if we assume that the causal model is faithful.

**Remark 2.5.1.** Note that there is a slight but harmless abuse of notation in this section. The intervention variable  $I_X$  takes different values in the augmented graph  $\mathcal{G}_{I_X}$  and the postintervention graph  $\mathcal{G}_{do(X)}$ . It takes values in  $\{idle, \{do(x)\}_{x\in X}\}$  and  $\{do(x)\}_{x\in X}$  in the two cases respectively. We nevertheless use the variable name to also denote the set of possible values it takes, but will be careful to specify which graph we are referring to whenever this happens.

## 2.5.2 Non-classical causal models

A typical laboratory experiment uses devices that take classical inputs (settings of knobs or buttons) and produce classical outputs (reading of a pointer or numbers on a screen). This classical, observed data may however be produced through quantum mechanisms. Bell [22] famously showed that local measurements on a shared quantum system can lead to stronger correlations than those on any shared classical system. In recent years, Wood and Spekkens [224] have reformulated Bell's Theorem in the language of Bayesian networks [159], showing that quantum correlations cannot be explained by faithful classical causal models, and therefore the framework of [159] must be suitably generalised for defining quantum causal models<sup>21</sup>. More generally, one can also consider post-quantum systems such as non-local or non-signaling boxes [201, 170], that produce stronger-than-quantum correlations. These can be seen as elements of a more general class of theories, generalised probabilistic theories (GPTs, see Section 2.4), of which quantum and classical theories emerge as particular cases. Hensen, Lal and Pusey (HLP) [115] were the first to define a generalised causal model that allows unobserved nodes to be either classical, quantum or more generally, GPT. This enables a characterisation of the theory-independent aspects of causal models and allows us to derive bounds on the correlations realisable in causal structures depending on the nature of their unobserved nodes, which has foundational as well as practical applications.

#### 2.5.2.1 Generalised causal structures and compatibility

**Definition 2.5.8** (Generalised causal structure). A generalised causal structure, denoted  $\mathcal{G}^G$  is a causal structure in which every observed node is associated with a classical random variable and every unobserved node is associated with a system belonging to a generalised probabilistic theory. Generalised causal structures include quantum  $\mathcal{G}^Q$  and classical causal structures  $\mathcal{G}^C$ . In the former case, each unobserved node is associated with a quantum system, and in the latter case, each unobserved node is associated with a classical random variable.

For a classical causal structure  $\mathcal{G}^C$ , the Markov factorization condition (2.45) provides a compatibility condition for a joint distribution with the causal structure. In case there are unobserved nodes, the observed distribution is simply obtained by taking a joint distribution over all the nodes that factorises according to the Markov condition, and then marginalising over the unobserved variables. If the unobserved systems are non-classical, this does not work since a joint distribution over all the nodes may no longer exist<sup>22</sup> and a more general compatibility condition that takes into account quantum and GPT mechanisms is required.

 $<sup>^{21}</sup>$ Unless one is willing to accept fine-tuned classical causal explanations that involve hidden influences propagating superluminally or retrocausally.

<sup>&</sup>lt;sup>22</sup>Quantum and GPT systems do not co-exist with the outcomes of measurements performed on them, therefore one cannot in general assign a joint probability distribution over quantum/GPT nodes and their children.

**Post-quantum/GPT causal structures:** The operational probabilistic framework of [52] provides a method of representing operations and transformations on GPT systems using circuit-like diagrams. Elements of this generalised circuit are called *tests*, which take as input a system and produce as output another system, and a classical outcome variable. Inputs and outputs of different tests are connected through *wires* which carry the associated systems along them. If a test has a trivial input system, then it is called a *preparation test* and if it has a trivial output system, it is called an *observation test*. Any test from the trivial system to itself is a probability distribution, hence probabilities can be generated by composing tests using wires. In the HLP framework [115], every node of a generalised causal structure is associated with a test. In the case of observed nodes, the output system of the test is simply a classical random variable and for unobserved nodes, this corresponds to a GPT system. Further, the systems described by the operational probabilistic framework [52] are *causal* in the sense that an outcome at an earlier time cannot depend on the choice of operation performed on the system at a later time. Then a distribution P over the observed nodes of a generalised causal structure  $\mathcal{G}^{G}$ is said to be *compatible* with  $\mathcal{G}^G$  if there exists a causal operational probabilistic theory, a test corresponding to each node and a system corresponding to each edge of  $\mathcal{G}$  in the theory that generate the observed distribution P. The formula for generating P in this manner is analogous to the Markov condition (2.45), where, instead of a product of conditional distributions  $P_{X_i|X_i^{\downarrow_1}}$ , we have a product of tests that map the unobserved parents of a node to its unobserved children, conditioned on the observed parents. This is called the generalised Markov condition, and the corresponding generalised causal structure is represented by a directed acyclic graph or a GDAG. The set of all distributions compatible with a generalised causal structure  $\mathcal{G}^G$  will be denoted by  $\mathscr{P}(\mathcal{G}^G)$ . Note that by construction these are distributions over observed nodes only.

Quantum causal structures: A GPT of particular interest is quantum theory and the causal structure framework of [115] includes quantum causal structures as a special case. There are however several frameworks for describing quantum causal structures that are compatible with each other, but differ essentially on how nodes and edges of the causal structure are identified with quantum systems and transformations. We will mainly stick to the description that follows from the generalised causal structures of [115], to provide a unified picture, but use a language and notation similar to that of [215] for convenience. We will discuss the relation to other approaches at the end of this section. For a quantum causal structure  $\mathcal{G}^Q$ , every exogenous node A i.e., those without any incoming edges correspond to a preparation of a density matrix  $\rho_A \in \mathscr{S}(\mathscr{H}_A)$ . In the case of observed nodes,  $\rho_A$  is classical. This can be seen as a preparation test, where the trivial input system is associated with the one dimensional Hilbert space  $\mathcal{H} = \mathbb{C}$ . Each directed edge corresponds to a quantum system and hence to a Hilbert space, which we will label by the starting and ending node of the edge. For example, if a node A only has two children Y and Z, then the Hilbert space of A will be factorised as  $\mathscr{H}_A = \mathscr{H}_{A_Y} \otimes \mathscr{H}_{A_Z}$ , where the subsystem  $\mathscr{H}_{A_Y}$  corresponds to the edge from A to Y, and  $\mathscr{H}_{A_Z}$  to the edge from A to Z. If A is a classical node, then the two factors  $A_Y$  and  $A_Z$  are taken to be copies of A, since classical information can be copied.

As in the HLP framework, each node is labelled by its output state which is classical for observed

nodes and quantum for unobserved nodes in  $\mathcal{G}^Q$ . Every observed node X represents the output statistics of a measurement performed on the incoming systems. If this node has unobserved parents then X can be seen as the result of a quantum measurement (implemented through a POVM) on those incoming quantum states, where the choice of measurement can possibly depend on the classical value of other observed parents of X. If X only has observed parents, then it can be seen as the output as a stochastic map on the incoming classical random variables. More generally, tests correspond to CPTP maps that map the joint state of all incoming edges to the joint state of all outgoing edges. Preparations, measurements (POVMs) and arbitrary classical stochastic maps can be seen as special cases of CPTP maps. A distribution, P over the observed nodes of a causal structure  $\mathcal{G}^Q$  is *compatible* with  $\mathcal{G}^Q$  if there exists a quantum state labelling each unobserved node (with subsystems for each unobserved edge) and CPTP maps, i.e., preparations or quantum transformations for each unobserved node as well as POVMs or stochastic maps for each observed node, that allow for the generation of P in accordance with the Born rule. We will denote the set of all observed distributions compatible with a quantum causal structure  $\mathcal{G}^Q$  by  $\mathscr{P}(\mathcal{G}^Q)$ , and for a classical causal structure  $\mathcal{G}^C$  by  $\mathscr{P}(\mathcal{G}^C)$ .

**Theory dependent and theory-independent aspects:** We already noted that in causal structures with non-classical nodes, a joint distribution over all nodes may no longer be defined. Further in classical causal structures, it is implicitly assumed that the complete information about a node is transmitted to each of its children. Then each node receives complete information about all its parents and performs a classical stochastic map to obtain the output. The stochastic nature of this map is taken to arise from lack of knowledge about additional "error variables", and once the parents and the error variables are given, the map corresponds to a deterministic function<sup>23</sup>. These aspects are not problematic in the classical case since classical information can be copied. However arbitrary states of quantum and GPT systems cannot be copied due to the *no-cloning theorem*. Hence, after a measurement on such a system, the measurement outcome and the initial state do not co-exist and one can no longer condition on unobserved systems in the same manner<sup>24</sup>. Neither can complete information about the state of an unobserved node be transmitted to all its children. Further, arbitrary quantum or GPT maps cannot be reduced to deterministic functions by including additional information. These theory-dependent differences mean that the sets of compatible correlations  $\mathscr{P}(\mathcal{G}^C), \ \mathscr{P}(\mathcal{G}^Q),$ and  $\mathscr{P}(\mathcal{G}^G)$  for a causal structure  $\mathcal{G}$  need not coincide in general. Causal structures that support such a gap between the sets of classical and non-classical correlations are termed as interesting causal structures, and [115] identify certain necessary conditions for a causal structure to be interesting. To do so, they prove the following *theory-independent* property of generalised causal structures, which is an analogue of Theorem 2.5.2, showing that the d-separation

 $<sup>^{23}</sup>$ Introducing suitable error variables and expressing the functional dependence of each node on its parents through structural equations is the basis of the structural causal model approach for classical causal structures. This is compatible with the Bayesian networks approach discussed in Section 2.5.1, and can been seen as providing underlying classical mechanisms to the probability assignments. Both approaches are covered in Pearl's book [159].

 $<sup>^{24}</sup>$ The quantum causal models of [6, 132] tackle this issue by introducing the notion of a conditional quantum state of an output conditioned on an earlier input. These are interpreted not as quantum states in the usual sense (since they correspond to systems at different times that do not coexist), but as a mathematical representations of the quantum channel transforming the input to the output system.

condition applies to classical, quantum and post-quantum causal structures alike. Identifying such theory-independent aspects is also useful because for instance, there is no consensus on the representation of states and transformations in arbitrary GPTs, even though this exists for specific GPTs such as *box world* (See Section 2.4).

**Theorem 2.5.3** (Hensen, Lal, Pusey 2014). Let  $\mathcal{G}^G$  be a generalised causal structure and let X, Y and Z be pairwise disjoint subsets of observed nodes in  $\mathcal{G}^G$ . If a probability distribution P is compatible with  $\mathcal{G}^G$ , then the d-separation  $X \perp^d Y | Z$  implies the conditional independence  $X \perp Y | Z$ . Conversely, if for every distribution P compatible with  $\mathcal{G}^G$ , the conditional independence  $X \perp Y | Z$  holds, then the d-separation  $X \perp^d Y | Z$  also holds in  $\mathcal{G}^G$ .

Using this, necessary conditions for a causal structure  $\mathcal{G}$  to be interesting are found in [115] by first identifying sufficient conditions under which the only restrictions on  $\mathscr{P}(\mathcal{G}^C)$  are those implied by d-separation, and then using Theorem 2.5.3 to conclude that such causal structures must be uninteresting i.e.,  $\mathscr{P}(\mathcal{G}^C) = \mathscr{P}(\mathcal{G}^Q) = \mathscr{P}(\mathcal{G}^G)$ . Simple examples of causal structures that satisfy this equivalence and are hence uninteresting, are those of Figures 2.9c-2.9e. In these causal structures, there are no restrictions implied by d-separation on the observed distribution  $P_{XY}$  and hence arbitrary distributions are compatible with these causal structures irrespective of the nature of the unobserved node  $\Lambda$ . Examples of causal structures that do not satisfy this equivalence and are hence *interesting* are the bipartite Bell ( $\mathcal{G}_B$ ) and Triangle ( $\mathcal{G}_T$ ) causal structures illustrated in Figure 2.6. In the former case, d-separation implies the independence of the inputs A and B as well as the no-signaling conditions  $(X \perp B \mid A \text{ and } Y \perp A \mid B)$  which are satisfied by all observed distributions  $\mathscr{P}(\mathcal{G}_B^C)$ ,  $\mathscr{P}(\mathcal{G}_B^Q)$  and  $\mathscr{P}(\mathcal{G}_B^G)$ . However,  $\mathscr{P}(\mathcal{G}_B^C)$  satisfy additional constraints given by Bell inequalities, which are not satisfied by all distributions in  $\mathscr{P}(\mathcal{G}_B^Q)$  and  $\mathscr{P}(\mathcal{G}_B^G)$ . In fact, there is also a gap between the sets  $\mathscr{P}(\mathcal{G}_B^Q)$  and  $\mathscr{P}(\mathcal{G}_B^G)$ , the former satisfies Tsirelson's inequalities while the latter does not. Example 2.5.4 describes the sets  $\mathscr{P}(\mathcal{G}_B^C)$  and  $\mathscr{P}(\mathcal{G}_B^C)$  for the bipartite Bell causal structure  $\mathcal{G}_B$ . That  $\mathcal{G}_T$  is also an interesting causal structure can be witnessed by suitably embedding the Bell scenario  $\mathcal{G}_B$  in the Triangle  $\mathcal{G}_T$  such that non-classicality of correlations in  $\mathcal{G}_T$  follows from that of  $\mathcal{G}_B$  [88] (this is explained in Section 4.3.1 of Chapter 4). The simplest causal structure that is interesting i.e., supports a classical-quantum gap is the *instrumental causal structure* of [158, 45, 203], which is essentially the causal structure of Figure 2.9d (which by itself is not interesting) but with an additional node and edge  $Z \longrightarrow X$ .

**Example 2.5.4** (Sets of compatible correlations in the bipartite Bell causal structure  $\mathcal{G}_B$ ). In the classical causal structure  $\mathcal{G}_B^C$ , the set of compatible (observed) distribution is obtained by assuming a joint distribution  $P_{\Lambda XYAB} \in \mathscr{P}_5$  over all nodes, that satisfies the Markov condition (2.45) and marginalising over the unobserved node  $\Lambda$ ,

$$\mathscr{P}(\mathcal{G}_B^C) \coloneqq \{ P_{XYAB} \in \mathscr{P}_4 | P_{XYAB} = \sum_{\Lambda} P_{\Lambda} P_A P_B P_{X|A\Lambda} P_{Y|B\Lambda} \}.$$
(2.55)

If  $\Lambda$  is a continuous random variable, the sum is replaced by an integral over  $\Lambda$ . This compatibility condition for the classical causal structure  $\mathcal{G}_B^C$  is in fact identical to the local-causality condition used in the derivation of Bell inequalities (see [39] for a comprehensive review). In the quantum causal structure  $\mathcal{G}_B^Q$ , the unobserved node  $\Lambda$  corresponds to a quantum state  $\rho_{\Lambda} \in \mathscr{S}(\mathscr{H}_{\Lambda}) = \mathscr{S}(\mathscr{H}_{\Lambda_X} \otimes \mathscr{H}_{\Lambda_Y})$ , and the observed nodes X and Y are associated with the POVMs  $\{E_A^X\}_X$  and  $\{E_B^Y\}_Y$ , that act on the subsystems  $\mathscr{H}_{\Lambda_X}$  and  $\mathscr{H}_{\Lambda_Y}$ , depending on the inputs A and B respectively to generate the output distribution.

$$\mathscr{P}(\mathcal{G}_B^Q) \coloneqq \{ P_{XYAB} \in \mathscr{P}_4 | P_{XYAB} = \operatorname{tr}\left( (E_A^X \otimes E_B^Y) \rho_\Lambda \right) P_A P_B \}.$$
(2.56)

#### 2.5.2.2 Interventions and counterfactuals in non-classical causal structures

Interventions in generalised causal structures can be modelled in the same manner as the classical case (Section 2.5.1.2), where a non-trivial intervention on an observed node X (or a subset of of nodes) introduces a new edge  $I_X \longrightarrow X$  and cuts off all other incoming arrows to X. This means that we can only intervene upon classical variables and not quantum systems. Note however that we can model situations involving the manipulation of quantum/GPT systems by allowing the choice of preparation, transformation or measurement on the system to depend on a classical variable. Then an intervention on such a variable will influence the map performed on the quantum/GPT system. Equation (2.51) relates the conditional probabilities  $P_{X_i|\text{par}(X_i)}$ for the pre and post intervention causal structures, but this can involve conditioning on possibly unobserved parents which is not possible in generalised causal structures. Equation (2.51) can however be generalised by specifying how the test (i.e., the causal mechanisms) corresponding to each node transforms under an intervention. In [115], each node is associated with a test that maps all joint state of the incoming (unobserved) subsystems to the joint of the outgoing (unobserved) subsystems, depending on the values of the observed parents. Under an intervention on the node X, the corresponding test would remain unchanged if  $I_X = idle$ , and would effectively correspond to a preparation test that prepares X in a specific value  $x \in X$  (deterministically) whenever  $I_X = do(x)$ , and traces out the systems associated with all other incoming edges.<sup>25</sup>

So far, the procedure is similar to the classical case, however, there is one crucial difference. Equation (2.52) which relates the pre and post intervention distributions and thereby provides a rule for counterfactual inference no longer holds for non-classical causal structures, since this equation is derived using the Markov compatibility condition (2.45), which does apply to this case. Since non-classical causal structures by construction include at least one unobserved node, one would expect that the observed post-intervention distribution may not always be determined by the observed pre-intervention distribution, as is the case also with classical causal structures involving unobserved nodes (see Example 2.5.3). [164] shows that it is impossible to derive any general inference rule relating the observed distributions in pre and post intervention quantum causal structures.<sup>26</sup> This means that it may not always be possible to deduce the post-intervention distribution counterfactually using the pre-intervention distribution, in-stead one would have to physically perform the intervention (i.e., an additional experiment,

<sup>&</sup>lt;sup>25</sup>Note that we can view do(x) effectively as a preparation test even though the node X has an input edge  $I_X \longrightarrow X$ , because a)  $I_X$  is just a hypothetical variable that is introduced to provide an agency to the intervention in the augmented graph approach, but interventions can be equivalently described by an alternative approach that does not introduce any additional variables [159], and b) The value of X is deterministically correlated with  $I_X$ , such that if do(x) is seen as a preparation on the exogenous variable  $I_X$ , this deterministically induces the preparation X = x on X.

<sup>&</sup>lt;sup>26</sup>The causal model framework of [164] is slightly different than what we have presented here, but nevertheless compatible with this picture. This is explained further in the next subsection.

after the original experiment that collected the pre-intervention statistics) to characterise this distribution and all the do-conditionals. However, [163] and [164] provide conditions for identifying quantum causal structures where an inference rule analogous to (2.52) does exist and fully counterfactual inference is possible. This is still a field of ongoing research, we will discuss interventions in non-classical causal structures in further detail in Chapter 6.

#### 2.5.2.3 Other approaches for causality beyond the classical setting

The main approach to quantum and post-quantum causal structures adopted in this thesis, that we have discussed so far is based on the framework of [115]. This approach retains d-separation and conditional independence criteria from the classical Bayesian networks approach [159] while generalising the Markov condition (2.45). There are a several other frameworks for quantum causal models that build on different aspects of Pearl's classical framework [159]. These frameworks are consistent with each other but can differ in scope due to how they formalise the notion of a causal model in quantum settings i.e., how the nodes and edges are represented, what compatibility conditions are used, and type of interventions/manipulations that are considered.

[165, 67, 6] have proposed generalisations of the Reichenbach principle [176] to the quantum setting. Among these, [165] is based on a new graph separation condition for quantum causal structures called q-separation, while [67, 6] are based on formulating quantum networks using quantum channels and their Choi states. [6] as well as an earlier work [132] construct quantum networks by representing quantum channels as conditional quantum states (analogous to conditional probabilities). The framework of [6] will employed in one of our results, we briefly outline it here and present the relevant details in Chapter 4 where it will be used. In [6], nodes correspond to systems and edges to transformations, in contrast to the HLP framework [115] where transformations (tests) occur at the nodes and the edges propagate systems. All nodes are taken to be quantum systems and a distinction between observed and unobserved nodes is not made. Then a conditional quantum state  $\rho_{A|par(A)}$  for a node A conditioned on its parents (analogous to conditional distributions  $P_{X|par(X)}$ ) is defined through the Choi-Jamiolkowski representation of the quantum channel that maps the quantum systems corresponding to par(A)to that corresponding to A. With this, a quantum Markov condition, completely analogous to the classical case (2.45) is defined. This also allows for the definition of a joint state  $\sigma_{A_1,\ldots,A_n}$ over all the nodes  $\{A_1, ..., A_n\}$ , even though the systems corresponding to these nodes may not coexist. This joint state  $\sigma$  characterises the structure of the channels connecting the various nodes and not the quantum states of the systems. Interventions correspond to CPTP maps that act on quantum nodes and the causal model provides a prescription for obtaining the post-intervention distribution given the joint state  $\sigma$  and the CPTP maps corresponding to the interventions, through a formula analogous to the Born rule.

In another recent framework for quantum causal models [163, 164] (also noted in the previous subsection), all nodes are taken to be classical, while edges correspond to propagating quantum systems that are acted upon by transformations that depend on the values of the classical

nodes.<sup>27</sup> Further, frameworks such as [181] have shown that quantum causal models can provide an information-theoretic advantage for causal inference by utilising quantum properties such as entanglement and superposition.

The discussion so far has focussed on the assumption that a fixed causal structure (between classical/quantum or post-quantum systems) exists, even if it may be unknown. More general approaches to non-classical notions of causality have considered scenarios where not only the systems in a causal structure, but the causal structure itself is of a quantum nature [111, 154, 230. In such frameworks, quantum superpositions of the causal and temporal orders between events are typically considered by dropping the assumption of a fixed background space-time or causal structure, and the study of such indefinite causal structures have garnered much research interest in recent years and shown to provide additional information-theoretic advantages [53, 9, 8, 10, 21, 107, 32, 155, 171, 51, 153, 18]. Here, analogous to Bell inequalities, which certify the non-classicality of correlations in a fixed causal structure, causal inequalities have been proposed for certifying the non-classicality of the causal structure itself [154]. Unlike the former which have been violated in several experiments, violations of the latter are modelled in several theoretical frameworks [154, 106, 32, 18], but whether causal inequalities can be physically violated remains a thriving open question. Further, scenarios where quantum systems are superposed/entangled over space as well as in time have been physically implemented. The so-called causal box framework [171] models information processing tasks where quantum states can be delocalised in states as well as time, in a manner compatible with our relativistic causality. This framework has applications for relativistic quantum cryptography as well as for foundational questions regarding causal inequalities, as we have shown in a previous work [209], and an ongoing work<sup>28</sup> (both of which are not included in this thesis). Finally, there also exist general frameworks for quantum causality that take into account gravitational effects, with the aim of developing an operational understanding of quantum gravity [111, 112, 230, 42].

# 2.6 Modal logic

In daily life and as part of the scientific method, we often use logic to reason about our observations and to predict the possibility of future observations. We do so by assigning truth values to different statements depending on what we observed and what we might already know. Modal logic is broadly speaking, a family of formal systems of logic that are associated with different modalities for assigning truth values to statements, such as "it is necessary that" (alethic), "it is known that" (epistemic), "it ought to be the case that" (deontic), or "it has been the case that" (temporal). In this thesis, we will focus on the branch of modal logic, namely *epistemic modal logic* that is applied for reasoning about knowledge. This is particularly useful in sce-

<sup>&</sup>lt;sup>27</sup>For example, three sequential operations on a quantum state are represented by the causal structure  $X \to Y \to Z$ , where X specifies the preparation of a quantum state  $\rho_X$  that travels to the next node where a CPTP map  $\mathscr{E}_Y$  labelled by Y acts on  $\rho_X$ , and finally a POVM labelled by Z,  $\{E^Z\}_Z$  acts on the output of the CPTP map  $\mathscr{E}_Y(\rho_X)$ .

<sup>&</sup>lt;sup>28</sup>Vilasini, V., del Rio, L. and Renner, R. Causality in definite and indefinite space-times. In preparation (2020).

narios (both classical and quantum) where where multiple agents reason about each others' knowledge to make logical deductions. Examples include games like poker, logical hat puzzles<sup>29</sup>, extended Wigner's friend experiments in quantum theory [85, 152] (Chapter 7) and in post-quantum theories [208] (Chapter 8), among others [41, 82, 133, 134].

Here we provide a brief summary including the main features of epistemic modal logic that are relevant for this thesis. Modal logic applies to most classical multi-agent setups, and can be seen as a compact mathematical way to capture some of intuitive laws commonly used for reasoning in classical settings. We will first review the basics of the standard framework [131], and then discuss the weaker versions of the axioms proposed recently in [152] for modelling scenarios involving multiple quantum agents. For further reading and a more in depth understanding of the modal logic framework, we refer the reader to [41, 101, 82, 133, 134].

## 2.6.1 Knowledge, possibilities and truth: Kripke structures

We will consider a finite set of agents  $\{A_1, ..., A_n\}$  who describe the world by means of primitive propositions  $\phi_1, \phi_2, ...$  belonging to a some set  $\Phi$ . These primitive propositions correspond to simple statements about the world, for example,  $\phi_1 =$  "Alice is a person" or  $\phi_2 =$  "Alice has a secret key". The framework of modal logic allows for compactly representing the knowledge that agents possess about these primitive statements as well more complex, chained statements about the knowledge of other agents, such as  $\tilde{\phi} =$  "Alice knows that Bob knows that Eve doesn't know the secret key k, and Alice further knows that k = 1". It also provides simple rules for combining such statements to perform logical deductions.

More formally, we consider a set  $\Sigma$  of possible states (or alternatives, or *worlds*<sup>30</sup>) as introduced by [131] in the context of modal logic: for example, in a world  $s_1 \in \Sigma$  the key value is k = 1and Eve does not know it, and in another world  $s_2 \in \Sigma$  Eve could know that k = 0. The truth value of a proposition  $\phi$  is then assigned depending on the possible world in  $\Sigma$ , and can differ from one possible world to another. The setup we will employ for modelling multiple reasoning agents is associated the following structure.

**Definition 2.6.1. (Kripke structure)** A Kripke structure M for n agents over a set of statements  $\Phi$  is a tuple  $\langle \Sigma, \pi, \mathcal{K}_1, ..., \mathcal{K}_n \rangle$  where  $\Sigma$  is a non-empty set of states, or possible worlds,  $\pi$  is an interpretation, and  $\mathcal{K}_i$  is a binary relation on  $\Sigma$ .

The interpretation  $\pi$  is a map  $\pi : \Sigma \times \Phi \to \{ \mathbf{true}, \mathbf{false} \}$ , which defines a truth value of a statement  $\phi \in \Phi$  in a possible world  $s \in \Sigma$ .

 $\mathcal{K}_i$  is a binary equivalence relation on a set of states  $\Sigma$ , where  $(s,t) \in \mathcal{K}_i$  if agent *i* considers world *t* possible given his information in the world *s*.

The truth assignment tells us if the proposition  $\phi \in \Phi$  is true or false in a possible world  $s \in \Sigma$ ;

<sup>&</sup>lt;sup>29</sup>For an analysis of classical hat puzzles in the epistemic modal logic language, see [152].

 $<sup>^{30}</sup>$ This is an entirely classical concept and is not to be confused with "many worlds" type interpretations of quantum theory. *World* here only represents potential alternative situations i.e. a classical list of possibilities, it does not mean that all these situations are actually realised.

for example, if  $\phi =$  "you are reading this PhD thesis" and s is the present world in which you have indeed chosen to read this thesis, then  $\pi(s, \phi) = \mathbf{true}$ . On the other hand, if  $\phi' =$  "There is a mammoth in the snow" then  $\pi(s, \phi') = \mathbf{false}$  in the current world where mammoths are (unfortunately) extinct, however in a world s' corresponding to the ice age  $\phi'$  can indeed be a true statement. Therefore the truth value of a statement in a given structure M might vary from one possible world to another; we will denote that  $\phi$  is true in world s of a structure M by  $(M, s) \models \phi$ , and  $\models \phi$  will mean that  $\phi$  is true in any world s of a structure M.

Naturally, agents need not possess complete information about a possible world they are in, and may consider other alternative worlds possible; for example, if Bob doesn't know whether Alice has a secret key, he can consider as possible both the world where Alice has such a key, and one where she doesn't. This situation is captured by binary relations  $\mathcal{K}_i$ , as formalised in the following definition.

**Definition 2.6.2** (Knowledge operators  $K_i$ ). We say that agent  $A_i$  "knows"  $\phi$  in a world  $s \in \Sigma$  i.e.,

$$(M,s) \vDash K_i \phi$$

if and only if for all possible worlds  $t \in \Sigma$  such that  $(s,t) \in \mathcal{K}_i$  (that is, all the worlds deemed possible by the agent, given their knowledge), it holds that  $(M,t) \models \phi$ .

Knowledge operators combined with the standard logical operators allow us to compactly express complex statements, for instance, our earlier example  $\tilde{\phi} =$  "Alice knows that Bob knows that Eve doesn't know the secret key k, and Alice further knows that k = 1" becomes

$$K_A \left[ (K_B \neg K_E k) \land k = 1 \right].$$

## 2.6.2 Reasoning about other's knowledge: Axioms of knowledge

The axioms of knowledge [95] provide certain rules for combining the statements produced by different agents and to compress them for deducing new statements. These axioms might appear to be common-sensical, but it is imperative to state them formally since not all of these axioms apply in the quantum world, that often surprises our daily common-sensical intuitions.

The distribution axiom allows agents combine statement which contain inferences:

Axiom 1 (Distribution axiom.). If an agent is aware of a proposition  $\phi$  and that another proposition  $\psi$  follows from  $\phi$ , then the agent can conclude that  $\psi$  holds:

$$(M,s) \vDash (K_i \phi \land K_i (\phi \Rightarrow \psi)) \Rightarrow (M,s) \vDash K_i \psi$$

Knowledge generalization rule permits agents use commonly shared knowledge:

Axiom 2 (Knowledge generalization rule.). All agents know all the propositions that are always valid in a structure:

$$(M,s) \vDash \phi \ \forall s \in \Sigma \Longrightarrow \vDash K_i \phi \ \forall i.$$

At first sight, this might appear to be a strong assumption. However, this depends on what  $\Sigma$  is chosen to be for a particular purpose, and characterises what statements can be assumed to be the basic, common knowledge of all agents. For example, we would find it reasonable to expect that the propositions  $\phi_1$ ="objects fall when we drop them",  $\phi_2$ ="guitars have strings",  $\phi_3$ ="we cannot signal from future to past" to be common knowledge, such that they hold in every  $s \in \Sigma$  that we wish to consider. However, if we wish to consider hypothetical worlds where guitars are wind instruments or where we can signal to the past (as physicists working in foundations, including ourselves often do), we can simply include a world  $s' \in \Sigma$  where the statements  $\phi_2$  or  $\phi_3$  are false, and hence do not correspond to the common knowledge of the agents. In fact, parts of Chapter 6, will indeed be set in such a hypothetical world where  $\phi_3$  is not true. One can also treat a theory (such as quantum theory) or aspects of the theory (such as the Born rule) to be the common knowledge of the agents that we wish to consider, as we will see in Chapter 8.

Axiom 3 (Truth axiom). If an agent knows that a proposition is true then the proposition is true,

$$(M,s) \vDash K_i \phi \Rightarrow (M,s) \vDash \phi$$

In philosophy, the truth axiom is often considered as a candidate for a property that distinguishes knowledge from belief.

Positive and negative introspection axioms highlight the ability of agents to reflect upon their knowledge:

Axiom 4 (Positive and negative introspection axioms.). Agents can perform introspection regarding their knowledge:

$$(M,s) \vDash K_i \phi \Rightarrow (M,s) \vDash K_i K_i \phi$$
 (Positive Introspection),  
 $(M,s) \vDash \neg K_i \phi \Rightarrow (M,s) \vDash K_i \neg K_i \phi$  (Negative Introspection).

Weaker version: The truth axiom assigns objective reality to all statements that agents know which can be particularly problematic for quantum settings. Hence a weaker alternative to the truth axiom was proposed in [152].

The trust structure governs the way the information is passed on between agents:

**Definition 2.6.3** (Trust). We say that an agent *i* trusts an agent *j* (and denote it by  $j \sim i$ ) if and only if

$$(M,s) \vDash K_i K_j \phi \Longrightarrow K_i \phi, \quad \forall \phi \in \Phi, s \in \Sigma$$

Note that trust is neither a symmetric nor a transitive relation and hence does not define a pre-order on the agents. This is also in line with what we might expect common-sensically; I

might trust my favourite news reporter (R) to deliver reliable news such that  $K_I K_R \phi \Rightarrow K_I \phi$ would hold in all worlds *s* relevant to me and for all statements  $\phi$  relating to the news (which I could take to characterise the sets  $\Sigma$  and  $\Phi$ ) i.e.,  $R \rightsquigarrow I$ . However R, who does not know me has no reason to trust what I say, even if it relates to the news hence  $K_I K_R \phi \Rightarrow K_I \phi$  or equivalently  $R \not \sim I$ . Similarly, R might have a trusted source S for acquiring the news, and S in turn may acquire that news through a source T whom she trusts i.e.,  $S \rightsquigarrow R$  and  $T \rightsquigarrow S$ . While R would trust the same information when relayed indirectly through the agent S, R may not know of T's existence and hence not trust T if he approached R directly, hence  $T \rightsquigarrow R$  does not necessarily hold.

While replacing truth with the notion of trust may suffice in standard quantum settings, as we will see in Chapters 7 and 8, it turns out to be incompatible with situations where reasoning agents are themselves modelled as physical systems of quantum or post-quantum theories [152].

# Part I

# Approaches to causality in non-classical theories

Correlation doesn't imply causation, but it does waggle its eyebrows suggestively and gesture furtively while mouthing 'look over there.'

- Randall Munroe

# CHAPTER 3

# Overview of techniques for analysing causal structures

In Section 2.5, we provided an overview of the frameworks used for modelling causal structures in classical, quantum and post-quantum theory. We motivated the problem of certifying the non-classicality of causal structures, which has become a focal point of research interest in quantum information by virtue of having deep foundational as well as practical implications. In this chapter, we outline some of the techniques used for this purpose. Section 3.1 and Section 3.2 provide an overview of the main probabilistic and entropic techniques employed in Chapters 4 and 5, taking as example the bipartite Bell causal structure (Figure 2.6a), which is a causal structure of particular interest in this thesis. These sections are largely based on the review sections of our published papers [206] and [207]. The main results of these papers are presented in Chapters 4 and 5. All of the above focus on the case of faithful and acyclic causal structures, as does the majority of literature in the field, and justifiably so, since our observations suggest that causal influences propagate in one direction i.e., from past to future. The present chapter will also focus only on the acyclic case. In Chapter 6, we develop a framework for modelling cyclic and fine-tuned causal structures in non-classical theories and will review the necessary preliminaries for cyclic causal structures there.

# **3.1** Certifying non-classicality in probability space

In the probability space characterisation of bipartite Bell causal structure  $\mathcal{G}_B$  (Figure 2.6a), the observed correlations are represented through the conditional distribution  $P_{XY|AB}$ . Each point in the probability space corresponds to such a conditional distribution and it will be convenient to denote the cardinalities |A|, |B|, |X| and |Y| as  $i_A$ ,  $i_B$ ,  $o_A$  and  $o_B$  (representing the inputs and outputs of the parties Alice and Bob) respectively. Each Bell scenario is defined by the 4-tuple  $(i_A, i_B, o_A, o_B)$ . In a given scenario we will be interested in the sets of *local* and *non-signaling* distributions, which are the sets of conditional distributions  $P_{XY|AB}$  corresponding to the sets

 $\mathscr{P}(\mathcal{G}_B^C)$  and  $\mathscr{P}(\mathcal{G}_B^G)$  (Section 2.5.2.1) of observed correlations  $P_{XYAB}$ . Following the notation of [202], we will express the distribution using a matrix, rather than a vector. For instance, in the case where all the variables take values in  $\{0,1\}$  and using P(xy|ab) as an abbreviation for  $P_{XY|AB}(xy|ab)$ , this is done as

$$P_{XY|AB} = \begin{bmatrix} P(00|00) & P(01|00) & P(00|01) & P(01|01) \\ P(10|00) & P(11|00) & P(10|01) & P(11|01) \\ P(00|10) & P(01|10) & P(00|11) & P(01|11) \\ P(10|10) & P(11|10) & P(10|11) & P(11|11) \end{bmatrix}$$
(3.1)

and the generalisation to larger alphabets is analogous (see, e.g., [202]). This format is convenient because it makes it easy to check whether a distribution is non-signaling, i.e., to check that  $P_{X|AB}$  is independent of B and that  $P_{Y|AB}$  is independent of A. Both the local and nonsignaling sets form convex polytopes, that are highly symmetric. In particular, such polytopes are invariant under local relabellings and/or relabelling parties. By *local relabellings* we mean combinations of relabelling the inputs (e.g.,  $A \mapsto A \oplus 1$ ) and outputs conditioned on the local input (e.g.,  $X \mapsto X \oplus \alpha A \oplus \beta$  where  $\alpha, \beta \in \{0, 1\}$  and  $\oplus$  denotes modulo-2 addition). One might also think about more general global relabellings that depend on both inputs (for instance maps of the form  $X \mapsto X \oplus \alpha A \oplus \beta B \oplus \gamma$  with  $\alpha, \beta, \gamma \in \{0, 1\}$ ), but these do not preserve the nonsignaling set in general so will not be considered here. The only global relabelling we consider is exchange of the two parties, which corresponds to transposing the distribution in the matrix notation of Equation (3.1) and always preserves non-signaling.

More specifically, the non-signaling set is restricted only by the conditional independences  $X \perp B \mid A$  and  $Y \perp A \mid B$  (no-signaling constraints)<sup>1</sup>, which follow from the d-separation relations in  $\mathcal{G}_B$  (Figure 2.6a), and naturally the positivity and normalisation conditions since these are valid probability distributions. These restrictions can be written out as linear inequality/equality constraints, and characterise the  $\mathcal{H}$  representation of the non-signaling polytope. The corresponding  $\mathcal{V}$ -representation is characterised by the extremal non-signaling distributions. On the other hand, the local set is the set of all observed distributions  $P_{XY\mid AB}$  that can arise when  $\Lambda$  is classical corresponds to the set of correlations that admit a local hidden variable model, i.e., the set of distributions that can be expressed in the form

$$P_{XY|AB} = \int_{\Lambda} d\Lambda \, P_{\Lambda} \, P_{A} \, P_{B} \, P_{X|A\Lambda} \, P_{Y|B\Lambda}. \tag{3.2}$$

Here,  $\Lambda$  can be any (possibly continuous) random variable. In the rest of the thesis, we will refer to such correlations either as *local* or as *classical* and denote the set of all such distributions  $\mathscr{L}$ . We also use  $\mathscr{L}^{(i_A, i_B, o_A, o_B)}$  to denote the local set in the  $(i_A, i_B, o_A, o_B)$  scenario.

The local set forms a convex polytope, which can be specified in terms of a finite set of Bell inequalities, each a necessary condition for classicality of the correlation. Along with the positivity and normalisation constraints, these define the  $\mathcal{H}$ -representation of  $\mathscr{L}^{(i_A, i_B, o_A, o_B)}$ . The  $\mathcal{V}$ -representation of the local polytope is defined by the set of all *local deterministic distributions* i.e., distributions  $P_{XY|AB}$  that are deterministic over X and Y for every value of A and B. There

<sup>&</sup>lt;sup>1</sup>The independence  $A \perp B$  (measurement independence) also follow from the graph, but this is not relevant when we are considering the conditional distributions  $P_{XY|AB}$ .

are  $o_A^{i_A} \cdot o_B^{i_B}$  distinct local deterministic distributions in a  $(i_A, i_B, o_A, o_B)$  scenario. In the matrix notation (3.1), each block corresponds to a fixed value for A and B. The local deterministic distributions expressed in this notation have exactly one non-zero entry (= 1) in each block in accordance with the no-signaling condition required by the causal structure. The set of all such distributions can be easily enumerated. Hence, the  $\mathcal{V}$ -representation of the local polytope (local deterministic distributions) and the  $\mathcal{H}$ -representation of the non-signaling polytope (nosignaling, positivity and normalisation constraints) can be easily found for a given scenario. The  $\mathcal{H}$ -representation of the local polytope (Bell inequalities) and the  $\mathcal{V}$ -representation of the non-signaling polytope (extremal non-signaling distributions) can in principle be found through facet and vertex enumeration respectively. However, this becomes computationally intractable for larger cardinalities of the observed variables. In the following, we describe the structure of these sets in more detail for the (2, 2, 2, 2) and (2, 2, 3, 3) bipartite Bell scenarios, which are of particular interest in this thesis.

# 3.1.1 The bipartite Bell scenario with 2 inputs and 2 outputs per party

In the (2, 2, 2, 2) scenario, there are eight extremal Bell inequalities (facets of the local polytope), which are equivalent under local relabellings to the following inequality [57, 202]

$$I_{\text{CHSH}} := P(X = Y | A = 0, B = 0) + P(X = Y | A = 0, B = 1) + P(X = Y | A = 1, B = 0) + P(X \neq Y | A = 1, B = 1) \le 3$$
(3.3)

We denote these by  $I_{\text{CHSH}}^k$  for  $k \in [8]$ , where  $I_{\text{CHSH}}^1 \coloneqq I_{\text{CHSH}}$  and [n] stands for the set  $\{1, 2, ..., n\}$  where n is a positive integer. This provides the facet description of the (2, 2, 2, 2) local polytope. One can also express  $I_{\text{CHSH}}$  in matrix form using

$$M_{\rm CHSH} = \frac{\begin{vmatrix} 1 & 0 & 1 & 0 \\ 0 & 1 & 0 & 1 \\ 1 & 0 & 0 & 1 \\ 0 & 1 & 1 & 0 \end{vmatrix},$$
(3.4)

so that the Bell inequality can be written tr  $(M_{\text{CHSH}}^T P) \leq 3$ , where P is the distrubution expressed in the matrix form (3.1) and T denotes the transpose.

In the vertex picture, the (2, 2, 2, 2) local polytope has 16 local deterministic vertices and the (2, 2, 2, 2) non-signaling polytope shares the vertices of the local polytope and has eight more: the Popescu-Rohrlich (PR) box and seven distinct local relabellings [202, 170]. The PR box distribution (discussed in Section 2.4) satisfies  $X \oplus Y = A.B$  and takes the following form in the current notation

$$P_{\rm PR} = \frac{\begin{vmatrix} \frac{1}{2} & 0 & \frac{1}{2} & 0 \\ 0 & \frac{1}{2} & 0 & \frac{1}{2} \\ \hline \frac{1}{2} & 0 & 0 & \frac{1}{2} \\ 0 & \frac{1}{2} & \frac{1}{2} & 0 \end{vmatrix}.$$
(3.5)
We denote the eight extremal non-signaling vertices equivalent under local relabellings to  $P_{\rm PR}$  by  $P_{\rm PR}^k$ ,  $k \in [8]$  where  $P_{\rm PR}^1 \coloneqq P_{\rm PR}$ . Note that the 8 CHSH inequalities  $\{I_{\rm CHSH}^k\}$  are in one-to-one correspondence with these 8 extremal non-signaling points i.e., each  $P_{\rm PR}^k$  violates exactly one CHSH inequality and each CHSH inequality is violated by exactly one  $P_{\rm PR}^k$ .

## 3.1.2 The bipartite Bell scenario with 2 inputs and 3 outputs per party

In the (2, 2, 3, 3) Bell scenario, there are two classes of Bell inequalities that completely characterize the local polytope: the CHSH inequalities and the  $I_{2233}$  inequalities [122, 64]. A representative example of the latter is

$$I_{2233} \coloneqq \left[ P(X = Y | A = 0, B = 1) + P(X = Y - 1 | A = 1, B = 1) + P(X = Y | A = 1, B = 0) + P(X = Y | A = 0, B = 0) \right] - \left[ P(X = Y - 1 | X = 0, B = 1) + P(X = Y | A = 1, B = 1) + P(X = Y - 1 | A = 1, B = 0) + P(X = Y + 1 | A = 0, B = 0) \right] \le 2,$$

$$(3.6)$$

where all the random variables take values in  $\{0, 1, 2\}$  and all additions and subtractions of the random variables are modulo 3. CHSH-type inequalities for the (2, 2, 3, 3) scenario can be obtained from those of the (2, 2, 2, 2) scenario (Equation (3.4)) through a procedure known as "lifting". The inequalities of the larger scenario are known as the *lifted CHSH inequalities* [167]. Evaluating the value of a lifted CHSH inequality attained by a given distribution in the (2, 2, 3, 3) scenario is equivalent to first coarse-graining the distribution by combining two outcomes into a single outcome, and then evaluating the corresponding CHSH inequality of the (2, 2, 2, 2) scenario. For instance, the lifted CHSH inequality corresponding to  $M_{\text{CHSH}}$  (3.4) and the coarse-graining of always mapping outcomes 1 and 2 to 1 is given in matrix form in the following equation, along with a representative inequality of the  $I_{2233}$  type.

$M_{ m CHSH}^{(2,2,3,3)}$ =	1 0 0 1 0	0 1 1 0 1	0 1 1 0 1	1 0 0 1	0 1 1 1 0	0 1 1 1 0	and	<i>M</i> <sub><i>I</i><sub>2233</sub> =</sub>	$     \begin{bmatrix}       1 \\       -1 \\       0 \\       1 \\       0     $	0 1 -1 -1 1	-1 0 1 0 -1	$     \begin{array}{c}       1 \\       0 \\       -1 \\       -1 \\       0     \end{array} $	-1 1 0 1 -1	$\begin{array}{c} 0 \\ -1 \\ 1 \\ 0 \\ 1 \end{array}$		(3.7	7)
	0	1	1	1	0	0			0	1	-1	0	-1	1			
	0	1	1	1	0	0			-1	0	1	1	0	-1			

The (2, 2, 3, 3) local polytope has a total of 1116 facets, 36 of which correspond to positivity constraints, 648 to CHSH facets (these are equivalent to first coarse-graining two of the outputs into one (for each party and each input) and then applying one of the eight (2, 2, 2, 2) CHSH inequalities), and the remaining 432 are  $I_{2233}$ -type [63] (we label these  $I_{2233}^i$  for  $i \in \{1, 2, \ldots, 432\}$  with  $I_{2233}^1 = I_{2233}$ ).

The facets of the non-signaling polytope correspond to positivity constraints, since the nosignaling constraints being equalities, only reduce the dimension of the polytope. Converting this facet description to the vertex description (e.g., using the PORTA software [55]) one can obtain all the vertices of the (2, 2, 3, 3) non-signaling polytope. This comprises 81 local deterministic vertices, 648 PR-box type vertices and 432 extremal non-signaling vertices (for each of the  $I_{2233}$  inequalities there is one of the latter that gives maximal violation). We call these new vertices the  $I_{2233}$ -vertices. The specific vertex that maximally violates (3.6) is

$$P_{\rm NL} := \begin{bmatrix} \frac{1}{3} & 0 & 0 & \frac{1}{3} & 0 & 0 \\ 0 & \frac{1}{3} & 0 & 0 & \frac{1}{3} & 0 \\ 0 & 0 & \frac{1}{3} & 0 & 0 & \frac{1}{3} \\ \frac{1}{3} & 0 & 0 & 0 & \frac{1}{3} \\ 0 & \frac{1}{3} & 0 & 0 & 0 & \frac{1}{3} \\ 0 & 0 & \frac{1}{3} & \frac{1}{3} & 0 & 0 \end{bmatrix}$$
(3.8)

The 432  $I_{2233}$  vertices of the (2, 2, 3, 3) non-signaling polytope are related to each other through local relabellings<sup>2</sup>. The  $I_{2233}$  inequalities are a special case of the CGLMP inequalities which correspond to facets of the local polytope  $\mathscr{L}^{(2,2,d,d)}$   $(d \ge 3)$  [64, 143]. For the d > 3 case however, there may in general be other new classes of inequalities apart from the CHSH and CGLMP inequalities.

## 3.2 Certifying non-classicality in entropy space

#### **3.2.1** Motivation for the entropic analysis of causal structures

In the bipartite Bell causal structure (Figure 2.6a), that we discussed above, the set of all joint conditional distributions  $P_{XY|AB}$  over the observed nodes X, Y, A, B that can arise when  $\Lambda$  is classical is relatively well understood. For fixed input and output sizes, it forms a convex polytope and hence membership can be checked using a linear program (although the size of the linear program scales exponentially with the number of inputs and the problem is NP-complete [169]). Because of this, the complete set of Bell inequalities characterizing these polytopes are unknown for |X|, |Y| > 3 or |A|, |B| > 5 [143, 15, 66]. In causal structures with more unobserved common causes (such as the Triangle causal structure of Figure 2.6b), the set of compatible correlations is not always convex, and hence not well understood. The inflation technique proposed in [223] has been used to derive some necessary constraints (or Bell inequalities) on the classical sets of correlations in such causal structures. This technique in principle certify whether or not a given distribution belongs to the classical set  $\mathscr{P}(\mathcal{G}^C)$  of a causal structure  $\mathcal{G}$  [149], and involves constructing a new causal structure, known as the inflation of the original

<sup>&</sup>lt;sup>2</sup>In general, equivalent points of the non-signaling polytope may be related by local relabellings or exchange of the two parties (which is a global operation). In the (2, 2, 3, 3) scenario there are  $2 \times (2 \times 6^2)^2 = 10368$  such operations, twice the number of local relabellings. To count these, note that for each party there are 2 ways to permute the inputs, and 6 ways to permute the outputs for each of the 2 inputs. All 432 extremal points of the  $I_{2233}$  type (those which maximally violate a  $I_{2233}$  inequality) can be generated using only local relabellings of  $P_{\rm NL}$ , and similarly all 648 extremal points of the CHSH type can be generated through local relabellings of  $P_{\rm PR}$  embedded in the (2, 2, 3, 3) scenario (by adding zero probabilities to the third outcome).

causal structure such that Bell inequalities for the inflated causal structure also hold for the original causal structure. However, the method does not tell us how to construct a suitable inflation of the causal structure in order to achieve this, or how large this inflation needs to be. Thus, in general, using the inflation technique becomes intractable in practice.

One approach to overcoming the difficulties of non-convex correlation sets is to analyse the problem in entropy space [225]. This has proven to be useful in a number of cases (see, e.g., [89, 44], or [215] for a detailed review), since the problem is convex in entropy space and the entropic inequalities characterising the relevant sets are independent of the number of measurement outcomes. These advantages have motivated the entropic analysis of causal structures and the entropy vector method is employed for this purpose. In causal structures with observed parentless nodes (such as the Bell scenario), this method can be supplemented with the post-selection technique [34] that can allow for the detection of non-classicality that was previously undetectable in entropy space. In the following, we outline the basics of the entropy vector method, with and without post-selection. For a more thorough overview of entropic techniques for analysing causal structures, we refer the reader to [215].

#### 3.2.2 The entropy vector method

The results of this thesis are related to certifying the gap between the classical and quantum sets of correlations in entropy space. For this, one must find necessary conditions satisfied by the set of classical entropies over the observed nodes, and look for quantum violations of these conditions. Hence the entropy vector method in this case only involves classical entropies. In the following, we describe the entropy vector method for the Shannon entropy H(), but the same concepts will be later applied to other measures such as the Tsallis or Rényi entropies.

**Definition 3.2.1** (Entropy vector). Given a joint distribution  $P_{X_1,\ldots,X_n} \in \mathscr{P}_n$  over n random variables  $X_1, X_2, \ldots, X_n$ , the *entropy vector* of  $P_{X_1,\ldots,X_n}$  is defined as a vector with  $2^n - 1$  components, each of which correspond to the entropy of an element of the powerset of  $\{X_1, X_2, \ldots, X_n\}$  (excluding the empty set), i.e., the entropy vector can be expressed as

$$(H(X_1), ..., H(X_n), H(X_1X_2), H(X_1X_3), ..., H(X_1X_2, ..., X_n)).$$

The entropy vector of a distribution  $P \in \mathscr{P}_n$  is denoted as  $\mathbf{H}(P) \in \mathbb{R}^{2^n-1}$ .

**Definition 3.2.2** (Entropy cone). Let  $\Gamma_n^*$  be set of all vectors in  $\mathbb{R}^{2^{n-1}}$  that are entropy vectors of a probability distribution  $P \in \mathscr{P}_n$ , i.e.,  $\Gamma_n^* = \{v \in \mathbb{R}^{2^{n-1}} : \exists P_{X_1,\dots,X_n} \text{ s.t. } v = \mathbf{H}(P_{X_1,\dots,X_n})\}$ . Its closure  $\overline{\Gamma_n^*}$  is known as the *entropy cone* and it includes vectors v for which there exists a sequence  $P_k \in \mathscr{P}_n$  such that  $\mathbf{H}(P_k)$  tends to v as  $k \to \infty$ .

Note that the conditional entropies and mutual informations can be encoded in the entropy vector description through the chain rule (2.20) which gives the relations H(X|Y) = H(XY) - H(Y), I(X:Y) = H(X) + H(Y) - H(XY) and I(X:Y|Z) = H(XZ) + H(YZ) - H(XYZ) - H(Z). The entropy cone  $\overline{\Gamma_n^*}$  is known to be a convex set for any n [227], but it is difficult to characterise and various approximations to it have been considered [215].

An outer approximation to the entropy cone, known as the *Shannon cone* is obtained by noting that the information-theoretic properties of entropies (discussed in Section 2.3.1) imply constraints on valid entropy vectors. These include non-negativity of the entropies, monotonicity (2.22) (i.e.,  $H(R) \leq H(RS)$ ) and submodularity (2.24) (also known as strong-subadditivity;  $H(RT) + H(ST) \geq H(RST) + H(T)$ ). As noted in Section 2.3.1, in the Shannon case, monotonicity and submodularity are equivalent to the non-negativity of the conditional entropy H(S|R) and the conditional mutual information I(R:S|T) respectively and hold for any three disjoint subsets R, S and T of  $\{X_1, \ldots, X_n\}$ .

**Definition 3.2.3** (Shannon constraints and the Shannon cone). The set of linear constraints comprising of non-negativity, monotonicity and submodularity are together known as the *Shannon constraints* and the set of vectors  $u \in \mathbb{R}^{2^{n-1}}$  obeying all the Shannon constraints form the convex cone known as the *Shannon cone*,  $\Gamma_n$ .

Following standard practice, we will include non-negativity implicitly, such that every entropy vector v belongs to the space  $\mathbb{R}_{\geq 0}^{2^n-1}$ . Excluding non-negativity, there are a total of  $n+n(n-1)2^{n-3}$ independent Shannon constraints for n variables [225]. By definition, the Shannon cone is an outer approximation to  $\overline{\Gamma_n^*}$  i.e.,  $\overline{\Gamma_n^*} \subseteq \Gamma_n$ .<sup>3</sup> Hence all entropy vectors derived from a probability distribution  $P \in \mathscr{P}_n$  obey the Shannon constraints but not all vectors  $u \in \mathbb{R}^{2^n-1}$  obeying the Shannon constraints are such that  $\mathbf{H}(P) = u$  for some joint distribution  $P \in \mathscr{P}_n$ .

**Entropic causal constraints:** The constraints on the entropy vectors mentioned so far are independent of the causal structure. We know from Section 2.5.1 that a causal structure imposes further constraints on the entropies of the random variables associated with its nodes. In the case of Shannon entropies and a classical causal structure  $\mathcal{G}^C$  over n nodes, all the constraints implied by the causal structure on the entropies can be derived from the following n constraints (of Equation (2.49), reproduced here for convenience), one for each node  $X_i$ 

$$I(X_i:X_i^{\ddagger}|X_i^{\downarrow_1}) = 0 \tag{3.9}$$

These encode the conditional independence of each node from its non-descendants given its parents. Then the set of all entropy vectors  $v \in \Gamma_n^*$  that also satisfy the *n* entropic causal constraints (3.9) for a classical causal structure  $\mathcal{G}^C$  are denoted as  $\Gamma_n^*(\mathcal{G}^C)$ . Similarly, the set of all entropy vectors in the outer approximation,  $v \in \Gamma_n$  that satisfy the causal constraints for  $\mathcal{G}^C$  are denoted as  $\Gamma_n(\mathcal{G}^C)$ . Then  $\Gamma_n^* \subseteq \Gamma_n$  by construction. Further, it can be shown (Lemma 9 of [215]) that  $\Gamma_n^*(\mathcal{G}^C)$  is indeed the set of all achievable entropy vectors in  $\mathcal{G}^C$  i.e.,

$$\Gamma_{n}^{*}(\mathcal{G}^{*}) = \{ v \in \mathbb{R}^{2^{n}-1}_{\geq 0} | \exists P = \Pi_{i=1}^{n} P_{X_{i} | X_{i}^{\downarrow_{1}}} \in \mathscr{P}_{n} \text{ s.t. } \mathbf{H}(P) = v \},$$
(3.10)

and that its closure  $\overline{\Gamma_n^*}(\mathcal{G}^C)$  is convex. Note that it need not be finitely generated. However, its outer approximation  $\Gamma_n(\mathcal{G}^C)$  is by construction defined through a finite set of inequalities i.e., the  $n + n(n-1)2^{n-3}$  Shannon constraints and the *n* causal constraints (3.9).

Given a causal structure  $\mathcal{G}$ , we wish to characterise the set of entropy vectors over the observed nodes that can arise in the classical causal structure  $\mathcal{G}^C$ . Say the causal structure  $\mathcal{G}$  has n nodes

<sup>&</sup>lt;sup>3</sup>For  $n \leq 3$ , the cones coincide, but for  $n \geq 4$  they do not [227].



Figure 3.1: The bipartite Bell causal structure and its post-selected version: (a) The bipartite Bell causal structure. The nodes A and B represent the random variables corresponding to independently chosen inputs, while X and Y represent the random variables corresponding to the outputs.  $\Lambda$  is an unobserved node representing the common cause of X and Y. (b) The post-selected bipartite Bell causal structure for the case of binary inputs. The observed nodes  $X_a$  represent the outputs when the input is  $a \in \{0, 1\}$  and likewise for  $Y_b$ . Note that  $X_0$  and  $X_1$  are never simultaneously observed and likewise  $Y_0$  and  $Y_1$ .

 $\{X_1, ..., X_n\}$  of which, without loss of generality, the first k are observed and the remaining n - k are unobserved. Then the marginalisation is performed by projecting the entropy cone in  $\mathbb{R}^{2^{n}-1}$  (over all the nodes) to its marginal entropy cone in  $\mathbb{R}^{2^{k}-1}$  (over the observed nodes only). The finitely generated outer approximation,  $\Gamma_n(\mathcal{G}^C)$  is often considered for this purpose since the cone  $\overline{\Gamma}_n^*(\mathcal{G}^C)$  is difficult to characterise. The projection of the cone  $\Gamma_n(\mathcal{G}^C)$  into the subspace of the k observed nodes defines the marginal cone  $\Gamma_k(\mathcal{G}^C)$ , and can be obtained through Fourier-Motzkin Elimination [221] (Section 2.2.2). This provides a  $\mathcal{H}$ -representation of the marginal cone  $\Gamma_k(\mathcal{G}^C)$ , and every marginal entropy vector over the observed nodes arising from a compatible distribution in the classical causal structure  $\mathcal{G}^C$  necessarily belongs to this cone. Since non-classical causal structures do not satisfy the initial assumption of the existence of the joint distribution/entropies, they may give rise to correlations that do not satisfy the marginal constraints on the observed nodes obtained through this procedure. A violation of any of the inequalities defining  $\Gamma_k(\mathcal{G}^C)$  by the entropy vector  $v = \mathbf{H}(P_Q)$  of a distribution  $P_Q$  compatible with the quantum causal structure  $\mathcal{G}^Q$ , certifies the non-classicality of  $P_Q$ .

#### **3.2.3** Post-selected causal structures

In causal structures with one or more observed parentless nodes, supplementing the entropy vector method with an additional technique can allow for tighter characterisations of the marginal scenario. This technique involves defining, for each given causal structure with at least one parentless node, a new causal structure called the post-selected causal structure. The general technique for doing this can be found in [215] (for example). Here, we will only consider the bipartite Bell causal structure (Figure 2.6a, and reproduced in Figure 3.1a for convenience) with two inputs per party and the post-selected version thereof.

The post-selected causal structure is obtained by removing the parentless observed nodes A

and B in the original causal structure 3.1a and replacing the descendants X and Y with two copies of each i.e.,  $X_{A=0}$ ,  $X_{A=1}$ ,  $Y_{B=0}$ ,  $Y_{B=0}$  such that the original causal relations are preserved and there is no mixing between the copies (this is shown in Figure 3.1b). It makes sense to do this in the classical case because classical information can be copied, so we can simultaneously consider the outcome X given A = 0 and that given A = 1. By contrast, in the quantum case the values of A correspond to different measurements that are used to generate X, and the associated variables  $X_{A=0}$  and  $X_{A=1}$  may not co-exist. It hence does not make sense to consider a joint distribution over  $X_{A=0}$  and  $X_{A=1}$  in this case. We therefore only consider the subsets of the observed variables that co-exist

$$\mathcal{S} \coloneqq \{X_0, X_1, Y_0, Y_1, X_0 Y_0, X_0 Y_1, X_1 Y_0, X_1 Y_1\},\tag{3.11}$$

where we use the short form  $X_0Y_0$  for the set  $\{X_0, Y_0\}$  etc. Any non-trivial inequalities derived for the co-existing sets in the classical case can admit quantum or GPT violations.

#### 3.2.4 Entropy vectors and post-selection

In [34], Braunstein and Caves derived a set of constraints on the post-selected causal structure of Figure 3.1b and showed that these constraints can be violated by quantum correlations. In fact, it is now known that in the absence of post-selection, the sets of entropies over the observed nodes in the classical and quantum causal structures,  $\mathcal{G}_B^C$  and  $\mathcal{G}_B^Q$  coincide and therefore, the non-classicality of  $\mathcal{G}_B$  cannot be detected entropically [214]. To discuss post-selected entropic inequalities, we first introduce the notion of *entropic classicality*. For every distribution  $P_{XY|AB}$ in the Bell causal structure (Figure 3.1a), we can associate an entropy vector  $v \in \mathbb{R}^8$  in the postselected causal structure (Figure 3.1b) whose components are the entropies of each element of the set  $\mathcal{S}$  (Equation (3.11)) distributed according to  $P_{X_aY_b} \coloneqq P_{XY|A=a,B=b}$ . Let **H** be the map that takes the observed distribution to its corresponding entropy vector in the post-selected causal structure.

**Definition 3.2.4** (Entropic classicality). An entropy vector  $v \in \mathbb{R}^8$  is *classical* with respect to the bipartite Bell causal structure (Figure 3.1a) if there exists a classical distribution  $P_{XY|AB} \in \mathscr{L}$  such that  $\mathbf{H}(P_{XY|AB}) = v$ , where  $\mathscr{L}$  is the local or classical polytope (Equation (3.2)). Further, a distribution  $P_{XY|AB}$  is *entropically classical* if there exists a classical distribution with the same entropy vector, i.e., if there exists a classical entropy vector v such that  $\mathbf{H}(P_{XY|AB}) = v$ .

The set of all classical entropy vectors forms a convex cone. We now review how the Braunstein Caves (BC) Inequalities are derived for the case when the observed parentless nodes Aand B are binary. In this case, the post-selected causal structure 3.1b imposes no additional constraints on the distribution (or entropies) of the observed nodes  $X_0$ ,  $X_1$ ,  $Y_0$  and  $Y_1$  because they share a common parent and thus any joint distribution over  $X_0$ ,  $X_1$ ,  $Y_0$  and  $Y_1$  can be realised in the causal structure 3.1b. By contrast, any correlations in the original causal structure 3.1a must obey the no-signaling constraints over the observed nodes A, B, X and Y since A does not influence Y and B does not influence X in this causal structure. The inequalities derived by Braunstein and Caves follow by applying monotonicity (2.22), strong subadditivity in unconditional form (2.24) and the chain rule (2.20) to the entropies of the observed variables  $\{X_0, X_1, Y_0, Y_1\}$  in the post-selected causal structure. The derived relations hold for the classical causal structure (and not necessarily for the quantum and GPT cases) because only in the classical case does it make sense to consider a joint distribution over these four variables that in the quantum and GPT cases do not co-exist (c.f. Section 3.2.3). The BC inequalities are entropic Bell inequalities i.e., they hold for every classical entropy vector in the post-selected causal structure 3.1b. There are four BC inequalities

$$I_{BC}^{1} \coloneqq H(X_{0}Y_{0}) + H(X_{1}) + H(Y_{1}) - H(X_{0}Y_{1}) - H(X_{1}Y_{0}) - H(X_{1}Y_{1}) \leq 0$$

$$I_{BC}^{2} \coloneqq H(X_{0}Y_{1}) + H(X_{1}) + H(Y_{0}) - H(X_{0}Y_{0}) - H(X_{1}Y_{0}) - H(X_{1}Y_{1}) \leq 0$$

$$I_{BC}^{3} \coloneqq H(X_{1}Y_{0}) + H(X_{0}) + H(Y_{1}) - H(X_{0}Y_{0}) - H(X_{0}Y_{1}) - H(X_{1}Y_{1}) \leq 0$$

$$I_{BC}^{4} \coloneqq H(X_{1}Y_{1}) + H(X_{0}) + H(Y_{0}) - H(X_{0}Y_{0}) - H(X_{0}Y_{1}) - H(X_{1}Y_{0}) \leq 0$$
(3.12)

It has been shown in [90] that these four inequalities are complete in the following sense (the lemma below is implied by Corollary V.3 in [90]).

**Lemma 3.2.1.** A distribution in the post-selected Bell scenario with binary A and B is entropically classical if and only if it satisfies the four BC inequalities (3.12).

Note that the BC inequalities (3.12) and this completeness result are independent of the cardinality of X and Y. However, a crucial point is that entropic classicality does not imply classicality of the associated distribution. For example, the entropy vector  $\mathbf{H}(P_{PR})$  associated with the maximally non-classical distribution  $P_{PR}$  (Equation (3.5)) satisfies all the BC inequalities is hence entropically classical. Since entropic inequalities are non-linear in the probabilities, it is nevertheless possible to have a convex combination of two entropically classical distributions that is entropically non-classical. This allows for the non-classicality of a distribution such as  $P_{PR}$  to be "activated" in entropy space by mixing with a classical distribution. Hence a natural question that arises is whether post-selected entropic inequalities can identify all non-classical distributions in this manner, after a suitable and reasonable post-processing operation (such as mixing with a classical distribution). This is known to be the case for (d, d, 2, 2) Bell scenarios with  $d \ge 2$  [46], but has been an open question for the non-binary outcome case. In Chapter 5 (which is based on [207]), we analyse this question for the 2 input, d outcome Bell scenarios or (2, 2, d, d) scenarios with  $d \ge 3$ , and our results suggest that the answer is negative in this case for both Shannon and Tsallis entropic inequalities. The Triangle causal structure (Figure 2.6b) on the other hand cannot be analysed using post-selection since it has no observed parentless nodes. We analyse this causal structure using Tsallis entropies, in the absence of post-selection in Chapter 4 (based on [206]). We leave further details of the entropic technique and of these results to Chapters 4 and 5.

## CHAPTER 4

## Entropic analysis of causal structures without post-selection

As discussed in Section 3.2, the main motivation for the entropic technique for analysing Causal structures stems from their ability to overcome certain problems that arise in probability space, such as the non-convexity of correlation sets and the drastic scaling of the number of Bell inequalities in the cardinality of the variables. The entropic method for analysing causal structures has proven to be useful in a number of cases (see, e.g., [90, 44], or [215] for a detailed review). However, it was shown in [214] that the entropy vector method with Shannon entropies cannot detect the classical-quantum gap for line-like causal structures, in the absence of post-selection. This includes causal structures such as the Bell scenario which are well known to support non-classical correlations. Although new Shannon entropic inequalities have been derived using this method (e.g., in the Triangle causal structure), no quantum violation of these have been found even when non-classical correlations are known to exist in these causal structures [216, 47]. Due to these limitations of Shannon entropies, it is natural to ask whether other entropic quantities could do better.

This chapter is predominantly based on work carried out with Roger Colbeck [206], while Appendix 4.5.3 reports observations from a side-project with Mirjam Weilenmann. The main question we address here is whether generalised entropic inequalities can overcome these limitations of Shannon entropies in the absence of post-selection. In particular we employ Tsallis entropies and analyse the Bell 2.6a and Triangle causal structures 2.6b. A motivation for considering Tsallis entropies for the task is that they are a family with an additional (real) parameter. The set of entropies for all possible values of this parameter conveys more information about the underlying probability distribution than a single member of the family and hence the ability to vary a parameter may give advantages for analysing causal structures. Tsallis entropies appear to be a good candidate since they satisfy monotonicity, strong subadditivity (in both conditional forms) and the chain rule which are desirable properties for their use

in the entropy vector method.<sup>1</sup> Tsallis entropies have been considered in the context of causal structures before [213] where they were shown to give an advantage over Shannon entropy in detecting the non-classicality of certain states in the post-selected Bell scenario 3.1b<sup>2</sup>. In this chapter we analyse causal structures in the absence of post-selection and consider post-selection in Chapter 5. Note however that the post-selection technique (c.f. Section 3.2.3) can only be applied to causal structures with at least one observed parentless node i.e., it can be applied to the Bell, but not to the Triangle causal structure.

## 4.1 Summary of contributions: in words and in poetry

The entropy vector method requires as input the constraints implied by a causal structure on the entropies of its nodes. Finding that these constraints in for Tsallis entropies cannot take the same form as the Shannon ones (Equation (3.9)), we derive the constraints on the classical Tsallis entropies that are implied by a given causal structure in Section 4.2. As an additional result, we generalise these entropic causal constraints to quantum Tsallis entropies for certain cases, in Appendix 4.5.1. In Section 4.2.2, we use these causal constraints in the entropy vector method with Tsallis entropies but find that the computational procedure becomes too time consuming even for simple causal structures such as the bipartite Bell scenario. Despite this limitation, we derive new Tsallis entropic inequalities for the Triangle causal structure in Section 4.3, using known Shannon entropic inequalities of [47] and our Tsallis constraints of Section 4.2. In Section 4.4, we discuss the reasons for the computational difficulty of this method, the drawbacks of using Tsallis entropies for analysing causal structure and identify potential future directions stemming from our work. In Appendix 4.5.2, we provide details of the Mathematica package LPASSUMPTIONS [62] that we developed and used for obtaining some of the computational results of this, and the next chapter. Additionally, in Appendix 4.5.3, we summarise the results obtained together with Mirjam Weilenmann where we also found limitations of using Rényi entropies for certifying non-classicality in causal structures. Below is a poetic summary of our paper [206] on which the current chapter is based, following which we present the results (in a more rigorous, non-poetic fashion).

About underlying physics we want to tell, Using observed correlations, as did Bell For a scenario that was blessed, In probability space, with convex correlation sets. Generally though, That may not be so.

So we move to entropy space,

<sup>&</sup>lt;sup>1</sup>Other examples of more general entropy measures such as the Rényi entropy [179] do not satisfy one or more of these properties, making it more difficult to get entropic constraints on them using the entropy vector method (c.f. Table 2.1).

<sup>&</sup>lt;sup>2</sup>Note that non-classicality cannot be detected Shannon entropically in the Bell causal structure (Figure 2.6a) without post-selection [214].

Putting convexity back in place, But we can not be too smug, For the Shannon cones aren't too snug.

So here we try our luck with Tsallis entropies, Hoping they give us tighter inequalities. On the way we find some constraints that are new, That follow from classical causal structures, and some quantum ones too

Then we use them in the method of entropy vectors. Alas, that seems too much for our little computers. But we have a cheeky way to put our constraints to use, Using known Shannon results, new Tsallis inequalities we produce.

Though we found nice math results along the way, Our Tsallis inequalities also seem hard to violate, Computationally costly to find tighter inequalities, A feature or a bug of the entropic technique?

## 4.2 Causal constraints and Tsallis entropy vectors

In Section 2.3.2, we discussed some of the general properties of Tsallis entropy that hold irrespective of the underlying causal structure over the variables. These include non-negativity, monotonicity (2.22), and strong subadditivity (2.24) which characterise the set of *Shannon constraints* (c.f. Definition 3.2.3). We summarise these properties here for convenience. For any joint distribution over the random variables involved the following properties hold.

1. **Pseudo-additivity** [68]: For two independent random variables X and Y i.e.,  $P_{XY} = P_X P_Y$ , and for all q, the Tsallis entropies satisfy

$$S_q(XY) = S_q(X) + S_q(Y) + (1 - q)S_q(X)S_q(Y).$$
(4.1)

Note that in the Shannon case (q = 1), we recover additivity for independent random variables.

- 2. Upper bound [94]: For  $q \ge 0$  we have  $S_q(X) \le \ln_q d_X$ . For q > 0 equality is achieved if and only if  $P_X(x) = 1/d_X$  for all x (i.e., if the distribution on X is uniform).
- 3. Monotonicity [71]: For all q,

$$S_q(X) \le S_q(XY). \tag{4.2}$$

4. Strong subadditivity [93]: For  $q \ge 1$ ,

$$S_q(XYZ) + S_q(Z) \le S_q(XZ) + S_q(YZ).$$

$$(4.3)$$

5. Chain rule [93]: For all q,

$$S_q(X_1, X_2, \dots, X_n | Y) = \sum_{i=1}^n S_q(X_i | X_{i-1}, \dots, X_1, Y).$$
(4.4)

The chain rules  $S_q(XY) = S_q(X) + S_q(Y|X)$  and  $S_q(XY|Z) = S_q(X|Z) + S_q(Y|XZ)$  emerge as particular cases and allow the Tsallis mutual informations (Equation (2.31)) to be written as follows.<sup>3</sup> Note that, due to the chain rules, the conditional version of the properties are also satisfied.

$$I_q(X:Y) = S_q(X) + S_q(Y) - S_q(XY),$$
  

$$I_q(X:Y|Z) = S_q(XZ) + S_q(YZ) - S_q(Z) - S_q(XYZ).$$
(4.5)

The causal structure imposes the causal Markov constraints on the joint probability distribution as well as entropic causal constraints over the variables involved (Section 2.5.1). We have seen in Section 3.2 that the entropy vector method requires as an input, the complete set of entropic causal constraints that encode all the conditional independences of the causal structure. In the case of Shannon entropies, for an n node causal structure, the n Shannon entropic causal constraints of Equation (3.9) imply all the conditional independences that follow from d-separation for that causal structure. To analyse causal structures using Tsallis entropies, we require analogous Tsallis entropic causal constraints.

A first observation is that Tsallis entropy vectors do not in general satisfy the causal constraints (Equation (3.9)) satisfied by their Shannon counterparts. For a concrete counterexample, consider the simple, three variable causal structure where Z is a common cause of X and Y, and where there are no other causal relations. In terms of Shannon entropies, the only causal constraint in this case is I(X : Y|Z) = 0. Taking X, Y and Z to be binary variables with possible values 0 and 1, the distribution  $P(xyz) = 1/4 \ \forall x \in X, y \in Y$  if z = 0 and P(xyz) = 0 otherwise, satisfies  $P(xy|z) = P(x|z)P(y|z) \ \forall x \in X, y \in Y$  and  $z \in Z$  but has a q = 2 Tsallis conditional mutual information of  $I_2(X : Y|Z) = \frac{1}{4}$ . Hence when using Tsallis entropies (and conditional Tsallis entropy as defined in Section 2.3.2), the causal constraint cannot be simply encoded by  $I_q(X : Y|Z) = 0$  for q > 1.

Given this observation, it is natural to ask whether there are constraints for Tsallis entropies implied by the causal Markov condition (Equation (2.45)). We answer this question with the following Theorems. Note that we will be using the notation  $d_X$  to denote the cardinality/alphabet size |X| of a random variable X in order the make the notation consistent with the quantum case (Section 4.5.1) where  $d_X$  will denote the dimension of the Hilbert space associated with the subsystem X. Then for classical states,  $d_X$  coincides with the cardinality of the corresponding random variable, which justifies this notation.

**Theorem 4.2.1.** If a joint probability distribution  $P_{XY}$  over random variables X and Y with alphabet sizes  $d_X$  and  $d_Y$  factorises as  $P_{XY} = P_X P_Y$ , then for all  $q \in [0, \infty)$ , the Tsallis mutual

<sup>&</sup>lt;sup>3</sup>Using the chain rule, the monotonicity and strong subadditivity relations (Equations (4.2) and (4.3)) are equivalent to the non-negativity of the unconditional and conditional Tsallis mutual informations. For q < 1, strong subadditivity does not hold in general [93], hence we often restrict to the case  $q \ge 1$  in what follows.

information  $I_q(X:Y)$  is upper bounded by

$$I_q(X:Y) \le f(q, d_X, d_Y),$$

where the function  $f(q, d_X, d_Y)$  is given by

$$f(q, d_X, d_Y) = \frac{1}{(q-1)} \left( 1 - \frac{1}{d_X^{q-1}} \right) \left( 1 - \frac{1}{d_Y^{q-1}} \right) = (q-1) \ln_q d_X \ln_q d_Y.$$

For  $q \in (0, \infty) \setminus \{1\}$ , the bound is saturated if and only if  $P_{XY}$  is the uniform distribution over X and Y.

*Proof.* The proof follows from the pseudo-additivity of Tsallis entropies (Property 1) and the upper bound (Property 2). Using these, for all  $q \ge 0$  and for all separable distributions  $P_{XY} = P_X P_Y$ , we have

$$I_q(X:Y) = S_q(X) + S_q(Y) - S_q(XY) = (q-1)S_q(X)S_q(Y) \le \frac{\left(1 - \frac{1}{d_X^{q-1}}\right)\left(1 - \frac{1}{d_Y^{q-1}}\right)}{q-1} = f(q, d_X, d_Y).$$
(4.6)

Whenever  $q \in (0, \infty) \setminus \{1\}$ , the bound is saturated if and only if  $P_{XY}$  is uniform over X and Y since, for these values of q,  $S_q(X)$  and  $S_q(Y)$  both attain their maximum values if and only if this is the case.

**Theorem 4.2.2.** If a joint probability distribution  $P_{XYZ}$  satisfies the conditional independence  $P_{XY|Z} = P_{X|Z}P_{Y|Z}$ , then for all  $q \ge 1$  the Tsallis conditional mutual information  $I_q(X : Y|Z)$  is upper bounded by

$$I_q(X:Y|Z) \le f(q, d_X, d_Y).$$

For q > 1, the bound is saturated only by distributions in which for some fixed value k the joint probabilities are given by  $P(xyz) = \begin{cases} \frac{1}{d_X d_Y} & \text{if } z = k \\ 0 & \text{otherwise} \end{cases}$  for all x, y and  $z^4$ .

*Proof.* Writing out  $I_q(X:Y|Z)$  in terms of probabilities we have

$$\begin{split} I_q(X:Y|Z) &= \frac{1}{q-1} \Big[ \sum_{xyz} P^q(xyz) + \sum_z P^q(z) - \sum_{xz} P^q(xz) - \sum_{yz} P^q(yz) \Big] \\ &= \frac{1}{q-1} \sum_z P^q(z) \Big[ \sum_{xy} P^q(xy|z) + 1 - \sum_x P^q(x|z) - \sum_y P^q(y|z) \Big] \\ &= \sum_z P^q(z) I_q(X:Y)_{P_{XY|Z=z}}. \end{split}$$

Using this and Theorem 4.2.1, we can bound  $I_q(X:Y|Z)$  as

$$\begin{split} \max_{P_{XYZ}=P_Z P_X|_Z P_Y|_Z} I_q(X:Y|Z) &= \max_{P_{XYZ}=P_Z P_X|_Z P_Y|_Z} \sum_z P(z)^q I_q(X:Y)_{P_{XY|Z=z}} \\ &\leq \max_{P_Z} \sum_z P(z)^q \max_{P_X|_Z=z} P_Y|_{Z=z} I_q(X:Y)_{P_{XY|Z=z}} \\ &= \max_{P_Z} \sum_z P(z)^q f(q,d_X,d_Y) = f(q,d_X,d_Y) \,. \end{split}$$

<sup>&</sup>lt;sup>4</sup>These distributions have deterministic Z and there is one such distribution for each value that Z can take.

The last step holds because for all q > 1,  $\sum_z P(z)^q$  is maximized by deterministic distributions over Z with a maximum value of 1 i.e., only distributions  $P_{XYZ}$  that are deterministic over Z saturate the upper bound of  $f(q, d_X, d_Y)$ . This completes the proof.

Two corollaries of Theorem 4.2.2 naturally follow.

**Corollary 4.2.1.** Let X, Y and Z be random variables with fixed alphabet sizes. Then for all  $q \ge 1$  we have

$$\max_{\substack{P_{XYZ} \\ P_{XY|Z} = P_{X|Z} P_{Y|Z}}} I_q(X:Y|Z) = \max_{\substack{P_{XY} \\ P_{XY} = P_X P_Y}} I_q(X:Y).$$

Furthermore, for q > 1, the maximum on the left hand side is achieved only by distributions in which for some fixed value k the joint probabilities are given by  $P(xyz) = \begin{cases} \frac{1}{d_X d_Y} & \text{if } z = k \\ 0 & \text{otherwise} \end{cases}$ , while the maximum on the right hand side occurs if and only if  $P_{XY}$  is the uniform distribution.

The significance of these new relations for causal structures is then given by the following corollary.

**Corollary 4.2.2.** Let  $P_{X_1...X_n}$  be a distribution compatible with the classical causal structure  $\mathcal{G}^{C}$  and X, Y and Z be disjoint subsets of  $\{X_1, \ldots, X_n\}$  such that X and Y are d-separated given Z. Then for all  $q \ge 1$  we have

$$I_q(X:Y|Z) \le f(q, d_X, d_Y),$$

where  $d_X$  is the product of  $d_{X_i}$  for all  $X_i \in X$ , and likewise for  $d_Y$ .

**Remark 4.2.1.** The results of this section can be generalised to the quantum case under certain assumptions i.e., as constraints on quantum Tsallis entropies implied by certain quantum causal structures (see Appendix 4.5.1 for details). Note that only constraints on the classical Tsallis entropy vectors derived in this section are required to detect the classical-quantum gap. Hence, Appendix 4.5.1 is not pertinent to the main results of this chapter but can be seen as additional results regarding the properties of quantum Tsallis entropies.

#### 4.2.1 Number of independent Tsallis entropic causal constraints

We saw previously that in the Shannon case (q = 1), the *n* conditions (3.9) of the form  $I(X_i : X_i^{\dagger}|X_i^{\downarrow_1}) = 0$  (i = 1, ..., n) imply all the independence relations that follow from the causal structure. In the Tsallis case however, the *n* conditions of the form  $I_q(X_i : X_i^{\dagger}|X_i^{\downarrow_1}) \leq f(q, d_{X_i}, d_{X_i^{\dagger}})$  (Corollary 4.2.2) do not do the same. In the bipartite Bell and Triangle causal structures, in the case where the dimension (cardinality) of each individual node is taken to be *d*, we find that there are 53 and 126 distinct Tsallis entropic inequalities that are implied by the respective causal structures. These are in one-to-one correspondence with the d-separation relations in the corresponding DAGs, and there are no redundancies in these constraints i.e., all of them are

required for characterising the conditional independences in the DAGs. In more detail, we used linear programming to show that each implication of d-separation yields a non-trivial entropic causal constraint for all q > 1 and d > 2 for the bipartite Bell and Triangle causal structures. By comparison, in these causal structures five and six independent Shannon entropic constraints imply all the others. As an illustration of the difference, in the Shannon case, I(A : BC) = 0implies I(A : B) = I(A : C) = 0, whereas the analogous implication does not hold in the Tsallis case in general: although  $I_q(A : BC) \leq f(q, d_A, d_{BC})$  implies  $I_q(A : B) \leq f(q, d_A, d_{BC})$ , it is not the case that  $I_q(A : BC) \leq f(q, d_A, d_{BC})$  implies  $I_q(A : B) \leq f(q, d_A, d_{BC})$ .

The number of distinct conditional independences (and hence the number of independent Tsallis constraints that follow from d-separation) in a DAG depends on the specific graph, however for any DAG  $\mathcal{G}_n$  with n nodes, the number of such constraints can be upper bounded by that of the n-node DAG where all n nodes are independent i.e., the n node DAG with no edges. The number of conditions in this DAG can be thought of as the number of ways of partitioning n objects into four disjoint subsets<sup>6</sup> such that the first two are non-empty and where the ordering of the first two does not matter. Therefore, there are at most  $\frac{1}{2}(4^n - 2 \times 3^n + 2^n)$  such conditions.

#### 4.2.2 Using Tsallis entropies in the entropy vector method

We used the causal constraints of Corollary 4.2.2 in the entropy vector method (Section 3.2) with the aim of deriving new quantum-violatable entropic inequalities for the Triangle causal structure (Figure 2.6b). To do so, we started with the variables A, B, C, X, Y, Z of the Triangle causal structure, the Shannon constraints and causal constraints satisfied by the Tsallis entropy vectors over these variables (Corollary 4.2.2) and used a Fourier-Motzkin (FM) elimination (Section 2.2.2) algorithm (from the PORTA software [55]) to eliminate the Tsallis entropy components involving the unobserved variables A, B, C and obtain the constraints on the observed nodes X, Y, Z. The Tsallis entropy vector for the six nodes has  $2^6 - 1 = 63$  components. The required marginal scenario with the observed nodes X, Y, Z has Tsallis entropy vectors with  $2^3 - 1 = 7$  components and in this case, the Fourier-Motzkin algorithm has to run 56 iterations, each of which eliminates one variable.

Starting with the full set of 126 Tsallis entropic causal constraints for the Triangle causal structure as well as the 246 independent Shannon constraints, the Fourier-Motzkin elimination algorithm did not finish within several days on a standard desktop PC and the number of intermediate inequalities generated grew to about 90,000 after 11 steps. Because of this we instead

<sup>&</sup>lt;sup>5</sup>For an explicit counterexample, consider  $P_{ABC} = \{\frac{3}{10}, 0, \frac{2}{10}, 0, \frac{1}{10}, \frac{1}{10}, \frac{2}{10}, \frac{1}{10}\}$  over binary A, B and C for which  $I_2(A:BC) = 9/25 < 3/8 = f(2,2,4)$  but  $I_2(A:B) = 13/50 > 1/4 = f(2,2,2)$ .

<sup>&</sup>lt;sup>6</sup>The four subsets correspond to the three arguments of the conditional mutual information and a set of 'leftovers'.

tried starting with a subset comprising 15 of the 126 Tsallis entropic causal constraints<sup>7</sup> i.e., 261 constraints on 63 dimensional vectors. We considered the case of q = 2 and where the six random variables are all binary. Again, in this case the algorithm did not finish after several days. We also tried starting with fewer causal constraints (for example, the six constraints analogous to the Shannon case) as well as using a modified code, optimised to deal with redundancies better but both of these attempts made no significant difference to this outcome.

Such a rapid increase of the number of inequalities in each step is a known problem with Fourier-Motzkin elimination where an elimination step over n inequalities can result in up to  $n^2/4$  inequalities in the output and running d successive elimination steps can yield a double exponential complexity of  $4(n/4)^{2^d}$  [221]. This rate of increase can be kept under control when the resulting set of inequalities has many redundancies. This happens in the Shannon case where the causal constraints are simple equalities and the system of 246 Shannon constraints plus 6 Shannon entropic causal constraints reduces to a system of just 91 independent inequalities is possible in general due to the nature of the causal constraints. The fact that the Tsallis entropic causal constraints are inequality constraints rather than equalities also contributes to the computational difficulty since each independent equality constraint in effect reduces the dimension of the problem by 1.

We also tried the same procedure on the bipartite Bell causal structure (Figure 2.6a), again for q = 2 and binary variables. Starting with the full set of 53 causal constraints, again resulted in the program running for over a week without nearing the end, and a similar result was obtained when starting only with 8–10 causal constraints. While starting with fewer causal constraints such as the 5 conditional independence constraints (one for each node) resulted in a terminating program, no non-trivial entropic inequalities were obtained (i.e., we only obtained those relating to the Shannon constraints or following directly from d-separation).<sup>8</sup>

# 4.3 New Tsallis entropic inequalities for the Triangle causal structure

Despite the limitations encountered in applying the entropy vector method to Tsallis entropies (Section 4.2.2), here we find new Tsallis entropic inequalities for the Triangle causal structure

<sup>&</sup>lt;sup>7</sup>These included the 6 that follow from "each node  $N_i$  is conditionally independent of its descendants given its parents" (denoted as  $N_i \perp^d N_i^{\dagger} | N_i^{\downarrow 1}$ ) and 9 more chosen arbitrarily from the total of 126 independent Tsallis constraints we found for the Triangle. The 6 former constraints for the Triangle (Figure 2.6b) are  $A \perp^d CXB$ ,  $B \perp^d CYA$ ,  $C \perp^d BZA$ ,  $X \perp^d YAZ | CB$ ,  $Y \perp^d XBZ | AC$  and  $Z \perp^d YCX | AB$ . An example of 9 more constraints for which the procedure did not work are  $X \perp^d Y | CB$ ,  $X \perp^d A | CB$ ,  $X \perp^d Z | CB$ ,  $Y \perp^d X | AC$ ,  $Y \perp^d B | AC$ ,  $Y \perp^d Z | AC$ ,  $Z \perp^d Y | AC$ ,  $Z \perp^d C | AB$  and  $Z \perp^d X | AB$ . We also tried some other choices and number of constraints but this did not lead to any improvement.

<sup>&</sup>lt;sup>8</sup>For example, we were able to obtain  $I_2(A:BY) \leq \frac{7}{16}$  and  $I_2(B:AX) \leq \frac{7}{16}$ , while, in the case of binary variables and q = 2, the independences in the DAG together with Theorem 4.2.1 imply  $I_2(A:BY) \leq \frac{6}{16}$  and  $I_2(B:AX) \leq \frac{6}{16}$ , which are the Tsallis entropic equivalents of the two no-signaling constraints.

for all  $q \ge 1$  by using known inequalities for the Shannon entropy [47] and the causal constraints derived in Section 4.2. Using the entropy vector method for Shannon entropies, the following three classes of entropic inequalities were obtained for the Triangle causal structure (Figure 2.6b) in [47].<sup>9</sup> Including all permutations of X, Y and Z, these yield 7 inequalities.

$$-H(X) - H(Y) - H(Z) + H(XY) + H(XZ) \ge 0, \tag{4.7a}$$

$$-5H(X) - 5H(Y) - 5H(Z) + 4H(XY) + 4H(XZ) + 4H(YZ) - 2H(XYZ) \ge 0, \quad (4.7b)$$

$$-3H(X) - 3H(Y) - 3H(Z) + 2H(XY) + 2H(XZ) + 3H(YZ) - H(XYZ) \ge 0.$$
(4.7c)

By replacing the Shannon entropy H() with the Tsallis entropy  $S_q()$  on the left hand side of these inequalities and minimizing the resultant expression over our outer approximation to the classical Tsallis entropy cone for the Triangle causal structure, one can obtain valid Tsallis entropic inequalities for this causal structure. More precisely, the outer approximation to the classical Tsallis entropy cone for the Triangle is characterised by the  $6 + 6(6 - 1)2^{6-3} = 246$ independent Shannon constraints (monotonicity and strong subadditivity constraints) and the 126 causal constraints (one for each conditional independence implied by the causal structure). To perform this minimization we used LPASSUMPTIONS [62] (Appendix 4.5.2), a Mathematica package that we developed for solving linear programs involving unspecified variables (other than those being optimised over), by implementing the simplex method (Section 2.2.3.2). In our case, we assumed that the dimensions of all the unobserved nodes (A, B and C) are equal to  $d_u$  and those of all the observed nodes (X, Y and Z) is  $d_o$ , and so the unspecified variables are  $q \ge 1$ ,  $d_u \ge 2$  and  $d_o \ge 2$ . We obtained the following Tsallis entropic inequalities for the Triangle.

$$-S_q(X) - S_q(Y) - S_q(Z) + S_q(XY) + S_q(XZ) \ge B_1(q, d_o, d_u),$$
(4.8a)

$$-5S_q(X) - 5S_q(Y) - 5S_q(Z) + 4S_q(XY) + 4S_q(XZ) + 4S_q(YZ) - 2S_q(XYZ) \geq B_2(q, d_o, d_u) \coloneqq \max \left( B_{21}(q, d_o, d_u), B_{22}(q, d_o, d_u) \right),$$
(4.8b)

$$-3S_q(X) - 3S_q(Y) - 3S_q(Z) + 2S_q(XY) + 2S_q(XZ) + 3S_q(YZ) - S_q(XYZ) \ge B_3(q, d_o, d_u), \quad (4.8c)$$

where,

$$B_1(q, d_o, d_u) = -\frac{1}{q-1} \left( 1 - d_o^{1-q} \right) \left( 2 - d_o^{1-q} - d_u^{1-q} \right), \tag{4.9a}$$

<sup>&</sup>lt;sup>9</sup>Note that a tighter entropic characterization was found in [216] based on non-Shannon inequalities, and that the techniques introduced here could also be applied to these.

$$B_{21}(q, d_o, d_u) = -\frac{1}{q-1} \left( 11 + d_u^{3-3q} + 6d_o^{2-2q} + 3d_o^{1-q}d_u^{1-q} - 6d_u^{1-q} - 15d_o^{1-q} \right),$$

$$B_{22}(q, d_o, d_u) = -\frac{1}{q-1} \left( 10 + d_o^{1-q}d_u^{3-3q} + 5d_o^{2-2q} + 2d_o^{1-q}d_u^{1-q} - 5d_u^{1-q} - 13d_o^{1-q} \right),$$
(4.9b)

$$B_3(q, d_o, d_u) = -\frac{1}{q-1} \left( 6 + d_o^{1-q} d_u^{2-2q} + 3d_o^{2-2q} + d_o^{1-q} d_u^{1-q} - 3d_u^{1-q} - 8d_o^{1-q} \right).$$
(4.9c)

Note that  $\lim_{q\to 1} B_1 = \lim_{q\to 1} B_2 = \lim_{q\to 1} B_3 = 0 \quad \forall d_u, d_o \ge 2$ , recovering the original inequalities for Shannon entropies (Equations (4.7a)–(4.7c)) as a special case.

In [182], an upper bound on the dimensions of classical unobserved systems needed to reproduce a set of observed correlations is derived in terms of the dimensions of the observed systems. In the case of the Triangle causal structure with  $d_X = d_Y = d_Z = d_o$  and  $d_A = d_B = d_C = d_u$  as considered here, the result of [182] implies that all classical correlations  $P_{XYZ}$  can be reproduced by using hidden systems of dimension at most  $d_o^3 - d_o$ . Since the dimension of the unobserved systems is unknown, it makes sense to take the minimum of the derived bounds over all  $d_u$ between 2 and  $d_o^3 - d_o$ . By taking their derivative, one can verify that for q > 1 each of the functions  $B_1$ ,  $B_{21}$ ,  $B_{22}$  and  $B_3$  is monotonically decreasing in  $d_o$  and  $d_u$ , and hence that the minimum is obtained for  $d_u = d_o^3 - d_o$  for any given  $d_o \ge 2$ . It follows that for all q > 1 and  $d_o \ge 2$  relations of the same form as Equations (4.8a)–(4.8c) hold, with the quantities on the right hand sides replaced by

$$B_{1}^{*}(q, d_{o}) = B_{1}(q, d_{o}, d_{o}^{3} - d_{o})$$

$$= \frac{-1}{q-1} \left( 2 + d_{o}^{2-2q} - 3d_{o}^{1-q} + d_{o}(-d_{o} + d_{o}^{3})^{-q} - d_{o}^{3}(-d_{o} + d_{o}^{3})^{-q} - d_{o}^{2-q}(-d_{o} + d_{o}^{3})^{-q} + d_{o}^{4-q}(-d_{o} + d_{o}^{3})^{-q} \right)$$

$$(4.10a)$$

$$B_{21}^{*}(q, d_{o}) = B_{21}(q, d_{o}, d_{o}^{3} - d_{o})$$

$$= \frac{-1}{q-1} \left( 11 + 6d_{o}^{2-2q} - 15d_{o}^{1-q} + (-d_{o} + d_{o}^{3})^{3-3q} - 6(-d_{o} + d_{o}^{3})^{1-q} + 3d_{o}^{1-q}(-d_{o} + d_{o}^{3})^{1-q} \right),$$

$$B_{22}^{*}(q, d_{o}) = B_{22}(q, d_{o}, d_{o}^{3} - d_{o})$$

$$= \frac{-1}{q-1} \left( 10 + 5d_{o}^{2-2q} - 13d_{o}^{1-q} + d_{o}^{1-q}(-d_{o} + d_{o}^{3})^{3-3q} - 5(-d_{o} + d_{o}^{3})^{1-q} + 2d_{o}^{1-q}(-d_{o} + d_{o}^{3})^{1-q} \right),$$
(4.10b)

$$B_{3}^{*}(q,d_{o}) = B_{3}(q,d_{o},d_{o}^{3}-d_{o})$$
  
$$= \frac{-1}{q-1} \left( 6 + 3d_{o}^{2-2q} - 8d_{o}^{1-q} + d_{o}^{1-q}(-d_{o}+d_{o}^{3})^{2-2q} - 3(-d_{o}+d_{o}^{3})^{1-q} + d_{o}^{1-q}(-d_{o}+d_{o}^{3})^{1-q} \right).$$
(4.10c)

A quantum violation of any of these bounds would imply that no unobserved classical systems of arbitrary dimension could reproduce those quantum correlations.

**Remark 4.3.1.** Because they are monotonically decreasing, the bounds for  $d_u = d_o^3 - d_o$  are not as tight as the  $d_u$ -dependent bounds for general q > 1. Nevertheless, as  $q \to 1$ , all the bounds  $B^*(q, d_o)$  tend to 0, reproducing the known result of [90] for the Shannon case.

**Remark 4.3.2.** In some cases it may be interesting to show quantum violations of these inequalities for low values of  $d_u$ , hence ruling out classical explanations with hidden systems of low dimensions, while possibly leaving open the case of arbitrary classical explanations. This would be interesting if it could be established that using hidden quantum systems allows for much lower dimensions than for hidden classical systems, for example.

#### 4.3.1 Looking for quantum violations

It is known that the Triangle causal structure (Figure 2.6b) admits non-classical correlations such as Fritz's distribution [88]. The idea behind this distribution is to embed the CHSH game in the Triangle causal structure such that non-locality for the Triangle follows from the non-locality of the CHSH game. To do so, C is replaced by the sharing of a maximally entangled pair of qubits, and A and B are taken to be uniformly random classical bits. The observed variables X, Y and Z in Figure 2.6b are taken to be pairs of the form  $X := (\tilde{X}, B)$ ,  $Y := (\tilde{Y}, A)$  and Z := (A, B), where  $\tilde{X}$  and  $\tilde{Y}$  are generated by measurements on the halves of the entangled pair with B and A used to choose the settings such that the joint distribution  $P_{\tilde{X}\tilde{Y}|BA}$  maximally violates a CHSH inequality. By a similar post-processing of other non-local distributions in the bipartite Bell causal structure (Figure 2.6a) such as the Mermin-Peres magic square game [147, 161] and chained Bell inequalities [34], one can obtain other non-local distributions in the Triangle that cannot be reproduced using classical systems. We explore whether any of these violate any of our new inequalities.

Since the values of  $B_i(q, d_o, d_u)$  are monotonically decreasing in  $d_o$  and  $d_u$ , if a distribution realisable in a quantum causal structure does not violate the bounds (4.8a)–(4.8c) for all  $q \ge 1$ and some fixed values of  $d_o$  and  $d_u$ , then no violations are possible for  $d'_o > d_o$ ,  $d'_u > d_u$ . We therefore take the smallest possible values of  $d_o$  and  $d_u$  when showing that a particular distribution cannot violate any of the bounds.

For Fritz's distribution [88], C is a two-qubit maximally entangled state, A and B are binary random variables while X, Y and Z are random variables of dimension 4, i.e., the actual observed dimensions are  $(d_X, d_Y, d_Z) = (4, 4, 4)$  in this case. Here we see that taking  $d_o = 4$  and the smallest possible  $d_u$  which is  $d_u = 2$ , the left hand sides of Equations (4.8a)-(4.8c) evaluated for Fritz's distribution do not violate the corresponding bounds  $B_i(q, d_o = 4, d_u = 2)$  for any  $q \ge 1$ . This means that it is not possible to detect any quantum advantage of this distribution (even over the case where the unobserved systems are classical bits) using this method, and automatically implies that it cannot violate the bounds  $B_i(q, d_o = 4, d_u)$  for  $d_u \ge 2$ .

We also considered the chained Bell and magic square correlations embedded in the Triangle

	smallest			
Scenario	Ineq. (4.8a) $(i = 1)$	Ineq. (4.8b) $(i = 2)$	Ineq. $(4.8c)$ $(i = 3)$	observed dim. $(d_o^{\min})$
<i>N</i> = 2	2	2	2	4
N = 3	3	2	3	6
N = 4	4	2	4	8
N = 5	5	2	5	10
N = 6	6	2	6	12
N = 7	7	2	7	14
N = 8	8	2	8	16
N = 9	9	3	9	18
N = 10	10	3	10	20
Magic Sq.	4	2	4	9

Table 4.1: Values of  $d^i$  for the chained Bell and magic square correlations embedded in the Triangle causal structure. The values of N correspond to the number of inputs per party in the chained Bell inequality, which always has two outputs per party (the N = 2 case corresponds to Fritz's distribution [88]). When embedded in the Triangle, the number of outcomes of the observed nodes are  $(d_X, d_Y, d_Z) = (2N, 2N, N^2)$ . The last column of the table gives the minimum of the observed node dimensions  $(d_X, d_Y, d_Z)$  for each N, which is simply 2N. For the magic square, the dimensions  $(d_X, d_Y, d_Z)$  are (12, 12, 9). In all cases, the minimum value of  $d^i$  such that the Inequalities (4.8a)– (4.8c) with bounds  $B_i(q, d_o = d^i, d_u = 2)$  are not violated for any  $q \ge 1$  is less than the minimum observed dimension  $d_o^{\min}$ , and hence no violations of (4.8a)–(4.8c) could be found for the relevant case with  $d_o = d_o^{\min}$ .

causal structure analogously to the case discussed above. For each of these, we define  $d^i$  to be the smallest value of  $d_o$  for which the bound  $B_i(q, d_o = d^i, d_u = 2)$  cannot be violated for any q > 1. The values of  $d^i$  are given in Table 4.1 for the different cases of the chained Bell correlations and the magic square. Since the values of  $d^i$  are always lower than the smallest of the observed dimensions in the problem, and due to the monotonicity of the bounds it follows that none of these quantum distributions violate any of our inequalities when the observed dimension is set to  $d_o^{\min}$ .

We further checked for violations of Inequalities (4.8a)-(4.8c) by sampling random quantum states for the systems A, B and C and random quantum measurements whose outcomes would correspond to the classical variables X, Y and Z. The value of q was also sampled randomly between 1 and 100. We considered the cases where the shared systems were pairs of qubits with 4 outcome measurements ( $d_X = d_Y = d_Z = 4$ ) and qutrits with 9 outcome measurements ( $d_X = d_Y = d_Z = 9$ ) but were unable to find violations of any of the inequalities even for the bounds with the  $d_o = 4$ ,  $d_u = 2$  (two qubit case) and  $d_o = 9$ ,  $d_u = 2$  (two qutrit case), i.e., the bounds obtained when the unobserved systems are classical bits.

**Remark 4.3.3.** In the derivation of Inequalities (4.8a)–(4.8c), we set the dimensions of the observed nodes X, Y and Z to all be equal and those of the unobserved nodes A, B and C to also all be equal. One could in principle repeat the same procedure taking different dimensions for all 6 variables but we found the computational procedure too demanding. However, Table 4.1

shows that even when we consider the bounds  $B_i(q, d_o, d_u)$  with  $d_o$  and  $d_u$  much smaller than the actual dimensions, known non-local distributions in the Triangle considered in Table 4.1 do not violate the corresponding Inequalities (4.8a)–(4.8c) for any  $q \ge 1$ . Since the bounds are monotonically decreasing in  $d_u$  and  $d_o$ , even if we obtained the general bounds for arbitrary dimensions of X, Y, Z, A, B and C, they would be strictly weaker than  $B_i(q, d^i, d_u = 2)$  $\forall i \in \{1, 2, 3\}, q \ge 1$  and can certainly not be violated by these distributions.

## 4.4 Discussion

We have investigated the use of Tsallis entropies within the entropy vector method to causal structures, showing how causal constraints imply bounds on the Tsallis entropies of the variables involved. Although Tsallis entropies for  $q \ge 1$  possess many properties that aid their use in the entropy vector method, the nature of the causal constraints makes the problem significantly more computationally challenging than in the case of Shannon entropy. This meant that we were unable to complete the desired computations in the former case, even for some of the simplest causal structures. Nevertheless, we were able to derive new classical causal constraints expressed in terms of Tsallis entropy by analogy with known Shannon constraints, but were unable to find cases where these were violated, even using quantum distributions that are known not to be classically realisable. This mirrors an analogous result for Shannon entropies [216].

One could also investigate whether other entropic quantities could be used in a similar way, and in Appendix 4.5.3 we have considered the use of Rényi entropies. The Rényi entropies of order  $\alpha$  do not satisfy strong subadditivity for  $\alpha \neq 0, 1$ , while the Rényi as well as the min and max entropies fail to obey the chain rules for conditional entropies. Thus, use of these in the entropy vector method, would require an entropy vector with components for all possible conditional entropies as well as unconditional ones, considerably increasing the dimensionality of the problem, which we would expect to make the computations harder. In some cases, not having a chain rule has not been prohibitive [217], but our results of Appendix 4.5.3 reveal significant limitations of Rényi entropies for analysing causal structure.

Further, one could consider using algorithms other than Fourier-Motzkin elimination to obtain non-trivial Tsallis entropic constraints over observed nodes starting from the Tsallis cone over all the nodes (see e.g., [99]). These could in principle yield solutions even in cases where FM elimination becomes intractable. However, we found that the FM elimination procedure became intractable even when starting out with only a small subset of the Tsallis entropic causal constraints for a simple causal structure such as the Bell one. This suggests that the difficulty is not only with the number of constraints, but also with their nature (in particular, that they are not equalities and depend non-trivially on the dimensions). Consequently, we bypassed FM elimination and used an alternative technique to obtain new Tsallis entropic inequalities for the Triangle causal structure (Section 4.3).

It is also worth noting that the following alternative definition of the Tsallis conditional entropy

was proposed in [3].

$$\tilde{S}_{q}(X|Y) = \frac{1}{1-q} \frac{\sum_{y} P(y)^{q} S_{q}(X|Y=y)}{\sum_{y} P(y)^{q}} = \frac{1}{1-q} \left( \frac{\sum_{x,y} P(xy)^{q}}{\sum_{y} P(y)^{q}} - 1 \right).$$
(4.11)

Using this definition, Tsallis entropies would satisfy the same causal constraints as the Shannon entropy (Equation (2.49)). However, the conditional entropies defined this way do not satisfy the chain rules of Equation (2.20) but instead obey a non-linear chain rule,  $S_q(XY) = S_q(X) + S_q(Y|X) + (1-q)S_q(X)S_q(Y|X)$  [3]. This would again mean that conditional entropies would need to be included in the entropy vector. Furthermore, since Fourier-Motzkin elimination only works for linear constraints, an alternative algorithm would be required to use this chain rule in conjunction with the entropy vector method.

That the inequalities for Tsallis entropy derived in this work depend on the dimensions of the systems involved could be used to certify that particular observed correlations in a classical causal structure require a certain minimal dimension of unobserved systems to be realisable. To show this would require showing that classically-realisable correlations violate one of the inequalities for some  $d_u$ . Such bounds would then complement the upper bounds of [182]. However, in some cases we know our bounds are not tight enough to do this. As a simple example, within the Triangle causal structure we tried taking  $X = (X_B, X_C)$ ,  $Y = (Y_A, Y_C)$  and  $Z = (Z_A, Z_B)$  with  $X_B = Z_B$ ,  $X_C = Y_C$  and  $Y_A = Z_A$  where each are uniformly distributed with cardinality D, for  $D \in \{3, \ldots, 10\}$ . In this case it is clear that the correlations cannot be achieved with classical unobserved systems with  $d_u = 2$ . Taking the bound with  $d_u = 2$  and  $d_o = D^2$  no violations of (4.8a)-(4.8c) were seen by plotting the graphs for  $q \in [1, 20]$ , for the range of D above. Hence, our bounds are too loose to certify lower bounds on  $d_u$  in this case.

While our analysis highlights significant drawbacks of using Tsallis entropies for analysing causal structures, it does not rule out the possibility of Tsallis entropies being able to detect the classical-quantum gap<sup>10</sup> in these causal structures, or others. To overcome the difficulties we encountered we would either need increased computational power, and/or the development of new, alternative techniques for analysing causal structures (with or without entropies).

## 4.5 Appendix

#### 4.5.1 Quantum generalisations of Theorems 4.2.1 and 4.2.2

The main results of this Chapter only required the Tsallis entropic causal constraints of Theorems 4.2.1 and 4.2.2, which were derived for classical causal structures. Here, we present addi-

<sup>&</sup>lt;sup>10</sup>Proving that Tsallis entropies are unable to do this would also be difficult. For instance, the proof of [214] that Shannon entropies are unable to detect the gap in line-like causal structures involves first characterising the marginal polytope through Fourier-Motzkin elimination, which itself proved to be computationally infeasible with Tsallis entropies even for the simplest line-like causal structure, the bipartite Bell scenario.

tional results that generalise these theorems to certain quantum causal structures. For these, we recap some of the properties of quantum Tsallis entropies encountered in Section 2.3.2.

Tsallis entropies as defined for classical random variables in Section 2.3.2 are easily generalised to the quantum case by replacing the probability distribution by a density matrix [119]. For a quantum system described by the density matrix  $\rho \in \mathscr{S}(\mathscr{H})$  on the Hilbert space  $\mathscr{H}$  and q > 0, the quantum Tsallis entropy is defined by

$$S_{q}(\rho) = \begin{cases} -\operatorname{Tr} \rho^{q} \ln_{q} \rho, & q \neq 1. \\ H(\rho), & q = 1. \end{cases}$$
(4.12)

where  $H(\rho) = -\operatorname{Tr} \rho \ln \rho$  is the von Neumann entropy of  $\rho$  and  $\ln_q(x) = \frac{x^{1-q}-1}{1-q}$  as in Section 2.3.2.<sup>11</sup>

Given a density operator  $\rho_{AB} \in \mathscr{S}(\mathscr{H}_A \otimes \mathscr{H}_B)$ , the conditional quantum Tsallis entropy of A given B can then be defined by  $S_q(A|B)_{\rho} = S_q(AB) - S_q(B)$ , the mutual information between A and B by  $I_q(A : B)_{\rho} = S_q(A) + S_q(B) - S_q(AB)$ , and for  $\rho_{ABC} \in \mathscr{S}(\mathscr{H}_A \otimes \mathscr{H}_B \otimes \mathscr{H}_C)$  the conditional Tsallis information between A and B given C is defined by  $I_q(A : B|C)_{\rho} = S_q(A|C) + S_q(B|C) - S_q(AB|C)$ . In this section we use  $d_S$  to represent the dimensions of the Hilbert space  $\mathscr{H}_S$ .

The following properties of quantum Tsallis entropies will be useful for what follows.

1. **Pseudo-additivity** [200]: If  $\rho_{AB} = \rho_A \otimes \rho_B$ , then

$$S_q(AB) = S_q(A) + S_q(B) + (1 - q)S_q(A)S_q(B).$$
(4.13)

- 2. Upper bound [13]: For all q > 0, we have  $S_q(A) \le \ln_q d_A$  and equality is achieved if and only if  $\rho_A = \mathbb{1}_A/d_A$ .
- 3. Subadditivity [13]: For any density matrix  $\rho_{AB}$  with marginals  $\rho_A$  and  $\rho_B$ , the following holds for all  $q \ge 1$ ,

$$S_q(AB) \le S_q(A) + S_q(B). \tag{4.14}$$

Using these we can generalize Theorem 4.2.1 to the quantum case. This corresponds to the causal structure with two independent quantum nodes and no edges in between them.

**Theorem 4.5.1.** For all bipartite density operators in product form, i.e.,  $\rho_{AB} = \rho_A \otimes \rho_B$  with  $\rho_A \in \mathscr{S}(\mathscr{H}_A)$  and  $\rho_B \in \mathscr{S}(\mathscr{H}_B)$ , the quantum Tsallis mutual information  $I_q(A : B)_{\rho}$  is upper bounded as follows for all q > 0

$$I_q(A:B)_{\rho} \le f(q, d_A, d_B),$$

<sup>&</sup>lt;sup>11</sup>Analogously to the classical case we keep it implicit that if  $\rho$  has any 0 eigenvalues these do not contribute to the trace.

where the function  $f(q, d_A, d_B)$  is given by

$$f(q, d_A, d_B) = \frac{1}{(q-1)} \left( 1 - \frac{1}{d_A^{q-1}} \right) \left( 1 - \frac{1}{d_B^{q-1}} \right) = (q-1) \ln_q d_A \ln_q d_B.$$

The bound is saturated if and only if  $\rho_{AB} = \frac{\mathbb{1}_A}{d_A} \otimes \frac{\mathbb{1}_B}{d_B}$ .

*Proof.* The proof goes through in the same way as the proof of Theorem 4.2.1 for the classical case (Properties 1 and 2 are analogous to those needed in the classical proof).  $\Box$ 

Next, we generalise Theorem 4.2.2 and Corollaries 4.2.1 and 4.2.2. This would correspond to the causal constraints on quantum Tsallis entropies implied by the common cause causal structure with C being a complete common cause of A and B (which share no causal relations among themselves). Here, one must be careful in precisely defining the conditional mutual information and interpreting it physically. For example, if the common case C were quantum and the nodes A and B were classical outcomes of measurements on C, then A, B and C do not coexist and there is no joint state  $\rho_{ABC}$  in such a case. This is a significant difference in quantum causal modelling compared to the classical case, and there have been several proposals for how do deal with it [132, 67, 6, 163]. In the following we consider two cases:

- 1. When C is classical, all 3 systems coexist and  $\rho_{ABC}$  can be described by a classicalquantum state (See Theorem 4.5.2).
- 2. When C is quantum, one approach is to view  $\rho_{ABC}$  not as the joint state of the 3 systems but as being related to the Choi-Jamiolkowski representations of the quantum channels from C to A and B (See Section 4.5.1.1) as done in [6].

The following Lemma proven in [125] is required for our generalization of Theorem 4.2.2 in the first case.

**Lemma 4.5.1** ([125], Lemma 1). Let  $\mathscr{H}_A$  and  $\mathscr{H}_Z$  be two Hilbert spaces and  $\{|z\}\}_z$  be an orthonormal basis of  $\mathscr{H}_Z$ . Let  $\rho_{AZ}$  be classical on  $\mathscr{H}_Z$  with respect to this basis i.e.,

$$\rho_{AZ} = \sum_{z} P(z) \rho_A^{(z)} \otimes |z\rangle \langle z|,$$

where  $\sum_{z} P(z) = 1$  and  $\rho_A^{(z)} \in S(\mathscr{H}_A) \ \forall z$ . Then for all q > 0,

$$S_q(AZ)_{\rho} = \sum_z P(z)^q S_q(\rho_A^{(z)}) + S_q(Z),$$

where  $S_q(Z)$  is the classical Tsallis entropy of the variable Z distributed according to  $P_Z$ .

Note that the above Lemma immediately implies that

$$S_q(A|Z)_{\rho} = \sum_{z} P(z)^q S_q(\rho_A^{(z)}).$$
(4.15)

**Theorem 4.5.2.** Let  $\rho_{ABC} = \sum_{c} P(c) \rho_{AB}^{(c)} \otimes |c\rangle \langle c|$ , where  $\rho_{AB}^{(c)} = \rho_{A}^{(c)} \otimes \rho_{B}^{(c)} \forall c$ , then, for all  $q \ge 1$ ,  $I_q(A:B|C)_{\rho_{ABC}} \le f(q, d_A, d_B)$ .

For q > 1 the bound is saturated if and only if  $\rho_{ABC} = \frac{\mathbb{1}_A}{d_A} \otimes \frac{\mathbb{1}_B}{d_B} \otimes |c\rangle \langle c|_C$ .

*Proof.* Using (4.15) we have,

$$\begin{split} I_q(A:B|C)_{\rho_{ABC}} &= S_q(A|C)_{\rho} + S_q(B|C)_{\rho} - S_q(AB|C)_{\rho} \\ &= \sum_c P(c)^q [S_q(\rho_A^{(c)}) + S_q(\rho_B^{(c)}) - S_q(\rho_{AB}^{(c)})] \\ &= \sum_c P(c)^q I_q(A:B)_{\rho_{AB}^{(c)}}. \end{split}$$

The rest of the proof is analogous to Theorem 4.2.2, where using the above, Theorem 4.5.1 and defining the set  $\mathcal{R} = \{\rho_{ABC} \in \mathcal{H}_A \otimes \mathcal{H}_B \otimes \mathcal{H}_C : \rho_{ABC} = \sum_c P(c)\rho_A^{(c)} \otimes \rho_B^{(c)} \otimes |c\rangle \langle c|\}$  we have,

$$\begin{aligned} \max_{\mathcal{R}} I_q(A:B|C)_{\rho} &= \max_{\mathcal{R}} \sum_{c} P(c)^q I_q(A:B)_{\rho_{AB}^{(c)}} \\ &\leq \max_{\{P(c)\}_c} \sum_{c} P(c)^q(c) \max_{\{\rho_A^{(c)}\}_c, \{\rho_B^{(c)}\}_c} I_q(A:B)_{\rho_{AB}^{(c)}} \\ &= f(q, d_A, d_B), \end{aligned}$$

where the last step follows because for all  $q \ge 1$ ,  $\sum_{c} P(c)^{q}$  is maximized by deterministic distributions over C with a maximum value of  $1^{12}$  and  $I_q(A:B)_{\rho_{AB}^{(c)}}$  for product states is maximised by the maximally mixed state over A and B for all c (Theorem 4.5.1). Thus, for q > 1, the bound is saturated if and only if  $\rho_{ABC} = \frac{\mathbb{1}_A}{d_A} \otimes \frac{\mathbb{1}_B}{d_B} \otimes |c\rangle \langle c|_C$  for some value c of C.

#### 4.5.1.1 A generalisation: when systems do not coexist

There is a fundamental problem with naively generalising classical conditional independences such as  $P_{XY|Z} = P_{X|Z}P_{Y|Z}$  to the quantum case by replacing joint distributions by density matrices: it is not clear what is meant by a conditional quantum state e.g.,  $\rho_{A|C}$  since it is not clear what it means to condition on a quantum system, specially when the (joint state of the) system under consideration and the one being conditioned upon do not coexist. There are a number of approaches for tackling this problem, from describing quantum states in space and time on an equal footing [117] to quantum analogues of Bayesian inference [132] and causal modelling [67, 6, 163]. In the following, we will focus on one such approach that is motivated by the framework of [6]. Central to this approach is the Choi-Jamiołkowski isomorphism [121, 54] from which one can define conditional quantum states.

 $<sup>^{12}\</sup>mathrm{For}~q>1$  such deterministic distributions are the only way to obtain the bound.



Figure 4.1: A circuit decomposition of the channel  $\mathscr{E} : \mathscr{S}(\mathscr{H}_{\mathbf{C}}) \to \mathscr{S}(\mathscr{H}_{\mathbf{A}} \otimes \mathscr{H}_{\mathbf{B}})$ when **C** is a *complete common cause* of **A** and **B**: If the map  $\mathscr{E}$  from the system *C* to the systems *A* and *B* can be decomposed as shown here, then *C* is a complete common cause of *A* and *B* ([6]). We build up our result step by step considering the channels given by  $\mathscr{E}_{\mathbf{I}}$  (unitary),  $\mathscr{E}_{\mathbf{II}}$  (unitary followed by local isometries) and  $\mathscr{E}_{\mathbf{III}} = \mathscr{E}$ .

**Definition 4.5.1** (Choi state). Let  $|\gamma\rangle = \sum_i |i\rangle_R |i\rangle_{R^*} \in \mathscr{H}_R \otimes \mathscr{H}_{R^*}$ , where  $\mathscr{H}_{R^*}$  is the dual space to  $\mathscr{H}_R$  and  $\{|i\rangle_R\}_i$ ,  $\{|i\rangle_{R^*}\}_i$  are orthonormal bases of  $\mathscr{H}_R$  and  $\mathscr{H}_{R^*}$  respectively. Given a channel  $\mathscr{E}_{R|S} : \mathscr{S}(\mathscr{H}_R) \to \mathscr{S}(\mathscr{H}_S)$ , the *Choi state of the channel* is defined by

$$\rho_{S|R} = (\mathscr{E}_{R|S} \otimes \mathcal{I})(|\gamma\rangle\langle\gamma|) = \sum_{ij} \mathscr{E}(|i\rangle\langle j|_R) \otimes |i\rangle\langle j|_{R^*}$$

Thus,  $\rho_{S|R} \in \mathscr{P}(\mathscr{H}_S \otimes \mathscr{H}_{R^*})$ .

Now, if a quantum system C evolves through a unitary channel  $\mathscr{E}_{I}(\cdot) = U'(\cdot)U'^{\dagger}$  to two systems A' and B' where  $U' : \mathscr{H}_{C} \to \mathscr{H}_{A'} \otimes \mathscr{H}_{B'}$ , it is reasonable to call the system C a quantum common cause of the systems A' and B'. Further, this would still be reasonable if one were to then perform local completely positive trace preserving (CPTP) maps on the A' and B' systems. By the Stinespring dilation theorem, these local CPTP maps can be seen as local isometries followed by partial traces, and the local isometries can be seen as the introduction of an ancilla in a pure state followed by a joint unitary on the system and ancilla. This is illustrated in Figure 4.1 and is compatible with the definition of quantum common causes presented in [6]. In other words, a system C can be said to be a complete (quantum) common cause of systems A and B if the corresponding channel  $\mathscr{E} : \mathscr{S}(\mathscr{H}_{C}) \to \mathscr{S}(\mathscr{H}_{A} \otimes \mathscr{H}_{B})$  can be decomposed as in Figure 4.1 for some choice of unitaries  $U', U_A, U_B$  and pure states  $|\phi\rangle_{E_A}, |\psi\rangle_{E_B}$ . Note that a more general set of channels fit the definition of quantum common cause in Ref. [6] than we use here; whether the theorems here extend to this case we leave as an open question.

In [6] it is shown that whenever a system C is a complete common cause of systems A and B then the Shannon conditional mutual information evaluated on the state  $\tau_{ABC^*} = \frac{1}{d_A} \rho_{AB|C}$ 

satisfies  $I(A:B|C^*)_{\tau} = 0$  where  $\rho_{AB|C}$  is the Choi state of the channel from C to A and B. We generalise this result to Tsallis entropies for  $q \ge 1$  for certain types of channels. We present the result in three cases, each with increasing levels of generality. These are explained in Figure 4.1 and correspond to the cases where the map from the complete common cause C to its children A and B is (I) unitary ( $\mathscr{E}_{I} = U'$ ); (II) unitary followed by local isometries ( $\mathscr{E}_{II}$ ); (III) Unitary followed by local isometries followed by partial traces on local systems ( $\mathscr{E}_{III} = \mathscr{E}$ ).

**Lemma 4.5.2.** Let  $\mathscr{E}_{I} : \mathscr{S}(\mathscr{H}_{C}) \to \mathscr{S}(\mathscr{H}_{A'} \otimes \mathscr{H}_{B'})$  be a unitary quantum channel i.e.,

 $\mathscr{E}_{\mathrm{I}}(\cdot) = U'(\cdot)U'^{\dagger},$ 

where  $U': \mathscr{H}_C \to \mathscr{H}_{A'} \otimes \mathscr{H}_{B'}$  is an arbitrary unitary operator. If  $\rho_{A'B'|C}$  is the corresponding Choi state, then the Tsallis conditional mutual information evaluated on the state  $\tau_{A'B'C^*} = \frac{1}{d_C}\rho_{A'B'|C} \in \mathscr{S}(\mathscr{H}_{A'} \otimes \mathscr{H}_{B'} \otimes \mathscr{H}_{C^*})$  satisfies

$$I_q(A':B'|C^*)_{\tau} = f(q, d_{A'}, d_{B'}) \qquad \forall q > 0.$$

*Proof.* The conditional mutual information  $I_q(A':B'|C^*)_{\tau}$  can be written as

$$I_q(A':B'|C^*)_{\tau} = \frac{1}{q-1} \left( \operatorname{Tr}_{A'B'C^*} \tau^q_{A'B'C^*} + \operatorname{Tr}_{C^*} \tau^q_{C^*} - \operatorname{Tr}_{A'C^*} \tau^q_{A'C^*} - \operatorname{Tr}_{B'C^*} \tau^q_{B'C^*} \right).$$
(4.16)

We will now evaluate every term in the above expression for the case where the channel that maps the C system to the A' and B' systems is unitary. In this case,  $\tau_{A'B'C^*}$  is a pure state and can be written as  $\tau_{A'B'C^*} = |\tau\rangle\langle\tau|_{A'B'C^*}$  where

$$|\tau\rangle_{A'B'C^*} = \frac{1}{\sqrt{d_C}} \sum_i U'|i\rangle_C \otimes |i\rangle_{C^*}.$$
(4.17)

This means that  $\operatorname{Tr}_{A'B'C^*} \tau^q_{A'B'C^*} = \operatorname{Tr}_{A'B'C^*} \tau_{A'B'C^*} \quad \forall q > 0$ . Since  $\tau_{A'B'C^*}$  is a valid quantum state, it must be a trace one operator and we have

$$\operatorname{Tr}_{A'B'C^*} \tau^q_{A'B'C^*} = 1 \quad \forall q > 0$$
(4.18)

Further, we have  $\tau_{C^*} = \operatorname{Tr}_{A'B'} \tau_{A'B'C^*} = \frac{\mathbbm{1}_{C^*}}{d_C}$  and hence

$$\operatorname{Tr}_{C^*} \tau_{C^*}^q = \frac{1}{d_C^{q-1}} = \frac{1}{d_{A'}^{q-1} d_{B'}^{q-1}}.$$
(4.19)

The second step follows from the fact that  $U': \mathscr{H}_C \to \mathscr{H}_{A'} \otimes \mathscr{H}_{B'}$  is unitary so  $d_C = d_{A'}d_{B'}$ .

Now, the marginals over A' and B' are  $\tau_{A'} = \operatorname{Tr}_{B'C^*} \tau_{A'B'C^*} = \frac{\mathbb{1}_{A'}}{d_{A'}}$  and  $\tau_{B'} = \operatorname{Tr}_{A'C^*} \tau_{A'B'C^*} = \frac{\mathbb{1}_{B'}}{d_{B'}}$ . By the Schmidt decomposition of  $\tau_{A'B'C^*}$ , the non-zero eigenvalues of  $\tau_{A'}$  are the same as those of  $\tau_{B'C^*}$ . Since the Tsallis entropy depends only on the non-zero eigenvalues,  $S_q(A') = S_q(B'C^*)$  and hence

$$\operatorname{Tr}_{B'C^*} \tau^q_{B'C^*} = d_{A'} \left( \frac{1}{d^q_{A'}} \right) = \frac{1}{d^{q-1}_{A'}}.$$
(4.20)

By the same argument it follows that

$$\operatorname{Tr}_{A'C^*} \tau^q_{A'C^*} = d_{B'} \left( \frac{1}{d^q_{B'}} \right) = \frac{1}{d^{q-1}_{B'}}.$$
(4.21)

Combining Equations (4.16)-(4.21), we have

$$I_q(A':B'|C^*)_{\tau} = \frac{1}{q-1} \left( 1 + \frac{1}{d_{A'}^{q-1}d_{B'}^{q-1}} - \frac{1}{d_{A'}^{q-1}} - \frac{1}{d_{B'}^{q-1}} \right) = f(q, d_{A'}, d_{B'}) \quad \forall q > 0.$$

$$(4.22)$$

**Lemma 4.5.3.** Let  $\mathscr{E}_{II} : \mathscr{S}(\mathscr{H}_C) \to \mathscr{S}(\mathscr{H}_{\tilde{A}} \otimes \mathscr{H}_{\tilde{B}})$  be a quantum channel of the form

$$\mathscr{E}_{\mathrm{II}}(\cdot) = (U_A \otimes U_B) \Big[ |\phi\rangle \langle \phi|_{E_A} \otimes U'(\cdot) U'^{\dagger} \otimes |\psi\rangle \langle \psi|_{E_B} \Big] (U_A \otimes U_B)^{\dagger},$$

where  $U': \mathcal{H}_C \to \mathcal{H}_{A'} \otimes \mathcal{H}_{B'}, U_A: \mathcal{H}_{E_A} \otimes \mathcal{H}_{A'} \to \mathcal{H}_{\tilde{A}} \text{ and } U_B: \mathcal{H}_{B'} \otimes \mathcal{H}_{E_B} \to \mathcal{H}_{\tilde{B}} \text{ are arbitrary unitaries and } |\phi\rangle_{E_A} \text{ and } |\psi\rangle_{E_B} \text{ are arbitrary pure states. If } \rho_{\tilde{A}\tilde{B}|C} \text{ is the corresponding Choi state, then the Tsallis conditional mutual information evaluated on the state } \tau_{\tilde{A}\tilde{B}C^*} = \frac{1}{d_C}\rho_{\tilde{A}\tilde{B}|C} \in \mathcal{S}(\mathcal{H}_{\tilde{A}} \otimes \mathcal{H}_{\tilde{B}} \otimes \mathcal{H}_{C^*}) \text{ satisfies}$ 

$$I_q(A:B|C^*)_{\tau} = f(q, d_{A'}, d_{B'}) \qquad \forall q > 0.$$

Proof. Note that the map  $\mathscr{E}_{\Pi}$  is the unitary map  $\mathscr{E}_{I}(\cdot) = U'(\cdot)U'^{\dagger}$  followed by local isometries  $V_A$  and  $V_B$  on the A' and B' systems respectively. Since the expression for the conditional mutual information  $I_q(\tilde{A} : \tilde{B}|C^*)_{\tau}$  can be written in terms of entropies, which are functions of the eigenvalues of the relevant reduced density operators, and since the eigenvalues are unchanged by local isometries, this conditional mutual information is invariant under local isometries. The rest of the proof is identical to that of Lemma 4.5.2 resulting in

$$I_q(\tilde{A}:\tilde{B}|C^*)_{\tau} = I_q(A':B'|C^*)_{\tau} = f(q, d_{A'}, d_{B'}) \qquad \forall q > 0.$$
(4.23)

For the last case where  $\mathscr{E}_{\text{III}}(\cdot) = \text{Tr}_{A''B''} \left[ (U_A \otimes U_B) \left[ |\phi\rangle \langle \phi|_{E_A} \otimes U'(\cdot)U'^{\dagger} \otimes |\psi\rangle \langle \psi|_{E_B} \right] (U_A \otimes U_B)^{\dagger} \right]$ , one could intuitively argue that tracing out systems could not increase the mutual information and one would expect that

$$I_q(AA'':BB''|C^*)_{\tau} \ge I_q(A:B|C^*)_{\tau}.$$
(4.24)

Since  $I_q(AA'': BB''|C^*)_{\tau} = I_q(A: B|C^*)_{\tau} + I_q(AA'': B''|BC^*)_{\tau} + I_q(A'': B|AC^*)_{\tau}$ , Equation (4.24) would follow from strong subadditivity used twice i.e.,  $I_q(AA'': B''|BC^*)_{\tau} \ge 0$  and  $I_q(A'': B|AC^*)_{\tau} \ge 0$ . However, it is known that strong subadditivity does not hold in general for Tsallis entropies for q > 1 [162]. Ref. [162] also provides a sufficiency condition for strong subadditivity to hold for Tsallis entropies. In the following Lemma, we provide another, simple sufficiency condition that also helps bound the Tsallis mutual information  $I_q(AA'': B|C)_{\tau}$  (or  $I_q(A:BB''|C)_{\tau}$ ) corresponding to the map  $\mathscr{E}_{\text{III}}$  where only one of A'' or B'' is traced out but not both.

**Lemma 4.5.4** (Sufficiency condition for strong subadditivity of Tsallis entropies). If  $\rho_{ABC}$  is a pure quantum state, then for all  $q \ge 1$  we have  $I_q(A:B|C)_{\rho} \ge 0$ .

Proof. We have

$$I_q(A:B|C) = S_q(AC) + S_q(BC) - S_q(ABC) - S_q(C).$$

Since  $\rho_{ABC}$  is pure we have  $S_q(ABC) = 0 \forall q > 0$  and (from the Schmidt decomposition argument mentioned earlier)  $S_q(AC) = S_q(B)$ ,  $S_q(BC) = S_q(A)$  and  $S_q(C) = S_q(AB)$ . Thus,

 $I_q(A:B|C) = S_q(A) + S_q(B) - S_q(AB) = I_q(A:B) \ge 0,$ 

which follows from subadditivity of quantum Tsallis entropies for  $q \ge 1$  [13]. In other words, for pure  $\rho_{ABC}$ , strong subadditivity of Tsallis entropies is equivalent to their subadditivity which holds whenever  $q \ge 1$ .

**Corollary 4.5.1.** Let  $\mathscr{E}_{III}^1 : \mathscr{S}(\mathscr{H}_C) \to \mathscr{S}(\mathscr{H}_{\tilde{A}} \otimes \mathscr{H}_B)$  be a quantum channel of the form

$$\mathscr{E}_{\mathrm{III}}^{1}(\cdot) = \mathrm{Tr}_{B''} \Big[ (U_A \otimes U_B) \Big[ |\phi\rangle \langle \phi|_{E_A} \otimes U'(\cdot) U'^{\dagger} \otimes |\psi\rangle \langle \psi|_{E_B} \Big] (U_A \otimes U_B)^{\dagger} \Big],$$

where  $U': \mathcal{H}_C \to \mathcal{H}_{A'} \otimes \mathcal{H}_{B'}, U_A: \mathcal{H}_{E_A} \otimes \mathcal{H}_{A'} \to \mathcal{H}_{\tilde{A}} \cong \mathcal{H}_A \otimes \mathcal{H}_{A''} \text{ and } U_B: \mathcal{H}_{B'} \otimes \mathcal{H}_{E_B} \to \mathcal{H}_{\tilde{B}} \cong \mathcal{H}_B \otimes \mathcal{H}_{B''} \text{ are arbitrary unitaries and } |\phi\rangle_{E_A} \text{ and } |\psi\rangle_{E_B} \text{ are arbitrary pure states. If } \rho_{\tilde{A}B|C} \text{ is the corresponding Choi state, then the Tsallis conditional mutual information evaluated on the state } \tau_{\tilde{A}BC^*} = \frac{1}{d_C}\rho_{\tilde{A}B|C} \in \mathscr{S}(\mathcal{H}_{\tilde{A}} \otimes \mathcal{H}_B \otimes \mathcal{H}_{C^*}) \text{ satisfies}$ 

$$I_q(\tilde{A}:B|C^*) \coloneqq I_q(AA'':B|C^*)_\tau \le f(q,d_{A'},d_{B'}) \qquad \forall q \ge 1.$$

*Proof.* Since  $I_q(AA'':BB''|C^*)_{\tau} = I_q(AA'':B|C^*)_{\tau} + I_q(AA'':B''|BC^*)_{\tau}$ , the purity of  $\tau_{\tilde{A}\tilde{B}C^*} = \tau_{AA''BB''C^*}$  and Lemma 4.5.4 imply that

 $I_q(AA'':BB''|C^*)_{\tau} \ge I_q(AA'':B|C^*)_{\tau}, \forall q \ge 1,$ 

or (equivalently) in more concise notation,

$$I_q(\tilde{A}:\tilde{B}|C^*)_\tau \ge I_q(\tilde{A}:B|C^*)_\tau \quad \forall q \ge 1.$$

Finally, using Lemma 4.5.3 we obtain the required result.

Now, for Equation (4.24) to hold, we do not necessarily need strong subadditivity. Even if  $I_q(A'':B|AC)_{\tau} \ge 0$  does not hold, Equation (4.24) would still hold if  $I_q(AA'':B''|BC)_{\tau} + I_q(A'':B|AC)_{\tau} \ge 0$ . This motivates the following conjecture.

**Conjecture 4.5.1.** Let  $\mathscr{E}_{III} : \mathscr{S}(\mathscr{H}_C) \to \mathscr{S}(\mathscr{H}_A \otimes \mathscr{H}_B)$  be a quantum channel of the form

$$\mathscr{E}_{\text{III}}(\cdot) = \text{Tr}_{A''B''} \Big[ (U_A \otimes U_B) \Big[ |\phi\rangle \langle \phi|_{E_A} \otimes U'(\cdot) U'^{\dagger} \otimes |\psi\rangle \langle \psi|_{E_B} \Big] (U_A \otimes U_B)^{\dagger} \Big],$$

where  $U': \mathscr{H}_C \to \mathscr{H}_{A'} \otimes \mathscr{H}_{B'}, U_A: \mathscr{H}_{E_A} \otimes \mathscr{H}_{A'} \to \mathscr{H}_A \otimes \mathscr{H}_{A''} \text{ and } U_B: \mathscr{H}_{B'} \otimes \mathscr{H}_{E_B} \to \mathscr{H}_B \otimes \mathscr{H}_{B''} \text{ are arbitrary unitaries and } |\phi\rangle_{E_A} \text{ and } |\psi\rangle_{E_B} \text{ are arbitrary pure states. If } \rho_{AB|C} \text{ is the corresponding Choi state, then the Tsallis conditional mutual information evaluated on the state } \tau_{ABC^*} = \frac{1}{d_C}\rho_{AB|C} \in \mathscr{S}(\mathscr{H}_A \otimes \mathscr{H}_B \otimes \mathscr{H}_{C^*}) \text{ satisfies}$ 

$$I_q(A:B|C^*)_{\tau} \le f(q, d_{A'}, d_{B'}) \qquad \forall q \ge 1.$$

Notice that in Corollary 4.5.1 and Conjecture 4.5.1, the bounds are functions of  $d_{A'}$  and  $d_{B'}$ and not of the dimensions of the systems A and B (those in the quantity on the left hand side). In the case that  $d_A \ge d_{A'}$  and  $d_B \ge d_{B'}$ , the fact that  $f(q, d_A, d_B)$  is a strictly increasing function of  $d_A$  and  $d_B \forall q \ge 0$  allows us to write  $I_q(\tilde{A} : B | C^*)_{\tau} \le f(q, d_{\tilde{A}}, d_B)$  and  $I_q(A : B | C^*)_{\tau} \le$  $f(q, d_A, d_B)$  under the conditions of Corollary 4.5.1 and Conjecture 4.5.1 respectively. However, if  $d_A \le d_{A'}$  and/or  $d_B \le d_{B'}$ , the bounds  $f(q, d_{\tilde{A}}, d_B)$  and  $f(q, d_A, d_B)$  are tighter than the bound  $f(q, d_{A'}, d_{B'})$  and so not implied. However, based on the several examples that we have checked, we further conjecture the following.

Conjecture 4.5.2. Under the same conditions as Conjecture 4.5.1

$$I_q(A:B|C^*)_{\tau} \le f(q, d_A, d_B) \qquad \forall q \ge 1.$$

Further, it is shown in [6] that if C is a complete common cause of A and B then the corresponding Choi state,  $\rho_{AB|C}$  decomposes as  $\rho_{AB|C} = (\rho_{A|C} \otimes \mathbb{1}_B)(\mathbb{1}_A \otimes \rho_{B|C})$  or  $\rho_{AB|C} = \rho_{A|C}\rho_{B|C}$  in analogy with the classical case where if a classical random variable Z is a common cause of the random variables X and Y, then the joint distribution over these variables factorises as  $p_{XY|Z} = p_{X|Z}p_{Y|Z}$ . Then we have that  $\tau_{ABC^*} = \frac{1}{d_C}\rho_{AB|C} = \frac{1}{d_C}\rho_{A|C}\rho_{B|C}$ . By further analogy with the classical results of Section 4.2, one may also consider instead a state of the form  $\hat{\sigma}_{ABCC^*} = \sigma_C \otimes \frac{1}{d_C}\rho_{A|C}\rho_{B|C} = \sigma_C \otimes \tau_{ABC^*}$ , where  $\sigma_C \in \mathscr{S}(\mathscr{H}_C)$ .<sup>13</sup> Note that  $\hat{\sigma}_{ABCC^*}$  is a valid density operator on  $\mathscr{H}_A \otimes \mathscr{H}_B \otimes \mathscr{H}_C \otimes \mathscr{H}_{C^*}$ .

**Lemma 4.5.5.** The state  $\hat{\sigma}_{ABCC^*} = \sigma_C \otimes \tau_{ABC^*}$  defined above satisfies

$$I_q(A:B|CC^*)_{\hat{\sigma}} \leq f(q,d_A,d_B),$$

whenever  $I_q(A: B|C^*)_{\tau} \leq f(q, d_A, d_B)$  holds for the state  $\tau_{ABC^*} = \frac{1}{d_A}\rho_{AB|C}$ , where  $\rho_{AB|C}$  represents the quantum channel from C to A and B and  $\sigma_C$  is the input quantum state to this channel.

*Proof.* Since  $\hat{\sigma}$  is a product state between the C and  $ABC^*$  subsystems, by the pseudo-additivity of quantum Tsallis entropies and the chain rule we have

$$\begin{split} I_q(A:B|CC^*)_{\hat{\sigma}} &= S_q(ACC^*) + S_q(BCC^*) - S_q(ABCC^*) - S_q(CC^*) \\ &= S_q(AC^*) + S_q(BC^*) - S_q(ABC^*) - S_q(C^*) \\ &- (q-1)S_q(C) \left(S_q(AC^*) + S_q(BC^*) - S_q(ABC^*) - S_q(C^*)\right) \\ &= (1 - (1 - q)S_q(C))I(A:B|C^*) \\ &= \operatorname{Tr}(\sigma_C^q)I(A:B|C^*) \,. \end{split}$$

Now let  $p_c$  be the distribution whose entries are the eigenvalues of  $\sigma_C$ . We have  $\text{Tr}(\sigma_C^q) = \sum_c p_c^q$ . Thus if q > 1,  $\sum_c p_c^q \leq 1$  with equality if and only if  $p_c = 1$  for some value of c. It follows that

$$I_q(A:B|CC^*)_{\hat{\sigma}} \leq I_q(A:B|C^*)_{\tau}.$$

Therefore, if  $I_q(A:B|C^*)_{\tau} \leq f(q, d_A, d_B)$ , we also have  $I_q(A:B|CC^*)_{\hat{\sigma}} \leq f(q, d_A, d_B)$ .

<sup>&</sup>lt;sup>13</sup>This is the analogue of the statement  $P_{ABC} = P_C P_{A|C} P_{B|C}$  for probability distributions.

### 4.5.2 LPASSUMPTIONS: A new linear program solver for Mathematica

Our new Tsallis entropic inequalities for the Triangle causal structure, presented in Section 4.3 were obtained using the LPASSUMPTIONS Mathematica package that we developed for solving linear programs involving unspecified variables. The package is available as open source on Github [62]. Here we provide a brief overview of the main functionality of this package and how it was used for obtaining the results presented earlier in this chapter.

Linear programs are discussed in Section 2.2.3.1 of the preliminaries, but we restate the standard form of a linear program (according to the convention followed in this thesis) here for completeness. A generic linear programming problem can be specified by giving a vector c, a matrix M and a vector b such that the problem corresponds to

 $\begin{array}{l} \text{Minimize } c.x\\ \text{Subject to } Mx \geq b\\ x \geq 0 \end{array}$ 

Any linear programming problem can be written in this form and this is the same form as used by Mathematica's inbuilt LinearProgramming function. If c, M and b are completely specified (i.e., consist of numerical values), LinearProgramming can already be used to solve the problem. The aim of our package [62] is to also be able to cope with the case where there are unspecified constants in vector c representing the objective function. For example, we might wish to optimize  $x_1+ax_2$ , where a is an unspecified constant in the range  $0 \le a \le 1$ , returning the solution for all values of a in the range. In our case, for the results of Section 4.3, we optimised the entropic expressions of Equations (4.7a)-(4.7c) with the Shannon entropies H() replaced by Tsallis entropies  $S_q()$  over our outer approximation to the classical Tsallis entropy cone for the Triangle causal structure. This outer approximation is characterised by the Shannon constraints and our Tsallis entropic causal constraints (Corollary 4.2.2). The latter involve unspecified variables other than those being optimised over, namely the dimensions/cardinality of the variables and this optimisation cannot be performed using Mathematica's inbuilt LinearProgramming function, and requires our LPASSUMPTIONS package.

LPASSUMPTIONS works by using the two phase simplex algorithm [70] (Section 2.2.3.2). The algorithm involves visiting adjacent vertices of the feasible region (a polyhedron), starting from an initial vertex, until the vertex corresponding to the optimal solution is reached. The coefficients of the objective function and the constraints are encoded in a tableau and at each step, the next vertex to be visited is decided by checking the sign of certain elements in this tableau, which would indicate whether or not moving to that vertex would improve the value of the objective function. In our case, these elements of the tableau can depend on the additional unspecified variables in the problem. LPASSUMPTIONS relies on Mathematica's Simplify[expr,assump] command to decide on on the sign of these elements, which will then tell it how to proceed through the computation. Here, expr is an inequality that checks for the sign and assump are the constraints on the unspecified variables that can be input by the user, along with any additional assumptions on the expressions that are made during the computation. If Mathematica is unable to decide whether the inequality is true or false, the algorithm splits into two cases, one in which it assumes expr is true, and the other in which it assumes it is false. It then carries on adding additional assumptions as necessary until termination (or the number of iterations is exceeded). Problems can arise if expr is true (or false) but Mathematica's Simplify is unable to determine this (see the examples notebook mentioned in the next paragraph for such a case).

The main function of the package, is LPAssumptions[c, M, b, assump, (Options)]. It takes as input c (a vector corresponds to the objective function), M (the constraint matrix), b (the constraint vector) and assump (any initial assumptions on the unspecified variables involved). The output is a set of pairs  $\{(constr_i, vec_i)\}_i$ , where  $constr_i$  specifies the set of constraints under which the optimum is achieved by the vector vec<sub>i</sub>. With no assumptions and no unspecified variables, provided the problem is feasible and bounded, the answer given by LPAssumptions should match that of LinearProgramming (up to the slightly different structure of the output). Note that the current version of our package can only handle linear programs in which all the additional, unspecified variables occur in the objective function. Linear programs where all such variables occur in the constraint vector can also be solved using our program by the following trick: convert the original LP to the desired form (variables appearing only in the objective vector c) by taking the dual of the LP, solve the dual LP using LPASSUMPTIONS and use the duality theorems of Section 2.2.3.1 to obtain a solution for the original LP. We had to use this trick for obtaining our results of Section 4.3.

A detailed explanation of the various functionalities and commands can be found in the user manual we have made available along with the package [62], and a Mathematica notebook (LPAssumptions\_examples.nb) with examples illustrating the use of this package is available at http://www-users.york.ac.uk/~rc973/LPAssumptions.html. In the examples, we also show how to compute the local weight (defined in Appendix 5.6.3, see also [228, 66]) of a non-signaling probability distribution where the distribution has unspecified parameters (specifically we consider a noisy PR-box with inefficient detectors, see Section IV D F of [166]). This package was also used in some of the results of [207] which are presented in Chapter 5.

#### 4.5.3 Analysing causal structures using conditional Rényi entropies

Given the limitations of Shannon as well as Tsallis entropies found in [214] and the work presented in this chapter [206], a natural question that arises is— what about other generalised entropies? We already noted that other entropies like the Rényi, min or max entropies do not satisfy the chain rule (Table 2.1). The chain rule is important for the entropy vector method since it allows conditional entropies to be written in terms of unconditional ones (Equation (4.4)) and hence these need not be included as independent components of the entropy vector. In the absence of a chain rule, one would need to include the conditional entropies involving all possible subsets of nodes as additional components of the entropy vector, thereby significantly increasing the dimensionality of the problem and one would expect this to make the computations harder. This was an important motivation behind the choice of Tsallis entropies for the work presented in this chapter. However, in some cases, it has been possible to obtain non-trivial results using the entropic technique even in the absence of a chain rule [217]. In a side project with Mirjam Weilenmann, we investigated the possibility of using Rényi entropies for analysing causal structures. For this, we considered the conditional Rényi entropies of Definition 2.3.9. Some of the properties of Rényi entropies and this conditional version were discussed in Section 2.3.3 and summarised in Table 2.1, these include: the non-negativity of the entropy i.e.  $H_{\alpha}(X) \ge 0$ ,  $\forall \alpha \ge 0$ , additivity i.e., if  $P_{XY} = P_X P_Y$ , then  $H_{\alpha}(XY) = H_{\alpha}(X) + H_{\alpha}(Y)$  $\forall \alpha \ge 0$  [179], monotonicity i.e.,  $H_{\alpha}(X) \le H_{\alpha}(XY) \ \forall \alpha \in (0,1) \cup (1,\infty)$  [136] and the conditional form of strong subadditivity, also known as data processing  $H_{\alpha}(X|Y) \ge H_{\alpha}(X|YZ) \ \forall \alpha \ge 0$  [120]. There are further properties of these entropies which are already known or can be easily derived, as we show below. Importantly, one of these (Property 4.5.3) is a generalised chain rule for the conditional Rényi entropies.

• Property 1: (Non-negativity of conditional entropy)  $H_{\alpha}(X|Y) \ge 0 \quad \forall \alpha \ge 1$ .

Proof. Since  $\sum_{x} P(x|y)^{\alpha} \leq \sum_{x} P(x|y) = 1 \quad \forall y \in Y \text{ and } \alpha \geq 1$ , we have that  $\sum_{y} P(y) \sum_{x} P(x|y)^{\alpha} \leq \sum_{y} P(y) = 1$  and hence  $\log(\sum_{y} P(y) \sum_{x} P(x|y)^{\alpha}) \leq 0 \quad \forall \alpha \geq 1$ . Thus  $H_{\alpha}(X|Y) \geq 0 \quad \forall \alpha \geq 1$ .  $\Box$ 

• **Property 2:** (Independence) If  $P_{XY} = P_X P_Y$ , then  $H_{\alpha}(X|Y) = H_{\alpha}(X) \quad \forall \alpha \ge 0$ . Note: This is a special case of Property 6 and will also be part of that in the implementation.

*Proof.* This directly follows from using  $P(xy) = P(x)P(y) \quad \forall x \in X, y \in Y$  in the definition of the conditional Rényi entropy (Equation (2.36)),

$$H_{\alpha}(X|Y) = \frac{1}{1-\alpha} \log\left(\sum_{y} P(y) \sum_{x} P(x)^{\alpha}\right) = \frac{1}{1-\alpha} \log\left(\sum_{x} P(x)^{\alpha}\right) = H_{\alpha}(X).$$

• **Property 3:** (Generalised chain rule)  $\forall \alpha \in (0, 1) \cup (1, \infty), \forall P_{XYZ}$ ,

$$H_{\alpha}(X|YZ) \ge H_{\alpha}(XZ|Y) - \log d_Z$$

This is derived for quantum Rényi entropies in [148] and follows as a special case for our classical definition.

• **Property 4:** (Conditional independence or entropic causal constraint) If  $P_{XY|Z} = P_{X|Z}P_{Y|Z}$ , then  $H_{\alpha}(X|YZ) = H_{\alpha}(X|Z)$  or equivalently,  $I_{\alpha}(X:Y|Z) = 0 \quad \forall \alpha \ge 0$ .

*Proof.* Using the conditional independence  $P_{XY|Z} = P_{X|Z}P_{Y|Z} \Rightarrow P(xyz) = P(x|z)P(y|z)P(z)$ =  $P(x|z)P(yz) \forall x \in X, y \in Y, z \in Z$  in the definition of the conditional Rényi entropy (Equation (2.36)) we have

$$H_{\alpha}(X|YZ) = \frac{1}{1-\alpha} \log(\sum_{xyz} P(xyz)^{\alpha} P(yz)^{1-\alpha})$$
  

$$= \frac{1}{1-\alpha} \log(\sum_{xyz} P(x|z)^{\alpha} P(yz)^{\alpha} P(yz)^{1-\alpha})$$
  

$$= \frac{1}{1-\alpha} \log(\sum_{xz} P(x|z)^{\alpha} \sum_{y} P(yz))$$
  

$$= \frac{1}{1-\alpha} \log(\sum_{xz} P(x|z)^{\alpha} P(z)) = H_{\alpha}(X|Z).$$
(4.25)

By the definition of the conditional mutual information,  $I_{\alpha}(X : Y|Z) = H(X|YZ) - H(Y|Z)$ , it follows that  $I_{\alpha}(X : Y|Z) = 0$  in this case.

• **Property 5:** (Monotonicity under conditioning)  $H_{\alpha}(X|Z) \leq H_{\alpha}(XY|Z) \quad \forall \alpha \geq 0.$ 

*Proof.* Noting that  $\sum_{y} P(xy|z)^{\alpha} \leq (\sum_{y} P(xy|z))^{\alpha} = P(x|z)^{\alpha} \quad \forall \alpha \geq 0$ , we have

$$H_{\alpha}(XY|Z) = \frac{1}{1-\alpha} \log(\sum_{z} P(z) \sum_{x} \sum_{y} P(xy|z)^{\alpha})$$
  

$$\geq \frac{1}{1-\alpha} \log(\sum_{z} P(z) \sum_{x} P(x|z)^{\alpha}) = H_{\alpha}(X|Z).$$

$$(4.26)$$

Using these constraints to define the outer approximation to the classical Rényi entropy cone for the bipartite Bell causal structure (Figure 3.1a), we applied the entropy vector method for this case. The range of  $\alpha$  for which all these constraints hold is  $\alpha \ge 1$ , and the following applies to this range. The program terminated and no non-trivial entropic inequalities were obtained, indicating that this outer approximation to the classical marginal Rényi entropy cone coincides with the quantum one for this causal structure. This is the same outcome as the Shannon case  $(\alpha = 1)$  found in [214], while for the Tsallis case analysed in the main part of this chapter, we found the problem to be computationally too costly and were hence unable to find any nontrivial Tsallis entropic inequalities. In the Rényi case, the program ran for several days (5-7 days on a regular desktop PC), probably owing to the dimension-dependency of the generalised chain rule (Property 4.5.3), suggesting that this method would be computationally costly for larger causal structures such as the Triangle (Figure 2.6b).

In the presence of post-selection, we saw that the Shannon entropic Braunstein-Caves inequalities (3.12) can be derived for the post-selected Bell causal structure of Figure 3.1b. These are derived by using the monotonicity (2.22) and strong-subadditivity (2.24) (unconditional form) on the entropies of the set  $\{X_0, X_1, Y_0, Y_1\}$ . In the Rényi case however, strong subadditivity does not hold in the unconditional form, and such inequalities cannot be derived analogously. Implementing the entropy vector method for Rényi entropies in the post-selected Bell causal structure, it was found by Mirjam that no non-trivial inequalities are obtained. We believe that this failure to detect non-classicality using Rényi entropies is because the generalised chain rule 4.5.3 is not a tight enough constraint, and as a consequence, the initial outer approximation to classical entropy cone is not a good one.

## CHAPTER 5

## Entropic analysis of causal structures with post-selection

 $\square$  HE results of Chapter 4 along with those of [214] have revealed important drawbacks of the  $\bot$  entropic technique for analysing causal structures in the absence of post-selection, using Shannon, Tsallis and well as Rényi entropies. In the present chapter, we analyse the postselected Bell causal structure (Figure 3.1b) using entropies. We have seen that post-selection allows for the derivation of non-trivial entropic inequalities (3.12) in causal structures such as the Bell scenario that do not support a classical-quantum gap in the absence of post-selection [214]. It is then natural to ask whether the non-classicality of a distribution can always be detected through post-selected entropic inequalities such as those of Equation (3.12). For the (d, d, 2, 2) Bell scenarios with  $d \ge 2$ , this is known to be the case [44] in the following sense. For every non-classical distribution in the (d, d, 2, 2) Bell scenario, there is a transformation that does not make any non-classical distribution classical, and such that the resulting distribution violates one of the BC entropic inequalities. The main purpose in this chapter is to investigate whether a similar result holds for non-binary outcomes. To do so, we need to specify a class of post-processing operations. The most general operations that we could consider are the non-classicality non-generating (NCNG) operations, i.e., those that do not map any classical distribution to a non-classical one. An interesting subset of these is the class of post-processings achievable through local operations and shared randomness (LOSR), which are physical in the sense that two separated parties with shared randomness could perform them.<sup>1</sup> Because of the difficulty of dealing with arbitrary NCNG operations, for the majority of our analysis we consider LOSR supplemented with the additional (NCNG) operation where the parties are exchanged (and convex combinations). We use LOSR+E to refer to this supplemented set.

We study the (2, 2, 3, 3) Bell scenario with LOSR+E post-processing operations, to see whether when applied to any non-classical distribution the result violates an entropic Bell inequality. We investigate this using both Shannon and Tsallis entropies. Our motivation for considering

<sup>&</sup>lt;sup>1</sup>In general NCNG operations used on the correlations prior to evaluating an entropic inequality need not be physical in this sense.

Tsallis entropies is that they are known to provide an advantage over the Shannon entropy in detecting non-classicality in the absence of post-processing [213] in the sense that there are non-classical distributions that violate Tsallis entropic inequalities but not the analogous Shannon-entropic inequality. In the (2, 2, 2, 2) case however, due to the result of [44], this advantage is less apparent when post-processings are considered. It is unclear whether or not this is also the case for the (2, 2, 3, 3) scenario, and hence we consider Tsallis entropies in this work. The present chapter is based on the paper [207] which is joint work with Roger Colbeck.

## 5.1 Summary of contributions: in words and in poetry

The result of [44] that Shannon inequalities are always sufficient for detecting distributions that violate the CHSH inequality in the (2, 2, 2, 2) scenario readily extends to the lifted-CHSH inequalities of the (2, 2, 3, 3) scenario (c.f. Corollary 5.4.1). We summarise this result in Section 5.2. Due to Corollary 5.4.1, we are particularly interesting in the region containing nonclassical distributions that satisfy all the CHSH-type inequalities in the (2,2,3,3) scenario. In Section 5.3, we find the vertex description of the convex polytope containing non-signaling distributions in the (2, 2, 3, 3) scenario that satisfy all the lifted-CHSH inequalities. We show that every non-classical distribution in this polytope violates exactly one  $I_{2233}$ -type inequality (Proposition 5.3.1) and identify non-classical distributions in this region that violate entropic inequalities. Further, in Section 5.4, we first consider a subset of the post-processing operations LOSR + E, that correspond to mixing with a classical distribution, and identify a family of non-classical distributions in  $\Pi_{CHSH}^{\{2,2,3,3\}}$  for which we conjecture that the non-classicality cannot be detected through arbitrary Shannon entropic inequalities and a class of Tsallis entropic inequalities. In Section 5.4.2, we show analytically that if these conjectures hold, then they also extend also the whole of LOSR + E. This suggests that post-selected entropic inequalities, both of Shannon and Tsallis types, are in general insufficient for detecting non-classicality in the bipartite Bell causal structure. Following the trend set in the previous chapter, a poetic summary of this chapter is provided before delving into the technical results.

Recall, we found it hard to certify, That a causal structure must be described, By systems that obey non-classical physics, Based on the entropies of the observed statistics.

But for some causal structures, we realise, That an additional technique does materialise. Post-selection provides a trick, To detect non-classicality that would otherwise escape our grip.

But does it allow us to identify, All the non-classical correlations that can arise? In some cases<sup>2</sup>, there's indeed a technique [44], But does it generalise? An answer we seek.

We consider a set of operations O you can do, On correlations before their entropies are put to test, For a subset M of O, our computations give a clue That the final entropies fail to reveal non-classicality, even at their best.

If our computations for M are true, We prove that they continue to be true for all of O, Along the way, we get some related results too, Through a minor geometric detour.

So it seems that post-selected entropic inequalities, Be it of Shannon or the Tsallis kind, Are insufficient for detecting non-classicality, Even if we post-process our data before we try.

## 5.2 Existing result for the (2,2,2,2) Bell scenario in entropy space

The current section summarises the relevant results of [44] regarding the sufficiency of entropic inequalities in the (2, 2, 2, 2) scenario. As previously mentioned, it is possible for a non-classical distribution to have the same entropy vector as a classical one and hence to be entropically classical. For example, the maximally non-classical distribution in probability space,  $P_{\rm PR}$  (Equation (3.5)) is entropically classical since it has the same entropy vector as the classical distribution

$$P_{\rm C} = \frac{\begin{vmatrix} \frac{1}{2} & 0 & \frac{1}{2} & 0\\ 0 & \frac{1}{2} & 0 & \frac{1}{2} \\ \frac{1}{2} & 0 & \frac{1}{2} & 0\\ 0 & \frac{1}{2} & 0 & \frac{1}{2} \end{vmatrix}$$
(5.1)

and hence cannot violate any of the BC inequalities<sup>3</sup>. However, the distribution  $\frac{1}{2}P_{\text{PR}} + \frac{1}{2}P_{\text{C}}$  maximally violates  $I_{\text{BC}}^4 \leq 0$  attaining a value of ln 2. That convex mixtures of non-violating distributions can lead to a violation is due to the fact that entropic inequalities are non-linear in the underlying probabilities (in contrast to the facet Bell inequalities in probability space).

<sup>&</sup>lt;sup>2</sup>Particularly, this refers to

Bell scenarios with output cardinality two,

Here we consider,

Cardinalities of three or larger.

 $<sup>{}^{3}</sup>P_{\rm PR}$  and  $P_{\rm C}$  are related by a permutation of the entries in the bottom right  $2 \times 2$  block and entropies are invariant under such permutations.
In [44], it was shown that such a procedure is possible for every non-classical distribution in the (d, d, 2, 2) Bell scenario with  $d \ge 2$  i.e., for every such distribution, there exists an LOSR transformation such that the resultant distribution violates a Shannon entropic BC inequality (3.12). Thus, non-classicality can be detected in this scenario by processing the observed correlations (in a way that cannot generate non-classicality) before using a BC entropic inequality on the result. In this sense the BC entropic inequalities provide a necessary and sufficient test for non-classicality in these scenarios.

In more detail, for the (2, 2, 2, 2) case this works as follows. First, one defines a special class of distributions, *isotropic distributions*, as follows for some  $k \in \{1, 2, ..., 8\} := [8]$  and  $\epsilon \in [0, 1]$ .

$$P_{\rm iso}^k = \epsilon P_{\rm PR}^k + (1 - \epsilon) P_{\rm noise}, \qquad (5.2)$$

where  $P_{\text{noise}}$  is white noise i.e., the distribution with all entries equal to 1/4. In the (2, 2, 2, 2) Bell scenario the isotropic distribution  $P_{\text{iso}}^k$  is non-classical if and only if  $\epsilon > 1/2$ . The LOSR transformation used in [44] involves first transforming the observed distribution into an isotropic distribution through a local depolarisation procedure that cannot generate non-classicality. Second, it is shown that for any non-classical isotropic distribution i.e., a  $P_{\text{iso}}^k$  with  $\epsilon > 1/2$ , there exists a classical distribution  $P_{\text{C}}^k$  such that the distribution  $P_{\text{E},v}^k = v P_{\text{iso}}^k + (1-v) P_{\text{C}}^k$  violates one of the BC entropic inequalities for sufficiently small v > 0. In particular, the value of  $I_{\text{BC}}^k$ for  $P_{\text{E},v}^k$  can be expanded for small v as

$$I_{\rm BC}^k \approx \frac{v}{\ln 4} [f(\epsilon) - (4\epsilon - 2)\ln v], \qquad (5.3)$$

where  $f(\epsilon)$  is a function of  $\epsilon$ , independent of v (see [44] for details). Thus for any  $\epsilon > 1/2$ , the corresponding isotropic distributions are non-classical and taking v arbitrarily small can make  $I_{BC}^k$  positive which is a violation of the entropic inequality. We summarise the main result of [44] for (2,2,2,2) Bell scenarios in the following Theorem (which is implicit in [44]).

**Theorem 5.2.1.** For every non-classical distribution,  $P_{XY|AB}$  in the (2,2,2,2) Bell scenario, there exists an LOSR transformation  $\mathcal{T}$ , such that  $\mathcal{T}(P_{XY|AB})$  violates one of the BC entropic inequalities (3.12).

One of the aims of the present chapter is to study whether this result extends to the case where the number of outcomes per party is more than two. In general, the (2, 2, d, d) Bell polytope for d > 2 has new, distinct classes of Bell inequalities and extremal non-signaling vertices other than the CHSH inequalities and the PR boxes. We analyse this problem for the d = 3 case, for which the following results pertaining to the (2, 2, 3, 3) scenario in probability space will be helpful.

# **5.3** New results for the (2,2,3,3) Bell scenario in probability space

In this section we compute the vertex description of the *CHSH-classical polytope*,  $\Pi_{CHSH}^{(2,2,3,3)}$ , i.e., the polytope whose facets are the 648 CHSH inequalities and the positivity constraints of the

(2, 2, 3, 3) Bell scenario. As previously mentioned, this will be the main region of interest in the remainder of this work since the non-classicality of distributions not belonging to this region can always be certified using Shannon-entropic inequalities (Corollary 5.4.1). The following result allows us to significantly speed up the vertex enumeration problem.

**Proposition 5.3.1.** Every non-classical distribution in  $\Pi^{(2,2,3,3)}_{CHSH}$  violates only one  $I_{2233}$  inequality.

*Proof.* Let  $i \in \{2, 3, ..., 432\}$  and consider the linear program that maximises the value of  $\epsilon \ge 0$  subject to there existing a non-signaling distribution that

- violates  $I_{2233}^1 \leq 2$  and  $I_{2233}^i \leq 2$  by at least  $\epsilon$ , i.e.,  $I_{2233}^1 \epsilon \geq 2$  and  $I_{2233}^i \epsilon \geq 2$ ;
- satisfies all CHSH-type inequalities.

We run over  $i \in \{2, 3, ..., 432\}$  and check that in all cases either the output of this linear program is  $\epsilon = 0$  (meaning that the two  $I_{2233}$  inequalities can be jointly saturated but not violated) or that the program is infeasible (the two  $I_{2233}$  inequalities cannot even be jointly saturated). By symmetry it follows that no pair of  $I_{2233}$  inequalities can be simultaneously violated when all the CHSH-type inequalities are satisfied.

Note that in the (2, 2, 3, 3) scenario there exist extremal non-signaling distributions that violate multiple Bell inequalities. For example, the distributions  $P_{\rm NL}$  (Equation (3.8)) and  $P_{\rm NL}^*$  (Equation (5.7)) violate  $I_{2233} \leq 2$  (c.f. Equation (3.6)) although only  $P_{\rm NL}$  violates it maximally. By symmetry,  $P_{\rm NL}$  also violates another  $I_{2233}$  inequality. In addition,  $P_{\rm NL}$  violates the CHSH-type inequality whose evaluation is equivalent to applying the output coarse-graining  $0 \mapsto 0, 1 \mapsto 1$ and  $2 \mapsto 1$  for each party and then evaluating (3.3). This is in contrast to the (2, 2, 2, 2) scenario where there is a one-to-one correspondence between the extremal non-signaling vertices and the CHSH inequalities in the sense that each such vertex violates exactly one CHSH inequality.<sup>4</sup>

Due to the symmetries, all the vertices of  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$  can be enumerated by first finding all the vertices for which the  $I_{2233}$  inequality of Equation (3.6) is saturated or violated i.e.,  $I_{2233} \ge 2$ , and taking the orbit of these vertices under local relabellings and exchange of the parties. The vertex enumeration for this case yields 47 extremal points of which 30 are the local deterministic points that saturate  $I_{2233} \ge 2$  and 17 are non-classical points that violate only this inequality. These are listed in Table 5.1. By applying all symmetries and removing duplicate vertices we find that  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$  has 7425 vertices (including the 81 local deterministic vertices).

<sup>&</sup>lt;sup>4</sup>Note that this correspondence breaks down in the (2, 2, 3, 3) scenario where it is possible for a CHSH-type vertex to violate multiple CHSH-type inequalities (these correspond to the same 2-outcome CHSH inequality after coarse-graining).

# **5.4** New results for the (2,2,3,3) Bell scenario in entropy space

We now investigate whether entropic inequalities are necessary and sufficient for non-classicality in the (2, 2, 3, 3) Bell scenario. In (d, d, 2, 2) scenarios with  $d \ge 2$ , only 2d Shannon entropic inequalities are required for the Shannon entropic characterisation of the scenario [46, 90]; in the (2, 2, 2, 2), these are the four inequalities of (3.12). It may at first seem surprising that these can always be used to decide whether a distribution is classical because the number of extremal Bell inequalities grows very rapidly in d in the (d, d, 2, 2) scenario [66], and deciding whether a distribution is classical is NP-complete [14]. The reduction in the number of inequalities in entropy space is compensated by the need to identify a suitable post-processing operation (of which there are uncountably many possibilities) in order to detect violations.

The first observation is a corollary of Theorem 5.2.1.

**Corollary 5.4.1.** Let  $P_{XY|AB}$  be a distribution in the (2,2,3,3) Bell scenario that violates at least one CHSH-type inequality. Then there exists an LOSR transformation  $\mathcal{T}$ , such that  $\mathcal{T}(P_{XY|AB})$  violates one of the BC entropic inequalities (3.12).

*Proof.* For each CHSH-type inequality in the (2, 2, 3, 3) scenario, there exists a coarse-graining in which two of the outcomes are mapped to one (for each party and each input) such that for any initial distribution in the (2, 2, 3, 3) scenario that violates the CHSH-type inequality the coarse-grained distribution violates one of the CHSH-inequalities in the (2, 2, 2, 2) scenario. Hence, for the given  $P_{XY|AB}$ , after applying the corresponding coarse-graining for the violated CHSH-type inequality, followed by the LOSR operation from Theorem 5.2.1 we violate one of the BC entropic inequalities.

This corollary means that we can limit our analysis to  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$ , the polytope in which all the CHSH inequalities are satisfied, and, in particular, the non-classical region of this. This is the region in which one of the  $I_{2233}$  inequalities is violated.

In going from the (2, 2, 2, 2) to (2, 2, 3, 3) scenario, a new class of inequalities (the  $I_{2233}$  inequalities) become relevant in probability space but the entropic characterisation remains unchanged, since entropic inequalities do not depend on the number of measurement outcomes. It is natural to ask whether all non-classical distributions in the (2, 2, 3, 3) scenario that satisfy all the CHSH inequalities cannot be certified entropically. However, this is not the case as shown by the following proposition.

**Proposition 5.4.1.** The polytope  $\Pi^{(2,2,3,3)}_{CHSH}$  is not entropically classical.



Figure 5.1: Plot of the  $I_{BC,q}^4$  value as a function of the Tsallis parameter q for the distribution  $p_e$  (Equation (5.4)). As seen from the plot, the distribution violates the BC inequality  $I_{BC,q}^4 \leq 0$  for q values between 1 (Shannon case) and just under 1.5 and the violation is maximum in the Shannon case, indicating that it is preferable to use Shannon rather than Tsallis entropies with q > 1 in this case.

*Proof.* Consider the distribution

$$P_e \coloneqq \frac{1}{50} \begin{bmatrix} 21 & 0 & 0 & 21 & 0 & 0 \\ 0 & 2 & 0 & 1 & 1 & 0 \\ 11 & 0 & 16 & 0 & 1 & 26 \\ \hline 31 & 0 & 0 & 20 & 1 & 10 \\ 1 & 1 & 0 & 1 & 0 & 1 \\ 0 & 1 & 16 & 1 & 1 & 15 \end{bmatrix}.$$
(5.4)

This is formed by mixing the non-local extremal point number 8 of  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$  (see Table 5.1) with the three local deterministic points 18, 26 and 47 with respective weights 1/10, 3/10, 1/5 and 2/5, and hence is in  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$ . It achieves a  $I_{\text{BC}}^4$  value of 0.0199733, in violation of  $I_{\text{BC}}^4 \leq 0$ , so is not entropically classical.

We remark that by mixing with more local deterministic distributions and varying the weights, larger violations of  $I_{BC}^4 \leq 0$  can be found; the distribution  $P_e$  used in the previous proposition was chosen for its relative simplicity. Interestingly, we find that the Shannon entropic BC inequalities appear to give the largest violation among the Tsallis entropic inequalities for  $q \geq 1$ when applied to  $P_e$ . This can be seen in Figure 5.1.

In light of Proposition 5.4.1, it is natural to ask whether the non-classicality of all distributions in  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$  can be detected through entropic inequalities. We find numerical evidence that suggests the contrary, i.e., that there are non-classical distributions in  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$  whose non-classicality cannot be detected through entropic inequalities using a general class of postprocessing operations, and hence that these entropic inequalities are not sufficient for detecting non-classicality in the (2, 2, 3, 3) scenario. Before presenting these results, we briefly overview the post-processing operations considered in this work. **Post-processing operations** In this chapter we study whether entropic inequalities can always detect non-classicality in the (2,2,3,3) Bell scenario. In order to do so we could in principle consider applying any NCNG operation to the distribution prior to evaluating the entropic inequality. However, due to the difficulty in dealing with arbitrary NCNG operations, we consider the subset of these corresponding to LOSR+E instead. In [222] it was shown that all LOSR operations can be generated by convex combinations of local deterministic operations. These can be thought of in the following way. Each party first does a deterministic function on their input, uses the result as the input to their device, then does a deterministic function on their input and the output of their device to form the final output. All such operations correspond to local relabellings and local coarse-grainings. Note that deterministic classical distributions can be formed as a special case of coarse-graining (a local deterministic distribution is formed when each party coarse-grains all of their outputs to one output for each of their inputs). For the distributions we consider for our main conjectures, it turns out that all the coarse-grainings give rise to local distributions (c.f. Proposition 5.4.5), so, by considering mixing with deterministic classical distributions, local relabelling and exchange of parties we can cover all LOSR+E operations. We hence start by separately considering mixing with classical distributions, and then consider relabelling and exchange of parties.

### 5.4.1 Mixing with classical distributions

Analogously to the (2, 2, 2, 2) case, we can define a family of distributions  $P_{iso,\epsilon}^{(2,2,3,3)} = \epsilon P_{\rm NL} + (1 - \epsilon)P_{noise}^{(2,2,3,3)}$ , where  $P_{noise}^{(2,2,3,3)}$  is the uniform distribution with all entries equal to 1/9 and  $\epsilon \in [0, 1]$ . This class of distributions is isotropic in the sense that the marginal distributions are uniform for each input of each party. In order to show the insufficiency of entropic inequalities, one needs to identify at least one non-classical distribution whose non-classicality cannot be detected through entropic inequalities. We will discuss this for the class  $P_{iso,\epsilon}^{(2,2,3,3)}$  and only consider distributions of this form in the rest of the chapter. Further, without loss of generality, we consider only the BC inequality  $I_{\rm BC}^4 \leq 0$  in what follows (by symmetry all the arguments will also hold for isotropic distributions corresponding to relabelled versions of  $P_{\rm NL}$  and the corresponding BC inequalities).

#### 5.4.1.1 Using Shannon entropy

In the entropic picture of the (2, 2, 3, 3) scenario, the 4 BC Inequalities (3.12) still hold (these are valid independently of the cardinality of the random variables). Again, analogously to the (2, 2, 2, 2) case, the maximally non-local distribution,  $P_{\rm NL}$  has the same entropy vector as the

classical distribution

$$P_{\rm C}^{(2,2,3,3)} = \frac{\begin{vmatrix} \frac{1}{3} & 0 & 0 & \frac{1}{3} & 0 & 0 \\ 0 & \frac{1}{3} & 0 & 0 & \frac{1}{3} & 0 \\ 0 & 0 & \frac{1}{3} & 0 & 0 & \frac{1}{3} \\ \hline \frac{1}{3} & 0 & 0 & \frac{1}{3} & 0 & 0 \\ 0 & \frac{1}{3} & 0 & 0 & \frac{1}{3} & 0 \\ 0 & 0 & \frac{1}{3} & 0 & 0 & \frac{1}{3} \\ \hline \end{vmatrix}$$
(5.5)

(amongst others). The distribution  $P_{\rm NL}$  is hence entropically classical. However, in contrast to the (2, 2, 2, 2) case, we have evidence suggesting that there are values of  $\epsilon$  for which  $P_{\rm iso,\epsilon}^{(2,2,3,3)}$ is non-classical, but such that the mixture  $vP_{\rm iso,\epsilon}^{(2,2,3,3)} + (1-v)P_{\rm L}$  is entropically classical for all classical distributions  $P_{\rm L}$  and all  $v \in [0, 1]$ , i.e., there exist non-classical distributions in the (2, 2, 3, 3) scenario for which mixing with classical distributions never gives rise to a non-classical entropy vector.

We begin by considering mixing  $P_{iso,\epsilon}^{(2,2,3,3)}$  with  $P_{C}^{(2,2,3,3)}$  in analogy with the treatment of the (2,2,2,2) case. Although we have not fully proven this, from our numerics, this mixing appears to be optimal in the sense that when it does not allow for entropic violations, no other mixing can either. This allows us to identify a range of  $\epsilon$  for which the mixture  $P_{iso,\epsilon}^{(2,2,3,3)}$  is non-classical, yet appears to remain entropically classical even when mixed with arbitrary classical distributions. We begin with two propositions whose proofs can be found in Appendix 5.6.3.

**Proposition 5.4.2.**  $P_{iso,\epsilon}^{(2,2,3,3)}$  is non-classical if and only if  $\epsilon > 1/2$ . Further, for  $\epsilon \le 4/7$ ,  $P_{iso,\epsilon}^{(2,2,3,3)}$  satisfies all the CHSH-type inequalities, while for  $\epsilon > 4/7$  it violates at least one CHSH-type inequality.

By analogy with the (2, 2, 2, 2) case, we consider the violation of  $I_{\rm BC}^4$  attainable by mixing  $P_{iso,\epsilon}^{(2,2,3,3)}$  with  $P_{\rm C}^{(2,2,3,3)}$ . We find that for  $\epsilon \in (1/2, 4/7]$ ,  $P_{iso,\epsilon}^{(2,2,3,3)}$  is non-classical but does not violate any of the BC inequalities. As shown in the above proposition, these distributions are in the CHSH-classical polytope  $\Pi_{\rm CHSH}^{(2,2,3,3)}$  and hence lie in our region of interest.

**Proposition 5.4.3.** For  $\epsilon \leq 4/7$ ,  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)} = vP_{\mathrm{iso},\epsilon}^{(2,2,3,3)} + (1-v)P_{\mathrm{C}}^{(2,2,3,3)}$  does not violate any of the BC entropic inequalities (3.12) for any  $v \in [0,1]$ . However, for all  $\epsilon > 4/7$ , there exists a  $v \in [0,1]$  such that the entropic inequality  $I_{\mathrm{BC}}^4 \leq 0$  is violated by  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)}$ .

The second part of this proposition already follows from Corollary 5.4.1 and Proposition 5.4.2. **Corollary 5.4.2.** For  $\epsilon \leq 4/7$ ,  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)} = vP_{\mathrm{iso},\epsilon}^{(2,2,3,3)} + (1-v)P_{\mathrm{C}}^{(2,2,3,3)}$  is entropically classical for all  $v \in [0, 1]$ .

*Proof.* This follows from Proposition 5.4.3 and Lemma 3.2.1.

While Proposition 5.4.3 shows that the proof strategy of [44] does not directly generalize to all non-classical distributions in the (2, 2, 3, 3) case, it does not rule out the possibility that

there may exist other mixings with classical distributions that could transform  $P_{iso,\epsilon}^{(2,2,3,3)}$  for  $\epsilon \in (1/2, 4/7]$  into a distribution that violates one of the BC inequalities. To investigate this, we can consider the polytope formed by mixing  $P_{iso,\epsilon}^{(2,2,3,3)}$  with classical distributions for some  $\epsilon \leq 4/7$ , i.e., the polytope  $\operatorname{Conv}(\{P_{iso,\epsilon}^{(2,2,3,3)}\} \cup \{P^{L,k}\}_k)$ , where  $\{P^{L,k}\}_k$  denotes the set of all (81) local deterministic vertices of the (2,2,3,3) Bell-local polytope,  $\{P_{iso,\epsilon}^{(2,2,3,3)}\}$  is a set with a single element and  $\operatorname{Conv}()$  denotes the convex hull. We considered several values of  $\epsilon \leq 4/7$  and numerically optimised the entropic expression  $I_{BC}^4$  over the non-classical region of this polytope<sup>5</sup> for each<sup>6</sup> and were unable to find violations. The optimization involves a non-linear objective function with linear constraints. Hence, it is possible that the numerical approach missed the global optimum. Nevertheless, this is evidence for the following conjecture and is presented in more detail in Appendix 5.6.2. Proposition 5.4.3 along with Figures 5.2 and evidence in Appendix 5.6.2 also suggest this conjecture.

**Conjecture 5.4.1.** Let  $\epsilon \leq 4/7$ . For all mixtures of the distribution  $P_{iso,\epsilon}^{(2,2,3,3)}$  with classical distributions in the (2,2,3,3) Bell scenario, the resulting distribution is entropically classical, *i.e.*, all distributions in Conv $(\{P_{iso,\epsilon}^{(2,2,3,3)}\} \cup \{p^{L,k}\}_k\})$  are entropically classical.

The interesting cases of Conjecture 5.4.1 are for non-classical distributions (i.e., for  $\epsilon > 1/2$ ), and the most relevant of these are those that can be achieved in quantum theory. The next remark addresses this case.

**Remark 5.4.1.** There exist non-classical quantum distributions that lie in the polytope  $\operatorname{Conv}(\{P_{iso,e=4/7}^{(2,2,3,3)}\} \cup \{P^{L,k}\}_k\})$  and in this case, our results suggest that the non-classicality of the corresponding distributions cannot be detected through entropic inequalities. Let  $P_{\text{QM}}$  be the quantum distribution from [64, Equation (14) with d = 3] with Bob's inputs relabelled. This violates  $I_{BC}^4 \leq 0$  through mixing with  $P_C^{(2,2,3,3)}$ . Consider then mixing  $P_{\text{QM}}$  with uniform noise  $P_{\text{noise}}^{(2,2,3,3)}$  to obtain  $P_{\text{mix}}(u) \coloneqq uP_{\text{QM}} + (1-u)P_{\text{noise}}^{(2,2,3,3)}$  ( $u \in [0,1]$ ). We found that for some values of u (e.g., u = 7/10),  $P_{\text{mix}}(u)$  is non-classical. Further,  $P_{\text{mix}}(u)$  is quantum achievable since it can be obtained from the density operator  $u|\psi'\rangle\langle\psi'| + (1-u)\frac{1}{9}$  (where  $|\psi'\rangle$  is the two qutrit state producing  $P_{\text{QM}}$ ) and the same quantum measurements that produce  $P_{\text{QM}}$  from  $|\psi'\rangle$ .

#### 5.4.1.2 Using Tsallis entropies

Given the results (Proposition 5.4.3 and Conjecture 5.4.1) of the previous section for Shannon entropic inequalities, a natural question is whether other entropic measures can provide an advantage over the Shannon entropy in detecting non-classicality. Here, we look at Tsallis entropies and find that similar results hold in this case as well, suggesting that Tsallis entropies also do not allow us to completely solve the problem.

 $<sup>^{5}</sup>$ To restrict to the non-classical region, it is sufficient to mix with a subset of these 81 locals— see Appendix 5.6.2 for more detail.

<sup>&</sup>lt;sup>6</sup>Note however that it is enough that these results hold for one value of  $\epsilon \in (1/2, 4/7]$  in order to conclude that entropic inequalities are not sufficient for detecting non-classicality in this scenario.

The properties of monotonicity, strong-subadditivity and the chain rule are sufficient to derive the BC inequalities, which hence also hold for Tsallis entropy when  $q \ge 1$ . Other generalized entropies such as Rényi or min/max entropies do not satisfy one or more of these properties in general and it is not clear whether the analogues of (3.12) hold for these. In the Rényi case, we saw in Appendix 4.5.3 that no analogous inequalities were obtained. Hence we focus on the case of Tsallis entropies in the remainder of this chapter, where for all  $q \ge 1$  we have

$$I_{BC,q}^{1} = S_{q}(X_{0}Y_{0}) + S_{q}(X_{1}) + S_{q}(Y_{1}) - S_{q}(X_{0}Y_{1}) - S_{q}(X_{1}Y_{0}) - S_{q}(X_{1}Y_{1}) \le 0$$

$$I_{BC,q}^{2} = S_{q}(X_{0}Y_{1}) + S_{q}(X_{1}) + S_{q}(Y_{0}) - S_{q}(X_{0}Y_{0}) - S_{q}(X_{1}Y_{0}) - S_{q}(X_{1}Y_{1}) \le 0$$

$$I_{BC,q}^{3} = S_{q}(X_{1}Y_{0}) + S_{q}(X_{0}) + S_{q}(Y_{1}) - S_{q}(X_{0}Y_{0}) - S_{q}(X_{0}Y_{1}) - S_{q}(X_{1}Y_{1}) \le 0$$

$$I_{BC,q}^{4} = S_{q}(X_{1}Y_{1}) + S_{q}(X_{0}) + S_{q}(Y_{0}) - S_{q}(X_{0}Y_{0}) - S_{q}(X_{0}Y_{1}) - S_{q}(X_{1}Y_{0}) \le 0$$
(5.6)

and we refer to these as the Tsallis entropic BC inequalities. Entropic classicality in Tsallis entropy space can be defined analogously to Definition 3.2.4, in terms of Tsallis entropy vectors over the set of variables S (Equation (3.11)). We say that a distribution is *q*-entropically classical if its entropy vector written in terms of the Tsallis entropy of order q is achievable using a classical distribution. In the case of the Shannon entropy, we used the fact (Lemma 3.2.1) that the BC Inequalities (3.12) are known to be necessary and sufficient for entropic classicality for 2-input Bell scenarios [90]. However, it is not clear if the result of [90] generalises to Tsallis entropies for q > 1. Thus our results in the Tsallis case are weaker than those for Shannon, being stated only for the BC inequalities. We leave the generalization to arbitrary Tsallis entropic inequalities as an open problem.

**Proposition 5.4.4.** For  $\epsilon \leq 4/7$ ,  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)} = vP_{\mathrm{iso},\epsilon}^{(2,2,3,3)} + (1-v)P_{\mathrm{C}}^{(2,2,3,3)}$  does not violate any of the Tsallis BC inequalities (5.6) for any  $v \in [0,1]$  and q > 1. However, for  $\epsilon > 4/7$  and every q > 1, there always exists a  $v \in [0,1]$  such that the entropic inequality  $I_{\mathrm{BC},q}^4 \leq 0$  is violated by  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)}$ .

We refer the reader to Appendix 5.6.3 for a proof of this Proposition. To investigate the extension to other mixings, we tried the same computational procedure (see Appendix 5.6.2) as in the Shannon case. We found no violation of the Tsallis entropic BC inequalities for any mixings of  $P_{iso,\epsilon}^{(2,2,3,3)}$  with classical distributions, for several values of q > 1 and  $\epsilon \in (1/2, 4/7]$ , leading to the following conjecture, which is similar to Conjecture 5.4.1.

**Conjecture 5.4.2.** Let  $\epsilon \leq 4/7$ . For all mixtures of the distribution  $P_{iso,\epsilon}^{(2,2,3,3)}$  with classical distributions in the (2,2,3,3) Bell scenario, the resulting distribution does not violate any of the Tsallis entropic BC inequalities for any q > 1, i.e., all distributions in  $Conv(\{P_{iso,\epsilon}^{(2,2,3,3)}\} \cup \{P^{L,k}\}_k)$  satisfy the Tsallis entropic BC Inequalities (5.6) for all q > 1.

Figure 5.2a, shows the values of  $\epsilon$  and v for which  $I_{BC,q}^4$  (for q = 1, 2, 8) evaluated with  $P_{E,\epsilon,v}^{(2,2,3,3)}$  is positive, which is also suggestive of this conjecture.

**Remark 5.4.2.** Any impossibility result for the (2, 2, 3, 3) scenario also holds in the (2, 2, d, d) case for d > 3 because the former is always embedded in the latter i.e., every distribution in the (2, 2, 3, 3) scenario has a corresponding distribution in all the (2, 2, d, d) scenarios with d > 3

which can be obtained by assigning a zero probability to the additional outcomes. Further, the entropic Inequalities (3.12) remain the same for all these scenarios as they do not depend on the cardinality of the random variables involved. Thus the existence non-classical distributions for the d = 3 case whose non-classicality cannot be detected by entropic inequalities implies the same result for all d > 3.

### 5.4.2 Beyond classical mixings

So far, we only considered mixing with classical distributions to obtain entropic violations and gave evidence that this does not work for some non-classical distributions in the (2, 2, 3, 3) scenario. This motivates us to study whether using arbitrary LOSR+E operations allows us to detect this non-classicality through entropic violations. We show in this section that if Conjectures 5.4.1 and 5.4.2 hold then they also hold for all LOSR+E operations. First consider the following example.

The maximum possible violation of the BC inequalities in the (2, 2, 2, 2) case is  $I_{BC}^4 = \ln 2$  [44]. This is derived by considering only Shannon inequalities within the coexisting sets, and the bound that the maximum entropy of a binary variable is  $\ln 2$ . An analogous proof holds in the (2, 2, 3, 3) case, except that the bound is then  $\ln 3$ . In the former case we have  $P_{E,\epsilon=1,\nu=1/2} = \frac{1}{2}P_{PR} + \frac{1}{2}P_{C}$ , which maximally violates  $I_{BC}^4 \leq 0$ , while in the latter case, one such distribution is formed by  $(P_{NL} + P_{NL}^* + P_{C}^{(2,2,3,3)})/3$ , where  $P_{NL}^*$  is another extremal non-local distribution:

$$P_{\rm NL}^{*} = \frac{\begin{vmatrix} \frac{1}{3} & 0 & 0 & \frac{1}{3} & 0 & 0 \\ 0 & \frac{1}{3} & 0 & 0 & \frac{1}{3} & 0 \\ 0 & 0 & \frac{1}{3} & 0 & 0 & \frac{1}{3} \\ \hline \frac{1}{3} & 0 & 0 & 0 & 0 & \frac{1}{3} \\ 0 & \frac{1}{3} & 0 & \frac{1}{3} & 0 & 0 \\ 0 & 0 & \frac{1}{3} & 0 & \frac{1}{3} & 0 \\ \hline \end{array}$$
(5.7)

Since the equal mixture  $(P_{\rm NL} + P_{\rm C}^{(2,2,3,3)})/2$  violates  $I_{\rm BC}^4 \leq 0$  non-maximally, one may be motivated to use the non-local distribution  $\tilde{P}_{\rm NL} = (P_{\rm NL} + P_{\rm NL}^*)/2$  in place of  $P_{\rm NL}$  in the definition of  $P_{\rm iso,\epsilon}^{(2,2,3,3)}$ , i.e., to take

$$\tilde{P}_{iso,\epsilon}^{(2,2,3,3)} = \epsilon \tilde{P}_{NL} + (1-\epsilon) P_{C}^{(2,2,3,3)}.$$

One could then consider whether for  $\epsilon \in (1/2, 4/7]$ ,  $\tilde{P}_{\mathrm{E},\epsilon,v}^{(2,2,3,3)} = v \tilde{P}_{\mathrm{iso},\epsilon}^{(2,2,3,3)} + (1-v) P_{\mathrm{C}}^{(2,2,3,3)}$  violates  $I_{\mathrm{BC}}^4 \leq 0$ . Interestingly, while  $\tilde{P}_{\mathrm{E},\epsilon,v}^{(2,2,3,3)}$  violates  $I_{\mathrm{BC}}^4 \leq 0$  for a larger range of v values whenever  $\epsilon > 4/7$ , it does not give any violation (for any value of v) when  $\epsilon \leq 4/7$ , and Propositions 5.4.2 and 5.4.3 also hold if  $\tilde{P}_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$  replaces  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$  (see Figure 5.2b for an illustration). The corresponding results also hold for the Tsallis case with q > 1, i.e., Proposition 5.4.4 also holds with  $\tilde{P}_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$  replacing  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$  (see also Figure 5.2). These suggest that mixing with relabellings in addition to mixing with classical distributions may also not help to violate entropic inequalities when  $\epsilon \leq 4/7$ .



**Figure 5.2:** The regions in the  $v - \epsilon$  plane where the Shannon entropic inequality  $I_{BC,1}^4 := I_{BC}^4 \leq 0$  (blue), the Tsallis entropic inequality,  $I_{BC,q}^4 \leq 0$  for q = 2 (orange) and q = 8 (red) are violated by the distributions (a)  $P_{E,\epsilon,v}^{(2,2,3,3)} = vP_{iso,\epsilon}^{(2,2,3,3)} + (1-v)P_C^{(2,2,3,3)}$ (b)  $\tilde{P}_{E,\epsilon,v}^{(2,2,3,3)} = v\tilde{P}_{iso,\epsilon}^{(2,2,3,3)} + (1-v)P_C^{(2,2,3,3)}$ . Here  $P_{iso,\epsilon}^{(2,2,3,3)} = \epsilon P_{NL} + (1-\epsilon)P_{noise}^{(2,2,3,3)}$  and  $\tilde{P}_{iso,\epsilon}^{(2,2,3,3)} = \epsilon(1/2P_{NL} + 1/2P_{NL}^*) + (1-\epsilon)P_{noise}^{(2,2,3,3)}$ . For both (a) and (b)  $I_{BC,q}^4 \leq 0$  is not violated when  $\epsilon \leq 4/7 \approx 0.5714$  but for  $\epsilon > 4/7$ , there is a violation of this inequality for a larger range of v values in the latter case, and also for a larger range in the q = 2 case as compared to the other two cases.

In the remainder of this section we consider the full set of LOSR+E operations. We first note that all input coarse-grainings of  $P_{iso,\epsilon}^{(2,2,3,3)}$  result in local distributions (there are no Bell inequalities if either of the parties have only one input). Similarly, considering output coarsegrainings, whenever three outcomes are mapped to one the resulting distribution is always classical because there are no Bell inequalities if one party always makes a fixed outcome for one of their inputs. We henceforth only consider coarse-grainings that take two outcomes to one. We can choose two of the three outcomes to combine into one for each party and each local input. For the four input choices  $\{A = 0, A = 1, B = 0, B = 1\}$ , there are 81, 108, 54 and 12 distinct coarse-grainings of this type when the outcomes of either 4, 3, 2 or 1 input choices are coarse-grained. Thus there are a total of 255 coarse-grainings that remain.

If we apply all such coarse-grainings to  $P_{iso,\epsilon}^{(2,2,3,3)}$ , this generates 255 possible distributions that we denote  $\{P_{\epsilon}^{\text{CG},i}\}_{i}$ ,  $i \in [255]$ . There are also 432 distinct local relabellings of  $P_{iso,\epsilon}^{(2,2,3,3)}$ , which we denote by  $\{P_{\epsilon}^{\text{R},j}\}_{j}$ ,  $j \in [432]$  (this set includes  $P_{iso,\epsilon}^{(2,2,3,3)} = P_{\epsilon}^{\text{R},1}$ ). Due to symmetries of  $P_{iso,\epsilon}^{(2,2,3,3)}$ , it turns out that exchanging parties can be achieved through local relabellings for these distributions, so we do not need to separately consider the exchange in our results pertaining to  $P_{iso,\epsilon}^{(2,2,3,3)}$ . The set of all distributions that can be achieved through a convex mixture of  $P_{iso,\epsilon}^{(2,2,3,3)}$ with its coarse-grainings, relabellings and classical distributions is a convex polytope  $\Pi_{\epsilon}$  for each  $\epsilon$  and is the convex hull of these 255 + 432 + 81 = 768 points, i.e.,

$$\Pi_{\epsilon} \coloneqq \operatorname{Conv}\left(\{P_{\epsilon}^{\operatorname{CG},i}\}_{i} \bigcup \{P_{\epsilon}^{\operatorname{R},j}\}_{j} \bigcup \{P^{\operatorname{L},k}_{\epsilon}\}_{k}\}\right) \,.$$

We present the results for the remaining coarse-grainings and relabellings separately below. Firstly, we show that the coarse-grainings  $\{P_{\epsilon}^{\text{CG},i}\}_i$  of  $P_{\text{iso},\epsilon}^{(2,2,3,3)}$  are classical if and only if  $\epsilon \leq 4/7$ .

**Proposition 5.4.5.** The distribution  $P_{\epsilon}^{\text{CG},i}$  is classical for all *i* if and only if  $\epsilon \leq 4/7$ .

This is intuitive because  $P_{iso,\epsilon}^{(2,2,3,3)}$  satisfies all the CHSH-type inequalities if and only if  $\epsilon \leq 4/7$ . Since coarse-grainings cannot generate non-classicality and correspond to reducing the number of outcomes, and since  $I_{2233}$  requires three outcomes, after coarse-graining the only relevant thing is whether there is a CHSH-violation. A full proof is given in Appendix 5.6.3.

Proposition 5.4.5 implies that  $\Pi_{\epsilon} = \operatorname{Conv}(\{P_{\epsilon}^{\mathrm{R},j}\}_{j} \cup \{P^{\mathrm{L},k}\}_{k}) \ \forall \epsilon \leq 4/7$ , and that it is not necessary to consider coarse-grainings for such values of  $\epsilon$ . Our next results are that if Conjectures 5.4.1 and 5.4.2 hold for  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$  for  $\epsilon \leq 4/7$ , then they continue to hold even when we consider arbitrary convex combinations with classical distributions and local relabellings of  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$ .

**Proposition 5.4.6.** Let  $\epsilon \leq 4/7$ . If Conjecture 5.4.1 holds, then every distribution in  $\Pi_{\epsilon}$  is Shannon entropically classical.

**Proposition 5.4.7.** Let  $\epsilon \leq 4/7$ . If Conjecture 5.4.2 holds, then every distribution in  $\Pi_{\epsilon}$  satisfies the Tsallis entropic BC Inequalities (5.6)  $\forall q > 1$ .

These are proven in Appendix 5.6.3 and give the following corollary.

**Corollary 5.4.3.** Let  $\epsilon \leq 4/7$ . If Conjectures 5.4.1 and 5.4.2 hold, then for any operation  $\mathcal{O}$  in LOSR+E,  $\mathcal{O}(P_{iso,\epsilon}^{(2,2,3,3)})$  does not violate a Shannon or Tsallis (q > 1) entropic BC inequality.

## 5.5 Discussion

We have provided evidence that there are distributions in the (2, 2, d, d)  $(d \ge 3)$  scenarios for which arbitrary LOSR+E operations do not enable detection of non-classicality with any Shannon entropic inequalities or Tsallis entropic BC inequalities. This is in contrast to the (2, 2, 2, 2)case [44], where non-classicality can always be certified using the Shannon BC inequalities and LOSR post-processings. The region where the BC inequalities can fail to detect non-classicality would contain non-classical distributions that satisfy all CHSH-type Bell inequalities. We found all the vertices that characterize this region for the d = 3 case and identified distributions in it that violate BC entropic inequalities. Thus, the set of all non-classical distributions in the (2, 2, 3, 3) scenario that cannot be certified through entropic inequalities under LOSR+E postprocessings is not characterized by the CHSH-type inequalities. Although we considered LOSR+E operations, a natural next question is to what extent the results can be generalized to more general NCNG operations. In particular, there could be a non-linear NCNG map that allows the entropic BC inequalities to detect a wider range of non-classical distributions. An interesting open question would be to see whether for any non-classical distribution of the form  $P_{iso,\epsilon}^{(2,2,3,3)}$  with  $1/2 < \epsilon \leq 4/7$  (conjectured to be entropically classical with respect to LOSR+E), one of these more general operations would allow its non-classicality to be detected entropically.

In [213] it was shown that (without post-processing) Tsallis entropic inequalities can detect non-classicality undetectable by Shannon entropic inequalities in the (2, 2, 2, 2) and (2, 2, 3, 3)Bell scenarios. In the presence of LOSR+E operations, we did not find any advantage of Tsallis entropies over the Shannon entropy in the (2, 2, 3, 3) Bell scenario. In fact, for some non-classical distributions such as that of Equation (5.4), the Shannon entropic inequalities appear to give the largest violations among Tsallis entropies with  $q \ge 1^7$ . On the other hand, for the family of distributions  $P_{iso,\epsilon}^{(2,2,3,3)}$ , our results suggest that the range of  $\epsilon$  for which post-processing via mixing with classical distributions enables non-classicality detection is the same for the Shannon as well as Tsallis entropic BC inequalities for q > 1. However, when entropic detection of nonclassicality is possible, using Tsallis entropy can make it easier to do this detection in the sense that there is a wider range of mixings that achieve this (see Figure 5.2).<sup>8</sup>

In conclusion, while the entropic approach for detecting non-classicality is useful in a number of scenarios, it is known to have disadvantages in others. In particular, in the absence of postselection we are not aware of any cases where entropic inequalities can be violated [214, 216, 206]. We found that the entropic approach also suffers drawbacks in the presence of post-selection as it may fail to detect non-classicality under a natural class of post-processing operations, both in the case of Shannon and Tsallis entropies. However, this method remains of use since in many cases non-classicality can be detected using it.

## 5.6 Appendix

# 5.6.1 *V*-representation of the polytope $\Pi_{CHSH}^{(2,2,3,3)}$

Table 5.1 enumerates the 47 extremal points of the polytope  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$  that saturate or violate the inequality  $I_{2233} \leq 2$  (3.6) while satisfying all the CHSH-type inequalities in the (2,2,3,3) Bell

<sup>&</sup>lt;sup>7</sup>Which corresponds to those for which the BC inequalities can be derived in the classical case

<sup>&</sup>lt;sup>8</sup>As a specific example, consider the distribution  $P_{\mathrm{E},\epsilon=0.7,v=0.4}^{(2,2,3,3)}$ . Figure 5.2a indicates that this distribution violates the Tsallis entropic inequality  $I_{\mathrm{BC},q=2}^4 \leq 0$  but does not violate the Shannon entropic inequality  $I_{\mathrm{BC}}^4 \leq 0$ . However, we can always further mix  $P_{\mathrm{E},\epsilon=0.7,v=0.4}^{(2,2,3,3)}$  with the classical distribution  $P_{\mathrm{C}}^{(2,2,3,3)}$  to obtain  $0.05P_{\mathrm{E},\epsilon=0.7,v=0.4}^{(2,2,3,3)} + 0.95P_{\mathrm{C}}^{(2,2,3,3)} = 0.02P_{\mathrm{iso},\epsilon}^{(2,2,3,3)} + 0.98P_{\mathrm{C}}^{(2,2,3,3)}$  which violates the Shannon entropic inequality  $I_{\mathrm{BC}}^4 \leq 0$ . This is also in agreement with the results of [213], since when mixing is not considered, we also find examples where it is advantageous to use Tsallis entropy in the (2,2,3,3) scenario.

scenario. The first 17 of these are non-classical while the remaining 30 are local deterministic vertices. Due to Proposition 5.3.1 and the symmetries of the scenario, the remaining vertices of  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$  can be generated by taking the orbit of these vertices under local relabellings and exchange of parties. In Table 5.1, each extremal point is given by a single 36 dimensional vector which corresponds to writing the point in the notation explained in Section 3.1 (a 6 × 6 matrix) and "flattening" it by writing one row after another in order.

#### 5.6.2 Evidence for Conjectures 5.4.1 and 5.4.2

In order to check for violations of the Shannon and Tsallis entropic inequalities  $I_{\rm BC}^4 \leq 0$ and  $I_{\rm BC,q}^4 \leq 0$  that could be obtained by mixing  $P_{\rm iso,\epsilon}^{(2,2,3,3)}$  (Equation (3.8)) with arbitrary classical distributions, we maximized the left hand sides of these inequalities over the polytope  $\operatorname{Conv}(\{P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}\} \cup \{P^{\mathrm{L},k}\}_k)$  for some  $\epsilon$  values in (1/2, 4/7] (such as  $\epsilon = 4/7, 5/9$ ) numerically using MATHEMATICA. Note that this polytope contains the local polytope where by definition, entropic inequalities cannot be violated. Thus we can simplify the optimization and increase its reliability by only optimizing over the non-classical part of the polytope  $\operatorname{Conv}(\{P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}\} \cup \{P^{\mathrm{L},k}\}_k)$ . We find this region as follows. For  $1/2 < \epsilon \leq 4/7$ , we know from Proposition 5.4.2 that  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$  is non-classical but does not violate any of the CHSH inequalities, while it violates  $I_{2233} \leq 2$  (3.6). By Proposition 5.3.1, this is the only Bell inequality that  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$  violates for this range of  $\epsilon$ . Thus, the non-classical part of the polytope  $\operatorname{Conv}(\{P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}\} \cup \{P^{\mathrm{L},k}\}_k)$  is the convex hull of  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$  and all the local deterministic points that satisfy  $I_{2233} = 2$ . These are the 30 local deterministic points of Table 5.1. Hence we only need to optimise over convex combinations of  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$  with these 30 points and not all 81 local deterministic points, which reduces the size of the optimization (number of variables) and increases the chances of it being effective in detecting entropic violations if there are any.

Performing the optimization as outlined above, we found the maximum value to always be non-positive for both Shannon case and the Tsallis case with q = 1.1, 2, 3, 10, 50. We obtained similar results when taking other values of  $\epsilon \leq 4/7$  in the distribution  $P_{iso,\epsilon}^{(2,2,3,3)}$  and also when considering the inequalities  $I_{BC,q}^i$  for  $i \in \{1, 2, 3\}$ . This suggests that no point in the polytope  $Conv(\{P_{iso,\epsilon}^{(2,2,3,3)}\} \cup \{P^{L,k}\}_k)$  violates any of the (Shannon or Tsallis entropic) BC inequalities for  $\epsilon \leq 4/7$ . For  $\epsilon > 4/7$ , some mixing of  $P_{iso,\epsilon}^{(2,2,3,3)}$  with  $P_C^{(2,2,3,3)}$  gives a distribution that violates  $I_{BC,q}^4 \leq 0 \ \forall q \geq 1$  (c.f. Proposition 5.4.3). Note that in the Shannon (q = 1) case, the range of values of the mixing parameter v for which a violation can be found becomes arbitrarily small as  $\epsilon$  approaches 4/7 from above (see Figure 5.2). This limits the effectiveness of numerical tests for q close to 1. For instance, in the Shannon case our program was not able to detect violations of  $I_{BC}^4 \leq 0$  for  $\epsilon < 4.2/7$  (even though our analytic argument shows that these are present), while it was for  $\epsilon \geq 4.00001/7$ , but not below. The reason for this difference is in line with what one might expect by comparing the plots in Figure 5.2, where for  $\epsilon > 4/7$  the range of values of v for which a violation is possible is larger in the q = 2 case. This highlights an advantage of using Tsallis entropy and gives us further confidence that for  $1/2 \leq \epsilon < 4/7$  there is no violation.

Number	Vertex
1	$\frac{1}{c}(1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 0, 1, 0, 2, 0, 1, 1, 0, 0, 0, 2, 0, 1, 1, 2, 0, 0)$
2	$\frac{1}{c}(1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 1, 0, 1, 0, 1, 0, 1, 1, 2, 0, 0, 0, 1, 1, 0, 2, 0, 1, 0, 1, 0, 0, 2, 1, 1, 0)$
3	$\frac{1}{2}(1, 1, 0, 2, 0, 0, 0, 1, 1, 0, 2, 0, 1, 0, 1, 0, 0, 2, 1, 0, 1, 0, 1, 1, 1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 1, 0)$
4	$\frac{1}{c}(2, 0, 0, 1, 0, 1, 0, 2, 0, 1, 1, 0, 0, 0, 2, 0, 1, 1, 1, 0, 1, 0, 1, 1, 1, 0, 1, 0, 1, 0, 1, 0, 1, 1, 1, 1, 0)$
5	$\frac{1}{4}(1, 0, 0, 1, 0, 0, 0, 1, 1, 1, 1, 0, 1, 0, 1, 0, 0, 2, 1, 0, 1, 0, 1, 1, 1, 1, 0, 1, 0, 1, 0, 0, 1, 1, 0, 0)$
6	$\frac{1}{4}(1, 0, 0, 1, 0, 0, 0, 1, 1, 1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 0, 0, 0, 1, 0, 1, 1, 0, 1, 0, 1, 0, 1, 0, 0, 2, 1, 1, 0)$
7	$\frac{1}{4}(1, 0, 0, 1, 0, 0, 0, 1, 1, 1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 0, 1, 0, 2, 0, 1, 1, 0, 1, 0, 1, 0, 0, 1, 1, 0, 0)$
8	$\frac{1}{4}(1, 0, 0, 1, 0, 0, 0, 2, 0, 1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 0, 0, 0, 1, 0, 1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 1, 0)$
9	$\frac{1}{4}(1, 1, 0, 1, 0, 1, 0, 1, 0, 0, 1, 0, 0, 0, 2, 0, 1, 1, 1, 0, 1, 0, 1, 1, 0, 1, 0, 0, 0, 1, 0, 1, 1, 1, 1, 0)$
10	$\frac{1}{4}(1, 1, 0, 1, 0, 1, 0, 1, 0, 0, 1, 0, 1, 0, 1, 0, 1, 1, 1, 0, 0, 0, 1, 0, 1, 1, 0, 0, 0, 1, 0, 1, 1, 1, 1, 0)$
11	$\frac{1}{\overline{\epsilon}}(1, 1, 0, 1, 0, 1, 0, 1, 0, 0, 1, 0, 1, 0, 1, 0, 1, 1, 2, 0, 0, 0, 1, 1, 0, 1, 0, 0, 0, 1, 0, 1, 1, 1, 1, 0)$
12	$\frac{1}{\overline{\epsilon}}(1, 1, 0, 1, 0, 1, 0, 1, 1, 0, 2, 0, 0, 0, 1, 0, 0, 1, 1, 0, 1, 0, 1, 0, 1, 0, 0, 0, 1, 0, 1, 1, 1, 1, 0)$
13	$\frac{1}{7}(1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 1, 0, 0, 0, 1, 0, 0, 1, 1, 0, 1, 0, 1, 0, 1, 0, 0, 0, 1, 0, 1, 1, 2, 0, 0)$
14	$\frac{1}{\overline{r}}(1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 1, 0, 0, 0, 1, 0, 0, 1, 1, 0, 1, 0, 1, 1, 0, 2, 0, 1, 0, 1, 0, 0, 1, 1, 0, 0)$
15	$\frac{1}{\overline{\epsilon}}(1, 1, 0, 2, 0, 0, 0, 1, 0, 0, 1, 0, 1, 0, 1, 0, 1, 1, 1, 0, 0, 0, 1, 0, 1, 1, 0, 1, 0, 1, 0, 1, 1, 1, 1, 0)$
16	$\frac{1}{5}(2, 0, 0, 1, 0, 1, 0, 1, 1, 1, 1, 0, 0, 0, 1, 0, 0, 1, 1, 0, 1, 0, 1, 1, 1, 1, 0, 1, 0, 1, 0, 1, 1, 0, 0)$
17	$\frac{1}{6}(2, 1, 0, 2, 0, 1, 0, 2, 1, 1, 2, 0, 1, 0, 2, 0, 1, 2, 2, 0, 1, 0, 2, 1, 1, 2, 0, 1, 0, 2, 0, 1, 2, 2, 1, 0)$
18	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0
19	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0
20	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0
21	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0
22	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0
23	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0
24	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0
25	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0)
26	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 0, 1, 1, 0, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0)
27	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 0, 1, 0, 1, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
28	(0, 0, 0, 0, 0, 0, 0, 0, 1, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
29	(0, 0, 0, 0, 0, 0, 0, 0, 1, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 1, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0)
30	(0, 0, 0, 0, 0, 0, 0, 0, 1, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
31	(0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
32	(0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
33	(0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
34	(0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0)
35	(0, 0, 0, 0, 0, 0, 0, 1, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
36	(0, 0, 0, 0, 0, 0, 0, 1, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
37	(0, 0, 0, 0, 0, 0, 1, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0)
38	(0, 0, 1, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
39	(0, 1, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
40	(0, 1, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
41	(0, 1, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
42	(1, 0, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
	(1, 0, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 1, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0)
44	(1, 0, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
45	(1, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
46	(1, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
47	[1, 0, 0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,

Table 5.1: The vertices of  $\Pi_{\text{CHSH}}^{(2,2,3,3)}$  that saturate or violate the  $I_{2233}$  Inequality (3.6). All the vertices of the polytope can be obtained from the vertices listed here through local relabellings or exchange of parties.

However, because of the form of our objective function, the optimisation methods available do not guarantee to find the global maximum. Thus, our findings only constitute evidence for the conjectures and are not conclusive. In general, finding global optima for non-linear, non-convex/concave functions is an open question. A potential avenue for proving these conjectures is using DC (difference of convex) programming [118] since our objective function being a linear combination of entropies is a difference of convex functions.

### 5.6.3 Proofs

For the proofs we need the concept of the local weight of a non-signaling distribution [228, 66]

**Definition 5.6.1.** The *local weight* of a non-signaling distribution  $P_{XY|AB}$  is the largest  $\alpha \in [0,1]$  such that we can write

$$P_{XY|AB} = \alpha Q_{XY|AB}^{\mathrm{L}} + (1 - \alpha) Q_{XY|AB}^{\mathrm{NL}},$$

where  $Q_{XY|AB}^{L}$  is an arbitrary local distribution and  $Q_{XY|AB}^{NL}$  is an arbitrary non-signaling distribution. We denote the local weight by  $l(P_{XY|AB})$ .

The local weight of a distribution can be found by linear programming.

**Proposition 5.4.2.**  $P_{iso,\epsilon}^{(2,2,3,3)}$  is non-classical if and only if  $\epsilon > 1/2$ . Further, for  $\epsilon \le 4/7$ ,  $P_{iso,\epsilon}^{(2,2,3,3)}$  satisfies all the CHSH-type inequalities, while for  $\epsilon > 4/7$  it violates at least one CHSH-type inequality.

*Proof.* The distribution  $P_{iso,\epsilon}^{(2,2,3,3)} = \epsilon P_{NL} + (1-\epsilon) P_{noise}^{(2,2,3,3)}$  can be written as follows.

$$P_{iso,\epsilon}^{(2,2,3,3)} = \begin{bmatrix} A & B & B|A & B & B \\ B & A & B|B & A & B \\ B & B & A|B & B & A \\ \hline A & B & B|B & A & B \\ B & A & B|B & B & A \\ B & B & A|A & B & B \end{bmatrix}$$
(5.8)

where  $A = (2\epsilon+1)/9$  and  $B = (1-\epsilon)/9$ . We used the LPASSUMPTIONS linear program solver [62] (see Appendix 4.5.2 of Chapter 4 for details) to find the local weight of  $P_{iso,\epsilon}^{(2,2,3,3)}$ , as a function of  $\epsilon$  to be

$$l(P_{iso,\epsilon}^{(2,2,3,3)}) = \begin{cases} 1 & 0 \le \epsilon \le \frac{1}{2} \\ 2(1-\epsilon) & \frac{1}{2} \le \epsilon \le 1 \end{cases}$$

which establishes the first part of the claim.

The second part can be confirmed by computing the value of each CHSH-type quantity for the distribution  $P_{iso,\epsilon}^{(2,2,3,3)}$  and determining that each has a saturating  $\epsilon$  of at most 4/7.

**Proposition 5.4.3.** For  $\epsilon \leq 4/7$ ,  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)} = vP_{\mathrm{iso},\epsilon}^{(2,2,3,3)} + (1-v)P_{\mathrm{C}}^{(2,2,3,3)}$  does not violate any of the BC entropic inequalities (3.12) for any  $v \in [0,1]$ . However, for all  $\epsilon > 4/7$ , there exists a  $v \in [0,1]$  such that the entropic inequality  $I_{\mathrm{BC}}^4 \leq 0$  is violated by  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)}$ .

*Proof.* Consider the function  $f: (0,1) \times (0,1) \to \mathbb{R}$  given by

$$f(\epsilon, v) \coloneqq 3(3 - 2(1 - \epsilon)v) \ln[3 - 2(1 - \epsilon)v] + 5(1 - \epsilon)v \ln[(1 - \epsilon)v] - (1 + 2\epsilon)v \ln[(1 + 2\epsilon)v] - (3 - (2 + \epsilon)v) \ln[3 - (2 + \epsilon)v] - 3\ln 9$$

where we implicitly extend the domain to  $[0,1] \times [0,1]$  by taking the relevant limit. The Shannon entropic expression  $I_{BC}^4(\epsilon, v)$  evaluated for the distribution  $P_{E,\epsilon,v}^{(2,2,3,3)} = v P_{iso,\epsilon}^{(2,2,3,3)} + (1-v) P_C^{(2,2,3,3)}$  (seen as a function of  $\epsilon$  and v) is then given as

$$I_{\rm BC}^4 = \frac{1}{3\ln[2]}f(\epsilon, v)$$

Thus all the following arguments for  $f(\epsilon, v)$  also hold for  $I_{BC}^4$ .

We first use that for c > 0 and  $a \in \mathbb{R}$  for sufficiently small v we have

$$\ln[c+av] = \ln[c] + \frac{av}{c} + O(v^2).$$

Using this we can expand  $f(\epsilon, v)$  for small v as

$$f(\epsilon, v) = (4 - 7\epsilon)v\ln[v] - (4 - 7\epsilon)v(1 + \ln[3]) + v(5(1 - \epsilon)\ln[1 - \epsilon] - (1 + 2\epsilon)\ln[1 + 2\epsilon]) + O(v^2).$$
(5.9)

Thus, since  $\lim_{v\to 0} v \ln[v] = 0$  we have  $\lim_{v\to 0} f[\epsilon, v] = 0$ .

We also have

$$\frac{\partial}{\partial v}f(\epsilon,v) = (4-7\epsilon)\ln[v] + 5(1-\epsilon)\ln[1-\epsilon] - (1+2\epsilon)\ln[1+2\epsilon] -6(1-\epsilon)\ln[3-2(1-\epsilon)v] + (2+\epsilon)\ln[3-(2+\epsilon)v].$$
(5.10)

Note that  $5(1-\epsilon)\ln[1-\epsilon] \leq 0$ ,  $-(1+2\epsilon)\ln[1+2\epsilon] \leq 0 \ \forall \epsilon \in [0,1]$ . Further, since  $3-2(1-\epsilon)v \geq 3-(2+\epsilon)v$  and both terms are positive,  $6(1-\epsilon) \geq (2+\epsilon) \ \forall \epsilon < 4/7$ , and using the fact that  $\ln[]$  is an increasing function, we have  $-6(1-\epsilon)\ln[3-2(1-\epsilon)v] + (2+\epsilon)\ln[3-(2+\epsilon)v] \leq 0 \ \forall \epsilon \in [0,4/7]$ ,  $v \in [0,1]$ . This in turn implies that

$$\frac{\partial}{\partial v} f(\epsilon, v) \le (4 - 7\epsilon) \ln [v] \qquad \forall \ 0 \le \epsilon \le 4/7, 0 \le v \le 1$$

Hence we can conclude that for  $\epsilon \leq 4/7$ ,  $\frac{\partial}{\partial v} f(\epsilon, v) < 0$  for all  $v \in [0, 1]$ . Thus,  $f(\epsilon, v)$  is zero at v = 0 and, for  $\epsilon \leq 4/7$ , decreases with v, implying that  $f(\epsilon, v) \leq 0 \quad \forall \epsilon \in [0, 4/7], v \in [0, 1]$ . Note that  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)} = v p_{\mathrm{iso},\epsilon}^{(2,2,3,3)} + (1-v) p_C^{(2,2,3,3)}$  does not violate any of the analogous inequalities  $I_{\mathrm{BC}}^i \leq 0$  for any  $i \in \{1, 2, 3\}, \epsilon, v \in [0, 1]$ . This is because for this distribution, we always

have  $H(X_0Y_0) = H(X_0Y_1) = H(X_1Y_0)$ ,  $H(X_0) = H(X_1) = H(Y_0) = H(Y_1)$ . Thus all three inequalities  $I_{BC}^1 \leq 0$ ,  $I_{BC}^2 \leq 0$  and  $I_{BC}^3 \leq 0$  reduce to  $H(X_1) + H(Y_1) - H(X_1Y_0) - H(X_1Y_1) \leq 0$ , which is always satisfied since  $H(X_1) \leq H(X_1Y_0)$  and  $H(Y_1) \leq H(X_1Y_1)$  by the monotonicity of Shannon entropy.

Further, using the expression for the derivative of  $f(\epsilon, v)$  with respect to v in Equation (5.10), we find that for  $\epsilon > 4/7$ ,  $\lim_{v \to 0} \frac{\partial}{\partial v} f(\epsilon, v) = \infty$ . Thus, since  $f(\epsilon, v) = 0$  for v = 0, sufficiently close to v = 0 there exists a v such that  $f(\epsilon, v) > 0$ . This proves the claim.

**Proposition 5.4.4.** For  $\epsilon \leq 4/7$ ,  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)} = vP_{\mathrm{iso},\epsilon}^{(2,2,3,3)} + (1-v)P_{\mathrm{C}}^{(2,2,3,3)}$  does not violate any of the Tsallis BC inequalities (5.6) for any  $v \in [0,1]$  and q > 1. However, for  $\epsilon > 4/7$  and every q > 1, there always exists a  $v \in [0,1]$  such that the entropic inequality  $I_{\mathrm{BC},q}^4 \leq 0$  is violated by  $P_{\mathrm{E},\epsilon,v}^{(2,2,3,3)}$ .

*Proof.* The Tsallis entropic expression  $I_{BC,q}^4(\epsilon, v)$  evaluated for the distribution  $P_{E,\epsilon,v}^{(2,2,3,3)} = vP_{iso,\epsilon}^{(2,2,3,3)} + (1-v)P_C^{(2,2,3,3)}$  (seen as a function of q,  $\epsilon$  and v) is given as

$$\begin{split} I_{\mathrm{BC},q}^4 &= \frac{1}{q-1} \Bigg[ 9 \Bigg( \frac{3-2(1-\epsilon)v}{9} \Bigg)^q + 15 \Bigg( \frac{(1-\epsilon)v}{9} \Bigg)^q - \frac{6}{3^q} - 3 \Bigg( \frac{3-(2+\epsilon)v}{9} \Bigg)^q - 3 \Bigg( \frac{(1+2\epsilon)v}{9} \Bigg)^q \Bigg] \\ &=: \frac{g(q,\epsilon,v)}{q-1} \end{split}$$

For q > 1, the following arguments for  $g(q, \epsilon, v)$  also hold for  $I^4_{\mathrm{BC},q}$ . Note that

$$\frac{\partial}{\partial v}g(q,\epsilon,v) = \frac{q}{3^{2q-1}} \Big[ -6(1-\epsilon)\Big(3-2(1-\epsilon)v\Big)^{q-1} + 5(1-\epsilon)\Big((1-\epsilon)v\Big)^{q-1} + (2+\epsilon)\Big(3-(2+\epsilon)v\Big)^{q-1} - (1+2\epsilon)\Big((1+2\epsilon)v\Big)^{q-1}\Big].$$
(5.11)

Then, since

$$6(1-\epsilon)\left(3-2(1-\epsilon)v\right)^{q-1} \ge (2+\epsilon)\left(3-(2+\epsilon)v\right)^{q-1} \qquad \text{and} \\ (1+2\epsilon)\left((1+2\epsilon)v\right)^{q-1} \ge 5(1-\epsilon)\left((1-\epsilon)v\right)^{q-1}$$

hold for all  $\epsilon \leq 4/7$ ,  $v \in [0, 1]$  and q > 1, we have

$$\frac{\partial}{\partial v}g(q,\epsilon,v) \le 0 \qquad \forall \epsilon \le 4/7, \ v \in [0,1], \ q > 1.$$

Since  $g(q, \epsilon, v = 0) = 0$ , this implies that  $g(q, \epsilon, v) \le 0 \quad \forall \epsilon \le 4/7, v \in [0, 1], q > 1$ . Hence, for  $\epsilon \le 4/7$  we cannot violate  $I_{BC,q}^4 \le 0$  for any  $v \in [0, 1], q > 1$ .

Analogously to the Shannon case,  $P_{\text{E},\epsilon,v}^{(2,2,3,3)} = v P_{\text{iso},\epsilon}^{(2,2,3,3)} + (1-v) P_{\text{C}}^{(2,2,3,3)}$  also does not violate any of the inequalities  $I_{\text{BC},q}^1 \leq 0$ ,  $I_{\text{BC},q}^2 \leq 0$  or  $I_{\text{BC},q}^3 \leq 0$  for any  $\epsilon, v \in [0,1]$  and q > 1 by the same argument as in Proposition 5.4.3.

Further, Equation (5.11) implies  $\lim_{v\to 0} \frac{\partial}{\partial v} g(q,\epsilon,v) = \frac{q}{3^q}(7\epsilon - 4)$  which is always positive for  $\epsilon > 4/7$ . Since  $g(q,\epsilon,v) = 0$  for v = 0, this allows us to conclude that for  $\epsilon > 4/7$ , there exists a v sufficiently close to v = 0 such that  $g(q,\epsilon,v) > 0$ . This establishes the claim.

**Proposition 5.4.5.** The distribution  $P_{\epsilon}^{\text{CG},i}$  is classical for all *i* if and only if  $\epsilon \leq 4/7$ .

*Proof.* For the "if" part of the proof, we calculated all the 255 coarse-grainings of  $P_{iso,\epsilon=4/7}^{(2,2,3,3)}$  and used a linear programming algorithm to find that all of these are local (their local weight equals 1). Since decreasing  $\epsilon$  in  $P_{iso,\epsilon}^{(2,2,3,3)}$  cannot increase the violation of any (probability space) Bell inequality and neither can coarse-grainings, it follows that if  $\epsilon \leq 4/7$ , all coarse-grainings of  $P_{iso,\epsilon}^{(2,2,3,3)}$  are classical.

For the "only if" part we need to show that if all coarse-grainings of  $P_{iso,\epsilon}^{(2,2,3,3)}$  are classical then  $\epsilon \leq 4/7$  or equivalently, if  $\epsilon > 4/7$ , there exists at least one coarse-graining of  $P_{iso,\epsilon}^{(2,2,3,3)}$  that is non-classical. Consider the coarse-graining that involves combining the second output with the first for all 4 input choices. For  $P_{iso,\epsilon}^{(2,2,3,3)}$  as in Equation (5.8), this coarse-graining gives

$$P_{\text{CG},\epsilon}^{(2,2,3,3)} = \begin{bmatrix} 2(A+B) & 0 & 2B & 2(A+B) & 0 & 2B \\ 0 & 0 & 0 & 0 & 0 & 0 \\ 2B & 0 & A & 2B & 0 & A \\ 2(A+B) & 0 & 2B & 3B+A & 0 & A+B \\ 0 & 0 & 0 & 0 & 0 & 0 \\ 2B & 0 & A & A+B & 0 & B \end{bmatrix},$$
(5.12)

where  $A = (2\epsilon + 1)/9$  and  $B = (1 - \epsilon)/9$ . The  $I_{2233}$  value or left hand side of Equation (3.6) for this distribution is 9A - 3B. For this to be classical, we require that  $9A - 3B \le 2$  which gives  $\epsilon \le 4/7$ . Again using the LPASSUMPTIONS linear program solver [62] we found the local weight of this distribution as a function of  $\epsilon$ , which gives the following.

$$l(P_{\mathrm{CG},\epsilon}^{(2,2,3,3)}) = \begin{cases} 1, & 0 \le \epsilon \le \frac{4}{7}, \\ \frac{1}{9}(17 - 14\epsilon), & \frac{4}{7} < \epsilon \le 1. \end{cases}$$

In other words, if  $\epsilon > 4/7$ , then the coarse-graining  $P_{\text{CG},\epsilon}^{(2,2,3,3)}$  of  $P_{\text{iso},\epsilon}^{(2,2,3,3)}$  violates the  $I_{2233}$  Inequality (3.6) and is hence non-classical. This concludes the proof.

We prove the following two propositions together as they only differ in one step.

**Proposition 5.4.6.** Let  $\epsilon \leq 4/7$ . If Conjecture 5.4.1 holds, then every distribution in  $\Pi_{\epsilon}$  is Shannon entropically classical.

**Proposition 5.4.7.** Let  $\epsilon \leq 4/7$ . If Conjecture 5.4.2 holds, then every distribution in  $\Pi_{\epsilon}$  satisfies the Tsallis entropic BC Inequalities (5.6)  $\forall q > 1$ .

*Proof.* If Conjectures 5.4.1 and 5.4.2 hold, then for any  $\epsilon \leq 4/7$ ,  $I_{BC,q}^i \leq 0$  holds  $\forall q \geq 1$ ,  $\forall i \in \{1, 2, 3, 4\}$  and for all distributions in the polytope  $Conv(\{P_{iso,\epsilon}^{(2,2,3,3)}\} \cup \{P^{L,k}\}_k\})$ .<sup>9</sup> We want to

<sup>&</sup>lt;sup>9</sup>Note that q = 1 covers the Shannon case.

show that this implies the same for the larger polytope that comprises the convex hull of not just  $P_{iso,\epsilon}^{(2,2,3,3)}$  and local deterministic distributions  $\{P^{L,k}\}_k$ , but also all the local relabellings of  $P_{iso,\epsilon}^{(2,2,3,3)}$ , i.e., the polytope  $\Pi_{\epsilon} = \text{Conv}(\{P_{\epsilon}^{R,j}\}_j \cup \{P^{L,k}\}_k)$ . While we considered  $P_{iso,\epsilon}^{(2,2,3,3)}$  in Propositions 5.4.3 and 5.4.4 and Conjectures 5.4.1 and 5.4.2, due to symmetries, these also apply to every relabelling of  $P_{iso,\epsilon}^{(2,2,3,3)}$ , i.e.,  $I_{BC,q}^i \leq 0 \ (\forall i \in \{1,2,3,4\})$  throughout every polytope in the set  $\{\operatorname{Conv}(\{P_{\epsilon}^{\mathrm{R},j}\} \cup \{P^{\mathrm{L},k}\}_k)\}_j$  (where j runs over all the local relabellings of  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$ )<sup>10</sup>. This is because for every input-output relabelling of  $P_{iso,\epsilon}^{(2,2,3,3)}$ , we can correspondingly relabel the inequality expression  $I_{\mathrm{BC},q}^4$  (for  $q \ge 1$ ) and the same arguments as Propositions 5.4.3 and 5.4.4 again hold, and similarly Conjectures 5.4.1 and 5.4.2 also follow for this case.<sup>11</sup> From this argument, it follows that if Conjectures 5.4.1 and 5.4.2 hold, then  $I^i_{\text{BC},q} \leq 0 \ \forall q \geq 1, \ \forall i \in \{1, 2, 3, 4\}$ everywhere in the union of the polytopes, i.e., everywhere in  $\bigcup_i \operatorname{Conv}(\{P_{\epsilon}^{\mathrm{R},j}\} \cup \{P^{\mathrm{L},k}\}_k)$ .

To conclude the proof it remains to show that  $\bigcup_i \operatorname{Conv}(\{P_{\epsilon}^{\mathrm{R},j}\} \cup \{P^{\mathrm{L},k}\}_k) = \prod_{\epsilon} \forall \epsilon \leq 4/7$ . This is established below. Then, Proposition 5.4.7 automatically follows while Proposition 5.4.6 follows from this and Lemma 3.2.1. 

**Proposition 5.6.1.**  $P_{\text{mix},\epsilon}^j \coloneqq \frac{1}{2} P_{\text{iso},\epsilon}^{(2,2,3,3)} + \frac{1}{2} P_{\epsilon}^{\text{R},j}$  is local  $\forall j \neq 1$  if and only if  $\epsilon \leq 4/7$ .

*Proof.* For the "if" part of the proof, we used a linear program to confirm that  $P^j_{\text{mix},\epsilon=4/7}$  is local<sup>12</sup>  $\forall j \neq 1$ . Since reducing  $\epsilon$  in  $P_{\min,\epsilon}^j$  cannot decrease the local weight, this also holds for  $\epsilon < 4/7.$ 

The "only if" part of the proof is equivalent to showing that  $\forall \epsilon > 4/7, \exists j$  such that  $P_{\min,\epsilon}^j$  is non-classical. Consider the particular local relabelling that corresponds to Alice swapping the outputs "1" and "2" only when her input is A = 1. Let the distribution obtained by applying this relabelling to  $P_{iso,\epsilon}^{(2,2,3,3)}$  be  $P_{\epsilon}^{R}$ . Then  $P_{mix,\epsilon} = \frac{1}{2}P_{iso,\epsilon}^{(2,2,3,3)} + \frac{1}{2}P_{\epsilon}^{R}$ . More explicitly,

$P_{\epsilon}^{\mathrm{R}} =$	A B A B B B	B A B B A	B B A B A B	A B B A B	B A B A B B	В В А В А	and	$p_{\mathrm{mix},\epsilon}$ =	A B A B B B	B A B *	B B A B *	A B B *	B A B A B B	B B A B *	,	(5.13)
	B	A	В	B	В	A			B	*	*	*	В	*		

where  $* = \frac{A+B}{2}$ ,  $A = (2\epsilon + 1)/9$  and  $B = (1-\epsilon)/9$ . Now consider the Bell inequality  $\text{Tr}(M^T P) \ge 1$ 

<sup>&</sup>lt;sup>10</sup>A note on notation:  $\{P_{\epsilon}^{\mathrm{R},j}\}$  is a set comprising a single element, while  $\{P_{\epsilon}^{\mathrm{R},j}\}_{j}$  is a set whose elements are the distributions for every j.

<sup>&</sup>lt;sup>11</sup>Note that output relabellings don't change the entropic expression but input relabellings (4 in number) can give either one of  $I_{\mathrm{BC},q}^i \leq 0$  for  $i \in \{1, 2, 3, 4\}$ . Thus for an output relabelling of  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$ , one can continue using  $I_{BC,a}^4$  in Propositions 5.4.3 and 5.4.4 and the following Conjectures while for input relabellings, one simply needs to relabel the inequalities accordingly and run the same arguments.  ${}^{12}j = 1$  is excluded since  $P^1_{\text{mix},\epsilon} = P^{(2,2,3,3)}_{\text{iso},\epsilon}$  which is non-classical for  $\epsilon > 1/2$ .

where P is an arbitrary distribution in the (2, 2, 3, 3) scenario and M is the following matrix.

$$M \coloneqq \begin{bmatrix} 0 & 1 & 1 & 0 & 1 & 1 \\ 1 & 0 & 1 & 1 & 0 & 1 \\ 1 & 1 & 1 & 0 & 0 & 0 \\ \hline 0 & 1 & 0 & 1 & 0 & 0 \\ 1 & 0 & 0 & 0 & 1 & 0 \\ 1 & 0 & 0 & 0 & 1 & 0 \end{bmatrix}$$
(5.14)

Then, the condition for  $P_{\min,\epsilon}$  to be non-classical with respect to this Bell inequality i.e.,  $\operatorname{Tr}(M^T P_{\min,\epsilon}) < 1$  gives us  $\epsilon > 4/7$ . Since  $P_{\epsilon}^{\mathrm{R}}$  is a local relabelling of  $P_{\mathrm{iso},\epsilon}^{(2,2,3,3)}$ , there exists a *j* such that  $P_{\epsilon}^{\mathrm{R}} = P_{\epsilon}^{\mathrm{R},j}$  and hence  $P_{\min,\epsilon} = P_{\min,\epsilon}^{j}$ . Thus we have shown that whenever  $\epsilon > 4/7$ ,  $\exists j$  such that  $P_{\min,\epsilon}^{j}$  is non-classical and hence  $P_{\min,\epsilon}^{j}$  is local for all *j* implies that  $\epsilon \leq 4/7$  which concludes the proof.

By symmetry, there is an analogue of Proposition 5.6.1 with  $P_{iso,\epsilon}^{(2,2,3,3)}$  replaced by  $P_{\epsilon}^{\mathbf{R},i}$  for any i, so we have the following corollary.

**Corollary 5.6.1.**  $\tilde{P}_{\min,\epsilon}^{i,j} \coloneqq \frac{1}{2} P_{\epsilon}^{\mathrm{R},i} + \frac{1}{2} P_{\epsilon}^{\mathrm{R},j}$  is local  $\forall j \neq i$  if and only if  $\epsilon \leq 4/7$ .

**Theorem 5.6.1** (Bemporad et. al 2001 [23]). Let  $\mathcal{P}$  and  $\mathcal{Q}$  be polytopes with vertex sets V and W respectively i.e.,  $\mathcal{P} = \operatorname{Conv}(V)$  and  $\mathcal{Q} = \operatorname{Conv}(W)$ . Then  $\mathcal{P} \cup \mathcal{Q}$  is convex if and only if the line-segment [v, w] is contained in  $\mathcal{P} \cup \mathcal{Q} \ \forall v \in V$  and  $w \in W$ .

Let  $\mathcal{P}_j = \operatorname{Conv}(\{P_{\epsilon}^{\mathrm{R},j}\} \cup \{P^{\mathrm{L},k}\}_k\})$  and  $\mathbb{P}$  be the set of polytopes  $\mathbb{P} \coloneqq \{\mathcal{P}_j\}_j$ . We use the above theorem to prove the final result that establishes Propositions 5.4.6 and 5.4.7.

**Lemma 5.6.1.** Let  $V_j$  be the vertex set of the polytope  $\mathcal{P}_j \in \mathbb{P}$  and  $V := \bigcup_j V_j$ . Then,  $\bigcup_{\mathcal{P}_j \in \mathbb{P}} \mathcal{P}_j = \operatorname{Conv}(\bigcup_i V_j) = \operatorname{Conv}(V) = \prod_{\epsilon} \forall \epsilon \leq 4/7$ .

Proof. By Corollary 5.6.1, for  $i \neq j$  we have that  $\frac{1}{2} \left( P_{\epsilon}^{\mathrm{R},i} + P_{\epsilon}^{\mathrm{R},j} \right) \in \mathcal{L}^{(2,2,3,3)} = \mathcal{P}_i \cap \mathcal{P}_j \ \forall \epsilon \leq 4/7$ . This implies that  $\alpha P_{\epsilon}^{\mathrm{R},i} + (1 - \alpha) P_{\epsilon}^{\mathrm{R},j} \in \mathcal{P}_i \cup \mathcal{P}_j \ \forall \epsilon \leq 4/7, \ \alpha \in [0,1]$ , i.e., the line segment  $[P_{\epsilon}^{\mathrm{R},i}, P_{\epsilon}^{\mathrm{R},j}]$  is completely contained in the union of the corresponding polytopes  $\mathcal{P}_i \cup \mathcal{P}_j$ . Note that all other line segments  $[v_i, v_j]$  with  $v_i \in V_i$  and  $v_j \in V_j$  are contained in  $\mathcal{P}_i \cup \mathcal{P}_j$  by construction since at least one of  $v_i$  or  $v_j$  would be a local-deterministic vertex. Therefore, by Theorem 5.6.1,  $\mathcal{P}_i \cup \mathcal{P}_j$  is convex  $\forall i, j$  and  $\epsilon \leq 4/7$ . We can then apply Proposition 5.6.1 and Theorem 5.6.1 to the convex polytopes  $\mathcal{P}_i \cup \mathcal{P}_j$  and  $\mathcal{P}_k$  and show that  $\mathcal{P}_i \cup \mathcal{P}_j \cup \mathcal{P}_k$  is convex  $\forall i, j, k$  and  $\epsilon \leq 4/7$ . Proceeding in this way, we conclude that  $\bigcup_{\mathcal{P}_i \in \mathbb{P}} \mathcal{P}_i$  is convex  $\forall \epsilon \leq 4/7$ , and hence  $\bigcup_{\mathcal{P}_j \in \mathbb{P}} \mathcal{P}_j = \operatorname{Conv}(\bigcup_i V_j) = \operatorname{Conv}(V) = \prod_{\epsilon}$ .

# CHAPTER 6

# Cyclic and fine-tuned causal models and compatibility with relativistic principles

The notion of causation is closely tied with that of space-time, but how are the two related? L Causation can be modelled operationally, without any reference to space-time, based only on observed correlations and interventions on systems [159]. It can be represented through directed graphs that are in principle independent from the partial order imposed by space-time. That the arrows of this directed graph are compatible with the light cone structure of space-time is an empirical observation that supports the notion of relativistic causality. For example, if an external intervention on a variable A produces an observable change in another variable B, then one would say that A affects B and call A a cause of B. Assigning space-time locations to the variables and requiring the effect B to always be in the future light cone of the cause A makes this causal relationship compatible with the partial order of Minkowski space-time. In general, a variable may jointly affect a set of variables without affecting individual variables in the set. Such scenarios correspond to causal models where the correlations are *fine-tuned* to hide certain causal influences. For certain embeddings of such causal models in spacetime (such as the *jamming* scenario considered in [105, 116]), this can lead to superluminal influence without superluminal signalling, or closed timelike curves that do not lead to an observable signalling to the past, some of these are strictly post-quantum features that do not arise in standard quantum theory. For a deeper understanding of causality in quantum and post-quantum theories, it is thus important to disentangle the two notions of causation– the operational one and the relativistic one, and to characterise the relationship(s) between them in these general scenarios. A rigorous framework for causally modelling such scenarios and formalising their relationship with a space-time structure is not available. Such an analysis can deepen our understanding of the principles of causality other than broad relativistic principles (such as no superluminal signalling) that apply in the quantum world. This chapter is based on joint work with Roger Colbeck that we hope to turn into a paper soon.

## 6.1 Summary of contributions: in words and in poetry

We develop an operational framework for analysing cyclic and fine-tuned causal models in non-classical theories, and their compatibility with a space-time structure. This in particular includes the jamming scenarios envisaged in [105, 116] where a party superluminally influences correlations between two other parties. We present the framework in two parts, the first concerns causal models (Section 6.3) and the second characterises the embedding of these causal models in a space-time structure (Section 6.4). In Sections 6.5.1 and 6.5.2, we derive necessary and sufficient conditions for the causal model to be compatibly embedded in a space-time structure and to rule out operationally detectable causal loops. Following this, in Section 6.6, we analyse the jamming scenario [105, 116] using causal modelling. We prove in Theorem 6.6.1 that if the shared joint system is classical and accessible to the parties, then the jamming scenario would indeed lead to superluminal signalling, contrary to what is claimed in [105, 116]. In Section 6.6.2, we analyse the results of [116] regarding such scenarios, providing counter examples to some of their claims, and discussing how our results provide possible resolutions. Keeping in theme with the many cycles and paradoxes that the reader will encounter in this thesis, we present them with a mildly self-referential poem before looping back into the technicalities.

What is causation? Can we always discover its presence? If it is not mere correlation, Then what is its essence?

What is it that orders the cause Before that what we dub the effect? What brings order to the chaos Telling us what we can or cannot affect?

Time<sup>1</sup> gives us a direction "forward". To pose questions without already knowing the answer. To drive causation through its invisible arrow, A rule that seems doomed to be followed.

Causality and space-time, At first, we disconnect. By actions and observable changes, We model cause and effect.

Space-time out of the way, we abstract away, Allowing causal loops, and hidden influences

<sup>1</sup>What is time, does it really "exist"? That is a question for another time, Another poem, and I must resist. That travel faster than light can propagate. But we'd like to avoid paradoxes That put our grandfathers' lives at stake. Requiring a joint probability distribution Keeps these inconsistencies at bay.

Bringing one foot back to reality, We embed the causal model in space-time Derive conditions for its compatibility. Align the actions and changes With the past and future, But this doesn't suffice to exorcise All the loops we set afoot earlier.

Some lurk, beyond the observable realm. Finely tuned parameters that don't reveal them. We can only rule out loops that manifest In observations that we can operationally test. So we critically analyse a previous claim [116] That specifies how all the loops can be tamed.

A plethora of questions begging for answers, Some to be addressed before publishing. Therefore, much like the work presented here, This poem too needs some fine-tuning.

# 6.2 Motivation for analysing cyclic and fine-tuned causal models

A ubiquitous assumption made in majority of the literature on causal modelling is that of *faithfulness*, which we briefly discussed in Section 2.5.1 (in particular, Example 2.5.1). This corresponds to the requirement that all the (conditional) independences observed in the correlations arising from a causal structure must be a consequence of graph separation properties (i.e., d-separation relations Definition 2.5.7) of that causal structure. In *unfaithful* causal models, the causal parameters are fine-tuned such that some causal influences exactly cancel out to yield additional independences beyond what is implied by d-separation. Fine-tuning creates problems for causal inference since it leads to correlations that do not faithfully reflect the causal relations and it is considered an undesirable property of causal models and often avoided in the literature, also on the grounds that fine-tuned causal models constitute a set of measure-zero. In fact quantum correlations in the Bell causal structure 3.1a can be explained using classical causal models if we allow for additional, fine-tuned causal influences (e.g., from A to X, see Figure 6.11) [224], but a faithful explanation using quantum causal models is often preferred.

However, there are a number of examples, as we will see below, which necessitate a fine-tuned explanation irrespective of whether the causal structure is classical and non-classical. These include certain everyday scenarios, cryptographic protocols as well as those arising in certain post-quantum theory. Moreover, another common assumption in the causality literature is that the causal structure is acyclic, this is justifiably so since our observations suggest that causal influences only propagate in one direction. Nevertheless, allowing fine-tuned causal influences makes possible cyclic causal structures that are compatible with minimal notions of relativistic causality, such as the impossibility of signalling superluminally at the observed level. Cyclic causal models have also been employed in the classical literature for describing systems with feedback loops [160, 84].

## 6.2.1 Friedman's thermostat and one-time pad

Consider a house with an ideal thermostat. Such a thermostat would maintain a constant inside temperature  $T_I$  throughout the year by adjusting the energy consumption E in accordance with the outside temperature  $T_O$ . An individual who does not know how a thermostat works might conclude that  $T_O$  and E which are correlated have a causal relationship between each other while the indoor temperature  $T_I$  is causally independent of everything else. However, an engineer who is more well-versed with the workings of a thermostat knows that both  $T_O$  and E exert a causal influence on  $T_I$ , and that these influences must perfectly cancel each other out for the thermostat to function ideally. The causal model in this case is fine-tuned since the independence of  $T_I$  from  $T_O$  and E does not correspond to a d-separation relation in the causal structure (Figure 6.1a). This thermostat analogy which is attributed to Milton Friedman [87], can be extended to a number of other scenarios such as the effect of fiscal and monetary policies on economic growth [183], or physical systems where several forces exactly balance out.

In cryptographic settings, examples that necessitate fine-tuning include one-time pad or the "traitorous lieutenant problem" [38]. Consider a general who wishes to relay an important secret message M to an ally and has two lieutenants available as messengers, but one of them is a traitor who might leak the message to enemies. Consider for simplicity that M is a single bit. The general could then adopt the following strategy: Depending on M = 0 or M = 1, two uniformly distributed bits  $M_1$  and  $M_2$  are generated such that  $M_1 = M_2$  or  $M_1 \neq M_2$ .  $M_1$  is given to the first and  $M_2$  to the second lieutenant to relay to the ally. Then the ally would receive  $M_1$  and  $M_2$  and can simply use modulo-2 addition  $\oplus$  to obtain  $M^*$  which would be identical to the original message  $M^* = M = M_1 \oplus M_2$  (Figure 6.1b). More importantly, the individual messages  $M_1$  and  $M_2$  contain no information about M and hence neither lieutenant has any information about the secret message. A similar protocol underlies the one-time pad where a message M is encrypted using a secret key K (both binary for this example) to produce an encrypted message  $M_E = M \oplus K$  which can be sent through a public channel as it will carry no information about the original message M if the key K is uniformly distributed and is kept private. Only a receiver of  $M_E$  who knows the key K can decrypt the message  $M = M_E \oplus K$ (Figure 6.1c). Hence fine-tuning of causal influences i.e., causation in the absence of correlation, is key to the working of such cryptographic protocols.



Figure 6.1: Causal structures for the motivating examples described in the main text: (a) Friedman's thermostat (b) Traitorous Lieutenant (c) One-time pad. In this chapter, we use squiggly arrows  $\longrightarrow$  to denote causal influence, as this will later be classified into solid and dashed arrows. Note that there may be additional causal influences, for example there can be a direct influence of the outside temperature  $T_O$  and/or the inside temperature  $T_I$  on the energy consumption E in (a), the latter would make it a cyclic causal model. Further, in examples like (b), we will later see that an additional common cause between  $M_1$  and  $M_2$  will be required to fully explain the correlations (c.f. Figure 6.4a).

Further, cyclic causal models have been analysed in the classical literature [160, 84] for the purpose of describing complex systems involving feedback loops. Note that the cyclic dependences here do not correspond to closed time-like curves since the variables under question are considered over a period of time— e.g., demand at time  $t_1$  influences the price at time  $t_2 > t_1$ , which in turn influences the demand at time  $t_3 > t_2$ . Therefore in order to characterise genuine closed time-like curves one must consider not only the pattern of causal influences, but also how the relevant variables are embedded in a space-time structure.

## 6.2.2 Jamming non-local correlations

Another example that involves fine-tuning, even though it has not been motivated or discussed in this context, is that of jamming non-local correlations introduced in [105]. Consider three, space-like separated parties, Alice, Bob and Charlie sharing a tripartite system  $\Lambda$  which they measure using measurement settings A, B and C, producing outcomes X, Y and Z respectively. Suppose that their space-time locations are such that Bob's future light cone entirely contains the joint future of Alice and Charlie, as shown in Figure 6.2. The standard no-signaling conditions forbid the input of each party from being correlated with the outputs of any subset of the remaining parties, in particular, the joint distribution  $P_{XYZ|ABC}$  satisfies  $P_{XZ|ABC} = P_{XZ|AC}$ . In [105] it is argued that a violation of this requirement does not lead to superluminal signalling in the space-time configuration of Figure 6.2, as long as X and Z are individually independent from B i.e.,  $P_{X|ABC} = P_{X|A}$  and  $P_{Z|ABC} = P_{Z|C}$ . This is because any influence that B exerts jointly on X and Z can only be checked when X and Z are brought together to evaluate the correlations  $P_{XZ|ABC}$ , which is only possible in their joint future, that is by construction contained in the future of B. This scenario allows Bob to jam the correlations between Alice and Charlie non-locally, hence the name.

In [116], where jamming is further analysed, the causal structure for such an experiment is represented by introducing a new random variable  $C_{XZ}$  associated with the set  $\{X, Z\}$  that encodes the correlations between its elements. Then B is seen as a cause of  $C_{XZ}$  but not as a cause of either X or Z. In general scenarios, this representation would require adding a new variable for every non-empty subset of the observed nodes, which can become intractable.<sup>2</sup> Further, this representation does not always correspond to what is physically going on— for instance, in the example of the traitorous lieutenant, this would introduce a new variable  $C_{M_1M_2}$ that is observably influenced by the general's original message M, while M would no longer be seen as a cause of  $M_1$  or  $M_2$ . However, we know that we physically generated  $M_1$  and  $M_2$ using<sup>3</sup> M, hence it is indeed a cause of at least one of them. Therefore, we aim to develop a new approach to causally modelling in a general class of fine-tuned and cyclic scenarios, using only the original variables/systems. The following proposition illustrates that the jamming scenario considered in [105, 116] necessarily corresponds to a fine-tuned causal model over the original variables. Here, jamming is considered in the context of multipartite Bell scenarios where the jamming variable is a freely chosen input of one of the parties. In the causal model approach adopted here, we will take free choice of a variable to correspond to the exogeneity of that variable in the causal structure.<sup>4</sup>

**Proposition 6.2.1.** Consider a tripartite Bell experiment where three parties Alice, Bob and Charlie share a system  $\Lambda$  which they measure using the setting choices A, B and C, producing the measurement outcomes X, Y and Z respectively. Let  $\mathcal{G}$  be any causal structure with only  $\{A, B, C, X, Y, Z\}$  as the observed nodes where A, B and C are exogenous. Then any joint distribution  $P_{XYZ|ABC}$  corresponding to the jamming correlations of [105, 116] defines a finetuned causal model over  $\mathcal{G}$ , irrespective of the nature (classical, quantum or GPT) of  $\Lambda$ .

*Proof.* Jamming allows Bob's input B to be correlated jointly with X and Z but not individually with X or Z. Hence jamming correlations in the tripartite Bell experiment of [105, 116] are characterised by the conditions  $B \perp X$  and  $B \perp Z$  while  $B \not\downarrow \{X, Z\}$ . Since B is exogenous (i.e., has no incoming arrows), the only way to explain the correlation between B and  $\{X, Z\}$  is through an outgoing arrow or a directed path from B to the set  $\{X, Z\}$  i.e., either an arrow from B to X, or from B to Z or both.<sup>5</sup> Since we require both independences  $B \perp X$  and  $B \perp Z$  to hold, at least one of these will not be a consequence of d-separation and hence the causal model must be fine-tuned in order to produce these correlations in the causal structure  $\mathcal{G}$ .  $\Box$ 

The simplest example of a jamming where  $B = X \oplus Z$  and all variables are binary uniformly

<sup>5</sup>If this were not the case, B would be d-separated from  $\{X, Z\}$  and therefore cannot be correlated with it.

<sup>&</sup>lt;sup>2</sup>In general, this representation would include up to  $2^n - 1$  observed variables whenever the original set of observed variables has n elements.

<sup>&</sup>lt;sup>3</sup>And possibly some additional information to explain the distribution over the individual variables. As we will see later in Figure 6.4a, a common cause  $\Lambda$  between  $M_1$  and  $M_2$  would also be required in such examples.

<sup>&</sup>lt;sup>4</sup>Free choice is often of a variable B is often defined through the condition that B can only be correlated with variables in its future [61]. We will discuss the relation between this notion of free choice and ours, namely taking B to be exogenous, later in the chapter (Section 6.6.2).



Figure 6.2: A tripartite Bell experiment: Three parties Alice, Bob and Charlie share a tripartite system  $\Lambda$ , they measure their subsystem using the freely chosen measurement settings A, B and C, producing the outcomes X, Y and Z respectively. (a) Space-time configuration for the jamming scenario [105, 116]: The measurement events of the three parties are space-like separated such that the future of Bob's input B contains the entire joint future (in blue) of Alice and Charlie's outputs X and Z. Here, B is allowed to signal to X and Z jointly (which can only be verified in the blue region) but not individually. In [116], a new variable  $C_{XZ}$  is introduced, located at the earliest point in the joint future of X and Z and representing the correlations between X and Z. (b) Causal structure for the usual tripartite Bell experiment. Note that in order to explain jamming correlations, we must either include additional causal arrows from B to X or B to Z or both (Proposition 6.2.1), or introduce a new node  $C_{XZ}$  with an incoming arrow from B [116]. We adopt the former approach in the rest of this chapter.

distributed (the remaining variables are irrelevant here), and we will revisit this example several times in this chapter. These are in fact the same correlations as the traitorous lieutenant example. However in the jamming case, the three variables involved are pairwise space-like separated and since B is exogenous, this corresponds to a situation where B superluminally influences the correlations between X and Z.

## 6.3 The framework, Part 1: Causal models

#### 6.3.1 Cyclic and fine-tuned causal models

In Section 2.5 we have reviewed the standard literature on acyclic and faithful causal models, both in the classical as well as non-classical cases. Here, following the motivation set out in the previous sections, we wish to relax the assumptions of acyclicity and faithfulness and extend these methods to cyclic and fine-tuned causal structures. While quantum cyclic causal models have been previously studied [18], these have only been analysed in the faithful case and are based on the split-node causal modelling approach of [6] (this approach is briefly discussed in

Section 2.5.2.3). This approach is not equivalent to the standard causal modelling approach such as [115] in the cyclic case, for example the former forbids faithful 2 node cyclic causal structures [18] but the latter does not, and the former admits a Markov factorisation (analogous to Equation (2.45)) while the latter does not in general (as we will see later in the chapter). To the best of our knowledge, there is no prior framework for causally modelling cyclic and unfaithful causal structures in the presence of quantum and post-quantum latent nodes, the lack of a Markov factorisation posing a particular difficulty. Here, we develop the bare bones of such a framework that will suffice for our main results. We note that there may be other, inequivalent ways to do the same.<sup>6</sup> We will define causal models in terms of minimal conditions that they must satisfy at the level of the observed nodes which are classical.

**Causal structure:** Causal structures will be represented using directed graphs, of which the directed acyclic graphs of Section 2.5 are special cases. Edges in these graphs will be denoted using  $\rightsquigarrow$  (unlike  $\rightarrow$  in previous chapters), as it will be useful to later classify these edges into solid  $\rightarrow$  and dashed  $\rightarrow$  ones based on certain operational conditions for detecting causation. These causal structures can have observed as well as unobserved nodes, where the former are classical random variables and the latter can be classical or non-classical systems. We will then assume the existence of such a causal structure (though it may be unknown) and use the following definition of *cause* that directly arises from this assumption.

**Definition 6.3.1** (Cause). Given a causal structure represented by a directed graph  $\mathcal{G}$ , possibly containing observed as well as unobserved nodes, we say that a node  $N_i$  is a *cause* of another node  $N_i$  if and only if there is a directed path  $N_i \longrightarrow N_j$  from  $N_i$  to  $N_j$  in  $\mathcal{G}$ .

**Observed distribution:** In classical acyclic causal models, the causal Markov condition (2.45) is used for defining the compatibility of the observed distribution with the causal structure [159]. In the non-classical case, the generalised Markov condition of [115] provides a compatibility condition (Section 2.5.2.1). However, in cyclic causal models, demanding such a factorisation will be too restrictive even in the classical case. For example, consider the simplest cyclic causal structure, the 2-cycle where  $X \longrightarrow Y$  and  $Y \longrightarrow X$ , with X and Y observed and X = Y. The Markov condition would imply that  $P_{XY} = P_{X|Y}P_{Y|X}$ . Since X = Y, the right hand side is a product of deterministic distributions, which forces  $P_{XY}$  to also be deterministic in order to be a valid distribution. Therefore, we instead use a weaker compatibility condition in terms of d-separation between observed nodes (Definition 2.5.7), a concept that also applies to non-classical causal structures (Theorem 2.5.3) and define compatibility of the observed distribution with a cyclic causal structure as follows within our framework.

**Definition 6.3.2** (Compatibility of observed distribution with a causal structure). Let  $\{X_1, ..., X_n\}$  be a set of random variables denoting the observed nodes of a directed graph  $\mathcal{G}$ , and  $P_{X_1,...,X_n}$  be a joint probability distribution over them. Then P is said to be *compatible* with  $\mathcal{G}$  if for all disjoint subsets X, Y and Z of  $\{X_1, ..., X_n\}$ ,

 $X \perp^d Y | Z \implies X \perp Y | Z$  i.e.,  $P_{XY|Z} = P_{X|Z} P_{Y|Z}$ ,

<sup>&</sup>lt;sup>6</sup>Based on a different condition for compatibility of a distribution with a causal structure, for example.

for the marginal distribution,  $P_{XYZ} = P_{XY|Z}P_Z$  over  $X \cup Y \cup Z$  of  $P_{X_1,\dots,X_n}$ .<sup>7</sup>

Definition 6.3.2 is essentially the *soundness* of d-separation (c.f. Theorem 2.5.2) and is satisfied by classical as well as non-classical causal models in the acyclic case [159, 115]. It is sometimes referred to as the *global directed Markov condition* [180] and holds in several classical cyclic causal models [160, 84] and well as non-classical acyclic causal models [115]. In Appendix 6.8.1, we provide an example of a quantum cyclic causal model where this holds. However there also exist cyclic causal models producing observed distributions that do not satisfy Definition 6.3.2, we discuss this further in the Appendix as well. There, we also present further motivation for the compatibility condition of Definition 6.3.2 in terms of the properties of the underlying causal mechanisms (e.g., functional dependences in the classical case, completely positive maps in the quantum case) and outline possible methods for identifying when this condition might hold for non-classical cyclic causal models. Even in the classical case, several inequivalent definitions of compatibility are possible (which become equivalent in the acyclic case) and [84] presents a detailed analysis of these conditions and the relationships between them. Such an analysis for the non-classical case is beyond the scope of the present thesis. For the rest of this chapter, we will only consider causal models that satisfy the compatibility condition 6.3.2.

We will work with the following minimal definition of a causal model in this chapter which is in terms of the graph and observed distribution only. Further details such as the functional relationships between classical variables, choice of quantum states/transformations, or generalised tests (Section 2.5.2.1) can also be included in the full specification of the causal model. These constitute the *causal mechanisms* of the model. Developing a complete and formal specification of these mechanisms and deriving the conditions for their compatibility with cyclic and fine-tuned causal models is a tricky problem, we outline possible ideas for this in Appendix 6.8.1 and leave the full problem for future work. The following minimal definition suffices for the purposes of the main results of this chapter.

**Definition 6.3.3** (Causal model). A causal model over a set of observed random variables  $\{X_1, ..., X_n\}$  consists of a directed graph  $\mathcal{G}$  over them (possibly involving classical/quantum/GPT unobserved systems) and a joint distribution  $P_{X_1,...,X_n}$  that is compatible with this graph according to Definition 6.3.2.

Definition 6.3.2 allows for fine-tuned distributions to be compatible with the causal structure since it only requires that d-separation implies conditional independence and not the converse. Fine-tuned causal models may in general have an arbitrary number of additional conditional independences that are not implied by the d-separation relations in the corresponding causal graph. The following lemma shows that some additional conditional independences that are not directly implied by d-separation can be derived using d-separation and other independences (not implied by d-separation) that may be provided.

<sup>&</sup>lt;sup>7</sup>Note that we only need to consider d-separation between observed sets of variables in this definition, however the paths being considered may involve unobserved nodes. For example, if the observed variables X and Y have an unobserved common cause  $\Lambda$ , then X and Y are not d-separated by the empty set since there is an unblocked path between X and Y through the unobserved common cause, and naturally we don't expect X and Y to be independent in this case.

**Lemma 6.3.1.** Let  $S_1$ ,  $S_2$  and  $S_3$  be three disjoint sets of RVs such that  $S_1 \perp S_2 | S_3$ . If S is a set of RVs that is d-separated from these sets in a directed graph  $\mathcal{G}$  containing  $S_1$ ,  $S_2$ ,  $S_3$  and S are nodes i.e.  $S \perp^d S_i \forall i \in \{1, 2, 3\}$ , then any distribution P that is compatible with  $\mathcal{G}$  also satisfies the following conditional independences,

$$(S_1 \cup S) \perp S_2 | S_3, \quad S_1 \perp (S_2 \cup S) | S_3 \text{ and } S_1 \perp S_2 | (S_2 \cup S).$$

A proof can be found in Appendix 6.8.2. Note that this lemma is trivial in the case of faithful causal models. This is because, the independence  $S_1 \perp S_2 | S_3$  implies the d-separation  $S_1 \perp^d S_2 | S_3$  in this case and combined with  $S \perp^d S_i$ , this would immediately imply the d-separations  $(S_1 \cup S) \perp S_2 | S_3, S_1 \perp (S_2 \cup S) | S_3$  and  $S_1 \perp S_2 | (S_2 \cup S)$ , which in turn imply the corresponding independences. This property is not so evident in the case of fine-tuned causal models but as we have shown, it nevertheless holds here. Specific examples of this property for fine-tuned causal models are discussed in Section 6.3.5.

## 6.3.2 Interventions and affects relations

We adopt the characterisation of interventions outlined in Section 2.5.2.2, following the augmented graph approach [159] presented in Section 2.5.1.2. The main intuition behind interventions in the acyclic case carries forth to the cyclic case as well, but we will not be allowed to use a Markov factorisation condition such as Equation (2.45) for classical cyclic causal structures or its non-classical analogue of [115] since this does not holds for cyclic causal structures in general. We summarise these concepts here for completeness. Given a generalised, cyclic causal structure  $\mathcal{G}$ , an intervention on an observed node X of  $\mathcal{G}$  is described as follows.

- An intervention variable  $I_X$  taking values in the set  $\{idle, \{do(x)\}_{x \in X}\}$  is introduced where  $I_X = idle$  corresponds to the original pre-intervention scenario and  $I_X = do(x)$  sets the variable X to a particular value  $x \in X$ , cutting off its dependence on all other parents in  $\mathcal{G}$ .
- Two new causal structures are introduced to keep track of what happens during such an intervention, the *augmented graph*  $\mathcal{G}_{I_X}$  and the *do-graph*  $\mathcal{G}_{do(X)}$ . The former is obtained from the original graph  $\mathcal{G}$  simply by adding  $I_X$  as a node an including an edge  $I_X \longrightarrow X$  (with the rest remaining unchanged). The latter corresponds to the case when a non-trivial intervention is performed i.e., when  $I_X \neq idle$ . Hence,  $\mathcal{G}_{do(X)}$  is obtained from  $\mathcal{G}_{I_X}$  by cutting off all incoming arrows to X, except the one from  $I_X$ , and  $I_X$  only takes values in  $\{do(x)\}_{x \in X}$  in  $\mathcal{G}_{do(X)}$ .
- The causal mechanisms involved in the causal model are updated accordingly i.e., they remain unchanged when  $I_X = idle$  and when  $I_X = do(x)$ , all causal mechanisms, except that of X remain unchanged while X is deterministically set to the value x (irrespective of the states of the subsystems corresponding to incoming edges to X in  $\mathcal{G}$ ). The do-conditional P(Y = y | do(x)) encodes the correlations between a variable Y and the intervened variable

X (or equivalently  $I_X$ , since X and  $I_X$  are perfectly correlated) in the post-intervention graph i.e.,  $P(Y = y|do(x)) \coloneqq P_{\mathcal{G}_{do(X)}}(Y = y|I_X = do(x))$  (c.f. Equation (2.51)), where  $P_{\mathcal{G}_{do(X)}}$  is compatible with the do-graph  $\mathcal{G}_{do(X)}$ .

This method naturally extends to simultaneous interventions on subsets of nodes (as described in Section 2.5.1.2). Only interventions on the observed nodes (which are classical) need to be considered since unobserved nodes are by construction experimentally inaccessible.<sup>8</sup>

Physically, the post-intervention distribution corresponds to additional empirical data that is collected in an experiment, that is in general different from the experiment generating the original, pre-intervention data. For example, when the latter involves passive observation of correlations between the smoking tendencies and presence of cancer in a group of individuals, the former may involve forcing certain individuals to take up smoking and then studying their chances of developing cancer. Such experiments may not necessarily be ethical but are nevertheless a physical possibility. In certain cases, it may be possible to fully deduce the post-intervention statistics counterfactually from the pre-intervention data alone (e.g., using a relation such as (2.52)), and the former experiment need not be actually performed, sparing us some ethical dilemmas.<sup>9</sup> In some cases however, this may not be possible (c.f. Example 2.5.3).

Therefore a complete specification of the post-intervention distribution in terms of the preintervention may not always be possible. However, the compatibility condition of Definition 6.3.2 allows for the derivation certain relationships between these distributions, in particular the three rules of Pearl's do-calculus [159]. These rules have been originally derived for classical causal models satisfying the causal Markov property (2.45) which we do not assume here. However, the derivation of these rules don't necessarily need the Markov property but only the weaker d-separation condition of Definition 6.3.2 assumed in our framework. We present such a derivation here for completeness, even though it bears some resemblance to the original proof of [159]. In the following, we will use  $\mathcal{G}_{\overline{X}}$  to denote the graph obtained by deleting all incoming edges to X in a graph  $\mathcal{G}$ , where X is some subset of the observed nodes. Similarly,  $\mathcal{G}_{\underline{X}}$  denotes the graph obtained by deleting all outgoing edges from a subset X of nodes in a graph  $\mathcal{G}$ .

**Theorem 6.3.1.** Given a causal model over a set S of observed nodes, associated causal graph  $\mathcal{G}$  and a distribution  $P_S$  compatible with  $\mathcal{G}$  according to Definition 6.3.2, the 3 rules of do-calculus [159] hold for interventions on this causal model.

#### • Rule 1: Ignoring observations

$$P(y|do(x), z, w) = P(y|do(x), w) \qquad if (Y \perp^d Z|X, W)_{\mathcal{G}_{\overline{\mathbf{v}}}}$$

$$(6.1)$$

<sup>&</sup>lt;sup>8</sup>Manipulation of non-classical systems such as quantum states, which are by construction unobserved, can be modelled by introducing observed classical variables that specify the choice of preparation/transformation or measurement acting on the system.

 $<sup>^{9}</sup>$ As discussed in Section 2.5, whenever this is possible, the causal model is called *identifiable* and in Example 2.5.3 we visited an example of a simple non-indentifiable scenario involving a classical latent common cause.

• Rule 2: Action/observation exchange

$$P(y|do(x), do(z), w) = P(y|do(x), z, w) \qquad if (Y \perp^d Z|X, W)_{\mathcal{G}_{\overline{X}, \underline{Z}}}$$
(6.2)

• Rule 3: Ignoring actions/interventions

$$P(y|do(x), do(z), w) = P(y|do(x), w) \qquad if (Y \perp^d Z|X, W)_{\mathcal{G}_{\overline{X}, \overline{Z(W)}}}, \tag{6.3}$$

where X, Y, Z and W are disjoint subsets of the observed nodes and Z(W) denotes the set of nodes in Z which are not ancestors of W.

A proof of the theorem is provided in Appendix 6.8.2. Note that considering the underlying causal mechanisms—e.g., the states, transformations, measurements involved in the original causal model should, in principle allow for the complete specification of the post-intervention distribution, even when it cannot be estimated from the observed correlations alone. To the best of our knowledge, this problem has not been studied in non-classical and cyclic causal models, we discuss this point in further detail in Appendix 6.8.1, providing examples of non-classical cyclic causal models where the post-intervention distribution can be calculated from the causal mechanisms.

Using these concepts, we now define the *affects relation* that is central to the results of this chapter. Intuitively, the condition states that when an intervention on a set X of variables produces a change in the distribution over another variable Y (everything else remaining unchanged), then we can say that X affects Y or that X signals to Y, in which case we have operationally identified that X is a cause of Y.

**Definition 6.3.4** (Affects relation). Consider a causal model over a set of random variables S with causal graph  $\mathcal{G}$  and a joint distribution P compatible with it. For  $X, Y \subseteq S$ , if there exists a value x of X such that

$$P(Y|do(x)) \neq P(Y),$$

then we say that X affects Y.

X affects Y implies that there must be a directed path from X to Y in  $\mathcal{G}$  (which is equivalent to X is a cause of Y, c.f. Definition 6.3.1) due to the following reason. If there were no directed paths from X to Y in  $\mathcal{G}$ , there would not be any in  $\mathcal{G}_{do(X)}$  which would make X and Y d-separated from each other in the latter and hence independent (due to Definition 6.3.2). But the affects relation asserts that the variables are not independent in the post-intervention graph which is a contradiction. However, the converse is not true. A causal relationship  $X \rightsquigarrow Y$ does not imply an affects relation X affects Y, in the presence of fine-tuning (as illustrated in the examples of Section 6.3.5). This motivates the following classification of the causal arrows  $\sim \rightarrow$  between observed nodes. The arrows  $\sim \rightarrow$  emanating from/pointing to an unobserved node cannot be operationally probed and hence need not be classified. **Definition 6.3.5** (Solid and dashed arrows). Given a causal graph  $\mathcal{G}$ , if two observed nodes X and Y in  $\mathcal{G}$  sharing a directed edge  $X \longrightarrow Y$  are such that X affects Y, then the causal arrow  $\longrightarrow$  between those nodes is called a *solid arrow*, denoted  $X \longrightarrow Y$ . Further, all arrows  $\longrightarrow$  between observed nodes in  $\mathcal{G}$  that are *not* solid arrows are called *dashed arrows*, denoted  $-\rightarrow$ . In other words,  $X \longrightarrow Y$  for any two RVs X and Y in  $\mathcal{G}$  implies that the X does not affect Y.

**Remark 6.3.1** (Exogenous nodes). Rule 2 of Theorem 6.3.1 has a particular implication that will be useful. It implies that for interventions on exogenous nodes X, the pre and post intervention distributions are identical. To see this, consider a causal graph that contains a directed path  $X \longrightarrow Y$  as a subgraph, and where X is an exogenous node. Then in the graph  $\mathcal{G}_{\underline{X}}$ , X becomes d-separated from all other nodes. Then the d-separation  $(Y \perp^d X)_{\mathcal{G}_{\underline{X}}}$  implies (by Rule 2) that P(y|do(x)) = P(y|x). Then if X and Y are correlated, i.e.,  $P(y|x) \neq P(y)$  this would imply by Definition 6.3.4 that X affects Y. If the path from X to Y was a single link, i.e.  $X \longrightarrow Y$ , then by Definition 6.3.5 this would further imply that this arrow must be a solid one (since X affects Y). Applying this to the graphs  $\mathcal{G}_{I_X}$  and  $\mathcal{G}_{do(X)}$ , where  $I_X$  is exogenous and correlated with X by construction (Equation (2.51)), we can conclude that the arrow from every intervention variable to the corresponding intervened variable must be a solid arrow, i.e.,  $I_X \longrightarrow X$ .

## 6.3.3 Causal loops

We distinguish between two types of causal loops that can arise in our framework. The first will be called an *affects causal loop* which is based on the affects relation of Definition 6.3.4 that corresponds to physically detectable causal relationships. The second will be called a *functional causal loop* and corresponds to a loop at the level of the causal mechanisms (which are functional dependences in the classical case). These function as causal loops at the underlying level even though they may not be detectable at the observable level.

**Definition 6.3.6** (Causal loops). A causal model associated with a causal structure  $\mathcal{G}$  over a set S of RVs is said to have an *affects causal loop* if there exist two distinct random variables X and Y in S such that X affects Y and Y affects X. Hence every affects causal loop corresponds to a directed cycle in  $\mathcal{G}$ . Any directed cycle in  $\mathcal{G}$  that is not an affects causal loop is called a *functional causal loop*.

Note that the *affects causal loop* is defined in terms of single RVs X and Y and not sets of RVs. Replacing X and Y by sets of observed variables  $S_1$  and  $S_2$  in this definition does not imply the existence of a directed cycle. For example, consider a causal structure  $\mathcal{G}$  with 4 nodes A, B, C and D, all of which are observed such that the only edges in  $\mathcal{G}$  are the solid arrows  $A \longrightarrow B$  and  $C \longrightarrow D$ , with A affects B and C affects D. Then, if  $S_1 = \{A, D\}$  and  $S_2 = \{B, C\}$  we have  $S_1$ affects  $S_2$  and  $S_2$  affects  $S_1$  even though  $\mathcal{G}$  is clearly acyclic. On the other hand, it is also possible to have a cyclic causal model that has no affects causal loops but admits cyclic affects relations in terms of subsets of variables (Example 6.3.5.3). Such loops fall into the category of functional causal loops, even though it might be possible to operationally detect their existence through simultaneous interventions on multiple nodes and additional conditions. We will discuss these further detail later in the chapter. In particular, we will see that such causal loops can be embedded in space-time without leading to superluminal signalling (Figure 6.10).

**Remark 6.3.2.** A causal modelling approach allows us to define causal loops without a reference to space-time. This would correspond to a closed time-like curve when the variables X and Y in the loop are each associated with single, but mutually distinct space-time points. Cyclic causal models have also been used to describe variables such as demand and supply that are studied over a period of time. These cyclic scenarios do not correspond to closed time-like curves but to physical feedback mechanisms.

## 6.3.4 Relationships between concepts

Due to the presence of fine-tuning and the introduction of the 2 types of causal arrows (solid and dashed), a number of concepts that are equivalent in faithful causal models are not equivalent for the causal models described in our framework. It will hence be illustrative to present some of relationships between the concepts relating to our causal models, before discussing the relation to space-time structure. These are illustrated in Figure 6.3. The reason for every implication is explained in the figure caption, and for every implication that fails, we provide a counter-example below. There are 11 implications in Figure 6.3 that do not hold. Some of these can be explained by the same counter-example or are immediately evident from the definitions. Therefore we first group these 11 cases based on the corresponding counter-example/argument needed for explaining them, in the end we will only need a few distinct counter-examples to cover all these cases.

- 1. Non-implications 1,2,4,9: These are covered by Example 6.3.1.
- 2. Non-implication 3: This is covered by Example 6.3.2.
- 3. Non-implication 6: This is covered by Example 6.3.3.
- 4. Non-implication 5: It is evident that "X is a cause of Y" does not imply  $X \dashrightarrow Y$ , since it can also be a cause through a solid arrow or an indirect cause.
- 5. Non-implications 7,8: These are just a consequence of the fact that correlation does not imply causation. Correlation between X and Y can arise when they share a common cause, even when neither X nor Y are a cause of each other.
- 6. Non-implication 10: It is evident that independence of X and Y does not imply that there is a dashed arrow between them, they can also be d-separated.
- 7. Non-implication 11: We can consider scenarios where two variables connected by a dashed arrow also share a common cause, and hence the dashed arrow does not by itself rule out the possibility of the variables being correlated.



Figure 6.3: Relationships between concepts relating to causal models: The black arrows denote implications while red (crossed out) arrows denote non-implications. The numbers label the counter-examples corresponding to each non-implication, which are explained in the main text. The equivalence between "X is a causal of Y" and " $\exists$  a directed path from X to Y" is by construction (Definition 6.3.1). This also justifies that the former is implied by  $X \longrightarrow Y$  and  $X \dashrightarrow Y$  (which are just specific instances of directed paths).  $X \longrightarrow Y$  implies X affects Y follows from Definition 6.3.5. That "X affects Y" implies " $\exists$  a directed path from X to Y" is explained in the paragraph following Definition 6.3.4. From this, the implication between "X affects Y" and "X is a cause of Y" also follows by Definition 6.3.1. Finally, an equivalence not shown in the figure is that X affects Y is equivalent to the condition that  $\exists x \in X$  such that  $P(Y|do(x)) \neq P(Y)$  (Definition 6.3.4), which is equivalent to saying that X signals to Y.

**Example 6.3.1** (Jamming). Consider the causal structure of Figure 6.4a where  $B \to A$ ,  $B \to C$  and the RVs A and C share an unobserved common cause  $\Lambda$ . By Definition 6.3.5 of the dashed arrows, we have B does not affect A and B does not affect C. Suppose that B affects the set  $\{A, C\}$ . When A, B and C are binary, a probability distribution compatible with this situation is one where all 3 RVs are uniformly distributed and correlated as  $B = A \oplus C$ , where  $\oplus$  stands for modulo-2 addition. Then, A and C individually carry no information about B but A and C jointly determine the exact value of B. In this case, B is a cause of A and of C but due to fine-tuning, this cannot be detected when considering B and A or B and C alone, but only when all 3 RVs are jointly accessed. The common cause is crucial to this example as explained in Figure 6.4a, and the causal structure compatible with the distribution and affects relations of this example is not unique. An alternative causal structure would be one where one of the dashed arrows  $B \to A$  or  $B \to C$  is dropped, as we will see in Section 6.8.

This example by itself makes no reference to space-time or the tripartite Bell scenario. However, for the particular embedding of the variables A, B and C in space-time where they are pairwise space-like separated and taken to correspond to the output of Alice, input of Bob and output of Charlie respectively, this becomes a special case of the tripartite jamming scenario of [105, 116] (Figure 6.2).<sup>10</sup> In the rest of the chapter, such examples, where an RV has dashed arrows to a set of RVs will be referred to as instances of "jamming" in accordance with the terminology of [105], irrespective of the space-time configuration. We will further discuss the relation of such causal models to space-time structure later in the chapter, and will also revisit Example 6.3.1 several times in this process.

**Example 6.3.2.** Consider a causal model over observed variables  $\{A, B, C, D\}$  associated with the causal graph  $\mathcal{G}$  given in Figure 6.4b. Here, there are no pairs of variables sharing an edge such that one of them affects the other. A correlation compatible with this graph is obtained by taking  $B = A \oplus C = D$  where all variables are binary and uniformly distributed. Here, B affects D even though there are no solid arrow paths from B to D.

**Example 6.3.3.** Consider the causal structure of Figure 6.5. Let the three variables S, E ad H be binary and correlated as  $H = S \oplus E$  and S = E. These relations imply that H = 0 deterministically while S = E. However, when we intervene on E, we can choose its value independently of S and whenever we choose  $E \neq S$ , we will see that H = 1 occurs with non-zero probability in other words there exists a value e such that  $P(H = 1|do(e)) \neq P(H = 1) = 0$  i.e., E affects H even though  $P_{EH} = P_E P_H$  in the original causal model. On the other hand, S does not affect H since S the exogeneity of S implies that P(H|do(S)) = P(H|S) always holds (Rule 1 of Theorem 6.3.1), and we have P(H|S) = P(H) by construction.<sup>11</sup>

<sup>&</sup>lt;sup>10</sup>Barring the slight change of notation: In Figure 6.2 A and C correspond to the inputs of Alice and Charlie while X and Z correspond to the outputs that are jammed by B. We don't make a distinction between inputs and outputs in general since we will also consider situations where the jamming variable is not exogenous for example.

<sup>&</sup>lt;sup>11</sup>This is essentially the example of Figure 2.7 (discussed in Section 2.5), where we have distinguished between solid and dashed arrows in accordance with Definition 6.3.5.


Figure 6.4: Some fine-tuned causal structures: (a) The jamming causal structure of Example 6.3.1. Note that the common cause  $\Lambda$  is essential to this example, because without  $\Lambda$ , A and C would be d-separated given B which would imply the conditional independence  $P_{AC|B} = P_{A|B}P_{C|B}$ . The dashed arrows would imply the independence of A and B as well as C and B and hence the observed distribution would factorise as  $P_{ABC} = P_A P_B P_C$ . Then no pairs of disjoint subsets of  $\{A, B, C\}$  would affect each other contrary to the original example. (b) Causal structure for Example 6.3.2 where B affects D even though there is no solid arrow path from B to D.



Figure 6.5: Affects relation does not imply correlation: This is a causal structure for Example 6.3.3 which demonstrates a scenario where E affects H even though  $P_{EH} = P_E P_H$ , i.e., solid arrows can also be fine-tuned and the ability to detect causation through an active intervention) does not imply that we will see correlation upon passive observation.

## 6.3.5 Identifying conditional independences and affects relations: Examples

Here we provide examples that better illustrate some of the definitions and rules of the framework laid out so far. In particular, how one can deduce the conditional independences and affects relations in a given causal model. We briefly summarise these below.

• Conditional independences: Given a causal graph  $\mathcal{G}$  with the set of observed nodes S, the conditional independences satisfied by the joint distribution  $P_S$  can be listed by a) using Definition 6.3.2 to list all the conditional independences implied by d-separation relations in  $\mathcal{G}$  and b) using Definition 6.3.5 of dashed arrows, and Lemma 6.3.1 to list further independences not implied by d-separation in  $\mathcal{G}$ . There may still be more conditional

independences in  $P_S$  that cannot be listed this way, for instance, in Example 6.3.1, we also have  $A \perp C$  but this neither follows from d-separation or the dashed arrow structure. Since we allow for fine-tuning, there could be arbitrarily many independences in P, but those mentioned above are sufficient for compatibility with the causal model.

• Affects relations: We have seen that a non-trivial intervention on a subset S of the observed nodes corresponds to a new causal graph  $\mathcal{G}_{do(S)}$  in which all the nodes contained in S are exogenous. Then the condition that S affects another subset R of the observed nodes (i.e., there exists a values s of S such that  $P(R|do(s)) \neq P(S)$ ), corresponds to the condition that S and R are not independent in the post-intervention causal structure  $\mathcal{G}_{do(S)}$ . Hence one can apply the procedure of identifying independences explained in the previous point to the post-intervention causal structure, to identify affects relations. Again, due to fine-tuning, this identification may not be exhaustive.

#### 6.3.5.1 Jamming (Figure 6.4a)

In the jamming causal structure  $\mathcal{G}^{jam}$  of Figure 6.4a and Example 6.3.1, Definition 6.3.2 does not impose any conditional independences on the observed distribution  $P_{ABC}$  since  $\Lambda$  is unobserved.<sup>12</sup> However, from Definition 6.3.5 of dashed arrows we have the independences  $A \perp B$ and  $C \perp B$ . Further, B affects  $\{A, C\}$  implies that  $B \not\downarrow \{A, C\}$  since B is exogenous here. Consider an intervention on A, the post-intervention causal structure  $\mathcal{G}^{jam}_{do(A)}$  only has the edges  $B \dashrightarrow C$  and  $\Lambda \leadsto C$  (along with  $I_A \longrightarrow A$  of course). The independences in  $\mathcal{G}^{jam}_{do(A)}$  due to dseparation are  $A \perp C$ ,  $A \perp B$ ,  $A \perp \{B, C\}$  and  $A \perp C \mid B$  and due to the dashed arrow, we have  $B \perp C$ . The independence  $A \perp C$  in  $\mathcal{G}^{jam}_{do(A)}$  implies that A does not affect C. Similarly, we can derive C does not affect A, A does not affect  $\{B, C\}$ , C does not affect  $\{A, B\}$  and  $\{A, C\}$  does not affect B. Further, using Lemma 6.3.1, we also have  $\{A, B\}$  does not affect C and  $\{B, C\}$ does not affect A. This is because in the causal structure  $\mathcal{G}^{jam}_{do(\{A,B\})}$ , A is d-separated from Band C, while B and C are independent of each other due to the dashed arrow connecting them. Using the lemma, this gives  $\{A, B\} \perp C$  in  $\mathcal{G}^{jam}_{do(\{A,B\})}$ , which in turn implies that  $\{A, B\}$  does not affect C.

#### 6.3.5.2 Fine-tuned collider (Figure 6.6a)

In the causal structure of Figure 6.6a, the independence  $A \perp C$  follows from Definition 6.3.2, while  $A \perp B$  and  $C \perp B$  follow from the dashed arrow structure. These are the same independences as the case in the previous example of jamming with unobserved  $\Lambda$  (where  $A \perp C$  was an additional independence in the jamming example but follows from d-separation in this case). Thus the distribution  $P_{ABC}$  from Example 6.3.1 is compatible with both the jamming (Figure 6.4a) as well the fine-tuned collider (Figure 6.6a) causal structures.<sup>13</sup> However inter-

<sup>&</sup>lt;sup>12</sup>If  $\Lambda$  in Figure 6.4a were observed, A and C would be d-separated given  $\{B, \Lambda\}$  and we would have the conditional independence  $P_{AC|B\Lambda} = P_{A|B\Lambda}P_{C|B\Lambda}$ .

<sup>&</sup>lt;sup>13</sup>Note that this is essentially the one-time pad example from earlier.



Figure 6.6: Some fine-tuned and/or cyclic causal structures: (a) A fine-tuned collider (b) A functional causal loop (c) An affects causal loop

ventions on the two causal structures yield different results. We have  $\{A, C\}$  affects B for the fine-tuned collider but not for the jamming case. We also have A affects  $\{B, C\}$  and C affects  $\{A, B\}$  for the fine-tuned collider even though A and C do not individually affect B due to the dashed arrow structure. This follows from the exogeneity of A and C and the joint correlations  $A = B \oplus C$ . Further,  $\{A, B\}$  does not affect C,  $\{B, C\}$  does not affect A and B does not affect  $\{A, C\}$  (in contrast with the jamming case where B affects  $\{A, C\}$ ).

#### 6.3.5.3 A functional causal loop (Figure 6.6b)

The cyclic causal structure of Figure 6.6b is a combination of jamming and the fine-tuned collider. The distribution  $P_{ABC}$  of Example 6.3.1 with  $B = A \oplus C$  is also compatible with this causal structure, but the affects relations differ. An intervention on A and C would cut off the incoming dashed arrows from B to A and C as well as those from  $\Lambda$ , and we would recover the fine-tuned collider of Figure 6.6a, giving  $\{A, C\}$  affects B. An intervention on B in the original causal structure would eliminate the collider and yield the jamming causal structure of Figure 6.4a, and we have B affects  $\{A, C\}$ . We also have A affects  $\{B, C\}$  and C affects  $\{A, B\}$  as in the collider. And as in the jamming case, we also get  $\{A, B\}$  does not affect C and  $\{B, C\}$  does not affect A due to Lemma 6.3.1. Further we have no pair of variables in  $\{A, B, C\}$  that affect each other and hence this does not correspond to an affects causal loop (Definition 6.3.6). Even though we have directed cycles and the cyclic affects relations B affects  $\{A, C\}$  and  $\{A, C\}$  affects B, the variables A, B and C can be embedded in Minkowski space-time such that this causal model does not lead to signalling superluminally, as we will see later (Figure 6.10).

#### 6.3.5.4 An affects causal loop (Figure 6.6c)

Consider the causal structure of Figure 6.6c over the variables A, B, C and D. Applying Definition 6.3.5 of solid arrows, to this causal structure we have A affects B, B affects C, C affects D and D affects A, which forms an affects causal loop. The conditional independences that follow from d-separation are  $A \perp C | \{B, D\}$  and  $B \perp D | \{A, C\}$  and any joint distribution

satisfying these would be compatible with the causal structure. To further illustrate the kind of causal loops allowed in this framework, consider the pairwise correlations A = B, B = C, C = D and  $D \neq A$ . Since this system of equations has no solutions, there exists no joint distribution  $P_{ABCD}$  from which the pairwise marginals producing these correlations can be obtained. Such examples correspond to grandfather type paradoxes and cannot be modelled in frameworks that demand the existence of a valid joint probability distribution over all variables involved in a causal loop. On the other hand, examples of solid arrow directed cycles where the functional dependences of the loop variables admit solutions such as A = B = C = D (with any probability) or the examples considered in [160] for other cyclic causal loops that do not involve any solid arrows, for example through a concatenation of structures such as that of Figure 6.4b. We discuss causal loops in more detail in Appendix 6.8.1, also in the case of quantum causal structures.

# 6.4 The framework, Part 2: Space-time

## 6.4.1 Space-time structure

We model space-time simply by a partially ordered set  $\mathcal{T}$  without assuming any further structure/symmetries. A particular example of  $\mathcal{T}$  is Minkowski space-time, where the partial order corresponds to the light-cone structure and the elements of  $\mathcal{T}$  can be seen space-time coordinates in some frame of reference. Our results will only depend on the order relations of  $\mathcal{T}$  and not on the representation of its particular elements. To make operational statements about  $\mathcal{T}$ , we must embed physical systems into it. In our case, we can only do so for the observed systems in the causal model which are random variables. We embed them in this space-time by assigning an element of  $\mathcal{T}$  to specify the space-time location of each variable, and refer to such variables as *ordered random variables* or ORVs. Here, the order of an ORV corresponds to that of the space-time  $\mathcal{T}$  (and not of the causal model).<sup>14</sup>

**Definition 6.4.1** (Ordered random variable (ORV)). Each ORV  $\mathcal{X}$  is defined by a pair  $\mathcal{X} := (R(\mathcal{X}), O(\mathcal{X}))$  where  $R(\mathcal{X})$  (equivalently denoted by the corresponding, non-calligraphic letter X) is a random variable and  $O(\mathcal{X}) \in \mathcal{T}$  specifies the location of  $\mathcal{X}$  with respect to a partially ordered set  $\mathcal{T}$ .

Notation: We use  $\prec, \succ$  and  $\not\prec \not\prec$  to denote the order relations for a given partially ordered set  $\mathcal{T}$ , where for  $\alpha, \beta \in \mathcal{T}, \alpha \not\prec \not\prec \beta$  corresponds to  $\alpha$  and  $\beta$  being unordered with respect to  $\mathcal{T}$ . This is not to be confused with  $\alpha = \beta$  which corresponds to the two elements being equal. These relations carry forth in an obvious way to ORVs and we say for example that 2 ORVs  $\mathcal{X}$  and  $\mathcal{Y}$  are ordered as  $\mathcal{X} \prec \mathcal{Y}$  iff  $O(\mathcal{X}) \prec O(\mathcal{Y})$ . Note however that  $\mathcal{X} = \mathcal{Y}$  is equivalent to X = Y and  $O(\mathcal{X}) = O(\mathcal{Y})$ .

<sup>&</sup>lt;sup>14</sup>This can be seen as an abstract version of space-time random variables.

Definition 6.4.2. The *future* of an ORV is the set

$$\mathcal{F}(\mathcal{X}) \coloneqq \{ \alpha \in \mathcal{T} : \alpha \succ O(\mathcal{X}) \}.$$

Then, we say that an ORV  $\mathcal{Y}$  lies in the future of an ORV  $\mathcal{X}$  iff  $O(\mathcal{Y}) \in \mathcal{F}(\mathcal{X})$ . In a slight abuse of notation, we will simply write this as  $\mathcal{Y} \in \mathcal{F}(\mathcal{X})$ , which is equivalent to  $\mathcal{X} \prec \mathcal{Y}$ .

Definition 6.4.3. The *inclusive future* of an ORV is the set

$$\overline{\mathcal{F}}(\mathcal{X}) \coloneqq \{ \alpha \in \mathcal{T} : \alpha \ge O(\mathcal{X}) \}.$$

Note that  $\mathcal{X} \in \overline{\mathcal{F}}(\mathcal{X})$  but  $\mathcal{X} \notin \mathcal{F}(\mathcal{X})$ , hence the name "inclusive" future. Further, any probabilities written in terms of ORVs should be understood as being probability distributions over the corresponding random variables.

## 6.4.2 Compatibility of a causal model with the embedding spacetime

We have encountered two types of order relations, one which is encoded in the arrows  $\rightsquigarrow$  of the causal structure and the other specified by the order relation  $\prec$  of the space-time  $\mathcal{T}$ . The former, which can include directed cycles corresponds to a pre-order while the latter is by construction a partial order. We now describe the compatibility condition will tell us when the order provided by a causal model is compatible with the order provided by the space-time  $\mathcal{T}$  at the operational level, here the classification of the arrows  $\rightsquigarrow$  into solid and dashed arrows will play a role. In an operational theory, we must be able to consider scenarios where a random variable may be copied, distributed or acted upon. For ordered random variables, we need to consider a set of locations in the embedding partial order at which copies of RVs can be accessed for further information processing. To this effect, we first propose the following definitions before defining compatibility.

**Definition 6.4.4** (Copy of a RV). A random variable X' is called a copy of a random variable X if X = X' and X affects X'. Hence every RV X is a copy of itself, we call this the *trivial* copy of X.

For ordered random variables  $\mathcal{X}$  and  $\mathcal{X}'$ , we will simply say  $\mathcal{X}'$  is a copy of  $\mathcal{X}$  whenever the corresponding RV  $X' \coloneqq R(\mathcal{X}')$  is a copy of the RV  $X \coloneqq R(\mathcal{X})$ .

**Definition 6.4.5** (Accessible region of an ORV). With each ordered random variable  $\mathcal{X} := (R(\mathcal{X}), O(\mathcal{X}))$  ordered with respect to some partially ordered set  $\mathcal{T}$ , we can associate a subset  $\mathcal{R}_{\mathcal{X}} \subseteq \mathcal{T}$  called its *accessible region*, such that any copy  $\mathcal{X}'$  of  $\mathcal{X}$  must belong to this set i.e.,  $O(\mathcal{X}') \in \mathcal{R}_{\mathcal{X}}$  for all copies  $\mathcal{X}'$  of  $\mathcal{X}$ , and  $\mathcal{R}_{\mathcal{X}}$  is the smallest subset of  $\mathcal{T}$  that has this property. We say that  $\mathcal{X}$  is accessible in  $R(\mathcal{X})$ .

The accessible region has the property that  $\mathcal{R}(\mathcal{X}') \subseteq \mathcal{R}(\mathcal{X})$  for all copies  $\mathcal{X}'$  of  $\mathcal{X}$  (since the copy of a copy is a copy), and  $\mathcal{X} \in \mathcal{R}_{\mathcal{X}}$  for all ORVs  $\mathcal{X}$ .

**Remark 6.4.1.** Definition 6.4.5 allows us to naturally associate accessible regions to sets of ORVs in terms of the accessible regions of their individual members. The accessible region of a set  $S = \{X_1, ..., X_k\}$  of ORVs is the smallest subset of  $\mathcal{T}$  within which all the ORVs in the set can be jointly accessed i.e.,  $\mathcal{R}_S = \bigcap_{X_i \in S} \mathcal{R}(X_i)$ . Each set S of ORVs can be thought of as being comprised of one copy of each of its members, and its accessible region is the region of space-time where a copy of each ORV in the set can be accessed.

**Definition 6.4.6** (Embedding). An embedding of a set of RVs S in a partially ordered set  $\mathcal{T}$  produces a corresponding set of ORVs S by assigning a location  $O(\mathcal{X}) \in \mathcal{T}$ , and an accessible region  $\mathcal{R}_{\mathcal{X}} \subset \mathcal{T}$  to each RV X, which defines the associated ORV  $\mathcal{X} = (X, O(\mathcal{X}))$ .

We now define what it means for a causal model to be compatible with the embedding partial order  $\mathcal{T}$  which captures the idea of no superluminal signalling with respect to the space-time  $\mathcal{T}$ . Intuitively, the definition corresponds to the condition that "it is possible to signal *everywhere* in the future and nowhere else". This is not the same as "it is possible to signal only to the future (but not necessarily to every point in the future)". We will see that this subtle difference plays a role in our results of the following sections (c.f. Remark 6.5.1).

**Definition 6.4.7** (Compatibility of a causal model with the embedding partial order (compat)). Let S be a set of ORVs ordered with respect to the partially ordered set  $\mathcal{T}$ . Then a causal model over S is said to be *compatible* with the partial order  $\mathcal{T}$  if the following conditions are satisfied:

- 1. For all  $\mathcal{X} \in \mathcal{S}$ ,  $\mathcal{R}_{\mathcal{X}} = \overline{\mathcal{F}}(\mathcal{X})$ , and
- 2. For all subsets of ORVs,  $S_1, S_2 \subseteq S$ , such that no proper subset of  $S_1$  affects  $S_2, S_1$  affects  $S_2 \Rightarrow \mathcal{R}_{S_2} \subseteq \mathcal{R}_{S_1}$ .

**Remark 6.4.2.** In the second statement above, the condition that no proper subset of  $S_1$  affects  $S_2$  is required because whenever a set S of ORVs affects another set  $S_2$ , any  $S_1 \supset S$  will also trivially affect  $S_2$ . In such situations, one is only interested in the non-trivial affects relation S affects  $S_2$  (given that no further proper subsets of S affect  $S_2$ ) and not  $S_1$  affects  $S_2$ , since  $S_1 \supset S$  could be arbitrarily large in this case.<sup>15</sup>

# 6.5 Necessary and sufficient conditions

In this section, we derive necessary and sufficient conditions for a causal model to be compatible with a partial order  $\mathcal{T}$  (Section 6.5.1) and for it to have no affects causal loops (Section 6.5.2). In [116], a set of necessary and sufficient conditions for no causal loops in the bipartite and

<sup>&</sup>lt;sup>15</sup>In the case of single variables,  $\mathcal{X}$  affects  $\mathcal{Y}$  implies that  $\mathcal{Y}$  must be in the future of  $\mathcal{X}$  in order to be compatible with the space-time. We don't require  $\mathcal{Y}$  to be in the future of every superset of  $\mathcal{X}$ , which would trivially affect it.

tripartite Bell scenarios have been proposed. The results of this section identify implicit assumptions in these claims [116] and provide necessary and sufficient conditions for compatibility with the space-time and for no affects causal loops for arbitrary causal structures. The other type of causal loops, functional causal loops cannot in general be detected operationally and therefore cannot be ruled out. We then discuss the relationships between the various conditions derived in our results in Section 6.5.3 and consider [116] in further detail in Section 6.6.2.

## 6.5.1 Conditions for compatibility with the embedding space-time

**Theorem 6.5.1.** [Necessary and sufficient conditions for compatibility with  $\mathcal{T}$ ] Given a causal model over a set of ORVs S, the following condition (cond) is necessary for the causal model to be compatible with the embedding partial order  $\mathcal{T}$  according to compat (Definition 6.4.7).

$$\forall subsets \ \mathcal{S}_1, \mathcal{S}_2 \subseteq \mathcal{S}, \ such \ that \ no \ proper \ subset \ of \ \mathcal{S}_1 \ affects \ \mathcal{S}_2, \\ \bigcap_{s_2 \in \mathcal{S}_2} \overline{\mathcal{F}}(s_2) \notin \bigcap_{s_1 \in \mathcal{S}_1} \overline{\mathcal{F}}(s_1) \quad \Rightarrow \quad \mathcal{S}_1 \ does \ not \ affect \ \mathcal{S}_2.$$

$$(6.4)$$

While **cond** alone is not sufficient for **compat**, along with the following additional requirement it provides a sufficient condition for **compat** 

$$\forall \mathcal{X} \in \mathcal{S}, \quad \overline{\mathcal{F}}(\mathcal{X}) \subseteq \mathcal{R}_{\mathcal{X}} \tag{6.5}$$

A proof of this theorem can be found in Appendix 6.8.2.

#### 6.5.2 Conditions for no affects causal loops

**Theorem 6.5.2.** [Necessary and sufficient condition for no affects causal loops] A necessary condition for a causal model over a set S of RVs to have no affects causal loops is that there exists an embedding of S in Minkowski space-time  $\mathcal{T}$  such that the corresponding ORVs S satisfies the condition **cond** of Equation (6.4). **cond** along with the additional assumption that any 2 distinct  $ORVs \mathcal{X}$  and  $\mathcal{Y}$  such that one affects the other, cannot share the same location in  $\mathcal{T}$ , are sufficient for having no affects causal loops in the causal model.

A proof of the above theorem can be found in Appendix 6.8.2. Note that a subset of the above conditions, would already be sufficient for no affects causal loops, since these causal loops are defined in terms of a chain of affects relations involving single variables (and not sets of variables). Suppose that for every pair of affects relations in a causal model, X affects Y where X and Y are single variables, we can ensure that the corresponding ordered random variables satisfy  $\mathcal{Y} \in \mathcal{F}(\mathcal{X})$ , this would already be sufficient to conclude that the causal model has no affects loops. However this is not sufficient for compatibility with the space-time since we may have X affects  $\{W, Z\}$  for some variables W and Z in the model that are not individually

affected by X such that  $\overline{\mathcal{F}}(\mathcal{W}) \cap \overline{\mathcal{F}}(\mathcal{Z}) \notin \overline{\mathcal{F}}(\mathcal{X})$ . Based on this, a third class of causal loops can be defined, those that can be operationally detected only when we consider affects relations of the form  $S_1$  affects  $S_2$  where at least one of  $S_1$  and  $S_2$  is a set of cardinality greater than one. An example would be the cyclic causal structure of Figure 6.6b where  $\{A, C\}$  affects B and Baffects  $\{A, C\}$ . With our current definition 6.3.6, this gets classified as a functional causal loop and not an affects loop. Prior to publication of this work, we will consider whether defining this third category of causal loops would be useful for deriving additional results. This does not, however, affect the results presented here.

Keeping this in mind and applying Theorem 6.5.2 to the bipartite and tripartite Bell experiments respectively, where the parties are space-like separated, we have Corollaries 6.5.1 and 6.5.2.

**Corollary 6.5.1** (Bipartite Bell scenario). Consider a causal model over the observed variables  $\{A, B, X, Y\}$  where A and B are exogenous, and X and Y share an unobserved common cause  $\Lambda$ . Suppose that  $\mathcal{T}$  corresponds to Minkowski space-time. Let the observed ORVs defined with respect to  $\mathcal{T}$  be such that  $A \nleftrightarrow \mathcal{B}, A \nleftrightarrow \mathcal{Y}, \mathcal{B} \nleftrightarrow \mathcal{X}, \mathcal{X} \nleftrightarrow \mathcal{Y}, A \prec \mathcal{X}$  and  $\mathcal{B} \prec \mathcal{Y}$  and the accessible region of every ORV coincides with its inclusive future. In this scenario, a set of sufficient conditions for the causal model to have no affects causal loops are

- 1.  $\mathcal{A}$  does not affect any subset of ORVs not containing  $\mathcal{X}$ .
- 2.  $\mathcal{B}$  does not affect any subset of ORVs not containing  $\mathcal{Y}$ .
- 3.  $\mathcal{X}$  and  $\mathcal{Y}$  do not affect any other ORVs.

1. and 2. are equivalent to the bipartite no signalling conditions (NS2) given the exogeneity of A and B.

$$P_{X|A}(x|a) \coloneqq \sum_{y} P_{X,Y|A,B}(x,y|a,b) = \sum_{y} P_{X,Y|A,B}(x,y|a,b') \quad \forall x, a, b, b',$$

$$P_{Y|B}(y|b) \coloneqq \sum_{x} P_{X,Y|A,B}(x,y|a,b) = \sum_{x} P_{X,Y|A,B}(x,y|a',b) \quad \forall y, a, a', b$$
(6.6)

Hence NS2 along with 3. are sufficient for no affects causal loops in this scenario.

**Corollary 6.5.2** (Tripartite Bell scenario). Consider a causal model over the observed variables  $\{A, B, C, X, Y, Z\}$  where A, B and C are exogenous, and X, Y and Z share an unobserved common cause  $\Lambda$ . Suppose that  $\mathcal{T}$  corresponds to Minkowski space-time. Let the observed ORVs defined with respect to  $\mathcal{T}$  be such that  $A < \mathcal{X}$ ,  $\mathcal{B} < \mathcal{Y}$ ,  $\mathcal{C} < \mathcal{Z}$ ,  $\overline{\mathcal{F}}(\mathcal{X}) \cap \overline{\mathcal{F}}(\mathcal{Z}) \subseteq \overline{\mathcal{F}}(\mathcal{B})$ , all other pairs of observed ORVs are unordered with respect to  $\mathcal{T}$  and the accessible region of every ORV coincides with its inclusive future. In this scenario, a set of sufficient conditions for the causal model to have no affects causal loops are

- 1. A does not affect any subset of ORVs not containing  $\mathcal{X}$ .
- 2.  $\mathcal{B}$  does not affect any subset of ORVs containing neither  $\mathcal{Y}$ , nor the set  $\{\mathcal{X}, \mathcal{Z}\}$ .

- 3. C does not affect any subset of ORVs not containing Z.
- 4.  $\mathcal{X}$ ,  $\mathcal{Y}$  and  $\mathcal{Z}$  do not affect any other ORVs.

Given the exogeneity of A, B and C, 1., 2. and 3. are equivalent to the relaxed tripartite no-signaling conditions (NS3') of [116].

$$P_{X,Y|A,B}(x,y|a,b) \coloneqq \sum_{z} P_{X,Y,Z|A,B,C}(x,y,z|a,b,c) = \sum_{z} P_{X,Y,Z|A,B,C}(x,y,z|a,b,c') \quad \forall x, y, a, b, c, c'$$

$$P_{Y,Z|B,C}(y,z|b,c) \coloneqq \sum_{z} P_{X,Y,Z|A,B,C}(x,y,z|a,b,c) = \sum_{z} P_{X,Y,Z|A,B,C}(x,y,z|a',b,c) \quad \forall y, z, a, a', b, c$$

$$P_{X|A}(x|a) \coloneqq \sum_{y,z}^{x} P_{X,Y,Z|A,B,C}(x,y,z|a,b,c) = \sum_{y,z}^{x} P_{X,Y,Z|A,B,C}(x,y,z|a,b',c') \quad \forall x,a,b,b',c,c'$$

$$P_{Z|C}(z|c) \coloneqq \sum_{x,y} P_{X,Y,Z|A,B,C}(x,y,z|a,b,c) = \sum_{x,y} P_{X,Y,Z|A,B,C}(x,y,z|a',b',c) \quad \forall z,a,a',b,b',c$$
(6.7)

Hence NS3' along with 4. are sufficient for no affects causal loops in this scenario.

An important point to note here is that the conditions are only sufficient and not necessary. This is because the Bell experiments considered in [116] correspond to particular embeddings of the causal model where the measurement events are space-like separated, while our Theorem 6.5.2only requires the existence of a suitable embedding which need not be this particular one. More explicitly, a violation of the no-signaling condition (6.6) in a Bell experiment is not necessarily a problem when the parties are not space-like separated, and if they are, they would need to be able to signal superluminally in more than one frame to create a causal loop from this ability to signal superluminally. But superluminal signalling in one frame need not imply the same for other frames. Hence the no-signaling conditions (6.6) and (6.7) are are not necessary for ruling out causal loops in the bipartite and tripartite Bell scenarios, contrary to the claim of [116]. We further discuss [116] in the context of our framework in Section 6.6.2 providing explicit counter examples to these claims. Furthermore, note that these corollaries only rule out affects causal loops and not all causal loops, they also do not guarantee compatibility with the space-time. For example, a causal model whereby  $\{A, X\}$  affects Y and  $\{B, Y\}$  affects X, A or X affecting Y and B or Y affecting X, would create a directed cycle or a functional causal loop between X and Y and also lead to signalling outside the future for the space-time configuration of Corollary 6.5.1. This is not ruled out by the conditions of the above corollaries, but can nevertheless be obtained from Theorems 6.5.1 and 6.5.2. We now present another noteworthy corollary that can be derived from Theorems 6.5.1 and 6.5.2, a proof of which can be found in Appendix 6.8.2.

**Corollary 6.5.3.** [No affects causal loops and compatibility with space-time] A necessary condition for a causal model over a set S of RVs to have no affects causal loops is that  $\exists$  an embedding of S in Minkowski space-time  $\mathcal{T}$  such that the causal model is compatible (Definition 6.4.7) with  $\mathcal{T}$ . This condition along with the additional assumption that any 2 distinct ORVs  $\mathcal{X}$  and  $\mathcal{Y}$  such that one affects the other, cannot share the same location in  $\mathcal{T}$ , are sufficient for having no affects causal loops in the causal model.

Remark 6.5.1. We have seen two distinct necessary and sufficient conditions for no causal loops: a) there exists an embedding in  $\mathcal{T}$  such that **cond** (Equation (6.4)) holds (Theorem 6.5.2), and b) there exists an embedding in  $\mathcal{T}$  such that the causal model is compatible with  $\mathcal{T}$  (Corollary 6.5.3). These two conditions namely cond and compatibility with  $\mathcal{T}$  are not equivalent, which follows from Theorem 6.5.1 due to the additional assumption required in one direction of the proof. The difference has to do with whether or not the accessible region coincides with the future for each ORV or is only a subset of it. The embedding for which a) holds may in general be different from the embedding for which b) holds, since an embedding also involves a specification of the accessible region (Definition 6.4.6). In the latter case, the accessible region must coincide with the future by definition of compatibility (Definition 6.4.7), while in the former case, it need not. In the case that only a) holds, we can have a jamming scenario (Figure 6.4a) where the jammed variables can be jointly accessed only within a strict subset  $\mathcal{R}_{\mathcal{AC}}$  of their joint future  $\overline{\mathcal{F}}(\mathcal{A}) \cap \overline{\mathcal{F}}(\mathcal{C})$ . Then it is enough if this subset is contained in the future of the jamming variable B and not the entire joint future. Note that modifying Definition 6.4.7 of compatibility to require  $\mathcal{R}_{\mathcal{X}} \subseteq \overline{\mathcal{F}}(\mathcal{X})$  instead of  $\mathcal{R}_{\mathcal{X}} = \overline{\mathcal{F}}(\mathcal{X})$ , would not alter the fact that Theorem 6.5.1 would require an additional assumption along with cond for one of the directions, it would only alter the direction for which this is required.

## 6.5.3 Relationships between the conditions

The results of Sections 6.5.1 and 6.5.2, along with related (non-)implications are summarised in Figure 6.7. All the implications are covered by the results of the previous sections as explained in the Figure. Here, we describe counter examples to establish the 7 non-implications of the Figure, which can be grouped as follows.

- 1. Non-implications I, III: While no affects causal loops implies cond by Theorem 6.5.1, it does not imply that  $O(\mathcal{X}) \neq O(\mathcal{Y})$  whenever  $\mathcal{X}$  affects  $\mathcal{Y}$ . This is because we can have an acyclic causal model with non-trivial affects relations that is embedded trivially in space-time, i.e., where all the variables are embedded at the same space-time location. This explains non-implication I. Further, this trivial embedding of any causal model is clearly compatible with the space-time, which explains non-implication III.
- 2. Non-implications IV, V: These are immediately established by embedding the 2-cycle affects loop (i.e., the two variable causal model with X affects Y affects X) in space-time such that X and Y are assigned the same space-time location  $O(\mathcal{X}) = O(\mathcal{Y})$ . Such causal loops between variables located at the same space-time point do not violate the compatibility condition 6.4.7 or lead to superluminal signalling, but they are rather uninteresting.
- 3. Non-implications II, VI: These are covered by the example provided in Remark 6.5.1: compatibility requires that the accessible region coincides with the future while cond only implies that the accessible region is a subset of the future.
- 4. Non-implications VII, VIII: This is because the accessible region (Definition 6.4.5) of an ordered random variable  $\mathcal{X}$  is defined only in terms of copies of  $\mathcal{X}$ . Therefore we



Figure 6.7: Illustrative summary of results pertaining to embeddings of causal models in a space-time.

can have an embedding for which  $\mathcal{X}$  affects another ORV  $\mathcal{Y}$  which is not a copy of itself and does not lie in its future, even if the accessible region of each variable is contained in the variable's future. Such an embedding would violate **cond**.

# 6.6 Theories with superluminal influence without superluminal signalling

## 6.6.1 Jamming and superluminal signalling

The jamming scenario analysed in the literature [105, 116] has been considered only in the context of multipartite Bell scenarios, in particular the tripartite case, as discussed in Section 6.2. There, the inputs of all the parties are considered to be freely chosen and the parties share a joint system  $\Lambda$  that provides the initial correlations for the Bell scenario. Additionally, jamming allows the input of one party to jointly signal to the outputs of a set of other parties, without signalling to them individually. In the causal modelling paradigm, we have formalised freely chosen variables as parentless nodes in the causal structure, the notion of signalling through the affects relation, and have modelled  $\Lambda$  as a common cause between outputs of all the parties. In the following, we will consider a particular class of jamming scenarios arising in the tripartite Bell experiment of [105, 116], those where Bob's input jams the outputs of Alice and Charlie, while the other inputs and outputs do not feature in the correlations. In this section, we simply denote these variables as B, A and C respectively.<sup>16</sup> In Figure 6.4a, we have explained that a common cause  $\Lambda$  between A and C is necessary for producing jamming correlations in a causal structure where the input B is parentless and the outputs A and C have no outgoing arrows. In the remainder of this section, we explore the possibility of superluminal signalling using such a causal model.<sup>17</sup>

It would be illustrative to sketch a simple protocol that can be used to obtain the jamming correlations of Example 6.3.1, that can lead to superluminal signalling in the space-time embedding considered in [105, 116]. This protocol captures the main intuition behind Theorem 6.6.1 that will follow, and is illustrated in Figure 6.8. It also illustrates that Figure 6.4a is not the unique causal structure that is compatible with the correlations and affects relations of Example 6.3.1.

A protocol for jamming: In Example 6.3.1, suppose that  $\Lambda$  is classical and all four variables, including  $\Lambda$  are binary and uniformly distributed. The output A is generated from  $\Lambda$  by flipping its value whenever B = 1 and outputting  $A = \Lambda$  otherwise i.e.,  $A = \Lambda \oplus B$ . C is set to always be equal to  $\Lambda$  irrespective of B. This immediately gives the required correlation  $B = A \oplus C$  and reproduces the affects relations of the example. Even though B does not affect A, it was physically sent to A and even though B jointly affects A and C, it was not sent to C. This corresponds to the causal structure of Figure 6.4a but with the arrow  $B \dashrightarrow C$  removed, as shown in Figure 6.8. Equivalently, the same correlations and affects relations can be generated by setting  $A = \Lambda$  and  $C = \Lambda \oplus B$  which would correspond to the causal structure of Figure 6.4a but with the arrow  $B \dashrightarrow A$  removed instead. In this example, without additional information, we cannot distinguish between the three causal structures where B has a dashed arrow to A, to C or to both. All 3 causal structures (also illustrated in Figure 6.12) are compatible with the correlations and the affects relations of the example.

Note that A and  $\Lambda$  together are enough to determine B. Therefore, in a space-time configuration such as that of [105, 116] where the space-time random variables  $\mathcal{A}$ ,  $\mathcal{B}$  and  $\mathcal{C}$  are pairwise spacelike separated, and  $\Lambda$  is in the common past of  $\mathcal{A}$  and  $\mathcal{C}$ , information about the freely chosen variable  $\mathcal{B}$  can already be available at the space-time location of  $\mathcal{A}$ . This is outside the future of  $\mathcal{B}$ , and can lead to superluminal signalling from the space-time location of  $\mathcal{B}$  to that of  $\mathcal{A}$ . More generally, note that  $\{B,\Lambda\}$  affects A but neither B nor  $\Lambda$  affect A (as denoted by the dashed arrows in Figure 6.8). By Theorem 6.5.1,  $\mathcal{A}$  must be in the joint future of  $\mathcal{B}$  and  $\Lambda$  for compatibility with the space-time. However, the space-like separation between  $\mathcal{A}$  and  $\mathcal{B}$  leads to a violation of this condition and hence to superluminal signalling, irrespective of the spacetime location of  $\Lambda$ . The following theorem (a proof of which can be found in Appendix 6.8.2)

<sup>&</sup>lt;sup>16</sup>Even though the naming convention is slightly different from the full tripartite Bell scenario of Figure 6.2

<sup>&</sup>lt;sup>17</sup>When causally modelling Bell scenarios, it is standard to take input variables to be exogenous and output variables to have no outgoing arrows, and be related by a common cause, as in Figures 3.1a and 6.2. However, one can explain the correlations and affects relations of the Bell scenario using alternative causal models whereby the inputs have incoming fine-tuned arrows, and/or outputs have outgoing fine-tuned arrows. These can also be analysed in our framework, but we will not consider them here.



Figure 6.8: A simple protocol that can lead to superluminal signalling in the jamming scenario of [105, 116]: (a) Space-time embedding of the variables A, B, C and  $\Lambda$  suggested in [105, 116], where the joint future of the space-time variables  $\mathcal{A}$  and  $\mathcal{C}$  (blue region) is contained in the future of  $\mathcal{B}$ . Here, B corresponds to the input of a party, Bob while A and C are the outputs of Alice and Charlie respectively. In this configuration, [105, 116] claim that there will be no superluminal signalling since  $\mathcal{A}$  and  $\mathcal{C}$  are jointly accessible only in the future of  $\mathcal{B}$  and  $B = A \oplus C$  can not be computed from A or C individually. However, if  $\Lambda$  is observed, then our protocol explained in the main text allows for superluminal signalling in this space-time configuration, contrary to this claim. In the protocol,  $C = \Lambda$  and  $B = A \oplus \Lambda$  freely chosen variable B can already be determined at the location of  $\mathcal{A}$  which is outside the future of  $\mathcal{B}$  and hence can lead to superluminal signalling. (b) The causal structure corresponding to the protocol. Note that this causal structure of Figure 6.4a even though both explain the same correlations and affects relations. (c) The correlations obtained from the protocol.

establishes the possibility of superluminal signalling using an observed  $\Lambda$  in a general class of jamming scenarios.

**Theorem 6.6.1.** Consider a causal model over four nodes A, B, C and  $\Lambda$ , all of which are observed. Let B be exogenous, A and C have no outgoing arrows and  $\Lambda$  be a common cause of A and C. Further, suppose that B affects the set  $\{A, C\}$  without affecting its individual elements i.e., B jams the correlations between A and C. This causal model is not compatible with Minkowski space-time  $\mathcal{T}$  i.e., leads to superluminal signalling when the variables A, B and C are embedded in  $\mathcal{T}$  as proposed in [105, 116], irrespective of the space-time location of  $\Lambda$ . The proposed embedding is:  $A \not\prec B$ ,  $B \not\prec f C$ ,  $A \not\prec f C$ ,  $\overline{\mathcal{F}}(A) \cap \overline{\mathcal{F}}(C) \subseteq \overline{\mathcal{F}}(B)$ , and the accessible region of each observed ORV coincides with its inclusive future.

**Remark 6.6.1.** The classical protocol described in this section can also be extended to jamming scenarios where Bob jams non-classical correlations between Alice and Charlie. For example, let Alice and Charlie share a bipartite state, measure it using the inputs A and C and obtain the outcomes X and Z, and let B be an input of Bob that may be sent to one of them. Suppose, depending on B, we would like Alice and Charlie to share either correlations arising from measurements on the Bell state  $|\psi_0\rangle = \frac{1}{\sqrt{2}}(|00\rangle + |11\rangle)$  or those arising from the same measurements on a different Bell state,  $|\psi_1\rangle = \frac{1}{\sqrt{2}}(|01\rangle + |10\rangle)$ . For this, Alice and Charlie simply need to share the first Bell state but one of the parties, say Alice must perform a controlled NOT

on her local subsystem controlled on the value of B, prior to the measurement. Then the parties would measure  $|\psi_0\rangle$  whenever B = 0 and  $|\psi_1\rangle$  whenever B = 1 as required. Similarly, B can be used to decide whether Alice and Charlie share one relabelling of a PR box (Equation (3.5)) or another.<sup>18</sup> Note however that in a tripartite Bell scenario where all the parties are noncommunicating, these correlations cannot be generated quantum mechanically or in boxworld, and therefore define a more general class of post-quantum theories [116, 185].

# 6.6.2 Comments on Horodecki and Ramanathan, Nat. Comms. 2019

Here, we address the claim of [116] (Propositions 2 and 3) that the no-signaling conditions NS2 (Equation (6.6)) and NS3' (Equation (6.7)) are necessary and sufficient for no causal loops in the bipartite and tripartite Bell scenarios respectively. The definition of no causal loops used by [116] in the proof of this statement is the following, which we quote verbatim below.

No causal loops occur, where a causal loop is a sequence of events, in which one event is among the causes of another event, which in turn is among the causes of the first event.

Here "events" correspond to measurement events, i.e., an input and output pair of space-time random variables both of which are assigned the same space-time location, and "cause" does not seem to be defined explicitly. A rigorous formalisation of these definitions is offered by the present framework— events correspond to ordered (or space-time) random variables (Definition 6.4.1), causation and signalling (which are non-equivalent) are defined in terms of a causal model (Definition 6.3.1 and 6.3.4) and different types of causal loops have been distinguished (Definition 6.3.6). Based on this, it appears that neither direction of these claims hold. In the following we provide concrete counter-examples to illustrate our points, and explain how our results provides possible resolutions. Since the definitions of [116] appear to be relatively ambiguous, we note that our framework may not be the only possible way to formalise them, but we are not aware of any other.

**Definition of causal loops:** As we have seen, fine-tuning allows for two distinct types of causal loops (Definition 6.3.6), the affects loops which can be detected at the observed level and the functional loops that may in principle be operationally undetectable (*functional causal loops*). An example of the latter in the bipartite Bell scenario is provided in Figure 6.9a, where the observed distribution  $P_{XYAB}$  over the parties' inputs A and B, and outputs X and Y

<sup>&</sup>lt;sup>18</sup>Similarly, one can devise protocols to realise other types of jamming correlations by allowing the underlying mechanisms (like the choice of local operation prior to measurement) to depend on B in a way that this does not reflect in the local statistics. Further one can also consider jamming correlations where Bob's output Y is non-trivially involved in the process, which we do not analyse here as it is not directly relevant to the results of this thesis.



Figure 6.9: Counter-examples to the claims of [116]: All variables are assumed to be binary in these examples. (a) Causal loops do not imply signalling: Let the dependence of X and Y on their parents be  $X = Y \oplus \Lambda$  and  $Y = X \oplus \Lambda$ , with X and Y being uniformly distributed.<sup>19</sup> This, along with the d-separation condition (Definition 6.3.2) yields  $P_{XYAB} = P_X P_Y P_A P_B$ , and the no-signaling conditions (6.6) trivially hold. Neverthe the the the the the the test and test when  $\Lambda$  is unobserved. (b) Signalling does not imply causal loops: When A and Y are embedded in space-time such that they are space-like separated, the causal model is not compatible (Definition 6.4.7) with the space-time and leads to superluminal signalling. However, it has no causal loops. (c) Instantaneous measurements are problematic: Let all the variables be uniformly distributed and related as  $X = A \oplus Y$  (X jams A and Y),  $A \oplus \Lambda = X$  (A and  $\Lambda$  collide into X). This gives  $Y = \Lambda$  (hence the solid arrow). If A and X are assigned the same space-time location by idealising the local measurement to be instantaneous, then no superluminal signalling ensues despite the causal loop. Note that even if we had a solid arrow 2-cycle with X affects A affects X, this need not pose a problem (either to the free choice definition used in [116] or to superluminal signalling) when they are embedded at the same location.

exhibits no correlations i.e.,  $P_{XYAB} = P_X P_Y P_A P_B$ , and there are no affects relations i.e., none of the causal influences can be operationally detected, when  $\Lambda$  is unobserved. Nevertheless, the causal structure involves a (functional) causal loop, which would qualify as a causal loop also according to the definition of [116]. Clearly we could never rule out such causal loops based on any operational conditions, especially using conditions that only restrict the correlations (such as NS2 and NS3'). We can only rule out causal loops of the affects type as done in Theorem 6.5.2, where have taken into account correlations as well as an operational notion of causation defined through active interventions. For this reason, even if we take "no causal loops" of [116] to mean "no affects causal loops", the no signalling conditions (NS2) (6.6) and (NS3') (6.7) are neither necessary nor sufficient, which we discuss below with explicit examples.

Sufficient conditions for no causal loops: The no-signaling conditions NS2 (6.6) and NS3' (6.7) are not sufficient for no causal loops (of either type) in the corresponding Bell scenarios, and there are a number of reasons for this. Firstly, in Theorem 6.6.1 we have shown that the jamming scenario in the space-time configuration of [116] can lead to superluminal signalling if the common cause  $\Lambda$  is an observed variable. The scenarios for which we have proven this result are a special case of the tripartite Bell experiment where the inputs of

Alice and Charlie, and the output of Bob are ignored, and they satisfy the tripartite nosignaling conditions NS3' of [116]. This shows that NS3' alone are not sufficient for ruling out superluminal signalling and subsequent causal loops that can arise, we have also seen that there exists an explicit protocol (Figure 6.8) that achieves superluminal signalling when  $\Lambda$  is observed. Furthermore, including this assumption still does not suffice to rule out causal loops, even in the bipartite Bell experiment, as shown in Corollaries 6.5.1 and 6.5.2. To rule out affects causal loops, we also require that the output variables do not affect any other variables. For an example of an affects causal loop that is not ruled out by the no-signaling conditions NS2 of the bipartite Bell scenario, simply consider a 2-cycle between the outputs X and Y such that  $X \longrightarrow Y$  and  $Y \longrightarrow X$  with X = Y. Since the inputs A and B play no role here, no-signaling is trivially satisfied despite the causal loop that allows X to affect Y and Y in turn to affect X. In addition, the inclusion of these extra assumptions still falls short for ruling out all causal loops and superluminal signalling. For the latter, as mentioned in Section 6.5.2, we require further assumptions involving affects relations such as  $\{A, X\}$  does not affect Y. We note that a previous work by Baumeler, Degorre and Wolf [20] also illustrates that non-classical, non-signaling correlations in the bipartite Bell scenario admit an alternative explanation using causal loops, pointing to insufficiency of the no-signaling conditions for ruling out causal loops.

**Necessary conditions for no causal loops:** We also noted in Section 6.6.1 that the nosignaling conditions are not necessary for no causal loops in the bipartite and tripartite Bell scenarios. Consider the causal structure of Figure 6.9b where we have the usual bipartite Bell scenario, and in addition that Alice's input A affects Bob's output Y. This violates the nosignaling conditions. The particular embedding of this causal model in Minkowski space-time  $\mathcal{T}$ where the space-time random variables  $\mathcal{A}$  and  $\mathcal{Y}$  are space-like separated is not compatible with the space-time and leads to superluminal signalling. However, the causal structure is clearly acyclic and hence the causal model has no causal loops (in this case both of the affects or the functional kind). Further, there clearly exists a compatible embedding of the causal model in  $\mathcal{T}$ , this would be one where  $\mathcal{Y}$  is taken to be in the future of  $\mathcal{A}$  (in accordance with Corollary 6.5.3). The fact that superluminal signalling does not imply causal loops has been pointed out in [116] itself. If superluminal signalling were allowed only in one special reference frame, this cannot be used to create causal loops.<sup>20</sup> The conclusion that the no-signaling conditions are necessary for no causal loops appears to be at odds with this observation, and the proof of [116] seems to be implicitly assuming that superluminal signalling in one frame would enable such signalling in all frames.

**Instantaneous measurements:** As an idealisation, measurements are considered to be instantaneous in [116], such that the input and output variables of each party are embedded at

 $<sup>^{20}</sup>$ Though we have not explicitly considered reference frames in this work, they could in principle be taken into account by introducing a partially ordered set for every choice of reference frame, such that the space-time is characterised by a collection of partially ordered sets. And compatibility of a causal model with the space-time (Definition 6.4.7) would correspond to compatibility (or no signalling outside the future) with all these partially ordered sets, such that signalling outside the light cone even in one frame would be a violation of compatibility.



Figure 6.10: A space-time embdedding for the cyclic causal model of Figure 6.6b: In the causal model, we have B affects  $\{A, C\}$  and  $\{A, C\}$  affects B. In this above space-time embedding, the future of the space-time random variable  $\mathcal{B}$  exactly coincides with the joint future of  $\mathcal{A}$  and  $\mathcal{C}$  (blue region) i.e., the causal model is compatible with the space-time (Definition 6.4.7) for this special embedding and does not lead to superluminal signalling even though it contains functional causal loops. Note that for affects causal loops, the only compatible embedding in space-time is the one where all the loop variables are assigned the same space-time location (c.f. Corollary 6.5.3).

the same space-time location. While this is an idealisation that is often made, in the present case in must be treated with caution, since causal loops (even of the affects type) between variables at the same space-time location do not necessarily lead to superluminal signalling (c.f. Theorem 6.5.2, Figure 6.9c). In the bipartite Bell scenario, when Alice's input and output are at the same space-time location, this allows for an affects causal loop  $A \longrightarrow X$  and  $X \longrightarrow A$ , while still satisfying free choice in the sense of: a free variable is not correlated with variables outside its (inclusive) future. This is yet another reason for the insufficiency for the no-signaling conditions for ruling out causal loops. From a physical standpoint, it would be natural to expect that in a causal theory, non-trivial operations cannot be performed instantaneously and some frameworks for causality, such as [171] explicitly forbid this. Note that the sufficiency proof of Theorem 6.5.2 explicitly forbids such instantaneous embeddings.

**Causal structure vs space-time structure:** It is important to point out that the directed graphs presented here differ from those of [116] in what they represent. In [116], arrows between space-time variables (which are always represented as solid)  $\mathcal{X} \to \mathcal{Y}$  represent the space-time structure i.e., that  $\mathcal{Y}$  is in the future of  $\mathcal{X}$ . This does not necessarily mean that  $\mathcal{X}$  is a cause of  $\mathcal{Y}$ . In this thesis, arrows are used to represent the causal structures such that  $\mathcal{X} \to \mathcal{Y}$  stands for  $\mathcal{X}$  is a cause of  $\mathcal{Y}$  (Definition 6.3.1). If the causal model is compatible with the space-time, this would imply that  $\mathcal{Y}$  is indeed in the future of  $\mathcal{X}$ , but apriory this need not be the case in our framework. [116] assumes that for any two space-time random variables  $\mathcal{X}$  is a cause of  $\mathcal{Y}$  implies that  $\mathcal{Y}$  is in the future of  $\mathcal{X}$ . Again the definition of "cause" is important here, since this need not hold for fine-tuned causes. For example if X is a fine-tuned cause of Y (i.e.,  $X \dashrightarrow Y$ ),  $\mathcal{Y}$  being in the future of  $\mathcal{X}$  is not necessary for ruling out superluminal signalling in such cases (Theorem 6.5.1). Formalising this notion in terms of the affects relation (introduced in Definition 6.3.4) we have  $\mathcal{X}$  affects  $\mathcal{Y}$  implies that  $\mathcal{Y}$  is in the (inclusive) future of  $\mathcal{X}$ . Note that this follows from our compatibility condition (Definition 6.4.7).

**Free choice:** Free choice of inputs in the Bell scenario is crucial to the arguments of [116] since without this, outputs of parties can be correlated with inputs of space-like separated parties through common causes, hence violating the no-signaling conditions without leading to causal loops. The commonly used definition of free choice proposed in [61] needs to be modified when considering jamming correlations in multipartite Bell scenarios as noted in [116]. The former requires each party's input to only be correlated with variables in its future while the latter allows Bob's input in the tripartite Bell experiment to be correlated with Alice and Charlie's outputs jointly even though the each output is space-like separated to the input variable. Accordingly, the definition will change depending on the scenario considered, and does not generalise easily. Here we have modelled free choice of a random variable by taking it to be exogenous in the causal graph, which generalises readily to arbitrary causal structures. Modelling free choice is not a major concern in this work since our main results, Theorems 6.5.1 and 6.5.2 do not assume any form of free choice and apply to arbitrary causal models.

**Remark 6.6.2.** It can be shown that defining free choice through exogeneity would recover the free choice definitions of [61, 116] in the Bell scenarios when the causal model is embedded compatibly in the space-time (using the results of Section 6.5.2). However, the converse is not true since the latter are defined in terms of correlations and the lack of correlation does not imply a lack of causation, specially when we allow for fine-tuning. Therefore the definitions are not equivalent. Other inequivalent definitions may also be possible. We leave for future work, a deeper exploration of the relationship(s) between the different notions of free choice that can arise in the presence of fine-tuning.

# 6.7 Discussion

**D-separation and affects relations:** The *affects relation* (Definition 6.3.4), based on the notion of interventions which is crucial for distinguishing between correlation and causation. In acyclic causal structures [159, 115] and in classical cyclic causal structures [84], existing frameworks completely describe how the post-intervention distribution can be calculated, from the observed distribution and/or the underlying causal mechanisms. In non-classical cyclic causal structures, such a characterisation is not available. In Section 6.3, we have used the d-separation condition (Definition 6.3.2) on the observed distribution to obtain a partial characterisation which suffices for the current purpose, but this does not fully specify the post-intervention distribution. In Appendix 6.8.1, we outline a possible method for doing so, given the underlying causal mechanisms. We note that this method may not always recover the d-separation condition 6.3.2. In the classical cyclic case, it has been shown that the d-separation condition is recovered whenever all the variables are discrete and the causal mechanisms of the model satisfy a certain property known as ancestrally unique solvability (discussed in Appendix 6.8.1). All the examples presented in the main chapter are either acyclic or cyclic and satisfy these properties. Therefore our main results are not impacted by making d-separation a defining condition. Defining the framework without this assumption, using the causal mechanisms as primitives would only generalise it, and could potentially be of independent interest as a general framework for causal modelling. Further, another observation made in Appendix 6.8.1 is that

the presence of causal loops could allow us to distinguish between a faithful, non-classical explanations vs unfaithful classical explanations (e.g., using non-local hidden variables) of quantum correlations, which cannot be operationally distinguished otherwise. Formalising this intuition would provide another interesting line of investigation stemming from this work.

**Causal loops and paradoxes:** A full formalisation of our causal model framework, in a manner described in the above paragraph would allow us to make a precise comparison with the two main types of closed timelike curves or CTCs that have been proposed in the literature namely, Deutsch's CTCs (DCTCs) [75] and post-selected CTCs (PCTCs) [25, 198, 137, 138]. These CTCs have provided insights into complexity classes in information theory, known to have different computing powers [2, 138] and to provide different resolutions to the grandfather and unsolved theorem paradoxes [138]. In our framework, grandfather type paradoxes are forbidden by the assumption that a valid joint probability distribution observed variables indeed exists. An example of a paradoxical scenario is a 2-cycle between X and Y where the influence  $X \longrightarrow Y$ defines the functional dependence Y = X and  $Y \longrightarrow X$  gives the dependence  $X \neq Y$ . These equations are mutually inconsistent and there is no joint distribution  $P_{XY}$  compatible with these dependences.<sup>21</sup> This consistency condition appears to be similar in spirit to that of PCTCs where the paradoxical scenario is never compatible with the post-selection i.e., occurs with zero probability. The unproved theorem paradox on the other hand, can depend on how the framework is formalised. For example, in classical cyclic causal models, an assumption regarding the unique solvability of the underlying functional dependences is often made. In particular, this could be seen as the requirement that any information involved in a loop (such as the unproved theorem) must be fully determined by the mechanisms of the causal model thereby eliminating the paradox of a proof that "came from nowehere".

**Causal inference in the presence of fine-tuning:** The *no fine-tuning* assumption, in some form or the other is required by all causal discovery algorithms, which are algorithms that infer an unknown causal structure from observed correlations, interventions and other additional information that may be available. Relaxations to the assumption have been considered where certain forms of fine-tuning have been allowed [226]. In the general case where arbitrary amount of fine-tuning is allowed, it can be fundamentally impossible to infer the causal structure since we have an arbitrary number of fine-tuned causal arrows that produce absolutely no observable effects. Take for instance the example of Figure 6.9a, where there is a fine-tuned causal loop even though no correlations can be seen between the observed variables even when we intervene on them. Therefore, to address the problem of causal inference in our framework i.e., in the presence of cycles, fine-tuning and non-classical systems, we would need to make certain assumptions on the kind of fine-tuning that is allowed. One reasonable assumption could be that whenever there is a fine-tuned arrow between two variables  $X \to Y$  that cannot be detected when considering those variables alone, the causal influence must become detectable

<sup>&</sup>lt;sup>21</sup>This is similar in structure to the Liar's paradox, e.g., "this statement is false" or more generally a liar cycle. Interestingly, this will crop up again in our discussion of multi-agent paradoxes and contextuality in Chapter 8. The striking similarities between such paradoxes in philosophy, time-travel, quantum contextuality and multi-agent scenarios have also been pointed out in [77].

when we consider additional variables along with X and Y, as is the case in the jamming and collider examples of Section 6.3.5. This is another interesting direction for future work.

**Jamming theories and physical principles:** An important motivation for analysing postquantum theories and jamming type scenarios constrained by relativistic principles is to gain deeper insights into the physical principles that single out quantum theory. One property of jamming theories that becomes suggested by our results is that they would involve some form of fine-tuning as well as variables and influences that must remain fundamentally inaccessible to observation. Even though jamming correlations in the space-like separated configuration of [105, 116] are post-quantum phenomena, they are similar in spirit to non-local hidden variable explanations of quantum theory such as the De broglie-Bohm pilot wave theory [27] which also involve fine-tuned influences. Here the pilot wave which is not operationally accessible leads to the operational predictions of quantum theory, but can involve non-local or superluminal influences at the ontological level. An interesting future direction would be to formalise the concept of jamming theories in a similar manner to generalised probabilistic theories and consider other physical properties thereof. One difficulty in doing this is that the post-quantum jamming scenario requires a particular space-time configuration, while GPTs can be formalised independently of a space-time structure. Disentangling the notions of causality and space-time through operational considerations, as done here, could be seen as first step towards such a more general characterisation of jamming theories.

**Indefinite causal orders:** Going beyond the notion of a fixed (but possibly unknown) causal structure, causal orders that may be fundamentally indefinite has been considered [154]. Here processes are said to be *causally separable* if they admit an explanation in terms of a classical mixture of fixed causal order processes, and *causally non-separable* otherwise. Even though our framework assumes a fixed causal and space-time structure, we note the following point regarding causally separable processes. Consider a scenario where the causal order  $X \longrightarrow Y$ or  $Y \longrightarrow X$  between two classical random variables is determined probabilistically by a third, binary variable A. Let the first causal order correspond to the dependence Y = X and the second correspond to the dependence  $X = Y \oplus 1$  for binary X and Y. This can be modelled in the causal structure of Figure 6.9a with the dashed arrows replaced with solid arrows, considering only the variables X, Y and A. The causal mechanisms would be  $X = \Lambda (Y \oplus 1) \oplus (\Lambda \oplus 1) E_X$ and  $Y = (\Lambda \oplus 1) X \oplus \Lambda E_Y$ , where  $E_X$  and  $E_Y$  are mutually independent, uniformly distributed binary variables. Conditioned on  $\Lambda = 0$ , X affects Y but Y does not affect X i.e., P(X|do(Y = $(y), \Lambda = 0) = P(X|\Lambda = 0) \forall y$  and similarly  $P(Y|do(X = x), \Lambda = 1) = P(Y|\Lambda = 1) \forall x$ . However, when  $\Lambda$  is not given, it can be checked that X affects Y and Y affects X which is an affects causal loop.<sup>22</sup> At first sight this might seem counter-intuitive to the idea that classical mixtures of causal orders should be implementable in a lab by flipping a coin to decide the order of two operations. Such a physical implementation corresponds to a situation where one operation (say, that which generates X) is performed at a time  $t_1$  and the other (which generates Y) at time  $t_2 > t_1$  when  $\Lambda = 0$  while the second operation is performed at  $t_1$  and first at  $t_2$  when  $\Lambda = 1$  i.e.,

<sup>&</sup>lt;sup>22</sup>Note that the contradictory conditions X = Y and  $Y = X \oplus 1$  are not a problem here since they never have to be satisfied simultaneously.

it is a directed acyclic graph over 4 variables, with  $X_{t_1} \rightarrow Y_{t_2}, Y_{t_1} \rightarrow X_{t_2}$ . This suggests the importance of the space-time embeddings in understanding the physicality if such processes. It would be interesting to analyse the quantum counterpart of this example, namely the quantum switch [53] (which implements a quantum controlled superposition of the orders) in a similar manner, through a suitable generalisation of our framework. Further, indefinite causal orders that violate so-called *causal inequalities* [154] have also been proposed in frameworks that do not assume a fixed background space-time structure (as we have assumed in the present work). However such frameworks appear to disallow certain types of causal loops that are allowed by ours, and it would also be interesting to consider whether genuine<sup>23</sup> causal inequality violations can be modelled in our framework.

**Indefinite space-time locations:** Physically, it is possible to implement scenarios where a quantum system is superposed not only between spatial locations, but also in time. These spatio-temporal superpositions are required in the physical implementation of the quantum switch mentioned above [173, 184, 171]. It would therefore be of interest to generalise our framework to also consider unobserved systems that may be space-time delocalised. The framework currently uses the standard quantum formalism and the Born rule which only apply to quantum states and composite systems defined at a single instant of time. Generalisations to the standard formalism that define quantum states over space and time have been proposed and a particularly relevant one for this purpose would be the causal box formalism of [171] which models quantum information processing mechanisms that act on such systems. In an ongoing collaboration with members of ETH Zurich<sup>24</sup>, we explore the relations between the causal box formalism and indefinite causality, outlining a way to recover a probability rule analogous to the Born rule in certain cases. This suggests one possible way to generalise the results of the current chapter to situations where the systems are delocalised in space and time.

# 6.8 Appendix

## 6.8.1 Do-conditionals from causal mechanisms

In Section 6.3 we have outlined how interventions and do-conditionals (i.e., the post intervention distribution) are defined in our framework, and Theorem 6.3.1 provides some conditions under which the post and pre intervention distributions can be related. Ideally though, one would expect that it should be possible to fully specify the post-intervention distribution if we are given all the underlying causal mechanisms involved in the causal structure. For example, in the classical case, the structural equations of the causal model [159] provide these causal

 $<sup>^{23}</sup>$ When no assumptions are made, causal inequalities, like Bell inequalities can be trivially violated. Identifying a genuine violation hence depends on the assumptions under which the causal inequality is derived, and these are not yet completely formalised for general scenarios.

<sup>&</sup>lt;sup>24</sup>Vilasini, V., del Rio, L. and Renner, R. *Causality in definite and indefinite space-times.* In preparation (2020). https://wdi.centralesupelec.fr/users/valiron/qplmfps/papers/qs01t3.pdf

mechasisms. Here for each node X in the causal structure, the dependence of X on its parents par(X) corresponds to a stochastic map, which can be written in terms of a deterministic function  $X = f_X(par(X), E_X)$  by including an additional exogenous and unobserved error variable  $E_X$  for each node X. This is called a *structural equation*. If the structural equations for all the nodes are known, then the complete post-intervention distribution can be calculated. This has been shown to be the case for classical cyclic causal models in [84]. An intervention do(x) on X, corresponds to updating the structural equation for X to X = x while keeping the remaining structural equations the same. Another important result for the classical case derived in [84] is that the d-separation property or the global directed Markov condition of Definition 6.3.2 is recovered whenever all the random variables are discrete and the structural equations of the causal model satisfy a property known as ancestrally unique solvability (anSEP). Roughly, this property demands that the structural equation for each node must admit a unique solution given the values of the node's ancestors. We need not define this concept formally for our purposes here.

Our goal would be to extend these ideas to quantum and post-quantum cyclic causal structures, where the causal mechanisms involve measurements and transformations on non-classical systems, which cannot be expressed using deterministic structural equations. In the non-classical case, it is unclear what conditions allow for the d-separation condition to be recovered. Even to make this question precise in the non-classical case, one would need to specify the analog of structural equations for such causal models which is an open problem. Here, we present a possible method for achieving this by explaining it using the following example and sketching how it might generalise to a larger class of causal models.

**Example 6.8.1** (A quantum cyclic causal model). Consider the cyclic variation of the bipartite Bell causal structure illustrated in Figure 6.11a. Let the common cause  $\Lambda$  correspond to the Bell state  $|\psi_{\Lambda}\rangle = \frac{1}{\sqrt{2}}(|00\rangle + |11\rangle)$ . Suppose that A and B are the settings of local measurements on the two subsystems such that when these variables take the value 0, it denotes a  $\sigma_Z$  or computational basis ( $\{|0\rangle, |1\rangle\}$ ) measurement on the associated subsystem, and the value 1 denotes a  $\sigma_X$  or Hadamard basis ( $\{|+\rangle, |-\rangle\}$ ) measurement. X and Y are the binary outcomes of these measurements where 0 and 1 for the Hadamard basis measurements denote the outcomes + and - respectively. The additional constraints coming from the causal loop are that B = X and A = Y. This specifies all the causal mechanisms, how do we calculate the observed distribution  $P_{XYAB}$ ?

A method based on post-selection: One method is to first calculate the observed correlations for the specified state and measurements in the original Bell scenario (Figure 3.1a), and then post-select on the observations that obey the loop conditions B = X and A = Y. More formally, this corresponds to transforming the original cyclic causal structure of Figure 6.11a to the acyclic causal structure of Figure 6.11b by cutting off the edges  $A \longrightarrow X$  and  $B \longrightarrow Y$  and replacing them with the edges  $A^* \longrightarrow X$  and  $B^* \longrightarrow Y$  by introducing two exogenous nodes  $A^*$ and  $B^*$ . Then the inputs  $A^*$  and  $B^*$  and outputs X and Y along with the shared system  $\Lambda$ define a Bell scenario, while the variables A = Y and B = X can simply be seen as local postprocessings of the outcomes. We can then calculate the observed probabilities for this acyclic causal structure using the Born rule, and post-select on  $A^* = A$  and  $B^* = B$ , which effectively

achieves the post-selection A = Y and B = X in the original Bell scenario (Figure 3.1a). These probabilities will not be normalised and one has to renormalise them to obtain the observed distribution  $P_{XYAB}$ . This is calculated in Figure 6.11 and can be used to find all the affects relations. An intervention on A would cut off the arrow from Y to A. We can immediately see that A does not affect X since A is effectively exogenous in the post-intervention causal structure and will be uncorrelated with X since  $\Lambda$  is the maximally entangled state. Similarly B does not affect Y. However,  $\{A, B\}$  affects  $\{X, Y\}$  since a joint intervention on A and B takes us back to the original Bell scenario in which these sets are correlated. We also have Xaffects B and Y affects A due to the loop conditions A = Y and B = X. And naturally, also  $\{X,Y\}$  affects  $\{A,B\}$ . In this example,  $\mathcal{G}_{do(A,B)}$  corresponds to a quantum causal structure (the Bell scenario) while  $\mathcal{G}_{do(X,Y)}$  is a simple classical causal structure for the causal structure  $\mathcal{G}$  of Figure 6.11a and the (observed) arrows  $\sim \rightarrow$  of the figure can be classified into dashed and solid arrows as:  $A \dashrightarrow X, B \dashrightarrow Y, X \longrightarrow B$  and  $Y \longrightarrow A$ . Note that this example corresponds to a functional causal loop as per Definition 6.3.6, and the observed distribution satisfies the d-separation condition of Definition 6.3.2. The post-intervention distribution is fully specified here because, all interventions (except the one on  $\Lambda$  alone) are associated with acyclic postintervention graphs and for interventions on the exogenous  $\Lambda$ , the post and pre-intervention distributions coincide.

Applying the method to fine-tuned explanations of non-classical correlations: It is known that certain non-classical correlations arising in the bipartite Bell causal structure cannot be obtained in the same causal structure if the common cause  $\Lambda$  was classical. However, these correlations can be easily generated in the classical, fine-tuned causal structure of Figure 6.11c, which differs from the original causal structure by the inclusion of fine-tuned causal influences from each party's input to the other party's output. We now explain how this is achieved and then apply the post-selection method explained above to create a causal loop in Figure 6.11c by adding  $X \longrightarrow B$  and  $Y \longrightarrow A$ . This will demonstrate that, even though the same nonclassical correlations and affects relations can be obtained in the original Bell causal structure and its fine-tuned classical counterpart 6.11c, the two causal structures behave differently in the presence of causal loops.

First consider the PR box (Equation (3.5)), which is one of the maximally non-classical correlations of the Bell causal structure. It is defined by the condition  $X \oplus Y = A.B$  where all the variables are binary. This is easily generated in the classical causal structure of Figure 6.11c by the structural equations  $\Lambda = E$ , Y = E and  $X = E \oplus A.B$  (where E is binary and uniformly distributed). Other non-classical correlations can be obtained by adding some "noise" to this PR box example. Let  $\Lambda = (E, F)$  correspond to two variables E and F both binary, and the former distributed uniformly. Then the structural equations Y = E and  $X = E \oplus F \oplus A.B$  for different distributions over the exogenous variable F correspond to the PR box mixed with different levels of noise.<sup>25</sup>

$$X = A.B \oplus E \oplus F,$$
  

$$Y = E.$$
(6.8)

Therefore, the causal mechanisms that allow us to produce non-classical correlations  $P_{XYAB}$  in the acyclic causal structure 6.11c are the functional dependences (6.8) along with a specification of the distributions over the exogenous variables E and F that constitute  $\Lambda$ . E is uniform while F can vary depending on the correlation to be generated. We now construct the causal loop by including the additional arrows  $X \longrightarrow B$  and  $Y \longrightarrow A$  and by effectively post-selecting on the loop condition A = Y and B = X. These, along with the causal mechanisms (6.8) of the acyclic case define the mechanisms for the cyclic causal structure. We will now see that these causal mechanisms are incompatible with each other. We have Y = E,  $X = E \oplus F \oplus A.B$ , A = Y and B = X, which gives  $X = E.X \oplus E \oplus F$  and Y = E. Therefore for (E, F) = (0,0), we have (X,Y) = (0,0) and for (E,F) = (0,1) we have (X,Y) = (1,0). However for (E,F) = (1,0)we get  $X = X \oplus 1$  which does not have a solution. For (E,F) = (1,1) we get X = X which is not a unique solution. Therefore if we demand unique solvability, we must require E = 0deterministically which contradicts the initial assumption that E is uniform. Even if we don't require uniqueness, we can not have (E,F) = (1,0) and forbidding this would make E and Fcorrelated and non-uniform.

Therefore, in the classical, fine-tuned explanation of the Bell correlations, adding the loop is not consistent with the causal mechanisms that generate the non-classical correlations in the absence of the loop— in particular, they are in conflict with the preparation of the exogenous variable  $\Lambda$ . If we have a consistent loop, then intervention on A and B will no longer recover the original non-classical correlations. This is in contrast to the faithful case analysed in Figure 6.11 (and explained previously in the text), when do(A, B) gives back the non-classical correlations of the Bell scenario. This suggests that certain (non-local) hidden variable explanations for quantum correlations (in a Bell experiment) can in principle be distinguished from the explanation provided by standard quantum mechanics in the presence of causal loops. We have only shown this for a particular set of functions or causal mechanisms for generating the former and it would be interesting to consider if this generalises, in particular to causal mechanisms provided by Bohmian mechanisms [27], a non-local hidden variable theory.

Generalising to other causal structures: The idea behind the post-selection method employed for Example 6.8.1 above can in principle be generalised to other non-classical, cyclic causal structures where every directed cycle includes at least one edge  $W \longrightarrow Z$  connecting classical nodes W and Z. The intuition is that cutting off such an edge in every direct cycle and replacing it with an edge  $W^* \longrightarrow Z$ , by introducing an additional, exogenous variable  $W^*$ would result in a directed acyclic graph (DAG). One can then apply the generalised causal model framework of [115] (reviewed in Section 2.5.2.1) to obtain the observed distribution in this DAG and then post-select on  $W = W^*$  for all the edges that were cut off. Then a way to recover

<sup>&</sup>lt;sup>25</sup>Note that the model can be symmetrised by including an additional, uniformly distributed binary variable G in the description of  $\Lambda = (E, F, G)$  and using the structural equations  $X = E \oplus (G \oplus 1)(A \cdot B \oplus F)$  and  $Y = E \oplus G(A \cdot B \oplus F)$ .

X		B A A A A A (a)		Ŷ	$ \begin{array}{c}         B \\         X \\         A^* \\         (b)                            $	A	Λ (c)	
	Х	Y	А	В	Measurements, outcomes	$P^{QM}_{XYAB}$	$P_{XYAB}$	
	0	0	0	0	$\sigma_Z \otimes \sigma_Z, \ (0,0)$	$\frac{1}{8}$	$\frac{1}{3}$	
	0	0	1	0	$\sigma_X \otimes \sigma_Z, (+,1)$	$\frac{1}{16}$	$\frac{1}{6}$	
	1	0	0	0	$\sigma_Z \otimes \sigma_X, (1, +)$	$\frac{1}{16}$	$\frac{1}{6}$	
	1	1	1	1	$\sigma_X \otimes \sigma_X, \ (-,-)$	$\frac{1}{8}$	$\frac{1}{3}$	
					(d)			

**Figure 6.11:** A cyclic quantum causal model: (a) A cyclic variation of the bipartite Bell causal structure (Figure 3.1a). (b) A method to calculate the observed distribution of (a) when  $\Lambda$  is non-classical involves this intermediate causal structure. This is obtained from (a) by copying the nodes A and B and removing the directed cycle as shown. This gives an acyclic causal structure for which the distribution  $P_{XYABA^*B^*}$  can be calculated using known methods. Then, post-selecting on  $A = A^*$  and  $B = B^*$  gives the distribution  $P_{XYAB}$  for the original cyclic causal structure of (a). (c) A classical causal, fine-tuned structure that can generate, all non-classical correlations of the bipartite Bell causal structure. Creating a causal loop in this case by adding the arrows  $X \longrightarrow B$  and  $Y \longrightarrow A$ does not lead to the same predictions as (a), which corresponds to adding these arrows to the original Bell causal structure. This method explained in the main text. (d) The table provides the observed distribution for Example 6.8.1 calculated using the proposed method. The only values of A, B, X and Y that are compatible with the loop conditions A = Y and B = X are those listed here, and the fifth column lists the measurements and outcomes that these values correspond to, according to Example 6.8.1.  $P_{XYAB}^{QM}$  denotes the probabilities of the measurements and outcomes listed in the fifth column calculated using the Born rule. These values are sub-normalised, and upon renormalisation, the observed distribution  $P_{XYAB}$  for the cyclic causal structure (a) is obtained. Note that the d-separation condition 6.3.2 is satisfied in this case.

the d-separation condition (using the result of [84]) would be to check whether these exists a classical causal model for the same cyclic causal structure that produces identical observed correlations and satisfies the anSEP property. Note that this classical causal model need not necessarily yield the same post-intervention distributions. In the example of Figure 6.11a, an intervention on A and B gives the Bell scenario, which as we know produces non-classical correlations that cannot be obtained in the corresponding classical causal model [224]. Finally, it would be interesting to compare this method with the framework of post-selected closed time-like curves [138].

**Remark 6.8.1.** We note that assuming the d-separation condition of Definition 6.3.2 as a primitive property of the framework rules out certain cyclic causal structures from being described in our current framework. In the classical case, these are precisely those cyclic causal models that do not satisfy an SEP or those involving continuous random variables (due to the result of [84]). A good example of such a causal model is that of [150], and [84] proposes a generalisation of d-separation called  $\sigma$ -separation through which they derive a generalised global directed Markov condition that applies to classical causal models involving continuous variables and/or do not satisfy an SEP. This reduces to d-separation in the acyclic case. Therefore, one option would be to replace d-separation with  $\sigma$ -separation in Definition 6.3.2 to generalise our framework for cyclic causal models.

## 6.8.2 Proofs

**Lemma 6.3.1.** Let  $S_1$ ,  $S_2$  and  $S_3$  be three disjoint sets of RVs such that  $S_1 \perp S_2 | S_3$ . If S is a set of RVs that is d-separated from these sets in a directed graph  $\mathcal{G}$  containing  $S_1$ ,  $S_2$ ,  $S_3$  and S are nodes i.e.  $S \perp^d S_i \forall i \in \{1, 2, 3\}$ , then any distribution P that is compatible with  $\mathcal{G}$  also satisfies the following conditional independences,

$$(S_1 \cup S) \perp S_2 | S_3, \quad S_1 \perp (S_2 \cup S) | S_3 \text{ and } S_1 \perp S_2 | (S_2 \cup S).$$

*Proof.* The conditional independence  $S_1 \perp S_2 | S_3$  stands for  $P_{S_1 S_2 | S_3} = P_{S_1 | S_3} P_{S_2 | S_3}$ , which implies

$$P_{S_1|S_2S_3} = P_{S_1|S_3}. \tag{6.9}$$

The 3 d-separation relations  $S \perp^d S_i$  for  $i \in \{1, 2, 3\}$  imply that S is d-separated from every subset of the union  $S_1 \cup S_2 \cup S_3$ . This implies the following independences by Definition 6.3.2 of compatibility of the distribution P with the causal model represented by  $\mathcal{G}$ ,

$$P_{S|S'} = P_S \quad \forall S' \subseteq S_1 \cup S_2 \cup S_3. \tag{6.10}$$

Now consider the conditional distribution  $P_{S_2|SS_1S_3}$ . We have,

$$P_{S_{2}|SS_{1}S_{3}} = \frac{P_{S_{2}SS_{1}S_{3}}}{P_{SS_{1}S_{3}}}$$

$$= \frac{P_{S_{3}}P_{S_{2}|S_{3}}P_{S_{1}|S_{2}S_{3}}P_{S|S_{1}S_{2}S_{3}}}{P_{SS_{1}S_{3}}}$$

$$= \frac{P_{S_{3}}P_{S_{2}|S_{3}}P_{S_{1}|S_{3}}P_{S}}{P_{S}P_{S_{1}S_{3}}}$$

$$= P_{S_{2}|S_{3}},$$
(6.11)

where we have used Equations (6.9) and (6.10) in the third line, noting that  $P_{S|S_1S_3} = P_S \Rightarrow P_{SS_1S_3} = P_S P_{S_1S_3}$ . Equation (6.11) is equivalent to  $P_{SS_1S_2|S_3} = P_{SS_1|S_3}P_{S_2|S_3}$  which denotes the conditional independence  $(S \cup S_1) \perp S_2|S_3$ . The conditional independence  $S_1 \perp (S \cup S_2)|S_3$  can be derived analogously due to the symmetry between  $S_1$  and  $S_2$ .

Finally, we have

$$P_{S_2|SS_3} = \frac{P_{S_2SS_3}}{P_{SS_3}} = \frac{P_{S_3}P_{S_2|S_3}P_{S|S_2S_3}}{P_SP_{S_3}} = P_{S_2|S_3}, \tag{6.12}$$

and similarly  $P_{S_1|SS_3} = P_{S_1|S_3}$ . Together with Equation (6.11), this implies  $P_{S_2|SS_1S_3} = P_{S_2|SS_3}$ . This is equivalent to  $P_{S_1S_2|SS_3} = P_{S_1|SS_3}P_{S_2|SS_3}$  which denotes the final conditional independence  $S_1 \perp S_2|(S \cup S_3)$ .

**Theorem 6.3.1.** Given a causal model over a set S of observed nodes, associated causal graph  $\mathcal{G}$  and a distribution  $P_S$  compatible with  $\mathcal{G}$  according to Definition 6.3.2, the 3 rules of do-calculus [159] hold for interventions on this causal model.

#### • Rule 1: Ignoring observations

$$P(y|do(x), z, w) = P(y|do(x), w) \qquad if (Y \perp^d Z|X, W)_{\mathcal{G}_{\overline{X}}}$$

$$(6.1)$$

#### • Rule 2: Action/observation exchange

$$P(y|do(x), do(z), w) = P(y|do(x), z, w) \qquad \text{if } (Y \perp^d Z|X, W)_{\mathcal{G}_{\overline{X}, \underline{Z}}}$$
(6.2)

#### • Rule 3: Ignoring actions/interventions

$$P(y|do(x), do(z), w) = P(y|do(x), w) \qquad if (Y \perp^d Z|X, W)_{\mathcal{G}_{\overline{X}, \overline{Z(W)}}}, \tag{6.3}$$

where X, Y, Z and W are disjoint subsets of the observed nodes and Z(W) denotes the set of nodes in Z which are not ancestors of W.

Proof. Rule 1: We first note that the graph  $\mathcal{G}_{do(X)}$  differs from  $\mathcal{G}_{\overline{X}}$  only by the inclusion of the additional nodes  $I_{X_i}$  and corresponding edge  $I_{X_i} \longrightarrow X_i$  for each  $X_i \in X$ . Therefore, the d-separation condition  $(Y \perp^d Z | X, W)_{\mathcal{G}_{\overline{X}}}$  for the latter implies the same condition  $(Y \perp^d Z | X, W)_{\mathcal{G}_{do(X)}}$  for the former. Using the d-separation condition of Definition 6.3.2 this implies the conditional independence of Y and Z given  $\{X, W\}$  for the distribution  $P_{\mathcal{G}_{do(X)}}$  compatible with  $\mathcal{G}_{do(X)}$ . Using the definition of the do-conditional (Equation (2.51)),  $P_{\mathcal{G}_{do(X)}}(y, z | x, w) \coloneqq$ P(y, z | do(x), w) = P(y | do(x), w) P(z | do(x), w). This conditional independence automatically gives the required Equation (6.1).

**Rule 2:**  $\mathcal{G}_{\overline{X},\underline{Z}}$  is the graph where all incoming arrows to X and outgoing arrows from Z are removed in  $\mathcal{G}$ . Hence, the d-separation condition  $(Y \perp^d Z | X, W)_{\mathcal{G}_{\overline{X},\underline{Z}}}$  implies that the only paths between Y and Z in the graph  $\mathcal{G}_{\overline{X}}$  that are not blocked by X and W are paths involving an

outgoing arrow from Z. These are precisely the paths that get removed in going from  $\mathcal{G}_{\overline{X}}$  to  $\mathcal{G}_{\overline{X},\underline{Z}}$ , resulting in the required d-separation there. The same statement holds for the graph  $\mathcal{G}_{do(X),I_Z}$  which corresponds to adding the nodes  $I_{Z_i}$  and edges  $I_{Z_i} \rightsquigarrow Z_i$  to  $\mathcal{G}_{do(X)}$  for each  $Z_i \in Z$ . The latter holds true since the addition of the  $I_{Z_i}$  nodes and  $I_{Z_i} \leadsto Z_i$  edges cannot create any additional paths between Z and Y that are left unblocked by X and W. This implies that the only paths between Y and the set  $I_Z \coloneqq \{I_{Z_i}\}_i$  not blocked by X and W in  $\mathcal{G}_{do(X),I_Z}$  are paths from  $I_Z$ , going through Z and involving an outgoing arrow from Z i.e., paths involving the subgraph  $I_Z \leadsto Z \leadsto \cdots$ ... All these paths would get blocked when conditioning additionally on Z. This gives (Y  $\perp^d$  $I_Z|X,W,Z)_{\mathcal{G}_{do(X),I_Z}}$ , which through the d-separation condition (Definition 6.3.2) and definition of the post-intervention distribution (Equation (2.51)), give the required Equation (6.2). In other words, once X, W and Z are given, Y does not depend on whether the given value z of Z was obtained through an intervention ( $I_Z = do(z)$ ) or passive observation (i.e., where  $I_{Z_i} = idle$ for all *i*, which is the causal model where no interventions are made on elements of Z).

**Rule 3:** The d-separation condition  $(Y \perp^d Z | X, W)_{\mathcal{G}_{\overline{X},\overline{Z(W)}}}$  implies  $(Y \perp^d Z | X, W)_{\mathcal{G}_{\overline{X},\overline{Z}}}$  since going from the first to the second graph involves removing additional incoming arrows to nodes in the set Z that don't belong to the subset  $Z(W) \subseteq Z$ , and if  $(P \perp^d Q | R)_{\mathcal{G}}$  holds for some sets P, Q, R of nodes in a graph  $\mathcal{G}$ , then the same d-separation,  $(P \perp^d Q | R)_{\mathcal{G}'}$  also holds in any graph  $\mathcal{G}'$  obtained from  $\mathcal{G}$  only by removing some of the edges. By the same argument as in the proof of Rule 1, this implies the same d-separation for the corresponding "do" graph, i.e.  $(Y \perp^d Z | X, W)_{\mathcal{G}_{do(X), do(Z)}}$ . Then, applying the compatibility condition of Definition 6.3.2 to  $P_{\mathcal{G}_{do(X), do(Z)}}$  gives the required Equation (6.3).

**Theorem 6.5.1.** [Necessary and sufficient conditions for compatibility with  $\mathcal{T}$ ] Given a causal model over a set of ORVs  $\mathcal{S}$ , the following condition (cond) is necessary for the causal model to be compatible with the embedding partial order  $\mathcal{T}$  according to compat (Definition 6.4.7).

$$\forall subsets \ \mathcal{S}_1, \mathcal{S}_2 \subseteq \mathcal{S}, \ such \ that \ no \ proper \ subset \ of \ \mathcal{S}_1 \ affects \ \mathcal{S}_2, \\ \bigcap_{s_2 \in \mathcal{S}_2} \overline{\mathcal{F}}(s_2) \notin \bigcap_{s_1 \in \mathcal{S}_1} \overline{\mathcal{F}}(s_1) \ \Rightarrow \ \mathcal{S}_1 \ does \ not \ affect \ \mathcal{S}_2.$$

$$(6.4)$$

While **cond** alone is not sufficient for **compat**, along with the following additional requirement it provides a sufficient condition for **compat** 

$$\forall \mathcal{X} \in \mathcal{S}, \quad \overline{\mathcal{F}}(\mathcal{X}) \subseteq \mathcal{R}_{\mathcal{X}} \tag{6.5}$$

Proof. Necessity of condition: For any two subsets  $S_1, S_2 \subseteq S$  such that no proper subset of  $S_1$ affects  $S_2$ , **compat** implies that  $\mathcal{R}_{S_2} \subseteq \mathcal{R}_{S_1}$  whenever  $S_1$  affects  $S_2$ . Further, by **compat**,  $\mathcal{R}_{\mathcal{X}} = \overline{\mathcal{F}}(\mathcal{X})$  for all ORVs  $\mathcal{X}$ . Using these along with the fact that for any subset  $S_i \subseteq S$ ,  $\mathcal{R}_{S_i} = \bigcap_{s_i \in S_i} \mathcal{R}_{s_i}$ (see Remark 6.4.1), we have  $\mathcal{R}_{S_i} = \bigcap_{s_i \in S_i} \overline{\mathcal{F}}(s_i)$  for all subsets  $S_i$  whenever **compat** holds. Hence

we have the following whenever **compat** holds, which is equivalent to Equation (6.4)

$$\mathcal{S}_1 \text{ affects } \mathcal{S}_2 \Rightarrow \bigcap_{s_2 \in \mathcal{S}_2} \overline{\mathcal{F}}(s_2) \subseteq \bigcap_{s_1 \in \mathcal{S}_1} \overline{\mathcal{F}}(s_1).$$

Sufficiency of condition + additional assumption: Applying the condition **cond** of Equation (6.4) to the case where  $\mathcal{Y} = \{\mathcal{X}\}$  (i.e.,  $\mathcal{Y}$  is any arbitrary copy of  $\mathcal{X}$ ), we have  $\mathcal{X}$  affects  $\mathcal{Y}$ (by Definition 6.4.4 of a copy). This gives  $\overline{\mathcal{F}}(\mathcal{X}') \subseteq \overline{\mathcal{F}}(\mathcal{X})$  and hence  $\mathcal{X}' \in \overline{\mathcal{F}}(\mathcal{X}) \forall$  copies  $\mathcal{X}'$  of  $\mathcal{X}$ . Since by Definition 6.4.5, the accessible region of  $\mathcal{X}$  is the smallest subset of  $\mathcal{T}$  containing all possible copies of  $\mathcal{X}$ , we must have  $\mathcal{R}_{\mathcal{X}} \subseteq \overline{\mathcal{F}}(\mathcal{X})$ . Hence **cond** by itself is not sufficient for **compat** as it does not require the accessible region to coincide with the future, only that it is a subset of the future. Along with the additional assumption that  $\mathcal{R}_{\mathcal{X}} \supseteq \overline{\mathcal{F}}(\mathcal{X}) \ \forall \mathcal{X} \in \mathcal{S}$ , this gives

$$\mathcal{R}_{\mathcal{X}} = \overline{\mathcal{F}}(\mathcal{X}), \quad \forall \mathcal{X} \in \mathcal{S}.$$
 (6.13)

Again, the accessible region of any subset  $S_i \subseteq S$  is  $\mathcal{R}_{S_i} = \bigcap_{s_i \in S_i} \mathcal{R}_{s_i}$  (Remark 6.4.1). Equation (6.13) then implies that

$$\mathcal{R}_{\mathcal{S}_i} = \bigcap_{s_i \in \mathcal{S}_i} \overline{\mathcal{F}}(s_i) \qquad \forall \mathcal{S}_i \subseteq \mathcal{S}$$
(6.14)

Hence we have that the condition of Equation (6.4) along with the additional requirement of Equation (6.5) imply Equation (6.13) and (6.14), where Equation (6.13) is the first condition for **compat** (Definition 6.4.7). Further, Equation (6.14) along with the condition Equation (6.4) then imply the second condition for **compat** which is

 $\forall \text{subsets } \mathcal{S}_1, \mathcal{S}_2 \subseteq \mathcal{S}, \text{ such that no proper subset of } \mathcal{S}_1 \text{ affects } \mathcal{S}_2, \mathcal{S}_1 \text{ affects } \mathcal{S}_2 \Rightarrow \mathcal{R}_{\mathcal{S}_2} \subseteq \mathcal{R}_{\mathcal{S}_1}.$ (6.15)

**Theorem 6.5.2.** [Necessary and sufficient condition for no affects causal loops] A necessary condition for a causal model over a set S of RVs to have no affects causal loops is that there exists an embedding of S in Minkowski space-time  $\mathcal{T}$  such that the corresponding ORVs S satisfies the condition **cond** of Equation (6.4). **cond** along with the additional assumption that any 2 distinct  $ORVs \mathcal{X}$  and  $\mathcal{Y}$  such that one affects the other, cannot share the same location in  $\mathcal{T}$ , are sufficient for having no affects causal loops in the causal model.

Proof. Necessity of condition: Consider a causal model over a set S of RVs that has no affects causal loops. Taking  $\mathcal{T}$  to be Minkowski space-time, embed the RVs in  $\mathcal{T}$  such that whenever an RV X affects another RV Y, then the corresponding ORVs satisfy  $\mathcal{Y} \in \mathcal{F}(\mathcal{X})$ . Since Xaffects Y implies that Y does not affect X (by the assumption of no affects causal loops), this is always possible for any pair of RVs. For embedding the remaining RVs, consider the following conditions for two sets  $S_1$  and  $S_2$  of the RVs,

- 1. No proper subset of  $S_1$  affects  $S_2$ ,
- 2. No proper subset of  $S_2$  is affected by  $S_1$ ,
- 3. No proper subset of  $S_2$  affects  $S_1$  and,
- 4. No proper subset of  $S_1$  is affected by  $S_2$ ,

Then embed the elements of  $S_1$  and  $S_2$  in  $\mathcal{T}$  as follows

• If conditions 1. and 2. hold and  $S_1$  affects  $S_2: \bigcap_{s_2 \in S_2} \overline{\mathcal{F}}(s_2) \subseteq \bigcap_{s_1 \in S_1} \overline{\mathcal{F}}(s_1)$ .

This automatically gives (since the sets are arbitrary):

- If conditions 3. and 4. hold and  $S_2$  affects  $S_1: \bigcap_{s_2 \in S_2} \overline{\mathcal{F}}(s_2) \supseteq \bigcap_{s_1 \in S_1} \overline{\mathcal{F}}(s_1)$ ,
- If conditions 1.-4. hold,  $S_1$  affects  $S_2$  and  $S_2$  affects  $S_1: \bigcap_{s_2 \in S_2} \overline{\mathcal{F}}(s_2) = \bigcap_{s_1 \in S_1} \overline{\mathcal{F}}(s_1)$ .

Such an embedding is always possible because the joint inclusive future of some set T of points in Minkowski space-time can be seen as the inclusive future of a single point  $L_T$ , which would be the earliest point that is in the future of all points in T, and the embedding imposes an order on the points  $L_T$  corresponding to a set. For example, for every pair of ORV sets  $S_1$  and  $S_2$  satisfying 1. and 2. and  $S_1$  affects  $S_2$ , the embedding requires  $L_{S_1} \leq L_{S_2}$ . For the third case where 1.-4. are satisfied and the sets affect each other, we would have  $L_{S_1} = L_{S_2}$ . Note that this is possible even when the RVs are assigned distinct locations in  $\mathcal{T}$ , since only the joint futures of sets need to coincide (this was the case in Figure 6.6b where B and  $\{A, C\}$  affect each other). In the case that  $S_1$  and  $S_2$  consist of single elements, this embedding clearly agrees with Xaffects Y implies  $\mathcal{Y} \in \mathcal{F}(\mathcal{X})$ , since no causal loops forbids the third case. In particular, for all subsets  $S_1$  and  $S_2$  satisfying the conditions 1. and 2., we have

$$S_1 \text{ affects } S_2 \Rightarrow \bigcap_{s_2 \in S_2} \overline{\mathcal{F}}(s_2) \subseteq \bigcap_{s_1 \in S_1} \overline{\mathcal{F}}(s_1)$$
 (6.16)

Now, suppose  $S_1$  and  $S_2$  are two sets of variables that satisfy 1. and  $S_1$  affects  $S_2$ . Then, either they also satisfy 2. or they do not satisfy 2., in which case there exists a proper subset  $S'_2 \,\subset S_2$  such that  $S_1$  affects  $S'_2$ . Without loss of generality, we can assume that  $S_1$  and  $S'_2$ satisfy both 1. and 2. (if not, we can repeat the argument for this case by taking a proper subset of  $S'_2$  that will satisfy these conditions). In other words  $S_1$  and  $S'_2 \subseteq S_2$  satisfy 1. and 2. and also the affects relation  $S_1$  affects  $S'_2$ . Therefore, for the embedding described above, we have  $\bigcap_{s'_2 \in S'_2} \overline{\mathcal{F}}(s'_2) \subseteq \bigcap_{s_1 \in S_1} \overline{\mathcal{F}}(s_1)$  due to Equation (6.16). Further,  $S_2 \supseteq S'_2$  implies that  $\bigcap_{s_2 \in S_2} \overline{\mathcal{F}}(s_2) \subseteq \bigcap_{s'_2 \in S'_2} \overline{\mathcal{F}}(s'_2)$ . Therefore for any  $S_1$  and  $S_2$ , which satisfy 1. and  $S_1$  affects  $S_2$ , we have  $\bigcap_{s_2 \in S_2} \overline{\mathcal{F}}(s_2) \subseteq \bigcap_{s_1 \in S_1} \overline{\mathcal{F}}(s_1)$ . In other words, for the embedding described above, we have the following which is equivalent to **cond** (Equation (6.4)).

 $\forall$  subsets  $S_1, S_2 \subseteq S$ , such that no proper subset of  $S_1$  affects  $S_2$ ,

$$\mathcal{S}_1 \text{ affects } \mathcal{S}_2 \Rightarrow \bigcap_{s_2 \in \mathcal{S}_2} \overline{\mathcal{F}}(s_2) \subseteq \bigcap_{s_1 \in \mathcal{S}_1} \overline{\mathcal{F}}(s_1).$$

For all other RVs (if any remain), the embedding can be arbitrary. For example, these may be superfluous RVs that are neither affected nor affect any other sets of RVs in the model.

Hence we have shown that if a causal model has no causal loops, there exists an embedding in Minkowski space-time  $\mathcal{T}$  such that **cond** is satisfied.

Sufficiency of condition+additional assumption: cond (Equation (6.4)) implies in particular that when the sets correspond to single elements  $\mathcal{X}$  and  $\mathcal{Y}$ ,  $\mathcal{X}$  affects  $\mathcal{Y}$  implies that  $\overline{\mathcal{F}}(\mathcal{Y}) \subseteq \overline{\mathcal{F}}(\mathcal{X})$  (i.e.,  $\mathcal{Y} \in \overline{\mathcal{F}}(\mathcal{X})$ ). Hence we can have a causal loop  $\mathcal{X}$  affects  $\mathcal{Y}$  and  $\mathcal{Y}$  affects  $\mathcal{X}$  only if  $\mathcal{X}$ and  $\mathcal{Y}$  share the exact same space-time location i.e.,  $O(\mathcal{X}) = O(\mathcal{Y})$ . The additional assumption forbids this and hence forbids causal loops.

**Corollary 6.5.3.** [No affects causal loops and compatibility with space-time] A necessary condition for a causal model over a set S of RVs to have no affects causal loops is that  $\exists$  an embedding of S in Minkowski space-time  $\mathcal{T}$  such that the causal model is compatible (Definition 6.4.7) with  $\mathcal{T}$ . This condition along with the additional assumption that any 2 distinct ORVs  $\mathcal{X}$  and  $\mathcal{Y}$  such that one affects the other, cannot share the same location in  $\mathcal{T}$ , are sufficient for having no affects causal loops in the causal model.

Proof. Necessity of condition: The proof is identical to that of Theorem 6.5.2, with the additional requirement that  $\overline{\mathcal{F}}(\mathcal{X}) = \mathcal{R}_{\mathcal{X}}$ . Note that this is a particular choice of embedding since an embedding involves an assignment of locations in  $\mathcal{T}$  to each RV along with an assignment of accessible regions (see Definition 6.4.6). In Theorem 6.5.1, we proved the sufficiency of **cond** (Equation (6.4)) along with the additional assumption  $\overline{\mathcal{F}}(\mathcal{X}) \subseteq \mathcal{R}_{\mathcal{X}}$  for compatibility of a causal model with the embedding  $\mathcal{T}$ . **cond** implies that  $\overline{\mathcal{F}}(\mathcal{X}) \supseteq \mathcal{R}_{\mathcal{X}}$  which along with  $\overline{\mathcal{F}}(\mathcal{X}) \subseteq \mathcal{R}_{\mathcal{X}}$ implies  $\overline{\mathcal{F}}(\mathcal{X}) = \mathcal{R}_{\mathcal{X}}$ , and this was required for the proof. Here, we already have  $\overline{\mathcal{F}}(\mathcal{X}) = \mathcal{R}_{\mathcal{X}}$ by our choice of embedding and Theorem 6.5.2 shows that no causal loops imply the existence of an embedding for which **cond** holds. Hence with the current choice of embedding, this also implies compatibility of the causal model with the embedding partial order  $\mathcal{T}$ .

Sufficiency of condition+additional assumption: The proof is identical to the that of the sufficiency part of Theorem 6.5.2.

**Theorem 6.6.1.** Consider a causal model over four nodes A, B, C and  $\Lambda$ , all of which are observed. Let B be exogenous, A and C have no outgoing arrows and  $\Lambda$  be a common cause of A and C. Further, suppose that B affects the set  $\{A, C\}$  without affecting its individual elements i.e., B jams the correlations between A and C. This causal model is not compatible with Minkowski space-time T i.e., leads to superluminal signalling when the variables A, B and C are embedded in T as proposed in [105, 116], irrespective of the space-time location of  $\Lambda$ . The proposed embedding is:  $A \not\prec B$ ,  $B \not\prec C$ ,  $A \not\prec C$ ,  $\overline{\mathcal{F}}(A) \cap \overline{\mathcal{F}}(C) \subseteq \overline{\mathcal{F}}(B)$ , and the accessible region of each observed ORV coincides with its inclusive future.

*Proof.* B affects  $\{A, C\}$  implies the existence of a directed path from B to at least one of A and C. Given that B is exogenous, A and C have no outgoing arrows and share a common cause  $\Lambda$ , this yields the 3 possible causal structures of Figure 6.12, where the arrows from  $\Lambda$  may be solid of dashed. In all these causal structures, we have the d-separation relations  $A \perp^d C | \{B, \Lambda\}$  and  $B \perp^d \Lambda$ . By Definition 6.3.2, these imply the conditional independences

$$A \perp C | \{B, \Lambda\} \quad \text{and} \quad B \perp \Lambda. \tag{6.17}$$



Figure 6.12: Possible causal structures for Theorem 6.6.1.

Further, the given affects relations along with the exogeneity of B imply that

$$B \not\downarrow \{A, C\}$$
 and  $B \perp A$  and  $B \perp C$ . (6.18)

We now show that in all the causal structures, irrespective of the classification of the outgoing arrows from  $\Lambda$  and the space-time location that it is assigned, the causal model is not compatible with Minkowski space-time with respect to the embedding given in the theorem statement, and hence leads to superluminal signalling. For this, our first step is to show, using Equations (6.17) and (6.18) that *B* affects at least one of  $\{A, \Lambda\}$  and  $\{C, \Lambda\}$ . Suppose that *B* affects neither of the sets, then by its exogeneity, this is equivalent to

$$B \perp \{A, \Lambda\}$$
 and  $B \perp \{C, \Lambda\}$  (6.19)

Then we have

$$P_{A|CB\Lambda} = P_{A|B\Lambda} = \frac{P_{AB\Lambda}}{P_{B\Lambda}} = \frac{P_{A\Lambda|B}P_B}{P_{B\Lambda}}$$
$$= \frac{P_{A\Lambda}P_B}{P_BP_{\Lambda}} = P_{A|\Lambda},$$
(6.20)

Where we have used  $A \perp C | \{B, \Lambda\}$  in the first line and  $B \perp \Lambda$  and  $B \perp \{A, \Lambda\}$  in the second line. Further,  $B \perp \{C, \Lambda\}$  implies that  $P_{BC\Lambda} = P_B P_{C\Lambda}$ . Using this in Equation (6.20),

$$P_{ABC\Lambda} = P_{A|\Lambda} P_{BC\Lambda} = P_B P_{A|\Lambda} P_{C\Lambda} \tag{6.21}$$

Summing over B and A, we obtain  $P_{AC} = \sum_{\Lambda} P_{A|\Lambda} P_{C\Lambda}$ . Using this in Equation (6.21) gives

$$P_{ABC} = P_B \sum_{\Lambda} P_{A|\Lambda} P_{C\Lambda} = P_B P_{AC}.$$
(6.22)

The last line is equivalent to  $B \perp \{A, C\}$ , which contradicts Equation (6.18). Therefore, given the conditions in the theorem statement, Equation (6.19) cannot hold i.e., B is correlated with at least one of  $\{A, \Lambda\}$  and  $\{C, \Lambda\}$ . Using the exogeneity of B, this implies that B affects at least one of these sets. For compatibility with the space-time, this affects relation implies the following constraint on the space-time embedding where the accessible region of each ORV is taken to coincide with its inclusive future.

$$\overline{\mathcal{F}}(\mathcal{A}) \cap \overline{\mathcal{F}}(\Lambda) \subseteq \overline{\mathcal{F}}(\mathcal{B}) \quad \lor \quad \overline{\mathcal{F}}(\mathcal{C}) \cap \overline{\mathcal{F}}(\Lambda) \subseteq \overline{\mathcal{F}}(\mathcal{B}).$$
(6.23)

Now, note that Equation (6.18) implies that

$$A \not \perp C | B \tag{6.24}$$

This is because if we had the contrary, i.e.,  $A \perp C|B$ , then along with  $B \perp A$  this gives  $P_{A|BC} = P_{A|B} = P_A$ . Using this along with  $B \perp C$  gives  $P_{ABC} = P_{A|BC}P_{BC} = P_AP_BP_C$ . This in turn implies  $B \perp \{A, C\}$ , which contradicts the first condition of Equation (6.18).

Next, we show that  $\{B, \Lambda\}$  affects A and  $\{B, \Lambda\}$  affects C, again by contradiction. Suppose that  $\{B, \Lambda\}$  does not affect A, this is equivalent to  $\{B, \Lambda\} \perp A$  (i.e.,  $P_{A|B\Lambda} = P_A$ ) since Band  $\Lambda$  are exogenous. Along with Equation (6.17), this implies that  $P_{A|BC\Lambda} = P_{A|B\Lambda} = P_A$ . Then  $P_{ABC} = \sum_{\Lambda} P_A P_{BC\Lambda} = P_A P_{BC}$ , which contradicts Equation (6.18). Therefore we must have  $\{B, \Lambda\}$  affects A. Similarly, one can show that  $\{B, \Lambda\}$  affects C must also hold under the assumptions given in the theorem statement. Now, we have four cases depending on whether or not  $\Lambda$  affects A or C, which we enumerate below.

• Case 1: A affects neither A nor C In this case, both the outgoing arrows from  $\Lambda$  would be dashed (by Definition 6.3.5). In this case, any compatible embedding of the causal model in space-time must satisfy (c.f. Theorem 6.5.1)

$$\mathcal{A} \in \overline{\mathcal{F}}(\mathcal{B}) \cap \overline{\mathcal{F}}(\Lambda) \quad \wedge \quad \mathcal{C} \in \overline{\mathcal{F}}(\mathcal{B}) \cap \overline{\mathcal{F}}(\Lambda).$$
(6.25)

• Case 2:  $\Lambda$  affects A but not C In this case we have  $\Lambda \longrightarrow A$  and  $\Lambda \dashrightarrow C$ , and a necessary condition for compatibly embedding this causal model in space-time is

$$\mathcal{A} \in \overline{\mathcal{F}}(\Lambda) \quad \wedge \quad \mathcal{C} \in \overline{\mathcal{F}}(\mathcal{B}) \cap \overline{\mathcal{F}}(\Lambda). \tag{6.26}$$

• Case 3:  $\Lambda$  affects C but not A In this case we have  $\Lambda \longrightarrow C$  and  $\Lambda \dashrightarrow A$  and compatibly with the space-time requires that

$$\mathcal{A} \in \overline{\mathcal{F}}(\mathcal{B}) \cap \overline{\mathcal{F}}(\Lambda) \quad \wedge \quad \mathcal{C} \in \overline{\mathcal{F}}(\Lambda).$$
(6.27)

• Case 4:  $\Lambda$  affects A as well as C Here we have  $\Lambda \longrightarrow A$  and  $\Lambda \longrightarrow C$  and compatibility with the space-time necessitates

$$\mathcal{A} \in \overline{\mathcal{F}}(\Lambda) \quad \wedge \quad \mathcal{C} \in \overline{\mathcal{F}}(\Lambda). \tag{6.28}$$

In Cases 1-3, we can immediately see that at least one of  $\mathcal{A}$  and  $\mathcal{C}$  must be in the future of  $\mathcal{B}$  to restrict the causal model from signalling superluminally. In Case 4, this follows from Equations (6.23) and (6.24)— since  $\mathcal{A}$  and  $\mathcal{C}$  must be in the future of  $\Lambda$ , the joint futures of each of these ORVs and  $\Lambda$  coincides with the future of the ORV itself. Therefore, the space-time compatibility condition is always violated by the embedding of [105, 116] where  $\mathcal{A}$ ,  $\mathcal{B}$  and  $\mathcal{C}$  are pairwise space-like separated, and for this embedding, the jamming causal model considered here, with observed  $\Lambda$  leads to superluminal signalling.

# Part II

Multi-agent paradoxes

The 'paradox' is only a conflict between reality and your feeling of what reality 'ought to be'.

- Richard Feynman

# CHAPTER 7

## Multi-agent paradoxes in quantum theory

**P**ROCESSING empirically acquired data and making inferences about the world form a crucial part of the scientific method. For a consistent description of the world, these inferences should be based on a sound system of logic— simple reasoning principles applicable to general situations, on which there is common agreement. For example, we would like to make inferences such as "if I know that a holds, and I know that a implies b, then I know that b holds" independently of the nature of a and b. When considering scenarios with several rational agents, inferences may involve reasoning about each other's knowledge. In such cases, we often use logical primitives such as, "If I know that she knows a, and I know that she arrived at a using a set of rules that we commonly agreed upon, then I know a". Examples include games like poker, complex auctions, cryptographic scenarios, and logical hat puzzles, where we must process complex statements of the sort "I know that she knows that he does not know a" based on some logical primitives, and the common knowledge of the agents. A system of logic that provides these simple and intuitive rules for multi-agent reasoning is modal logic, which we outlined in Section 2.6.

On the other hand, when agents (such as ourselves) describe the world through physical theories, we would like to be able to model the agents also as physical systems of the theory, in order to develop a more complete understanding of the physical world. In particular, we would like our theory to model at least those parts of the agent that are responsible for storing and processing empirical data, such as their memory. When that theory is quantum mechanics, it turns out that these two desiderata (applying standard rules of logic to reason about each other's knowledge, and modelling agents' memories as physical systems) are incompatible with agents' empirical observations. This incompatibility (or "paradox") was first formalised by Frauchiger and Renner, in a thought experiment where agents who can measure each others memories (modelled as quantum systems) and reason about shared and individual knowledge may reach conclusions that contradict their own observations [85].

The FR paradox is originally presented in terms of a prepare and measure scenario where
parties exchange quantum states and thereby generate quantum correlations between each other. However it can be equivalently described by an entanglement-based scenario where the parties do not communicate but pre-share suitable quantum correlations, as illustrated in Figure 7.1. Here, we present the entanglement version of the paradox (Section 7.1) as this will be the most convenient form for generalising the paradox to post-quantum settings as we will do in Chapter 8. The Frauchiger-Renner thought experiment has fuelled an enormous volume of discussion and debate in the scientific community that is still ongoing. In Section 7.2, we aim to provide a brief overview of this debate and an outlook on this matter, which will also serve to motivate the results of the next chapter where we generalise the analysis beyond quantum theory.

## 7.1 An entanglement version of the Frauchiger-Renner thought experiment

#### 7.1.1 The assumptions

The main result of Frauchiger and Renner establishes the impossibility of a physical theory  $\mathcal{T}$  to simultaneously satisfy three assumptions, denoted by  $\mathbf{Q}$ ,  $\mathbf{C}$  and  $\mathbf{S}$ . In [152], it was pointed out that the FR argument involved an additional implicit assumption,  $\mathbf{U}$ . We present these four assumptions below as stated in [85] and [152], before we proceed to demonstrate their incompatibility with the entanglement version of FR's argument.

The first assumption  $\mathbf{Q}$  pertains to the validity of certain predictions of quantum mechanics. For FR's argument, the full framework of quantum theory need not be assumed, but only requires a weaker version of the Born rule that is applicable for making deterministic predictions.

**Assumption** (**Q**). A theory  $\mathcal{T}$  that satisfies **Q** allows any agent Alice to reason as follows. Suppose Alice is in the vicinity of a system S associated with the Hilbert space  $\mathscr{H}_S$ , and she knows that "The system S is in a state  $|\psi\rangle_S \in \mathscr{H}_S$  at time  $t_0$ ". Furthermore, suppose that Alice also knows that" the value x is obtained by a measurement of S w.r.t. the family  $\{\pi_x^{t_0}\}_{x\in X}$  of Heisenberg operators relative to time  $t_0$ , and the measurement is completed at time t". If  $\langle \psi | \pi_{\xi}^{t_0} | \psi \rangle = 1$  for some  $\xi \in X$ , then Alice can conclude that "I am certain at time  $t_0$  that  $x = \xi$  at time t".

The second assumption  $\mathbf{C}$ , originally referred to as "self-consistency", pertains to how agents reason about each other's knowledge.

**Assumption** (C). A theory  $\mathcal{T}$  that satisfies C allows any agent Alice to reason as follows. If Alice has established that "I am certain that an agent Bob, upon reasoning using theory  $\mathcal{T}$ , is certain that  $x = \xi$  at time t" then Alice can conclude "I am certain that  $x = \xi$  at time t." The third assumption,  $\mathbf{S}$  (originally referred to as "single-world") pertains to the intuition that an agent experiences a single outcome when they perform or witness a measurement.

**Assumption** (S). A theory  $\mathcal{T}$  that satisfies S disallows any agent Alice from making both the statements "I am certain at time  $t_0$  that  $x = \xi$  at time t." and "I am certain at time  $t_0$  that  $x \neq \xi$  at time t", where x is a value that can be observed at time  $t > t_0$ .

The fourth assumption U, requires a bit more explanation. Suppose an agent Alice, has a system S prepared in the state  $|\psi\rangle_S = \frac{1}{\sqrt{2}}(|0\rangle + |1\rangle)$ . Alice measures S in the Z basis,  $\{|0\rangle, |1\rangle\}$ , and records the outcome x of the measurement in her memory A (which was initialised to the state  $|0\rangle_A$ . From Alice's perspective, x is a uniformly distributed classical bit. Suppose also that (rather idealistically) that this measurement happens in Alice's lab which is a closed system consisting of the subsystems S and A that does not leak any information to the environment. Now, consider how an outside agent Bob who models S and A both as quantum systems would describe the measurement process. Since Alice's lab is a closed system, Bob knows nothing about Alice's measurement outcome x and would describe the evolution of systems A and S in the lab through a unitary map  $U_{AS}$ . Further, due to the perfect correlations between the measurement outcome and the memory state that records it, this unitary would be a coherent controlled NOT with the S as control and A as target. That is, Bob describes Alice's measurement through the evolution,

$$\frac{1}{\sqrt{2}} \left( |0\rangle_S + |1\rangle_S \right) \otimes |0\rangle_A \xrightarrow{U_{AS}} \frac{1}{\sqrt{2}} \left( |0\rangle_S |0\rangle_A + |1\rangle_S |1\rangle_A \right).$$
(7.1)

Hence, Bob would see Alice's memory A to be entangled with her measured system S. The FR thought experiment assumes that outside agents such as Bob, who model an inside agent such as Alice as a quantum system would model measurements performed by the latter in through such reversible evolutions. This was noted in [152], and is encapsulated in the assumption **U** proposed by therein.

**Assumption** (U). A theory  $\mathcal{T}$  that satisfies U allows any agent Bob to model measurements performed by any other agent Alice as reversible evolutions in Alice's lab — for example, a unitary evolution  $U_{AS}$  of the joint state of Alice's memory A and the system S measured by her.

#### 7.1.2 The paradox: incompatibility of the assumptions

One may have already spotted a tension between the assumptions  $\mathbf{S}$  and  $\mathbf{U}$  which describe the "inside" and "outside" perspectives of a quantum measurement i.e., the subjective experience of a single classical outcome for Alice, vs the entangled superposition state of Alice and her system as seen by Bob. This is precisely the tension that Wigner originally illustrated in his thought experiment. In fact, Wigner's thought experiment essentially corresponds to the two agent scenario described in the previous section (paragraph preceding assumption  $\mathbf{U}$ ), where the inside observer Alice plays the role of "Wigner's friend" and the outside observer Bob who



(a) Inside perspective

(b) Outside perspective

Figure 7.1: An entanglement-based version of the Frauchiger-Renner setting [85] from different perspectives. Alice and Bob (inside agents) share a Hardy state  $|\Psi\rangle_{PR} = (|00\rangle + |10\rangle + |11\rangle)/\sqrt{3}$ , measure each their qubit (P and R respectively) and update their memories A and B accordingly. Their labs are contained inside the labs of the outside observers Ursula and Wigner, who can measure the systems APand RB respectively. The paradox arises when one tries to combine the inside and outside perspectives of quantum measurements on an entangled system into a single perspective. (a) From their viewpoints, Alice and Bob measure their halves of  $|\Psi\rangle_{PR}$ in the Z basis  $\{|0\rangle, |1\rangle\}$  to obtain the outcomes a and b. They then perform a classical CNOT (i.e., classical copy) to copy their classical outcome into their memories A and B both initialised to  $|0\rangle$ . (b) Ursula and Wigner perceive Alice and Bob's memory updates as implementing quantum CNOTs on A controlled by P and B controlled by R respectively. The resultant joint state is  $|\Psi\rangle_{APRB} = (|0000\rangle + |1100\rangle + |1111\rangle)/\sqrt{3}$ . Hence, they see quantum correlations between the systems and memories of the inside agents. Later, they measure the joint systems AP and RB in the "X basis"  $\{|ok\rangle = (|00\rangle - |11\rangle)/\sqrt{2}, |fail\rangle = (|00\rangle + |11\rangle)/\sqrt{2}\}$  to obtain the outcomes u and w respectively. If they obtain u = w = ok, the agents can reason about each others' knowledge to arrive at the paradoxical chain of statements  $u = w = ok \Rightarrow b = 1 \Rightarrow a = 1 \Rightarrow w = fail$ . In Chapter 8, we extend this scenario to box world where Alice and Bob share a PR box instead of the Hardy state and find a suitable memory update operation and measurements for the parties such that a stronger version of the paradox is recovered, independently of the outcomes obtained.<sup>1</sup>

describes Alice as a quantum system plays the role of Wigner. By including the assumptions  $\mathbf{Q}$  and  $\mathbf{C}$  which introduce the element of reasoning about the knowledge of other agents, the FR experiment elevates this apparent tension to an explicit mathematical contradiction between the four assumptions. The following entanglement version of the thought experiment is entirely equivalent to the original prepare and measure version, but simplifies the description.

<sup>&</sup>lt;sup>1</sup>This figure is taken from our paper [208] which is joint work with Nuriya Nurgalieva and Lídia del Rio, and the credits for this pretty illustration goes to Nuriya.

**Experimental setup.** The experiment involves four agents, whom we call Alice, Bob, Ursula and Wigner<sup>2</sup>. The experiment starts with Alice and Bob sharing the following bipartite state

$$|\psi\rangle_{PR} = \frac{1}{\sqrt{3}} \Big( |0\rangle_P |0\rangle_R + |1\rangle_P |0\rangle_R + |1\rangle_P |1\rangle_R \Big), \tag{7.2}$$

where P and R denote the subsystems held by Alice and Bob respectively (for reasons that will become more apparent in the next chapter). This state is known as a *Hardy state* due to its particular relevance in Hardy's paradox [108, 109]. Let A and B denote the subsystems that correspond to Alice's and Bob's memory respectively, where they store the outcome of any measurement that they may perform. Suppose that Alice's lab is located inside the lab of another agent, Ursula who can perform joint measurements on Alice's system P and her memory A. Similarly, let Bob's lab be located inside Wigner's lab, such that Wigner can perform joint measurements on Bob's system R and his memory B. Alice's and Bob's labs are isolated such that no information about their measurement outcomes leaks out.

#### **The protocol:** The protocol is the following:

- t=1 Alice measures her half (P) of joint state  $|\psi\rangle_{PR}$ , in the  $\{|0\rangle, |1\rangle\}$  (i.e., Z or computational) basis, and stores the outcome a in her memory A.
- **t=2** Bob measures his half (*R*) of joint state  $|\psi\rangle_{PR}$ , also in the *Z* basis, and stores the outcome *b* in his memory *B*. By the assumption **U**, the final joint state of Alice and Bob's systems and memories after their measurements at times t = 1 and t = 2, as described from the outside would be

$$|\psi\rangle_{PARB} = \frac{1}{\sqrt{3}} \Big( |0\rangle_P |0\rangle_A |0\rangle_R |0\rangle_B + |1\rangle_P |1\rangle_A |0\rangle_R |0\rangle_B + |1\rangle_P |1\rangle_A |1\rangle_R |1\rangle_B \Big)$$
(7.3)

t=3 Ursula jointly measures the systems P and A in Alice's lab, in the  $\{|ok\rangle, |fail\rangle\}$  basis defined below and obtains the outcome u.

$$|ok\rangle_{PA} \coloneqq \frac{1}{\sqrt{2}} \left( |0\rangle_{P}|0\rangle_{A} - |1\rangle_{P}|1\rangle_{A} \right)$$
  
$$|fail\rangle_{PA} \coloneqq \frac{1}{\sqrt{2}} \left( |0\rangle_{P}|0\rangle_{A} + |1\rangle_{P}|1\rangle_{A} \right)$$
  
(7.4)

t=4 Wigner jointly measures the systems R and B in Bob's lab, also in the  $\{|ok\rangle, |fail\rangle\}$  basis and obtains the outcome w.

$$|ok\rangle_{RB} \coloneqq \frac{1}{\sqrt{2}} \Big( |0\rangle_R |0\rangle_B - |1\rangle_R |1\rangle_B \Big)$$
  
$$fail\rangle_{RB} \coloneqq \frac{1}{\sqrt{2}} \Big( |0\rangle_R |0\rangle_B + |1\rangle)R |1\rangle_B \Big)$$
(7.5)

<sup>&</sup>lt;sup>2</sup>In accordance with the naming convention of [152, 208].

t=5 Ursula and Wigner compare the outcomes of their measurements. If they were both "ok", they halt the experiment. Otherwise, they reset the timer and all systems to the initial conditions, and repeat the experiment. Note that there is a non-zero probability for both Ursula and Wigner to obtain the outcome "ok" when measuring the subsystems PA and RB in the state  $|\psi\rangle_{PARB}$  of Equation (7.3) i.e.,

$$P(u = w = ``ok''|\psi_{PARB}) = |(\langle ok|_{PA} \otimes \langle ok|_{RB}) \cdot |\psi\rangle_{PARB}|^2 = \frac{1}{12}.$$
(7.6)

The measurement basis for each agent mentioned above are agreed on beforehand and the agents do not communicate once the experiment begins, except for the communication between Ursula and Wigner at t = 5. We now discuss how the agents reason about each other's knowledge to arrive at the contradiction, given that the experiment halted i.e., post-selecting on the condition that

$$u = w = ok$$

#### Agents' reasoning:

1. Ursula reasons about Bob: Rewriting the joint state  $|\psi\rangle_{PARB}$  as follows

$$\begin{aligned} |\psi\rangle_{PARB} &= \sqrt{\frac{2}{3}} |fail\rangle_{PA} |0\rangle_{R} |0\rangle_{B} + \frac{1}{\sqrt{3}} |1\rangle_{P} |1\rangle_{A} |1\rangle_{R} |1\rangle_{B}, \\ &= \sqrt{\frac{2}{3}} |fail\rangle_{PA} |0\rangle_{R} |0\rangle_{B} + \frac{1}{\sqrt{6}} |fail\rangle_{PA} |1\rangle_{R} |1\rangle_{B} - \frac{1}{\sqrt{6}} |ok\rangle_{PA} |1\rangle_{R} |1\rangle_{B}, \end{aligned}$$
(7.7)

Ursula at t = 3 reasons (using **Q**) on obtaining the outcome u = ok, that she knows with certainty that Bob must have obtained the outcome b = 1 at t = 2. This is because Ursula knows that her (normalised) post-measurement state on obtaining the outcome u = ok would be  $|ok\rangle_{PA}|1\rangle_{R}|1\rangle_{B}$ , and there is unit probability of Bob obtaining b = 1 for his measurement on B given this outcome. Thus Ursula (at t = 3) knows that

$$u = ok \Rightarrow b = 1$$

2. Bob reasons about Alice: From the original form of the joint state (7.3), Bob reasons using  $\mathbf{Q}$ , on obtaining the outcome b = 1 at time t = 2 that he knows with certainty that Alice would have obtained the outcome a = 1 at t = 1 i.e., Bob (at t = 2) knows that

$$b = 1 \Rightarrow a = 1.$$

3. Alice reasons about Wigner: Rewriting the joint state as

$$|\psi\rangle_{PARB} = \frac{1}{\sqrt{3}}|0\rangle_P|0\rangle_A|0\rangle_R|0\rangle_B + \sqrt{\frac{2}{3}}|1\rangle_P|1\rangle_A|fail\rangle_{RB},\tag{7.8}$$

Alice, on obtaining the outcome a = 1 at t = 1, uses **Q** to reason about what Wigner would observe at t = 4 when he measures Bob's lab. Similar to previous steps, she would conclude with certainty that Wigner observes w = fail i.e., Alice (at t = 1) knows that

$$a = 1 \Rightarrow w = fail$$

Note that in each of the steps above, **S** is implicitly used. These statements reflect the knowledge and reasoning of different agents, and are combined using the assumption **C** to give the following paradoxical chain of statements when u = w = ok is observed at t = 4.

$$u = w = ok \Rightarrow b = 1 \Rightarrow a = 1 \Rightarrow w = fail$$
(7.9)

In other words, whenever the experiment halts with u = w = ok, the agents can make deterministic statements about each other's reasoning and measurement outcomes, and thereby conclude that Alice had predicted w = fail with certainty. This is in conflict with the assumption **S** which forbids w from taking both the values with certainty.

In the modal logic language, the assumption  $\mathbf{C}$  can be seen as encoding the distribution axiom (Axiom 1) i.e., "If an agent knows a statement  $\phi$  and also that  $\phi \Rightarrow \psi$ , then they know  $\psi$ ", along with the concept of trust (Definition 2.6.3) i.e., an agent A trusts another agent B if and only if "A knows that B knows  $\phi$  implies that A knows  $\phi$ ". In [152], it is shown that the FR paradox can be reformulated as: the Kripke structure satisfying the knowledge axioms of modal logic described in Section 2.6 is incompatible with agents who apply quantum theory to model each other and reason using that logical system. They show that this is the case even if the truth axiom (Axiom 3) is replaced with the weaker notion of a trust structure (Definition 2.6.3). We refer the reader to [152] for a detailed analysis of this paradox in terms of the axioms of modal logic and the trust relations at different times. These ideas will also become clearer in Chapter 8 where we derive an analogous paradox in box world, a generalised probabilistic theory (see Section 2.4 for an overview of GPTs) by generalising the quantum theory dependent assumptions of FR to the general setting, in the language of modal logic. We now proceed to discuss some of the implications of FR's result and its relation to other no-go theorems.

**Remark 7.1.1.** It is important to note here, also as stressed in the original paper [85] that an "agent" need not necessarily correspond to a conscious observer in such thought experiments. An agent simply refers to any entity capable of performing measurements on other systems and applying simple deductive principles to process empirical information. For example, a small quantum computer would in principle fulfil the desiderata for agency in such settings. Hence, FR-type thought experiments could possibly be implementable in the near future, with progress in scalable quantum computing, even though they cannot be implemented with current technology.

## 7.2 Discussion

#### 7.2.1 Intuition and relation to Hardy's paradox

The intuition behind the FR paradox described above comes from Bell's theorem, combined with the unitary description of the measurement from the outside perspective (Equation (7.1)). The entanglement of the shared Hardy state  $|\psi\rangle_{PR}$  implies the inability to assign simultaneous values to the outcomes corresponding to different measurement settings [109], the FR experiment corresponds to a situation where all the choices of measurements are actually implemented in a single run of the experiment since the unitarity of the measurement map (7.1) allows the outside agents to reverse the effect of the inside agents' measurement before performing their measurement. This also means that the final joint state  $|\psi\rangle_{PABB}$  that Ursula and Wigner measure contains the initial correlations of  $|\psi\rangle_{PR}$  as evident from Equations (7.2) and (7.3). This leads to the paradoxical situation that the outcomes of all four parties cannot admit simultaneous value assignments. FR's paradox can be viewed as a manifestation of Hardy's proof of Bell's theorem [109] which is not based on the violation of Bell inequalities, but on deriving a logical contradiction between the prediction of quantum mechanics for certain entangled states (such as (7.2)) and the assumption of local hidden variables for the outcomes of possible measurements. Hardy derives such a contradiction for almost all bipartite entangled states in [109].<sup>3</sup> It can be shown that in the 4 agent setting of FR (Figure 7.1), replacing the shared state  $|\psi\rangle_{PR}$  by any one state from the class of Hardy states, and modifying the definition of the  $\{|ok\rangle, |fail\rangle\}$  basis accordingly (also as proposed in [109]), one can elevate Hardy's argument for the outcomes of potentially unperformed measurements in the regular bipartite Bell scenario to the outcomes of the four measurements (one per agent) actually performed by in an FR-type setup. Hence the FR paradox is closely related to Hardy's paradox [108], as was also pointed out in [174]. We leave the formal derivation of this and a more general characterisation of the states and measurements required for deriving FR-type paradoxes in N-agent experiments to future work.

### 7.2.2 Relation to other no-go theorems

There are other no-go theorems in a similar vein as FR, that are based on an extension of Wigner's thought experiment. All these theorems illustrate the incompatibility between an objective assignment of values to the measurement outcomes of all agents in scenarios where quantum predictions are extended to the level of agents. The most notable of these is Brukner's no-go theorem [36] which states that the following four assumptions are incompatible: 1. quantum theory is universally valid (also when applied to observers), 2. locality i.e., the setting of one party doesn't affect the outcome of the other, 3. freedom of choice 4. the ability to jointly assign truth values to observed outcomes of all the observers. Brukner derives the CHSH in-

 $<sup>^{3}</sup>$ This class excludes the maximally entangled Bell states which are too symmetric to derive a logical contradiction akin to the chain (7.9).

equalities from these assumptions 2.-4. and the well known violation of these Bell inequalities in quantum theory demonstrates the aforementioned incompatibility of the assumptions 1.-4. Assumption 1 roughly corresponds to the assumptions  $\mathbf{Q}$  and  $\mathbf{U}$  of FR, with the difference being that the former would need to assume full validity of the Born rule rather than the weaker, deterministic version given by  $\mathbf{Q}$ . Assumption 4 is similar in motivation as the assumption  $\mathbf{C}$ of FR, with the difference being that C relates to combining deterministic statements made by agents to assign values to all their outcomes. In this sense,  $\mathbf{C}$  is a stronger assumption than 4. which does not require the value assignments to be deterministic. Assumptions 2 and 3 are different from the FR setting, where the measurement choice for each party is fixed. One could say that FR's no-go theorem is to Brukner's what Hardy's is to Bell's. The former, in both cases are based on logical arguments, that can be made without reference to inequalities but require certain probabilities to be deterministic i.e., have 0 or 1 values. For example, in the FR case, we require P(b=0|u=ok) = P(a=0|b=1) = P(w=ok|a=1) = 0 in order to make deterministic logical inferences, which we cannot have for measurements on the maximally entangled Bell states (as per Hardy's argument [109]). However the latter could still be certified as incompatible with Brukner's assumptions since it violates CHSH inequalities.

[30] proposed a modified version of Brukner's theorem, noting a limitation of the latter. They pointed out that the Bell inequalities derived through assumptions 2.-4., can also be derived using the assumptions 3. (free choice) and Kochen-Spekker non-contextuality [127], without reference to the objectivity of the agents' observations, which is supposed to be the subject of such theorems. They show that Bell inequality violations are in general, not sufficient to demonstrate a contradiction between quantum theory and the existence of observer independent or objective facts about the world, and derive a new set of inequalities from weaker assumptions that are indeed sufficient for this purpose. The difference is that, Brukner's version includes hypothetical measurements that may not have been actually performed in the considered experimental run, while [30] only considers measurement outcomes that were actually observed by the agents, and tests for the objectivity (i.e., possibility of assigning universal truth values) of the observed outcomes. The exact connections between FR's paradox and these no-go theorems is not known. But these examples suggest that contextuality would be a necessary but not sufficient condition to derive FR-type paradoxes. We discuss the relation to contextuality in Chapter 8, Section 8.5.4 and leave a characterisation of the set of all contextual scenarios leading to FR-type paradoxes to future work. Such a characterisation would require a general definition of an FR-type logical paradox which is currently lacking, and to the best of our knowledge the only examples of such logical multi-agent paradoxes are the original FR paradox [85] and our box world version [208] (Chapter 8). Generalising the study of such paradoxes to arbitrary theories as attempted in [208] (Chapter 8) provides a starting point for addressing such questions.

#### 7.2.3 Interpretations of quantum theory

When treating agents as physical systems, the divide between the "observer" and the "observed" becomes subjective— we always put ourselves in the former category, but may be put in the latter category by another observer. Wigner [220] was among the first to point out through his

famous thought experiment, that this leads to conceptual issues in the quantum case, and this problem is sometimes colloquially referred to as the *shifty split*. FR's extension of the Wigner's friend scenario formalises this conceptual argument as a no-go theorem, thereby providing a concrete footing for the longstanding debates regarding the measurement problem and in turn, interpretations of quantum theory, which represent the different stances that one can take on the measurement problem. We begin this discussion by noting the importance of differentiating between objections against the FR result vs objections against specific assumptions used in deriving the result. The result itself, demonstrates an incompatibility between a certain set of assumptions and subsumes the latter type of objections— it leaves open the choice of which assumption(s) among Q, C, S and U must be dropped to resolve the paradox, it is agnostic to the choice itself. This allows for a classification of the different interpretations of quantum theory based on the choice they make, and also indicates that any interpretation that satisfies all four assumptions should be ruled out. It appears that the main interpretation(s) under threat of being ruled out are those presented in standard textbooks which albeit ambiguous, seem to be trying to have the cake and eat it too [174]. Possible resolutions to the apparent paradox involve dropping one or more of the assumptions.

**Dropping Q and/or U:** Objections against FR such as "you cannot put agents in superpositions", or "the outside description of the measurement (7.1) should lead to a mixed state and not a pure, entangled state" do not refute the validity of the theorem but that of the assumptions U and/or Q about extending quantum theory to agents. This is one possible resolution allowed by the theorem.<sup>4</sup> Further, in [196], Anthony Sudbery analyses Bohmian mechanics [72, 27, 78] in the light of FR's result, arguing that the interpretation satisfies all assumptions, except possibly U regarding the particular evolution of the labs. This would imply that Bohmian mechanics does not agree with the predictions of quantum theory at all scales, as is commonly believed. Veronika Baumann and Stefan Wolf [19] have provided an interesting analysis of the thought experiment in the relative state formalism [81, 219], considering different ways of describing the evolution of the labs and showing that these can lead to different predictions. They also compare other interpretations in this regard and show that the relative state formalism admitting unitary and universal quantum description deviates from the standard Born rule.

**Dropping C:** Certain versions of the Copenhagen interpretation [113, 28] also turn out to problematic for similar reasons as the textbook approach, and Matthew Leifer argues (for example, in this talk) that only a *perspectival* version of the interpretation, which gives up the assumption  $\mathbf{C}$  would survive. QBism [91] and relational quantum mechanics are also perspectival in this sense and propose a similar resolution— measurement results are real only from the

<sup>&</sup>lt;sup>4</sup>Similar misinterpretations can occur with Bell's theorem. For example, if we interpret Bell's theorem as telling us that there is fundamental uncertainty in quantum theory, that cannot be explained away by conditioning on classical information. A realist would argue that this is not true, since we can have fine-tuned, classical explanations i.e., non-local HV theories. Stating Bell's theorem as a no-go result, this argument is subsumed: you can either drop "realism" or "locality", you have a choice, the theorem doesn't tell you which one to pick.

subjective perspective of specific agents, and not objective properties of the world.<sup>5</sup> Further, a theory that satisfies all assumptions except  $\mathbf{C}$ , would call for a revision of our understanding of causal and logical inference. The latter, since dropping  $\mathbf{C}$  implies a break down of standard rules of classical modal logic [152]. The former, due to the following reason. As we have seen in Part I of this thesis, standard approaches to causal inference depict causal structures as having certain observed and certain unobserved nodes. The former are classical variables corresponding to the observed measurement statistics. In theories where agents as modelled as quantum systems according to  $\mathbf{Q}$  and  $\mathbf{U}$ , whether measurement outcomes are seen as classical variables or as entangled quantum subsystems is subjective, which challenges existing models for causal inference (both in classical and quantum theory). This suggests that causal structure itself may be subjective in such scenarios. It would be interesting to study causal and logical inference in such scenarios and understand how these differ from situations where agents are modelled as classical systems.

**Dropping S:** This appears to be an intuitive assumption but a violation of this is not necessarily in conflict with the quantum formalism. It is argued that interpretations such as manyworlds (and its numerous variations) [76, 74] as well as the relative state formalism mentioned earlier, do not satisfy this assumption [85]. In the former, measurement is seen as leading to a branching into different possible worlds such that every possible outcome of the measurement occurs in a corresponding world. Whether many-worlds type interpretations satisfy the other assumptions, may depend on the particular variation being considered, and how the branching is defined therein.

**Other assumptions:** Other implicit assumptions of the FR experiment have been pointed out [197, 152]. Some of these assumptions relate to idealizations such as the ability to perfectly prepare pure states and perform perfect projective measurements. While it is true that a relaxation of such assumptions may no longer result in a deterministic, logical paradox, these would still result in a conceptual problem regarding the incompatibility of universal quantum theory and objectivity of measurement results, akin to those pointed out by [36, 30]. Further, one can argue that most interpretations would allow for such idealizations to be made, which are common place in theoretical studies. It would be interesting to analyse whether some of the other implicit assumptions in the result pointed out in [197] are no longer needed in the entanglement version of the experiment presented here. This discussion by no means does full justice to the massive volume of discussions and literature generated by the FR result. We refer the reader to the original paper [85] as well as [152] for more in-depth discussions on interpretations.

<sup>&</sup>lt;sup>5</sup>However, as pointed out in [152, 85], it may be difficult to fully analyse such multi-agent experiments in QBism which tends to limit its focus on the experience and actions of single agents.

### 7.2.4 Outlook

In conclusion, we draw yet another analogy with Bell's theorem [22]. The theorem has resulted in significant conceptual and practical advancement by ruling out certain classical descriptions of the world (i.e., local hidden variables) due to their incompatibility with our observations, which agree with quantum theory. It has also faced several objections and criticism over the years, and generated several alternate derivations/versions [57, 56, 109, 102, 224] which reveal different facets of quantum theory that deviate from our understanding of the classical world. In a similar manner, theorems such as those of Frauchiger and Renner have deep foundational implications for the nature of objectivity, agents' experience, logical reasoning and causal inference. Critically examining the assumptions behind such theorems and characterising general scenarios where such incompatibilities can arise are likely to reveal important connections between these fundamental concepts and the role that quantum theory plays in them. Finally, as previously noted, FR-type thought experiments could possibly be implemented in the near future depending on the advancements in scalable quantum computing, and don't necessarily require the agents to be conscious observers [85]. This would in principle allow for the interpretations of quantum theory to be subject to experimental tests. The caveat being that a clear cut answer would require a loophole-free implementation. This would be significantly more difficult, as we have seen in the case of Bell's theorem where the first experimental test [86] was realised within a decade but the first loophole free tests [98, 114, 189] were only realised more than 5 decades after the original theorem [22] was proposed.

# CHAPTER 8

## Multi-agent paradoxes beyond quantum theory

THE Frauchiger-Renner thought experiment [85] reveals an incompatibility between extending quantum theory to reasoning agents and certain simple rules of logical deduction [152], which we have discussed in Chapter 7. Our goal here is to understand whether this incompatibility is a peculiar feature of quantum theory, or whether modelling reasoning agents using other physical theories can also lead to such contradictions. Previous works have analysed such multi-agent paradoxes only in quantum theory, and a theory-independent analysis could possibly reveal more fundamental aspects these examples. For example, is the reversibility (in the quantum case, unitarity) of the global measurement transformation a necessary property for such paradoxes or the entanglement of the shared state (in the quantum case, the Hardy state), or both? Generalising this study beyond the quantum setting would not only help us identify the features of physical theories responsible for such paradoxes, but may also shed some light on the structure of logic in non-classical theories. Here, we investigate this question within the landscape of generalized probabilistic theories [110, 16]. This chapter is based on our paper [208], which is coauthored by Nuriya Nurgalieva and Lídia del Rio.

### 8.1 Summary of contributions: in words and in poetry

In Section 8.2, we generalize the Frauchiger-Renner conditions using the laguage of modal logic, so that they can be applied to any physical theory. In particular, in Section 8.2.4, we introduce a way to describe an agent's measurement from the perspective of other agents in generalised probabilistic theories (GPT). Finally, in Section 8.4 we derive a logical inconsistency akin to one found in [85], using a setup where agents share a PR box, a maximally non-local resource in box world (a particular GPT). The paradox found is stronger than the quantum one, in

the sense that it does not rely on post-selection.<sup>1</sup>. Another important point brought to light by our version of the paradox is that the reversibility of the measurement map (akin to the assumption **U** in the quantum case) is not necessary for deriving such paradoxes in general. In Section 8.5, we provide a detailed discussion which includes a comparison of our results with the quantum case (Section 8.5.1), its implications for measurements in box world (Section 8.5.2), relationships between multi-agent paradoxes and contextuality (Section 8.5.4), and the plentiful scope for future work. Based on the knowledge of the summary sections of previous chapters, the reader might logically reason and thereby consider possible a world *s* where the present chapter also contains a poetic summary. This is indeed that world.

We talk, so we reason. We reason about what we know, We reason about what others know And if we trust them, make their knowledge our own.

We learn, so we store. We store our knowledge in a part of our memory. We model that memory by a physical theory. But if that theory is quantum, this leads to an inconsistency [85].

We learn, so we wonder. Which theories lead to such inconsistensies, Between reasoning agents and their memories modelled "physically"? Proposing a mathematical map for update of memories, An example we find in box world, a GPT, where using a PR box, agents find a stronger paradox.

We answer questions, so we question more. What properties of theories lead to these paradoxes galore? Must be contextuality, which forms of it, we are not yet sure That, in future work we will explore!

## 8.2 Generalized reasoning, memories and measurements

Here we generalize the assumptions underlying the Frauchiger-Renner result to general physical theories. The conditions can be instantiated by each specific theory. This includes but is not limited to theories framed in the approach of generalized probabilistic theories (GPTs) [110]. In some theories, like quantum mechanics and box world (a GPT), we will find these four conditions to be incompatible, by finding a direct contradiction in examples like the Frauchiger-Renner

<sup>&</sup>lt;sup>1</sup>The joint state and the probability distributions of the original Frauchiger-Renner paradox are akin to those of Hardy's paradox [109]. For a comparison of Hardy's paradox and PR box and why the latter allows for a contradiction without post-selection, see [4]. An entanglement version of the Frauchiger-Renner experiment and it's relation to our extension is explained in Figure 7.1.

experiment or the PR-box experiment described in Section 8.4. In other theories (like classical mechanics and Spekkens' toy theory [195]) these four conditions may be compatible. A complete characterization of theories where one can find these paradoxes is the subject of future work.

#### 8.2.1 Reasoning about knowledge

This condition is theory-independent. It tells us that rational agents can reason about each other's knowledge in the usual way. This is formalized by a weaker version of *epistemic modal logic*, which we explain in the following. For the full derivation of the form used here see [152], and for an overview of the modal logic framework, see Section 2.6.

We start with a simple example. The goal of modal logic is to allow us to operate with chained statements like "Alice knows that Bob knows that Eve doesn't know the secret key k, and Alice further knows that k = 1," which can be expressed as

$$K_A \left[ (K_B \neg K_E k) \land k = 1 \right],$$

where the operators  $K_i$  stand for "agent *i* knows." If in addition Alice trusts Bob to be a rational, reliable agent, she can deduce from the statement "I know that Bob knows that Eve doesn't know the key" that "I know that Eve doesn't know the key", and forget about the source of information (Bob). This is expressed as

$$K_A(K_B \neg K_E k) \implies K_A \neg K_E k.$$

We should also allow Alice to make deductions of the type "since Eve does not know the secret key, and one would need to know the key in order to recover the encrypted message m, I conclude that Eve cannot know the secret message," which can be encoded as

$$K_A[(\neg K_E \ k) \land (K_i \ m \implies K_i \ k, \ \forall \ i)] \implies K_A \neg K_E \ m.$$

Generalizing from this example, this gives us the following structure.

**Definition 8.2.1** (Reasoning agents). An experimental setup with multiple agents  $A_1, \ldots, A_N$  can be described by knowledge operators  $K_1, \ldots, K_N$  and statements  $\phi \in \Phi$ , such that  $K_i \phi$  denotes "agent  $A_i$  knows  $\phi$ ." It should allow agents to make *deductions*<sup>2</sup>, that is

$$K_i[\phi \land (\phi \implies \psi)] \implies K_i \ \psi.$$

Furthermore, each experimental setup defines a *trust relation* between agents (Definition 2.6.3): we say that an agent  $A_i$  trusts another agent  $A_j$  (and denote it by  $A_j \rightsquigarrow A_j$ ) iff for all statements  $\phi$ , we have

$$K_i(K_j \phi) \Longrightarrow K_i \phi.$$

<sup>&</sup>lt;sup>2</sup>Note that this is the distribution axiom, (Axiom 1).

**Remark 8.2.1** (One human  $\neq$  one agent). Note that in general 'one human  $\neq$  one agent.' For example, consider a setting where we know that Alice's memory will be tampered with at time  $\tau$  (much like the original Frauchiger-Renner experiment, or the sleeping beauty paradox [80]). We can define two different agents  $A_{t<\tau}$  and  $A_{t>\tau}$  to represent Alice before and after the tampering— and then for example Bob could trust pre-tampering (but not post-tampering) Alice,  $A_{t<\tau} \sim B$ .

**Remark 8.2.2** (Complexity cost of reasoning). Note that in general, even the most rational physical agents may be limited by bounded processing power and memory capacity, and will not be able to chain an indefinite number of deductions within sensible time scales. That is, these axioms for reasoning are an idealization of absolutely rational agents with unbounded processing power (see [1] for an overview of this and related issues). If we would like modal logic to apply to realistic, physical agents, we might account for a cost (in time, or in memory) of each logical deduction, and require it to stay below a given threshold, much like a resource theory for complexity. However, in the examples of this chapter, agents only need to make a handful of logical deductions, and these complexity concerns do not play a significant role.

### 8.2.2 Physical theories as common knowledge

This condition is to be instantiated by each physical theory, and is the way that we incorporate the physical theory into the reasoning framework used by agents in a given setting. If all agents use the same theory to model the operational experiment (like quantum mechanics, special relativity, classical statistical physics, or box world), this is included in the *common knowledge* shared by the agents. For example, in the case of quantum theory, we have that "everyone knows that the probability of obtaining outcome  $|x\rangle$  when measuring a state  $|\psi\rangle$  is given by  $|\langle x|\psi\rangle|^2$ , and everyone knows that everyone knows this, and so on."

**Definition 8.2.2** (Common knowledge). We model a physical theory shared by all agents  $\{A_i\}_i$  in a given setting as a set  $\mathbb{T}$  of statements that are common knowledge shared by all agents, i.e.

$$\phi \in \mathbb{T} \iff (\{K_i\}_i)^n \phi, \quad \forall \ n \in \mathbb{N},$$

where  $(\{K_i\}_i)^n$  is the set of all possible sequences of n operators picked from  $\{K_i\}_i$ . For example,  $(K_1 K_5 K_1 K_2) \in (\{K_i\}_i)^4$  and stands for "agent  $A_1$  knows that agent  $A_5$  knows that agent  $A_1$  knows that agent  $A_2$  knows."

Note that the set  $\mathbb{T}$  of common knowledge may include statements about the settings of the experiment, as well as complex derivations.<sup>3</sup> To find our paradoxical contradiction, we may only need a very weak version of a full physical theory: for example Frauchiger and Renner only require a possibilistic version of the Born rule, which tells us whether an outcome will be observed with certainty [85]. This will also be the case in box world.

<sup>&</sup>lt;sup>3</sup>One can also alternatively model a physical theory as a subset  $\mathbb{T}_P$  of the set  $\mathbb{T}$  of common knowledge,  $\mathbb{T}_P \subseteq \mathbb{T}$ , in the case when details of experimental setup are not relevant to the theoretical formalism.

#### 8.2.3 Agents as physical systems

In operational experiments, a reasoning agent can make statements about systems that she studies; consequently, the theory used by the agent must be able to produce a description or a model of such a system, namely, in terms of a set of states. For example, in quantum theory a two-state quantum system with a ground state  $|0\rangle$  and an excited state  $|1\rangle$  (*qubit*) can be fully described by a set of states  $\{|\psi\rangle\}$  in the Hilbert space  $\mathbb{C}^2$ , where  $|\psi\rangle = \alpha|0\rangle + \beta|1\rangle$  with  $\alpha, \beta \in \mathbb{C}$  and  $|\alpha|^2 + |\beta|^2 = 1$ . Other examples of theories and respective descriptions of states of systems include: GPTs, where e.g. a generalised bit (*gbit*) is a system completely characterized by two binary measurements which can be performed on it [16] (a review of GPTs can be found in Section 2.4); algebraic quantum mechanics, with states defined as linear functionals  $\rho: A \to \mathbb{C}$ , where A is a C\*-algebra [211]; or resource theories with some state space  $\Omega$ , and epistemically defined subsystems [73, 130].

**Definition 8.2.3** (Systems). A "physical system" (or simply "system") is anything that can be an object of a physical study<sup>4</sup>. A system can be characterized, according to the theory  $\mathbb{T}$ , by a set of possible states  $\mathscr{P}_S$ . In addition, a system is associated with a set of allowed operations,  $\mathcal{O}_S: \mathscr{P}_S \mapsto \mathscr{P}_S$  on these states.

**Definition 8.2.4** (Parallel composition). For any two systems  $S_1$  and  $S_2$ , the union of the two defines a new system  $S_1 \cup S_2$  or simply  $S_1S_2$ . The operator  $\parallel$  denotes parallel composition of states and operations such that  $P_{S_1} \parallel P_{S_2} \in \mathscr{P}_{S_1S_2}$  whenever  $P_{S_1} \in \mathscr{P}_{S_1}$  and  $P_{S_2} \in \mathscr{P}_{S_2}$  and similarly,  $O_{S_1} \parallel O_{S_2} \in \mathcal{O}_{S_1S_2}$  whenever  $O_{S_1} \in \mathcal{O}_{S_1}$  and  $O_{S_2} \in \mathcal{O}_{S_2}$ . In other words, the state  $P_{S_1} \parallel P_{S_2}$  of  $S_1S_2$  can be prepared by simply preparing the states  $P_{S_1}$  and  $P_{S_2}$  of the individual systems  $S_1$  and  $S_2$  and the operation  $O_{S_1} \parallel O_{S_2}$  can be implemented by locally performing the operations  $O_{S_1}$  and  $O_{S_2}$  on the individual systems.

We assume no further structure to this operator. Note also that we do not assume that a given composite system can be split into/described in terms of its parts even though combining individual systems in this manner allows us to define certain states of composite systems<sup>5</sup>. Now we introduce agents into the picture.

**Definition 8.2.5** (Agents). A physical setting may be associated with a set  $\mathcal{A}$  of agents. An agent  $A_i \in \mathcal{A}$  is described by a knowledge operator  $K_i \in \mathcal{K}_{\mathcal{A}}$  and a physical system  $M_i \in \mathcal{M}_{\mathcal{A}}$ , which we call a "memory." Each agent may study other systems according to the theory  $\mathbb{T}$ . An agent's memory  $M_i$  records the results and the consequences of the studies conducted by  $A_i$ . The memory may be itself an object of a study by other agents.

<sup>&</sup>lt;sup>4</sup>We strive to be as general as possible and do not suppose or impose any structure on systems and connections between them; in particular, we don't make any assumptions about how composite systems are formally described in terms of their parts.

 $<sup>{}^{5}</sup>$ In fact, in box world, we can consider operations on two initial systems that transform it into a new, larger system that can no longer be seen as being made up of 2 smaller systems. We call this "supergluing", see Section 8.5 for a discussion.

#### 8.2.4 Measurements and memory update

Here we consider measurements both from the perspective of an agent who performs them, and that of another agent who is modeling the first agent's memory. In an experiment involving measurements, each agent has the subjective experience of only observing one outcome (independently of how others may model her memory), and we can see this as the definition of a measurement: if there is no subjective experience of observing a single outcome, we don't call it a measurement. We can express this experience as statements such as  $\phi_0 =$  "The outcome was 0, and the system is now in state  $|0\rangle$ ." We explain this further after the formal definition.

**Definition 8.2.6** (Measurements). A measurement is a type of study that can be conducted by an agent  $A_i$  on a system S, the essential result of which is the obtained "outcome"  $x \in \mathcal{X}_S$ . If *witnessed* by another agent  $A_j$  (who knows that  $A_i$  performed the measurement but does not know the outcome), the measurement is characterized by a set of propositions  $\{\phi_x\} \in \Phi$ , where  $\phi_x$  corresponds to the outcome x, satisfying:

- $K_j(K_i(\exists x \in \mathcal{X}_S : K_i \phi_x)),$
- $K_j K_i \phi_x \Longrightarrow K_j K_i \neg (\phi_y), \quad \forall \ y \neq x.$

The first condition tells us that  $A_i$  knows that from  $A_i$ 's perspective, she must have observed one outcome  $x \in X$ , and  $A_i$  would have used this knowledge to derive all the relevant conclusions, as expressed by the proposition  $\phi_x$ . For example, if the measurement represents a perfect Z measurement of a qubit,  $\phi_0$  may include statements like "the qubit is now in state  $|0\rangle$ ; before the measurement it was not in state  $|1\rangle$ ; if I measure it again in the same way, I will obtain outcome 0" and so on. Note that this condition does not imply that the measurement outcome stored in  $A_i$ 's memory is classical for  $A_j$ . In fact, in the quantum case  $A_j$  may see  $A_i$ 's memory as a quantum system entangled with the system that  $A_i$  measured. Despite this,  $A_i$  knows that from  $A_i$ 's perspective, this outcome appears to be classical, which is what the first condition captures. The second condition implements  $A_i$ 's experience of observing a single outcome, and the fact that the outside agent  $A_i$  knows that this is the case from  $A_i$ 's perspective. If  $A_i$  observes x, they conclude that the conclusions  $\phi_u$  that they would have derived had they observed a different outcome y are not valid and  $A_j$  knows that  $A_i$  would do so. In the previous example, they would know that it does not hold  $\phi_1 =$  "the qubit is now in state  $|1\rangle$ ; before the measurement it was not in state  $|0\rangle$ ; if I measure it again I will see outcome 1." This condition also ensures that the conclusions  $\{\phi_x\}_x$  are mutually incompatible, i.e. that the measurement is tightly characterized.

A measurement of another agent's memory is also an example of a valid measurement. In other words, agent  $A_j$  can choose  $A_i$ 's lab, consisting of  $A_i$ 's memory and another system S (which  $A_i$  studies), as an object of her study. Thus, any agent's memory can be modelled by the other agents as a physical system undergoing an evolution that correlates it with the measured system. In quantum theory, this corresponds to the unitary evolution

$$\left(\sum_{x=0}^{N-1} p_x |x\rangle_{\text{system}}\right) \otimes |0\rangle_{\text{memory}} \to \sum_{x=0}^{N-1} p_x \underbrace{|x\rangle_{\text{system}} \otimes |x\rangle_{\text{memory}}}_{=: |\tilde{x}\rangle_{SM}}.$$
(8.1)

The key aspect here is that the set of states of the joint system of observed system and memory,  $\mathscr{P}_{SM} = \operatorname{span}\{|x\rangle_{\operatorname{system}} \otimes |x\rangle_{\operatorname{memory}}\}_{x=0}^{N-1}$  is post-measurement isomorphic to the set of states  $\mathscr{P}_S$ system alone. That is, for every transformation  $\epsilon_S$  that you could apply to the system before the measurement, there is a corresponding transformation  $\epsilon_{SM}$  acting on the  $\mathscr{P}_{SM}$  that is operationally identical. By this we mean that an outside observer would not be able to tell if they are operating with  $\epsilon_S$  on a single system before the measurement, or with  $\epsilon_{SM}$  on system and memory after the measurement. In particular, if  $\epsilon_S$  is itself another measurement on Swithin a probabilistic theory, it should yield the same statistics as post-measurement  $\epsilon_{SM}$ . For a quantum example that helps clarify these notions, consider S to be a qubit initially in an arbitrary state  $\alpha|0\rangle_S + \beta|1\rangle_S$ . An agent Alice measures S in the Z basis and stores the outcome in her memory A. While she has a subjective experience of seeing only one possible outcome, an outside observer Bob could model the joint evolution of S and A as

$$(\alpha|0\rangle_S + \beta|1\rangle_S) \otimes |0\rangle_A \rightarrow \alpha|0\rangle_S|0\rangle_A + \beta|1\rangle_S|1\rangle_A.$$

Suppose now that (before Alice's measurement) Bob was interested in performing an X measurement on S. This would have been a measurement with projectors  $\{|+\rangle + |_S, |-\rangle - |_S\}$ , where  $|\pm\rangle_S = \frac{1}{\sqrt{2}}(|0\rangle_S \pm |1\rangle_S)$ . However, he arrived too late: Alice has already performed her Z measurement on S. If now Bob simply measured X on S, he would obtain uniform statistics, which would be uncorrelated with the initial state of S. So what can he do? It may not be very friendly, but he can measure S and Alice's memory A jointly, by projecting onto

$$|+\rangle_{SA} = \frac{1}{\sqrt{2}} (|0\rangle_{S}|0\rangle_{A} + |1\rangle_{S}|1\rangle_{A})$$
$$|-\rangle_{SA} = \frac{1}{\sqrt{2}} (|0\rangle_{S}|0\rangle_{A} - |1\rangle_{S}|1\rangle_{A}),$$

which yields the same statistics of Bob's originally planned measurement on S, had Alice not measured it first. In fact, this is precisely the  $\{|ok\rangle, |fail\rangle\}$  basis in which the outside observers, Ursula and Wigner measure, in the Frauchiger-Renner case (c.f. Section 7.1). Ideally, this equivalence should also hold in the more general case where the observed system may have been previously correlated with some other reference system: such correlations would be preserved in the measurement process, as modelled by the "outside" observer Bob.

There are many options to formalize this notion that "every way that an outside observer could have manipulated the system before the measurement, he may now manipulate a subspace of 'system and observer's memory,' with the same results." A possible simplification to restrict our options is to take subsystems and the tensor product structure as primitives of the theory, which may apply to GPTs [16] but not for general physical theories (like field theories; for a discussion see [130]). Here, we will for now restrict ourselves to this case, and leave a more general formulation of this condition as future work. We also restrict ourselves to informationpreserving measurements (excluding for now those where some information may have leaked to an environment external to Alice's memory), which are sufficient to derive the contradiction.



(a) Alice's perspective. The measurement in Z basis performed by Alice, who writes the classical result down to her memory A.



(b) Bob's perspective on Alice performing a measurement. The memory update of Alice, after she measures the system S in Z basis, as seen from the point of view of the outside observer, corresponding to the memory update U.





(d) Bob's perspective on performing a measurement after Alice. Bob performing a measurement in  $\tilde{X}$  basis of systems S and A, after Alice's memory update U.

The measurement and memory update in quantum theory Figure 8.1: from different perspectives. From Alice's point of view, the measurement of the system S either in Z basis yields a classical result, which she records to her memory A, performing a classical CNOT (Figure 8.1a). For an outside observer, Bob who is not aware of Alice's measurement result, Alice's memory is entangled with the system and the CNOT is a quantum entangling operation which corresponds to the memory update U (Figure 8.1b). Further, there is no classical measurement outcome from Bob's perspective even though he knows that Alice would perceive one. If Bob had access to the system S prior to the measurement by A, and wanted to measure it in X basis  $(\{|+\rangle_S, |-\rangle_S\})$ , he would have to perform an operation  $\mathscr{E}_S[Z, X]$  (and then copy the classical result into his memory B) (Figure 8.1c). If the system S was initially in a state  $|\psi\rangle = |+\rangle_S$ , then a proposition which would correspond to this operation is  $\phi[\mathscr{E}_S[Z,X](|\psi\rangle_S)] =$  "s = +". However, if the measurement in Z is already performed by A and the result is written to her memory, the whole process described by Bob as a memory update u, and in order to comply his initial wish to measure S only, he can perform an operation  $\mathscr{E}_{SA}[Z,X]$  on S and A together instead, which is a measurement in  $\{|+\rangle_{SA}, |-\rangle_{SA}$  basis (Figure 8.1d). A proposition which this operation yields is  $\phi[\mathscr{E}_{SM_i} \circ u(|\phi\rangle_S \otimes |0\rangle_A)] =$ "sa = +" (as  $|\xi\rangle_{SA} = |+\rangle_{SA}$ ), which naturally follows from "s = +", given the structure of the memory update u.<sup>6</sup>

**Definition 8.2.7** (Information-preserving memory update). Let  $\mathscr{P}_S$  be a set of states of a system S that is being studied by an agent  $A_i$  with a memory  $M_i$ , and  $\mathscr{P}_{SM_i}$  be a set of states

<sup>&</sup>lt;sup>6</sup>This figure is taken from our paper [208] and was made by coauthors.

of the joint system  $SM_i$ . If for a given initial state  $Q_{M_i}^{in} \in \mathscr{P}_{M_i}$  of the memory, there exists a corresponding map  $U^Q : \mathscr{P}_{SM_i} \to \mathscr{P}_{SM_i}$  ( $\in \mathcal{O}_{SM_i}$ ) that satisfies the following conditions (1) and (2), then  $U^Q$  is called an *information-preserving memory update*.

1. Local operations on S before the memory update can be simulated by joint operations on S and  $M_i$  after the update. That is, for all  $P_S \in \mathscr{P}_S$ ,  $O_S \in \mathcal{O}_S$ ,  $A_j \in \mathcal{A}$ ,  $\phi$ , there exists an operation  $O_{SM_i} \in \mathcal{O}_{SM_i}$  such that

$$K_j \phi[O_S(P_s)] \Rightarrow K_j \phi[O_{SM_i} \circ U^Q(P_S \parallel Q_{M_i}^{in})],$$

where  $\phi[\ldots]$  are arbitrary statements that depend on the argument.

2. The memory update does not factorize into local operations. That is, there exist no operations  $O'_S \in \mathcal{O}_S$  and  $O'_{M_i} \in \mathcal{O}_{M_i}$  such that

$$U^Q = O'_S \parallel O'_{M_i}$$

Condition (1) was explained in previous paragraphs. Condition (2) is required because the trivial map which entails doing nothing to the system and memory (i.e., the identity) satisfies Condition (1) even though such an operation should certainly not be regarded as a memory update. Condition (2) requires that  $U^Q$  does not factorise into local operations over S and  $M_i$  is required in order to rule out such trivial operations that cannot be taken to represent a memory update. See Figure 8.1 for an example of  $U^Q$  in the quantum case where it is a reversible unitary operation and the initial state of the memory,  $Q_{M_i}^{in}$  is  $|0\rangle_{M_i}$ . In general, the memory update map  $U^Q$  need not be reversible; for example, in box world it is an irreversible transformation, as we will see later.

Note that it is enough to consider the memory update map  $U^Q$  corresponding to a particular choice of initial state  $Q_{M_i}^{in}$  since the map  $U^{Q'}$  corresponding to any other state  $Q_{M_i}^{in} \in \mathscr{P}_{M_i}$  can be obtained by first locally transforming the memory state into  $Q_{M_i}^{in}$  and then applying  $U^Q$ . Thus without loss of generality, we will consider only specific initial states in the chapter and drop the label Q on this map, simply calling it U. For example, in the quantum case, it is enough to consider the memory update with the memory initialised to the state,  $|0\rangle_{M_i}$ .

The characterization of measurements introduced in this section is rather minimal. In physical theories like classical and quantum mechanics, measurements have other natural properties that we do not require here. Two striking examples are "after her measurement, Alice's memory becomes correlated with the system measured in such a way that, for any subsequent operation that Bob could perform on the system, there is an equivalent operation he may perform on her memory" and "the correlations are such that there exists a joint operation on the system and Alice's memory that would allow Bob to conclude which measurement Alice performed." While these properties hold in the familiar classical and quantum worlds, we do not know of other physical theories where measurements can satisfy them, and they require Bob to be able to act independently on the system and on Alice's memory, which may not always be possible. For example, we will see that in box world, these two subsystems become *superglued* after

Alice's measurement, and that Bob only has access to them as a whole and not as individual components.<sup>7</sup> As such, we will not require these properties out of measurements, for now. We revisit this discussion in Section 8.5.

## 8.3 Agents, memory and measurement in box world

In Section 2.4, we reviewed Barrett's framework [16] for generalised probabilistic theories, in particular describing states, transformations and measurements in box world, a particular GPT. We will now instantiate our general conditions for agents, memories and measurements (definitions definitions 8.2.5 to 8.2.7) in box world. As there is no physical theory for the dynamics behind box world, there is plenty of freedom in the choice of implementation. In principle each such choice could represent a different physical theory leading to the same black-box behaviour in the limit of a single agent with an implicit memory. This is analogous to the way in which different versions of quantum theory (Bohmian mechanics, collapse theories, unitary quantum mechanics with von Neumann measurements) result in the same effective theory in that limit.

**Definition 8.3.1** (Agents in box world). Let  $\mathbb{T}$  be the theory that describes box world, according to [16] (see also Section 2.4). As per definition 8.2.5, an agent  $A_i \in \mathcal{A}$  is described by a knowledge operator  $K_i \in \mathcal{K}_{\mathcal{A}}$  and a physical memory  $M_i \in \mathcal{M}_{\mathcal{A}}$ .

We will focus on the case where the memory consists of bit or gbits. Each agent may study other systems according to the theory  $\mathbb{T}$ . An agent's memory records the results and the consequences of the studies conducted by them, and may be an object of a study by other agents.

It is worth mentioning that boxes do not correspond to physical systems, but to input/output functions that can only be evaluated once. As such, the post-measurement state of a physical system is described by a whole new box. The notion of an individual system itself, as we will see, may be unstable under measurements— some measurements *glue* the system to the observer's memory, in a way that makes individual access to the original system impossible.

**Measurement: observer's perspective.** From the point of view of the observer who is measuring (say Alice), making a measurement on a system corresponds simply to running the box whose state vector encodes the measurement statistics. Alice may then commit the result of her measurement to a physical memory, like a notebook where she writes 'I measured observable X and obtained outcome a.' To be useful, this should be a memory that may be consulted later, i.e. it could receive an input Y ='start: open and read the memory', and output the pair (X, a). In the language of GPTs, this means that Alice, from her own perspective, prepares a new box with a trivial input Y ='start' and two outputs (X', a'), with the behaviour  $\vec{P}(X', a'|Y) = \delta_{X,X'}\delta_{a,a'}$ , which depends on her observations. She may later run this box (look at her notebook) and recover the measurement data. The exact dimension of the box will depend

 $<sup>^7\</sup>mathrm{Thus}$  the state-space  $\mathscr{P}_{SM_i}$  can also contain such "super-glued states".



(a) Generally, in GPTs with some notion of subsystems, Ursula can think of the physical system measured by Alice, and Alice's memory pre-measurement as two boxes, which Ursula could in principle run if Alice chose not to measure (left). From Ursula's perspective, Alice's measurement corresponds to a transformation of the two initial boxes that results in a new, postmeasurement box available to Ursula, whose behaviour will depend on the concrete physical theory.



(b) In box world, suppose the two initial systems correspond to gbits, and Alice's memory is initialized as shown. If we want to preserve the global system dimensions, then the rules for allowed transformations (Definition 8.2.7) limit the statistics of Ursula's final box to be of the form shown in the right (Appendix 8.6.1). The asterisks represent arbitrary values, which will depend on the choice of implementation of Alice's measurement.

Figure 8.2: Memory update after a measurement: an outsider's perspective. Here Alice makes a measurement of a system (blue, top) at time t = 1 and stores her outcome in her memory (pink, bottom). The question is how an outsider, Ursula, models Alice's measurement. Note that this transformation need not be reversible, in fact this is true of the box world case.<sup>8</sup>

on how Alice perceives and models her own memory; for example it could consist of two bits, or two gbits, or, if we think that before the measurement she stored the information about the choice of observable elsewhere, it could be a single bit or gbit encoding only the outcome. We leave this open for now, as we do not want to constrain the theory too much at this stage.

**Measurements: inferences.** To see the kind of inferences and conclusions that an agent can take from a measurement in box world, it's convenient to look at the example where Alice and Bob share a PR box (Figure 2.4). Suppose that Alice measured her half of the box with input X = 1 and obtained outcome a = 0. From the PR correlations,  $XY = a \oplus b$ , she can conclude that if Bob measures Y = 0, he must obtain b = 0, and if he measures Y = 1, he must obtain b = 1. This is independent of whether Bob's measurement happens before or after Alice (or even space-like separated). She could reach similar deterministic conclusions for her other

<sup>&</sup>lt;sup>8</sup>This figure is taken from our paper [208] and was made by Lídia del Rio.

choice of measurement and possible outcomes. In the language of Definition 8.2.6, we have

$$\begin{split} \phi_{X=0,a=0} &= ``[Y=0 \implies b=0] \land [Y=1 \implies b=0]", \\ \phi_{X=0,a=1} &= ``[Y=0 \implies b=1] \land [Y=1 \implies b=1]", \\ \phi_{X=1,a=0} &= ``[Y=0 \implies b=0] \land [Y=1 \implies b=1]", \\ \phi_{X=1,a=1} &= ``[Y=0 \implies b=1] \land [Y=1 \implies b=0]". \end{split}$$

Measurement: memory update from an outsider's perspective. Next we need to model how an outside agent, Ursula, models Alice's measurement, in the case where Alice does not communicate her outcome to Ursula.<sup>9</sup> Suppose that all agents share a time reference frame, and Alice makes her measurement at time t = 1. From Ursula's perspective, in the most general case, this will correspond to Alice preparing a new box, with some number of inputs and outputs, which Ursula can later run (Figure 8.2a). The exact form of this box will depend on the underlying physical theory for measurements: in the quantum case it corresponds to a box with the measurement statistics of a state that's entangled between the system measured and Alice's memory, as we saw. In classical mechanics, it will correspond to perfect classical correlations between those two subsystems. In the other extreme, we could imagine a theory of very destructive measurements, where after Alice's measurement, the physical system she had measured would vanish. From Ursula's perspective, this could be modelled by a box with a void associated distribution. Now suppose that we would like to have a physical theory where the dimension of systems is preserved by measurements: for example, if the system that Alice measures is instantiated by a box with binary input and output (e.g. a gbit, or half of a PR-box), and Alice's memory, where she stores the outcome of the measurement is also represented as a gbit, then we would want the post-measurement box accessible to Ursula to have in total two binary inputs and two binary outputs (or more generally, four possible inputs and four possible outputs). Note that this is not a required condition for a theory to be *physical* per se— it is just a familiar rule of thumb that gives some persistent meaning to the notion of subsystems and dimensions. In such a theory that supports box world correlations, we find that the allowed statistics of Ursula's box must satisfy the conditions of Figure 8.2b (proof in Appendix 8.6.1). These conditions still leave us some wiggle room for possible different implementations.

**Measurements:** information-preserving memory update. In order to find a multiagent paradox, we will need a model of memory update that is information-preserving, in the sense of Definition 8.2.7. This does not imply that Alice's transformation (as seen by Ursula) be reversible. Firstly, it is known that all reversible dynamics in box world is trivial and cannot correlate product states [103], hence the memory update map (Definition 8.2.7) in box world cannot be reversible. Further, we find that in general, the memory update (from the outside perspective) can *glue* two subsystems such that Ursula will only be able to address them as a whole (since separating them could lead to a violation of no-signaling). But the relevant fact

<sup>&</sup>lt;sup>9</sup>Naming convention: as we will see in Section 8.4, the proposed experiment feature two "internal" agents, Alice and Bob, who will in turn be measured by two "external" agents, Ursula and Wigner. In the example of Section 8.2, the internal agent was Alice and the external Bob, so that their different pronouns could help keep track of whose memory we were referring to.



(a) Alice measures a system (inner blue box) and stores the outcome in her memory (inner pink box). From Ursula's perspective, this appears as though Alice simply wired the outputs of the two boxes with a controlled-NOT gate, so that the measurement output is copied to the output of the memory. When Ursula later runs the outer green box, she provides two inputs, which go through this circuit, resulting in two identical outputs.



(b) In an information-preserving model, for every measurement X on the initial state of the system measured by Alice, there exists a measurement  $\tilde{X}$  that Ursula can do jointly on the system and Alice's memory that reproduces the same statistics. This is achieved, for example, by Ursula fixing her second input to 0, and simply discarding the second (trivial) output as shown.

Figure 8.3: Information-preserving memory update. This (trivial) physical implementation of Alice's measurement in box world satisfies the conditions of Figure 8.2b and is information-preserving (Definition 8.2.7). The crucial detail is that Ursula is not allowed to open her box (in green) and access the circuitry inside. Note that is Ursula can perform arbitrary measurements on the joint system then the pre and post measurement states give different statistics, the former is a product state while the latter isn't. Further, there are other possibilities for modelling measurements— this is the simplest one that still allows us to derive the paradox. Details and proofs in Appendix 8.6.1.<sup>10</sup>

is that Ursula can apply some post-processing in order to obtain a new box with the same behaviour as the pre-measurement system that Alice observed. In Figure 8.3 we give an example of a model that satisfies these conditions, in addition to the conditions of Figure 8.2b. It is a minimal implementation among many possible, which already allows us to derive such a paradox beyond quantum theory. We further discuss some of the limitations and alternatives to this model in Section 8.5. What is important here (and proven in Appendix 8.6.1) is that this model generalizes to the case where Alice measures half of a bipartite state, like a PR box. That is, suppose that Alice and Bob share a PR box. Imagine that at time t = 1 Alice makes her measurement X, obtaining (from her perspective) an outcome a, and that Bob makes his measurement Y at time t = 2, obtaining outcome b. As usual, if Alice and Bob were to communicate at this point, they would find that  $XY = a \oplus b$ , and indeed the propositions  $\phi_{X,a}$  and  $\phi_{Y,b}$  that represent their subjective measurement experience would hold. But now suppose that Alice and Bob do not get the chance to communicate and compare their input and outputs; instead, at time t = 3, an observer Ursula, who models Alice's measurement as in Figure 8.3a, runs the box corresponding to Alice's half of the PR box and Alice's memory, and applies the post-processing of Figure 8.3b. Ursula's input is  $\tilde{X}$  and her output is  $\tilde{a}$ . Then the

 $<sup>^{10}</sup>$ This figure is taken from our paper [208] and was made by Lídia del Rio.

claim is that  $\tilde{X}Y = \tilde{a} \oplus b$ : that is, Ursula and Bob effectively share a PR box. This is proven in Appendix 8.6.1. We now have all the ingredients needed to find a multi-agent epistemic paradox in box world.

### 8.4 Finding the paradox

We now describe a scenario in box world where reasoning, physical agents reach a logical paradox.

**Experimental setup.** The proposed thought experiment is similar in spirit to the one proposed by Frauchiger and Renner [85], with the PR box playing the role of the Hardy state (c.f. Figure 7.1). Alice and Bob share a PR box (the corresponding box world state is given in Figure 2.4b); they each will measure their half of the PR box and store the outcomes in their local memories. Let Alice's lab be located inside the lab of another agent, Ursula's lab such that Ursula can now perform joint measurements on Alice's system (her half of the PR box) and memory, as seen in the previous section. Similarly, let Bob's lab be located inside Wigner's lab, such that Wigner can perform joint measurements on Bob's system and memory. We assume that Alice's and Bob's labs are isolated such that no information about their measurement outcomes leaks out. The protocol is the following:

- t=1 Alice measures her half of the PR box, with measurement setting X, and stores the outcome a in her memory A.
- t=2 Bob measures his half of the PR box, with measurement setting Y, and stores the outcome b in his memory B.
- t=3 Ursula measures the box corresponding to Alice's lab (as in Figure 8.3b), with measurement setting  $\tilde{X} = X \oplus 1$ , obtaining outcome  $\tilde{a}$ .
- t=4 Wigner measures the box corresponding to Bob's lab, with measurement setting  $\tilde{Y} = Y \oplus 1$ , obtaining outcome  $\tilde{b}$ .

Agents can agree on their measurement settings beforehand, but should not communicate once the experiment begins. The trust relation, which specifies which agents consider each other to be rational agents (as opposed to mere physical systems), is

$$A_{t=1,2} \Leftrightarrow B_{t=1,2}$$
$$B_{t=2,3} \Leftrightarrow U_{t=3}$$
$$U_{t=3,4} \Leftrightarrow W_{t=4}$$
$$W_{t=4} \Leftrightarrow A_{t=1}.$$

Note that the time information is important here as it tells us that the trust relations hold for the pre-tampering versions of the "inside" agents, Alice and Bob (who are measured by the "outside" agents Ursula and Wigner). The common knowledge  $\mathbb{T}$  shared by all four agents includes the PR box correlations, the way the external agents model Alice and Bob's measurements, and the trust structure above.

**Reasoning.** Now the agents can reason about the events in other agents' labs. We take here the example where the measurement settings are X = Y = 0,  $\tilde{X} = \tilde{Y} = 1$ , and where Wigner obtained the outcome  $\tilde{b} = 0$ ; the reasoning is analogous for the remaining cases.

1. Wigner reasons about Ursula's outcome. At time t = 4, Wigner knows that, by virtue of their information-preserving modelling of Alice and Bob's measurements, he and Ursula effectively shared a PR box<sup>11</sup>. He can therefore use the PR correlations  $\tilde{X}\tilde{Y} = \tilde{a} \oplus \tilde{b}$  to conclude that Ursula's output must be 1,

$$K_W(\tilde{b}=0 \implies \tilde{a}=1).$$

2. Wigner reasons about Ursula's reasoning. Now Wigner thinks about what Ursula may have concluded regarding Bob's outcome. He knows that at time t = 3, Ursula and Bob effectively shared a PR box<sup>11</sup>, satisfying  $\tilde{X}Y = \tilde{a} \oplus b$ , and that therefore Ursula must have concluded

$$K_W K_U (\tilde{a} = 1 \implies b = 1).$$

3. Wigner reasons about Ursula's reasoning about Bob's reasoning. Next, Wigner wonders "What could Ursula, at time t = 3, conclude about Bob's reasoning at time t = 2?" Well, Wigner knows that she knows that Bob knew that at time t = 2 he effectively shared a PR box with Alice, satisfying  $XY = a \oplus b$ , and therefore concludes

$$K_W K_U K_B (b = 1 \implies a = 1).$$

4. Wigner reasons about Ursula's reasoning about Bob's reasoning about Alice's reasoning. Now Wigner thinks about Alice's perspective at time t = 1, through the lenses of Bob (at time t = 2) and Ursula (t = 3). Back then, Alice knew that she obtained some outcome a, and that Wigner would model Bob's measurement in an information-preserving way, such that Alice (at time t = 1) and Wigner (of time t = 4) share an effective PR box<sup>11</sup>, satisfying  $X\tilde{Y} = a \oplus \tilde{b}$ , which results, in particular, in

$$K_W K_U K_B K_A (a = 1 \implies \tilde{b} = 1).$$

5. Wigner applies trust relations. In order to combine the statements obtained above, we need to apply the trust relations described above, starting from the inside of each propo-

<sup>&</sup>lt;sup>11</sup>See Appendix 8.6.1 for a proof.

sition, for example,

$$K_W K_U K_B K_A (a = 1 \implies b = 1)$$
  

$$\implies K_W K_U K_B (a = 1 \implies \tilde{b} = 1) \qquad (\because A \rightsquigarrow B)$$
  

$$\implies K_W K_U (a = 1 \implies \tilde{b} = 1) \qquad (\because B \rightsquigarrow U)$$
  

$$\implies K_W (a = 1 \implies \tilde{b} = 1) \qquad (\because U \rightsquigarrow W)$$

and similarly for the other statements (where  $A \sim B$  denotes B trusts A, c.f. Definition 2.6.3), so that we obtain

$$K_W [(\tilde{b} = 0 \implies \tilde{a} = 1) \land (\tilde{a} = 1 \implies b = 1) \land (b = 1 \implies a = 1) \land (a = 1 \implies \tilde{b} = 1)]$$
$$\implies K_W (\tilde{b} = 0 \implies \tilde{b} = 1).$$

We could have equally taken the point of view of any other observer, and from any particular outcome or choice of measurement, and through similar reasoning chains reached the following contradictions,

$$\begin{split} K_A[(a=0 \implies a=1) \land (a=1 \implies a=0)], \\ K_B[(b=0 \implies b=1) \land (b=1 \implies b=0)], \\ K_U[(\tilde{a}=0 \implies \tilde{a}=1) \land (\tilde{a}=1 \implies \tilde{a}=0)], \\ K_W[(\tilde{b}=0 \implies \tilde{b}=1) \land (\tilde{b}=1 \implies \tilde{b}=0)]. \end{split}$$

### 8.5 Discussion

We have generalized the conditions of the Frauchiger-Renner theorem and made them applicable to arbitrary physical theories, including the framework of *generalized probability theories*. We then applied these conditions to the GPT of box world and found an experimental setting that leads to a multi-agent logical paradox.

### 8.5.1 Comparison with the quantum thought experiment

We showed that box world agents reasoning about each others' knowledge can come to a deterministic contradiction, which is stronger than the original paradox, as it can be reached without post-selection, from the point of view of every agent and for any measurement outcome obtained by them.

**Post-selection.** In contrast to the original Frauchiger-Renner experiment of [85], no post-selection was required to arrive at this contradictory chain of statements as, in fact, all the

implications above are symmetric, for example

$$\tilde{a} = 0 \iff b = 0 \iff a = 0 \iff \tilde{b} = 0 \iff \tilde{a} = 1.$$

As a result, one can arrive at a similar (symmetric) paradoxical chain of statements irrespective of the choice of agent and outcome for the first statement. In other words, irrespective of the outcomes observed by every agent, each agent will arrive at a contradiction when they try to reason about the outcomes of other agents. This is because, as shown in [4], the PR box exhibits strong contextuality and no global assignments of outcome values for all four measurements exists for any choice of local assignments. In contrast, the original paradox of [85] admits the same distribution as that of Hardy's paradox [109]. It is shown in [4] that this distribution is an example of logical contextuality where for a particular choice of local assignments (the ones that are post-selected on in the original Frauchiger-Renner experiment), a global assignment of values compatible with the support of the distribution fails to exist, but this is not true for all local assignments. This makes the paradox even stronger in box world, since it can be found without post-selection and by any of the agents, for any outcome that they observe. In particular, the paradox would already arise in a single run of the experiment. For a simple method to enumerate all possible contradictory statements that the agents may make, see the analysis of the PR box presented in [4].

**Reversibility of the memory update map** As mentioned previously, the memory update map U in the quantum case is quantum CNOT gate which is a unitary and hence reversible. In box world however, this map cannot be reversible since it is known that all reversible maps in box world map product states to product states [103] and hence no reversible U in box world could satisfy Definition 8.2.7 of an information-preserving memory update. The map we propose here for box world is clearly irreversible as it leads to correlations between the initially uncorrelated system and memory.

#### 8.5.2 Physical measurements in box world

Since we lack a physical theory to explain how measurements and transformations are instantiated for generalised non-signaling boxes, and only have access to their input/output behaviour, all allowed transformations consist of pre- and post-processing. In the quantum case, we have in addition to a description of possible input-output correlations, a mathematical framework for the underlying states producing those correlations, the theory of von Neumann measurements and transformations as CPTP maps. In box world, introduction of dynamical features (for example, a memory update algorithm) is less intuitive and requires additional constructions. In the following, we outline the main limitations we found.

**Systems vs boxes.** In quantum theory, a system corresponds to a physical substrate that can be acted on more than once. For example, Alice could measure a spin first in the Z basis and then in X basis (obviously with different results than if she had measured first X and

then Z). The predictions for each subsequent measurement are represented by a different box in the GPT formalism, such that each box encodes the current state of the system in terms of the measurement statistics of a tomographically complete set of measurements. After each measurement, the corresponding box disappears, but quantum mechanics gives us a rule to compute the post-measurement state of the underlying system, which in turn specifies the box for future measurements. On the other hand, the default theory for box world lacks the notion of underlying physical systems and a definite rule to compute the post-measurement vector state of something that has been measured once. Indeed, Equations 2.44a-2.44c tell us that postmeasurement states is only partially specified: for instance, if the measurement performed was fiducial, we know that the block corresponding to that measurement in the post-measurement state would have a "1" corresponding to the outcome obtained and "0" for all other outcomes in the block. However, we still have freedom in defining the entries in the remaining blocks. Our model proposes a possible physical mechanism for updating boxes (which could be read as updating the state of the underlying system), but so far only for the case where we compare the perspectives of different agents, and we leave it open whether Alice has a subjective update rule that would allow her to make subsequent measurements on the same physical system.

**Verifying a measurement.** In our simple model, the external observer Ursula has no way to know which measurement Alice performed, or whether she measured anything at all— the connection between Alice's and Ursula's views is postulated rather than derived from a physical theory. Indeed, Alice could have simply wired the boxes as in Figure 8.3a without actually performing the measurement, and Ursula will not know the difference: she obtains the same joint state of Alice's memory and the system she measured. In contrast, consider the case of quantum mechanics with standard von Neumann measurements. There, Alice's memory gets entangled with the system, and the post-measurement state depends on the basis in which Alice measured her system. For example, if Alice's qubit S starts off in the normalised pure state  $|\psi\rangle =$  $\alpha|0\rangle_S + \beta|1\rangle_S$  and her memory M initialised to  $|0\rangle_M$ , the initial state of her system and memory from Ursula's perspective is  $|\Psi\rangle_{SM}^{in} = [\alpha|0\rangle_S + \beta|1\rangle_S] \otimes |0\rangle_M = [(\frac{\alpha+\beta}{\sqrt{2}})|+\rangle_S + (\frac{\alpha-\beta}{\sqrt{2}})|-\rangle_S)] \otimes |0\rangle_M$  If Alice measures the system in the Z basis, the post-measurement state from Ursula's perspective is  $|\Psi\rangle_{SM}^{out,Z} = \alpha |0\rangle_S |0\rangle_M + \beta |1\rangle_S |1\rangle_M$ , which is an entangled state. If instead, Alice measured in the Hadamard (X) basis, the post-measurement state would be  $|\Psi\rangle_{SM}^{out,X} = (\frac{\alpha+\beta}{\sqrt{2}})|+\rangle_S|0\rangle_M + (\frac{\alpha-\beta}{\sqrt{2}})|-\rangle_S|1\rangle_M$ . Clearly the measurement statistics of  $|\Psi\rangle_{SM}^{in}$ ,  $|\Psi\rangle_{SM}^{out,Z}$  and  $|\Psi\rangle_{SM}^{out,X}$  are different and Ursula can thus (in principle, with some probability) tell whether or not Alice performed a measurement and which measurement was performed by her. In the absence of a physical theory backing box world, we can still lift this degenerancy between the three situations (Alice didn't measure, she measured X = 0, or she measured X = 1) by adding another classical system to the circuitry of 8.3a: for example, a trit that stores what Alice did, and which Ursula could consult independently of the glued box of system and Alice's memory. However, we'd still have a postulated connection between what's stored in this trit and what Alice actually did, and not one that is physically motivated.

Supergluing of non-signaling boxes. For the memory update circuit (from Ursula's perspective) of Figure 8.3a, and the initial state of Equation (8.3), the final state would be

(following the notation of Section 2.4)  $\vec{P}_{fin}^{SM} = (p \ 0 \ 0 \ 1 - p|p \ 0 \ 0 \ 1 - p|q \ 0 \ 0 \ 1 - p|q)$  $q|q=0=0=1-q)_{SM}^{T}$ . Note that while the reduced final state of S does not depend on the input X' to M, the reduced final state on Alice's memory M,  $\vec{P}_{fin}^M$  clearly depends on the input X of the system S if  $p \neq q$ . If X = 0,  $\vec{P}_{fin}^M = (p \ 1-p|p \ 1-p)^T$  and if X = 1,  $\vec{P}_{fin}^M = (q \ 1-q|q \ 1-q)^T$ , i.e., the systems S and M are signalling. This is expected since there is clearly a transfer of information from S to M during the measurement as seen in Figure 2.5. However, this means that the state  $\vec{P}_{fin}^{SM}$  is not a valid box world state of 2 systems S and M but a valid state of a single system SM i.e., after Alice performs her wiring/measurement, it is not possible to physically separate Alice's system S from her memory M from Ursula's perspective. For if this were possible, there would be a violation of the no-signaling principle and the notion of relativistic causality. In quantum theory, on the other hand it is always possible to perform separate measurements on Alice's system and her memory even after she measures. We call this feature supergluing of post-measurement boxes, where it is no longer possible for Ursula to separately measure S or M, but she can only jointly measure SM as though it were a single system. Note that this is only the case for  $p \neq q$  and in our example with the PR-box (Section 8.4), p = q = 1/2 and  $\vec{P}_{fin}^{SM}$  remains a valid bipartite non-signaling state in this particular, finetuned case of the PR box and there is no supergluing in the particular example described in Section 8.4.

A glass half full. The above-mentioned features of the memory update in box world are certainly not desirable, and not what one would expect to find in a physical theory with meaningful notions of subsystems. An optimistic way to look at these limitations is to see them as providing us with further intuition for why PR boxes have not yet been found in nature. One of the main contributions of this chapter is the finding that despite these peculiar features of box world and the fact that it has no entangling bipartite joint measurements (a crucial step in the original quantum paradox), a consistent outside perspective of the memory update exists such that with our generalised assumptions, a multi-agent paradox can be recovered. This indicates that the reversibility of dynamics akin to quantum unitarity is not necessary to derive this kind of paradox. We suspect that contextuality, and the existence of a suitable information-preserving memory update (Definition 8.2.7) are necessary conditions for deriving such paradoxes. We discuss the relation to contextuality further in Section 8.5.4, and leave a detailed analysis to future work.

**Other models for physical measurements.** Ours is not the first attempt at coming up with a (partial) physical theory that reproduces the statistics of box world. Here we review the approach of Skrzypczyk et al. in [193]. There the authors consider a variation of box world that has a reduced set of physical states (which the authors call *genuine*), which consists of the PR box and all the deterministic local boxes. The wealth of box world state vectors (i.e. the non-signaling polytope, or what we could call epistemic states) is recovered by allowing classical processing of inputs and outputs via classical wirings, as well as convex combinations thereof. In contrast, box world takes all convex combinations of maximally non-signaling boxes (of which the PR box is an example) to be genuine physical states; this becomes relevant as we require the allowed physical operations to map such states to each other. For the restricted

state space of [193], the set of allowed operations is larger than in box world, particularly for multipartite settings. For example, there we are allowed maps that implement the equivalent of entanglement swapping: if Bob shares a PR box with Alice, and another with Charlie, there is an allowed map that he can apply on his two halves which leaves Alice and Charlie sharing a PR box, with some probability. It would be interesting to try to model memory update in this modified theory, to see if (1) there is a more natural implementation of measurements within the extended set of operations, and (2) whether this theory allows for multi-agent paradoxes.

### 8.5.3 Characterization of general theories

While we have shown that a consistency paradox, similar to the one arising in the Frauchiger-Renner setup, can also be adapted for the box world in terms of GPTs, it still remains unclear how to characterize all possible theories where it is possible to find a setup leading to a contradiction. It seems that contextuality is a key property of such theories, this is discussed in more detail in Section 8.5.4. Another central ingredient seems to be information-preserving models for physical measurements such as our memory update of Definition 8.2.7, which allow us to replace counter-factuals with actual measurements, performed in sequence by different agents.

**Beyond standard composition of systems.** Additionally, it is still an open problem to find an operational way to state the outside view of measurements (and a memory update operation), for theories without a prior notion of subsystems and a tensor rule for composing them. This will allow us to search for multi-agent logical paradoxes in field theories, for example. One possible direction is to use notions of effective and subjective locality, as outlined for example in [130].

### 8.5.4 Relation to contextuality

Multi-agent logical paradoxes involve chains (or possibly more general structures) of statements that cannot be simultaneously true in a consistent manner. Contextuality, on the other hand, can often be expressed in terms of the inability to consistently assign definite outcome values to a set of measurements [126, 194].

The examples of Frauchiger-Renner in quantum theory and the the present one in box world, both arise in contextual theories. Given this, and the strong links between these, Hardy's paradox [108, 109] and logical contextuality [4] (also pointed out in Section 7.2)— our hypothesis is that a contextual physical theory, when applied to systems that are themselves reasoning agents, can give rise to logical multi-agent paradoxes. The fact that such theories may allow a very different description of a measurement process from the points of views of an agent performing the measurement vs an outside agent (who analyses this agent and her system together) also has an important role to play in these paradoxes. In the quantum case this is closely linked to the measurement problem, the problem of reconciling unitary dynamics (outside view) and non-unitary "collapse" (inside view). The existence of a connection between multi-agent paradoxes and contextuality is hard to miss, but it is the nature of this connection that is unknown i.e., are all proofs of multi-agent logical paradoxes proofs of contextuality, or vice-versa? Based on our results, and the analysis of the Frauchiger-Renner experiment presented in Section 7.2, the hunch is that contextuality is necessary, but not sufficient for deriving such paradoxes: specific forms of contextuality, combined with suitable conditions on the memory update might be required. These questions will be formally addressed in future work. Nevertheless, in the following, we provide an overview of further connections and some more specific open questions in this direction.

Liar cycles. In [4] relations between logical paradoxes and quantum contextuality are explored; in particular, the authors point out a direct connection between contextuality and a type of classic semantic paradoxes called *Liar cycles* [65]. A Liar cycle of length N is a chain of statements of the form:

$$\phi_1 = "\phi_2 \text{ is true}", \phi_2 = "\phi_3 \text{ is true}", \dots, \phi_{N-1} = "\phi_N \text{ is true}", \phi_N = "\phi_1 \text{ is false}".$$
 (8.2)

It can be shown that the patterns of reasoning which are used in finding a contradiction in the chain of statements above are similar to the reasoning we make use of in FR-type arguments, and can also be connected to the cases of PR box (which corresponds to a Liar cycle of length 4) and Hardy's paradox. This further suggests that multi-agent paradoxes are closely linked to the notion of contextuality.

**Relation to logical pre-post selection paradoxes.** In [175], it has been shown that every proof of a logical pre-post selection paradox is a proof of contextuality. The exact connection between FR-type paradoxes and logical pre-post selection paradoxes is not known and this would be an interesting avenue to explore which would also provide insights into the relationship between FR paradoxes and contextuality.

## 8.6 Appendix

### 8.6.1 Proofs

#### 8.6.1.1 Memory update for a single lab

In this section, we describe how a box world agent would measure a system and store the result in a memory. From the perspective of an outside observer (who does not know the outcome of the agent's measurement), we describe the initial and final states of the system and memory before and after the measurement as well as the transformation that implements this memory update in box world. In the quantum case, any initial state of the system S is mapped to an isomorphic joint state of the system S and memory M (see Equation (8.1)) and hence the memory update map that maps the former to the latter (an isometry in this case<sup>12</sup>) satisfies Definition 8.2.7 of an information-preserving memory update. We will now characterise the analogous memory update map in box world and show that it also satisfies Definition 8.2.7.

**Theorem 8.6.1.** In box world, there exists a valid transformation u that maps every arbitrary, normalized state  $\vec{P}_{in}^S$  of the system S to an isomorphic final state  $\vec{P}_{fin}^{SM}$  of the system S and memory M and hence constitutes an information-preserving memory update (Definition 8.2.7).

*Proof.* To simplify the argument, we will describe the proof for the case where S and M are gbits. For higher dimensional systems, a similar argument holds, this will be explained at the end of the proof.

We start with the system in an arbitrary, normalized gbit state  $\vec{P}_{in}^S = (p \ 1 - p|q \ 1 - q)^T$ (where the subscript T denotes transpose and  $p, q \in [0, 1]$ ) and the memory initialised to one of the 4 pure states<sup>13</sup>, say  $\vec{P}_{in}^M = \vec{P}_1 = (1 \ 0|1 \ 0)^T$ . Then the joint initial state,  $\vec{P}_{in}^{SM} =$  $(p \ 1-p|q \ 1-q)_S^T \otimes (1 \ 0|1 \ 0)_M^T$  of the system and memory can be written as follows, where  $P_{in}(a = i, a' = j|X = k, X' = l)$  denotes the probability of obtaining the outcomes a = i and a' = j when performing the fiducial measurements X = k and X' = l on the system and memory respectively, in the initial state  $\vec{P}_{in}^{SM}$ .

$$\vec{P}_{in}^{SM} = \begin{pmatrix} P_{in}(a=0,a'=0|X=0,X'=0) \\ P_{in}(a=1,a'=0|X=0,X'=0) \\ P_{in}(a=1,a'=0|X=0,X'=0) \\ P_{in}(a=1,a'=1|X=0,X'=0) \\ P_{in}(a=0,a'=1|X=0,X'=1) \\ P_{in}(a=1,a'=0|X=0,X'=1) \\ P_{in}(a=1,a'=0|X=1,X'=0) \\ P_{in}(a=1,a'=0|X=1,X'=0) \\ P_{in}(a=1,a'=0|X=1,X'=0) \\ P_{in}(a=1,a'=0|X=1,X'=0) \\ P_{in}(a=0,a'=1|X=1,X'=1) \\ P_{in}(a=1,a'=0|X=1,X'=1) \\ P_{in}(a=1,a'=0|X=1,X'=1) \\ P_{in}(a=1,a'=1|X=1,X'=1) \\ P_{in}(a=$$

The rest of the proof proceeds as follows: we first describe a final state  $\vec{P}_{fin}^{SM}$  of the system and memory and a corresponding memory update map U that satisfy Definition 8.2.7 of a generalized information-preserving memory update. Then, we show that this map can be seen an allowed box world transformation which completes the proof.

<sup>&</sup>lt;sup>12</sup>An isometry since it introduces an initial pure state on M, followed by a joint unitary on SM.

<sup>&</sup>lt;sup>13</sup>It does not matter which pure state the memory is initialized in, a similar argument applies in all cases.

If an agent performs a measurement on the system, the state of the memory must be updated depending on the outcome and the final state of the system and memory after the measurement must hence be a correlated (i.e., a non-product) state. Although the full state space of the 2 gbit system SM is characterised by the 4 fiducial measurements  $(X, X') \in \{(0,0), (0,1), (1,0), (1,1)\}$ , Definition 8.2.7 allows us to restrict possible final states to a useful subspace of this state space that contain correlated states of a certain form. The definition requires that for every map  $\mathscr{E}_S$  on the system before measurement, there exists a corresponding map  $\mathscr{E}_{SM}$  on the system and memory after the measurement that is operationally identical. Thus it suffices if the joint final state  $\vec{P}_{fin}^{SM}$  belongs to a subspace of the 2 gbit state space for which only 2 of the 4 fiducial measurements are relevant for characterising the state, namely any 2 fiducial measurements on  $P_{fin}^{SM}$  that are isomorphic to the 2 fiducial measurements on  $\vec{P}_{in}^{S}$ . Note that by definition of fiducial measurements, the outcome probabilities of any measurement can be found given the outcome probabilities of all the fiducial measurements and without loss of generality, we will only consider the case where the agents perform fiducial measurements on their systems.

A natural isomorphism between fiducial measurements on  $\vec{P}_{in}^S$  and those on  $\vec{P}_{fin}^{SM}$  to consider here (in analogy with the quantum case) is:  $X = i \Leftrightarrow (X, X') = (i, i)$ ,  $\forall i \in \{0, 1\}$  i.e., only consider the cases where the fiducial measurements performed on S and M are the same. Now, in order for the states to be isomorphic or operationally equivalent, one requires that performing the fiducial measurements (X, X') = (i, i) on  $\vec{P}_{fin}^{SM}$  should give the same outcome statistics as measuring X = 0 on  $\vec{P}_{in}^S$ . This can be satisfied through an identical isomorphism on the outcomes:  $a = i \Leftrightarrow (a, a') = (i, i)$ ,  $\forall i \in \{0, 1\}$ . Then the final state of the system and memory,  $\vec{P}_{fin}^{SM}$  will be of the form

$$\vec{P}_{fin}^{SM} = \begin{pmatrix} P_{fin}(a=0,a'=0|X=0,X'=0) \\ P_{fin}(a=1,a'=0|X=0,X'=0) \\ P_{fin}(a=1,a'=0|X=0,X'=0) \\ P_{fin}(a=1,a'=1|X=0,X'=1) \\ P_{fin}(a=0,a'=0|X=0,X'=1) \\ P_{fin}(a=1,a'=0|X=0,X'=1) \\ P_{fin}(a=1,a'=0|X=1,X'=0) \\ P_{fin}(a=0,a'=0|X=1,X'=0) \\ P_{fin}(a=1,a'=0|X=1,X'=0) \\ P_{fin}(a=1,a'=0|X=1,X'=1) \\ P_{fin}(a=0,a'=0|X=1,X'=1) \\ P_{fin}(a=0,a'=0|X=1,X'=1) \\ P_{fin}(a=1,a'=0|X=1,X'=1) \\ P_{fin}(a=1,a'=0|X=1,X'=1) \\ P_{fin}(a=1,a'=0|X=1,X'=1) \\ P_{fin}(a=1,a'=0|X=1,X'=1) \\ P_{fin}(a=1,a'=0|X=1,X'=1) \\ P_{fin}(a=1,a'=0|X=1,X'=1) \\ P_{fin}(a=1,a'=1|X=1,X'=1) \\ P_{fin}(a=1,a'=1|X=1,X'$$

where \* are arbitrary, normalised entries and where  $P_{fin}(a = i, a' = j | X = k, X' = l)$  denotes the probability of obtaining the outcomes a = i and a' = j when performing the fiducial measurements X = k and X' = l on the system and memory respectively, in the final state  $\tilde{P}_{fin}^{SM}$ . This final state can be compressed since the only relevant and non-zero probabilities in  $\tilde{P}_{fin}^{SM}$  occur when X = X' and a = a'. We can then define new variables  $\tilde{X}$  and  $\tilde{a}$  such that  $X = X' = i \Leftrightarrow \tilde{X} = i$  and  $a = a' = j \Leftrightarrow \tilde{a} = j$  for  $i, j \in \{0, 1\}$  and  $\tilde{P}_{fin}^{SM}$  can equivalently be written as in Equation (8.5) which is clearly of the same form as  $P_{in}^S$ .

$$\vec{P}_{fin}^{SM} \equiv \begin{pmatrix} P(\tilde{a} = 0 | \tilde{X} = 0) \\ P(\tilde{a} = 1 | \tilde{X} = 0) \\ P(\tilde{a} = 0 | \tilde{X} = 1) \\ P(\tilde{a} = 1 | \tilde{X} = 1) \end{pmatrix}_{SM} = \begin{pmatrix} p \\ 1 - p \\ q \\ 1 - q \end{pmatrix}_{SM}$$
(8.5)

Hence the initial state of the system,  $\vec{P}_{in}^S = (p \ 1-p|q \ 1-q)^T$  (which is an arbitrary gbit state) is isomorphic to the final state of the system and memory,  $\vec{P}_{fin}^{SM}$  (as evident from Equation (8.5)) with the same outcome probabilities for X = 0, 1 and  $\tilde{X} = 0, 1$ . This implies that for every transformation  $\mathscr{E}_S$  on the former, there exists a transformation  $\mathscr{E}_{SM}$  on the latter such that for all outside agents  $A_j$  and for all  $p, q \in [0, 1]$  (i.e., all possible input gbit states on the system),  $K_j \phi[\mathscr{E}_S(\vec{P}_{in}^S)] \Rightarrow K_j \phi[\mathscr{E}_{SM} \circ \vec{P}_{fin}^{SM}]$ , where  $\vec{P}_{SM}^{fin} = U(\vec{P}_S^{in})$ . Thus any map U that maps  $\vec{P}_{in}^{SM} = \vec{P}_S^{in} \otimes P_{in}^M$  to  $\vec{P}_{SM}^{fin}$  satisfies Definition 8.2.7.

We now find a valid box world transformation that maps the initial state  $\vec{P}_{in}^{SM}$  (Equation (8.3)) to any final state of the form  $\vec{P}_{fin}^{SM}$  (Equation (8.4)) which would correspond to the memory update map U.

For higher dimensional systems S with n > 2 fiducial measurements,  $X \in \{0, ..., n-1\}$  and k > 2 outcomes taking values  $a \in \{0, ..., k-1\}$ , let  $b_n$  and  $b_k$  be the number of bits required to represent n and k in binary respectively. Then the memory M would be initialized to  $b_k$  copies of the pure state  $\vec{P}_{in,n}^M = (1 \quad 0|...|1 \quad 0)_M^T$  which contains n identical blocks (one for each of the n fiducial measurements). One can then perform the procedure of Figure 8.4 "bitwise" combining each output bit with one pure state of M and apply the same argument to obtain the result. For

<sup>&</sup>lt;sup>14</sup>The output wires of boxes carry classical information after the measurement.

the specific case of the memory update transformation of Figure 8.3a, this would correspond to a bitwise CNOT on the output wires of S and M.

#### 8.6.1.2 Memory update for two labs sharing initial correlations

So far, we have considered a single agent measuring a system in her lab. We can also consider situations where multiple agents jointly share a state and measure their local parts of the state, updating their corresponding memories. One might wonder whether the initial correlations in the shared state are preserved once the agents measure it to update their memories (clearly the local measurement probabilities remain unaltered as we saw in this section). The answer is affirmative and this is what allows us to formulate the Frauchiger-Renner paradox in box world as done in the Section 8.4, even though a coherent copy analogous to the quantum case does not exist here.

**Theorem 8.6.2.** Suppose that Alice and Bob share an arbitrary bipartite state  $\vec{P}_{in}^{PR}$  (which may be correlated), locally perform a fiducial measurement on their half of the state and store the outcome in their local memories A and B. Then the final joint state  $\vec{P}_{fin}^{\tilde{A}\tilde{B}}$  of the systems  $\tilde{A} := PA$  and  $\tilde{B} := RB$  as described by outside agents is isomorphic to  $\vec{P}_{in}^{PR}$  with the systems  $\tilde{A}$  and  $\tilde{B}$  taking the role of the systems P and R i.e., local memory updates by Alice and Bob preserve any correlations initially shared between them.

*Proof.* In the following, we describe the proof for the case where the bipartite system shared by Alice and Bob consists of 2 gbits, however, the result easily generalises to arbitrary higher dimensional systems by the argument presented in the last paragraph of the proof of Theorem 8.6.1.

Let  $\vec{P}_{in}^{PR}$  be an arbitrary 2 gbit state with entries  $P_{in}(ab = ij|XY = kl)$   $(i, j, k, l \in \{0, 1\})$ , which correspond to the joint probabilities of Alice and Bob obtaining the outcomes a = i and b = jwhen measuring X = k and Y = l on the P and R subsystems when sharing that initial state. Let  $X', a' \in \{0, 1\}$  and  $Y', b' \in \{0, 1\}$  be the fiducial measurements and outcomes for the memory systems A and B (also gbits) respectively. We describe the measurement and memory update process for each agent separately and characterise the final state of Alice's and Bob's systems and memories after the process as would appear to outside agents who do not have access to Alice and Bob's measurement outcomes. This analysis does not depend on the order in which Alice and Bob perform the measurement as the correlations are symmetric between them, so without loss of generality, we can consider Bob's measurement first and then Alice's.

Suppose that Bob's memory *B* is initialised to the state  $\vec{P}_{in}^B = \vec{P}_1^B = (1 \ 0|1 \ 0)_B^T$ . Then the joint initial state of the Alice's and Bob's system and Bob's memory as described by an agent Wigner outside Bob's lab is  $\vec{P}_{in}^{PRB} = \vec{P}_{in}^{PR} \otimes \vec{P}_1^B$ . This can be expanded as follows where  $P_{in}(abb' = ijk|XYY' = lmn)$  represents the probability of obtaining the binary outcomes a = i, b = j, b' = k when performing the binary fiducial measurements X = l, Y = m, Y' = n on the initial state  $\vec{P}_{in}^{PRB}$ .
$$\vec{P}_{in}^{PRB} = \begin{pmatrix} P_{in}(abb' = 000|XYY' = 000) \\ P_{in}(abb' = 010|XYY' = 000) \\ P_{in}(abb' = 010|XYY' = 000) \\ P_{in}(abb' = 101|XYY' = 000) \\ P_{in}(abb' = 101|XYY' = 000) \\ P_{in}(abb' = 110|XYY' = 000) \\ P_{in}(abb' = 111|XYY' = 000) \\ P_{in}(abb' = 111|XYY' = 000) \\ P_{in}(abb' = 111|XYY' = 000) \\ P_{in}(abb' = 000|XYY' = 111) \\ P_{in}(abb' = 001|XYY' = 111) \\ P_{in}(abb' = 010|XYY' = 111) \\ P_{in}(abb' = 010|XYY' = 111) \\ P_{in}(abb' = 100|XYY' = 111) \\ P_{in}(abb' = 110|XYY' = 111) \\ P_{in}(abb' = 111|XYY' = 111) \\ P_{in}(abb' = 111|XY' = 11) \\ P_{in}(abb' = 111|XY' = 111) \\ P_{in}(abb'$$

 $\vec{P}_{in}^{PRB}$  has 8 blocks  $G_{XYY'}$ , one for each value of (X, Y, Y') and is a product state with 4 equal pairs of blocks,  $G_{000}^{in} = G_{001}^{in}, G_{010}^{in} = G_{011}^{in}, G_{100}^{in} = G_{101}^{in}, G_{110}^{in} = G_{111}^{in}$  since both measurements on the initial state of B give the same outcome.

Now, the outside observer Wigner will describe the transformation on RB through the memory update map U of Figure 8.4. Let  $\vec{P}_{fin}^{PRB}$  be the final state that results by applying this map to the systems RB in the initial state  $\vec{P}_{in}^{PRB}$ . Any transformation on a system characterised by n fiducial measurements with k outcomes each can be represented by a  $nk \times nk$  block matrix where each block is a  $k \times k$  matrix (see [16] for further details), for the system RB, n = k = 4and the memory update map  $U_{RB}$  would be a 16 × 16 block matrix of the following form where each  $U_{ij}$  is a 4 × 4 matrix.

$$U_{RB} = \begin{pmatrix} U_{11} & \cdot & \cdot & U_{14} \\ \hline \cdot & & & \cdot \\ \cdot & & & \cdot \\ \hline \cdot & & & \cdot \\ \hline U_{41} & \cdot & \cdot & U_{44} \end{pmatrix}_{RB}$$

Here, the first 4 rows decide the entries in the first block of the transformed matrix, the next 4, the second block and so on. Noting that the memory update transformation (Figure 8.4) merely permutes elements within the relevant blocks (and does not mix elements between different blocks), the only non-zero blocks of  $U_{RB}$  are the diagonal ones  $U_{ii}$ . Further, by the same argument as in Theorem 8.6.1, the only relevant entries in the transformed state are when the same fiducial measurement is performed on Bob's system R and memory B i.e., only cases where Y = Y'. The remaining measurement choices maybe arbitrary for the final state (just as



Figure 8.4: Classical circuit decomposition of the memory update map U as a box world transformation: The blue box represents the final state of the system S and memory M after the memory update characterised by the fiducial measurements X and X' and the outcomes a, a'. Let U be the memory update map that maps the initial state  $\vec{P}_{in}^{SM}$  to a final state  $\vec{P}_{fin}^{SM}$ . Noting that we only need to consider the case of X = X' since the for  $X \neq X'$ , the entries of  $\vec{P}_{fin}^{SM}$  can be arbitrary, the action of  $\mathcal{T}$ is equivalent to the circuit shown here i.e., 1) Choose  $X_1 = X(=X')$  and perform this fiducial measurement on the initial state of the system  $\vec{P}_{in}^S$  to obtain the outcome  $a_1$ . 2) Fix  $X_2 = 0$  (or  $X_2 = 1$ ) and perform this fiducial measurement on the initial state of the memory  $\vec{P}_{in}^M = (1 \quad 0|1 \quad 0)_M^T$  to obtain the outcome  $a_2$ . 3) Set  $a = a_1$ . 4) If  $a_1 = 1$ , set  $a' = a_2$ , otherwise set  $a' = a_2 \oplus 1$ , where  $\oplus$  denotes modulo 2 addition.

they are for  $X \neq X'$  in Equation (8.4)). This means that among the 4 diagonal blocks, only 2 of them are relevant. The 4 fiducial measurements on RB are YY' = 00, 01, 10, 11 and in that order, only the first and fourth are relevant since they correspond to Y = Y'. Within these relevant blocks (in this case  $U_{11}$  and  $U_{44}$ ), the operation is a CNOT on the output b' controlled by the output b and we have the following matrix representation of the memory update map U of Figure 8.4<sup>15</sup>.

$$U_{RB} = \begin{pmatrix} CN & 0 & 0 & 0 \\ \hline 0 & * & 0 & 0 \\ \hline 0 & 0 & * & 0 \\ \hline 0 & 0 & 0 & CN \end{pmatrix}_{RB}, \qquad CN = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 1 \\ 0 & 0 & 1 & 0 \end{pmatrix}$$
(8.7)

<sup>&</sup>lt;sup>15</sup>The memory update map corresponding to the circuit of Figure 8.3a is a specific case of this map where the arbitrary blocks \* are also equal to CN

where 0 represents the  $4 \times 4$  null matrix and blocks labelled \* can be arbitrary. The final state  $\vec{P}_{fin}^{PRB}$  as seen by Wigner is then

$$\vec{P}_{fin}^{PRB} = (\mathcal{I}_P \otimes U_{RB})\vec{P}_{in}^{PRB} = (\mathcal{I}_P \otimes U_{RB})\left[\vec{P}_{in}^{PR} \otimes \left(\frac{1}{0}\right)_{R}\right], \tag{8.8}$$

where  $\mathcal{I}_P$  is the identity transformation on the P system. Since the CN blocks are the only relevant blocks in  $u_{RB}$  and each block of  $\vec{P}_{in}^{PRB}$  has the same pattern of non-zero and zero entries (Equation (8.6)), it is enough to look at the action of  $\mathcal{I}_P \otimes CN$  on the first block  $G_{000}^{in}$ of  $\vec{P}_{in}^{PRB}$ . Noting that  $\mathcal{I}_P$  is a 2 × 2 identity matrix, we have

where  $P_{fin}(abb' = ijk|XYY' = lmn)$  represents the probability of obtaining the outcomes a = i, b = j, b' = k when performing the fiducial measurements X = l, Y = m, Y' = n on the final state  $\vec{P}_{fin}^{PRB}$  and  $G_{000}^{fin}$  is the first block of this final state. Clearly the only non-zero outcome probabilities are when b = b' and this allows us to compress the final state by defining  $\tilde{b} = i \Leftrightarrow b = b' = i$  for  $i \in \{0, 1\}$  and we have the following.

$$(\mathcal{I}_P \otimes CN)G_{000}^{in} \equiv \begin{pmatrix} P_{in}(ab = 00|XY = 00) \\ P_{in}(ab = 01|XY = 00) \\ P_{in}(ab = 10|XY = 00) \\ P_{in}(ab = 11|XY = 00) \end{pmatrix} = \begin{pmatrix} P_{fin}(a\tilde{b} = 00|XYY' = 000) \\ P_{fin}(a\tilde{b} = 01|XYY' = 000) \\ P_{fin}(a\tilde{b} = 10|XYY' = 000) \\ P_{fin}(a\tilde{b} = 11|XYY' = 000) \end{pmatrix} = G_{00}^{in}$$

Here  $G_{00}^{in}$  is the first block of the initial state  $\vec{P}_{in}^{PR}$  and we have that the first block of the final state of PRB is equivalent (up to zero entries) to the first block of the initial state over PR alone or  $G_{000}^{fin} = G_{00}^{in}$ . Among the 8 blocks of  $\vec{P}_{fin}^{PRB}$ , only the 4 blocks  $G_{000}^{fin}, G_{011}^{fin}, G_{100}^{fin}$  and  $G_{111}^{fin}$  are

the relevant ones (since Y = Y' for these) and we can similarly show that  $G_{011}^{fin} \equiv G_{01}^{in}, G_{100}^{fin} \equiv G_{10}^{in}$ and  $G_{111}^{fin} \equiv G_{11}^{in}$  for the remaining 3 relevant blocks. Defining  $\tilde{Y} = i \Leftrightarrow Y = Y' = i$  for  $i \in \{0, 1\}$ , we obtain

$$\vec{P}_{fin}^{PRB} = \vec{P}_{fin}^{P\tilde{B}} = \vec{P}_{fin}^{P\tilde{B}} = \frac{\vec{P}_{fin}^{PR}(a\tilde{b} = 0|X\tilde{Y} = 00)}{P_{fin}(a\tilde{b} = 10|X\tilde{Y} = 00)} \\ \vec{P}_{fin}(a\tilde{b} = 10|X\tilde{Y} = 00) \\ \vec{P}_{fin}(a\tilde{b} = 11|X\tilde{Y} = 00) \\ \vec{P}_{fin}(a\tilde{b} = 00|X\tilde{Y} = 01) \\ \vec{P}_{fin}(a\tilde{b} = 01|X\tilde{Y} = 01) \\ \vec{P}_{fin}(a\tilde{b} = 01|X\tilde{Y} = 01) \\ \vec{P}_{fin}(a\tilde{b} = 10|X\tilde{Y} = 01) \\ \vec{P}_{fin}(a\tilde{b} = 10|X\tilde{Y} = 01) \\ \vec{P}_{fin}(a\tilde{b} = 01|X\tilde{Y} = 10) \\ \vec{P}_{fin}(a\tilde{b} = 01|X\tilde{Y} = 11) \\ \vec{P}_{fin}(a\tilde{b} = 10|X\tilde{Y} = 11) \\ \vec{P}_{fin}(a\tilde{b} = 11|X\tilde{Y} = 11) \\$$

Equation (8.9) shows that final state  $\vec{P}_{fin}^{P\tilde{B}}$  of Alice's system P, Bob's system R and Bob's memory B after Bob's local memory update is isomorphic to the initial state  $\vec{P}_{in}^{PR}$  shared by Alice and Bob, having the same outcome probabilities as the latter for all the relevant measurements. Thus the initial correlations present in  $\vec{P}_{in}^{PR}$  are preserved after Bob locally updates his memory according to the update procedure of Figure 8.4. One can now repeat the same argument for Alice's local memory update taking  $\vec{P}_{fin}^{P\tilde{B}} \otimes (1 \ 0|1 \ 0)_A^T$  to be the initial state and by analogously defining  $\tilde{s} = i \Leftrightarrow s = s' = i$  for  $s \in \{a, X\}, i \in \{0, 1\}$ , we have the required result that the final state after both parties perform their local memory updates (as described by outside agents Ursula and Wigner) is isomorphic and operationally equivalent to the initial state shared by the parties before the memory update.

$$\vec{P}_{fin}^{PARB} = \vec{P}_{fin}^{\tilde{A}\tilde{B}} \equiv \vec{P}_{in}^{PR} \tag{8.10}$$

## CHAPTER 9

## Conclusions and outlook

This two part thesis proposes new techniques for analysing acyclic and cyclic causal structures as well as multi-agent logical paradoxes in non-classical theories. By developing these techniques, we have derived interesting preliminary insights and have identified a number of possible directions for future work on these aspects of quantum and post-quantum theories, which have both foundational as well as practical applications.

The results reported in Part I pertain to the entropic analysis of causal structures (Chapters 4 and 5) and to the connections between the notions of causation and space-time (Chapter 6). In the former case, we have developed a systematic method for analysing causal structures using generalised entropies, providing, in particular, a new set of Tsallis entropic inequalities for the Triangle causal structure. We have also extended our analysis to post-selected causal structures and identified families of distributions whose non-classicality can be detected in Bell scenarios with non-binary outcomes. Further, the properties of classical and quantum Tsallis entropies derived in Chapter 4 can be potentially useful beyond the analysis of causal structures, since Tsallis entropies are widely employed in other areas of information theory as well as non-extensive statistical physics.

On the other hand, our results also reveal significant drawbacks of the entropic technique for certifying the non-classicality of causal structures, both with and without the additional post-selection method. We identified the main reasons for the computational intractability of the entropy vector method using Tsallis entropies. Although we were able to circumvent this issue to derive new Tsallis entropic inequalities for the Triangle scenario, no quantum violations of these inequalities were found even by known non-classical distributions in the scenario. We noted that Rényi entropies do not seem to overcome these limitations either, due to the lack of linear constraints required to construct a good initial outer approximation to the Rényi entropy cone. In post-selected causal structures, we provided numerical and analytical evidence suggesting that both Shannon and Tsallis entropic inequalities are in general insufficient for detecting non-classicality in Bell scenarios with binary inputs and three or more outcomes per party.

Limitations for distinguishing between classical and non-classical correlations in causal structures using Shannon entropies have been previously encountered in the absence of post-selection [214, 215]. By showing that similar limitations arise even for generalised entropies and in the presence of post-selection, our results shed further light on the debate regarding the usefulness of the entropic method for this purpose. It appears that the advantages of working in entropy space such as convexity and better scaling of the number of bounding inequalities come at the cost of losing valuable information about the non-classical nature of the correlations, even in post-selected causal structures. Our results illuminate some of the merits as well as limitations of the entropic technique and point to the need for developing novel methods for understanding causation in non-classical theories.

In general, this is a difficult task—fully characterising the bounding inequalities for the set of classical correlations in a given causal structure is known to be an NP-complete problem [169]. Therefore, increase in the computational power of current technologies and improvements to the performance of variable elimination algorithms would naturally increase the set of scenarios that can be characterised. For example, developing a variable elimination algorithm (such as Fourier-Motzkin) capable of dealing with systems of non-linear constraints would allow for tighter entropic characterisations through the inclusion of non-linear inequalities. One can also consider other algorithms for polyhedral projection that can sometimes produce results even when Fourier-Motzkin Elimination (FME) becomes intractable [99]. We have provided a method to bypass the FME to derive generalised entropic inequalities in cases where Shannon entropic inequalities are known. Our Tsallis inequalities thus derived depend on the cardinality of the variables in the Triangle causal structure. Hence one could, in principle, investigate dimension dependent distinctions between classical and quantum correlations using these, while also taking into account the new class of non-local distributions recently discovered in the Triangle network [178, 129]. Other possibilities include employing the inflation technique [223] in combination with the entropic method, constructing more efficient algorithms (than the random search typically employed) for finding quantum violations, and considering an alternative definition of the Tsallis conditional entropies [3] that satisfy simpler causal constraints but violate the chain rule. These methods could lead to improvements but are not guaranteed to be computationally tractable.

Cryptographic and other information-processing applications have greatly benefited from the characterisation of the sets of correlations in the Bell scenario [79, 145, 17, 168, 60, 59], which are relatively well understood due to their convexity. Often, these applications also go beyond the Bell scenario and involve networks of parties sharing multiple sources of entanglement, where the set of observed correlations in probability space is no longer convex. The entanglement swapping protocol [26, 229] is an important example, which also plays a role in the construction of quantum repeater networks [35] used in quantum communication scenarios and the quantum internet. Therefore, exploring different techniques for certifying non-classicality in and beyond the Bell causal structure, and understanding their advantages and limitations as done in this thesis is also of practical importance. At the foundational level, this reveals fundamental distinctions between the principles of causality that manifest themselves in classical and quantum regimes. The information-theoretic approach to causal structures also allow us to consider post-quantum theories and identify physical principles other than relativistic ones that single out quantum theory among this spectrum of theories.

In Chapter 6, we have further explored this fundamental aspect by developing a causal modelling framework for a general class of physical theories, and formalising relationships between causality, space-time and relativistic principles (such as the inability to signal superluminally) therein. This allowed us to analyse scenarios involving superluminal influences and causal loops that nevertheless do not lead to superluminal signalling at the operational level. In particular, our framework provides an initial tool set for causally modelling post-quantum theories admitting jamming non-local correlations [105, 116], which have not been rigorously analysed before, and for identifying further physical principles underlying such theories.

Moreover, these results also generate scope for interesting future investigations into causality and space-time structure. One promising avenue would be to extend this work to construct a more complete framework for causally modelling cyclic and fine-tuned causal structures in non-classical theories, which is likely to be of independent interest. Furthermore, this work can aid an operational analysis of closed time-like curves (CTCs) in the presence of fine-tuned causal influences, and an investigation of whether fine-tuned and non fine-tuned interpretations of quantum theory behave differently in the presence of CTCs. Yet another direction is to examine simulations of indefinite causal structures [111, 154, 11] using CTCs and fine-tuning, and analyse their embeddings within a fixed space-time structure. Indefinite causal structures can in theory be implemented through a quantum superposition of gravitating masses [230], they have also been claimed to be physically implemented by superposing the orders of quantum operations on a target system [173, 184]. The space-time information associated with these operations plays an important role in distinguishing between the two implementations  $[156]^1$ . Further, general relativity is known to admit CTC solutions [204, 100] even in the absence of quantum effects, while certain fundamental principles of quantum theory such as the nocloning principle and linearity may no longer hold in the presence of CTCs [75]. A deeper understanding of causality and space-time structure in non-classical theories, in the presence of causal loops would hence provide insights not only into quantum theory, but potentially also into its unification with general relativity, which has been a longstanding open problem in theoretical physics. The thesis provides a small contribution towards these larger goals.

Part II of this thesis contributes to an understanding of multi-agent logic in scenarios where reasoning agents model each other's memories as systems of a physical theory. Generalising the conditions of the Frauchiger-Renner quantum experiment to generalised probabilistic theories allowed us to derive a stronger paradox in the particular GPT of box world. Our result reveals that reversibility of the memory update akin to quantum unitarity is not necessary for deriving such multi-agent paradoxes. Rather, a logical form of contextuality along with an appropriate notion of an information-preserving memory update appear to be more fundamental for deriving a deterministic paradox. This work paves the way for a more general characterisation of Frauchiger-Renner type paradoxes, their relationships to contextuality and the structure of multi-agent logic in quantum and post-quantum theories.

Apart from the implications for logic, these multi-agent paradoxes also challenge standard approaches to causal modelling that we have discussed in the first part of this thesis. In these

<sup>&</sup>lt;sup>1</sup>As we also find in: Vilasini, V., del Rio, L. and Renner, R. *Causality in definite and indefinite space-times*. In preparation (2020). https://wdi.centralesupelec.fr/users/valiron/qplmfps/papers/qs01t3.pdf

settings, a system, such as a measurement outcome that appears classical to one agent may be modelled as an entangled subsystem (entangled with the first agent's memory) by another agent. Therefore the distinction between observed and unobserved systems in a causal structure can become subjective, and a detailed investigation of causality in these general settings is yet to be undertaken.

The fact that the agents do not necessarily model the measurement outcomes of other agents as classical systems implies that one cannot assume the existence of a joint distribution over all these outcomes. As we have also seen in Part I of this thesis, no joint distribution can in general be assigned to non-coexisting systems in a non-classical causal structure. It is precisely this inability to make a global assignment of values to all measurement outcomes that results in the apparent paradox, suggesting that these outcomes cannot be seen as objective facts about the world, but only subjective to the agents who observed them. In both the quantum and box-world versions of the paradox, the states and measurements involved produce non-classical correlations in the bipartite Bell causal structure. The treatment of agents as systems of a non-classical physical theory promotes this non-classical correlations can lead to such paradoxes. A better understanding of non-classical correlations in causal structures would hence enhance our knowledge about these general settings involving complex quantum systems that are capable of logical deductions.

With sufficient technological progress in creating and manipulating stable many-qubit superpositions, the physical implementation of multi-agent paradoxes may be a distinct possibility in the near future, since a relatively small quantum computer would fulfill the role of an agent for these purposes. There are global efforts towards achieving such macroscopic superpositions, both for the practical purposes of scalable quantum computing and for pushing the horizons of fundamental physics by probing the quantum effects of gravity. These make it imperative to develop more general theoretical frameworks for modelling multi-agent paradoxes as well as causality and space-time structure in non-classical theories. We believe there are several deep connections between causality and multi-agent logical paradoxes in non-classical theories that are yet to be discovered, and hope that the results presented in both parts of this thesis contribute towards a more complete physical framework or theory in the future.

## Bibliography

- [1] AARONSON, S. Why philosophers should care about computational complexity. arXiv:1108.1791 (2011).
- [2] AARONSON, S., AND WATROUS, J. Closed timelike curves make quantum and classical computing equivalent. Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences 465, 2102 (2008), pp. 631–647.
- [3] ABE, S., AND RAJAGOPAL, A. Nonadditive conditional entropy and its significance for local realism. *Physica A: Statistical Mechanics and its Applications 289*, 1 (2001), pp. 157–164.
- [4] ABRAMSKY, S., BARBOSA, R. S., KISHIDA, K., LAL, R., AND MANSFIELD, S. Contextuality, Cohomology and Paradox. 24th EACSL Annual Conference on Computer Science Logic (CSL 2015) 41 (2015), Ed. Stephan Kreutzer, pp. 211–228.
- [5] ACIN, A., AND MASANES, L. Certified randomness in quantum physics. *Nature 540*, 7632 (2016), pp. 213–219.
- [6] ALLEN, J.-M. A., BARRETT, J., HORSMAN, D. C., LEE, C. M., AND SPEKKENS, R. W. Quantum common causes and quantum causal models. *Physical Review X* 7, 3 (2017), p. 031021.
- [7] ARAKI, H., AND LIEB, E. H. Entropy inequalities. Communications in Mathematical Physics 18, 2 (1970), pp. 160–170.
- [8] ARAÚJO, M., BRANCIARD, C., COSTA, F., FEIX, A., GIARMATZI, C., AND BRUKNER, Č. Witnessing causal nonseparability. New Journal of Physics 17, 10 (2015), p. 102001.
- [9] ARAÚJO, M., COSTA, F., AND BRUKNER, Č. Computational Advantage from Quantum-Controlled Ordering of Gates. *Physical Review Letters* 113, 25 (2014), p. 250402.
- [10] ARAÚJO, M., FEIX, A., NAVASCUÉS, M., AND BRUKNER, Č. A purification postulate for quantum mechanics with indefinite causal order. *Quantum 1* (2016), p. 10.
- [11] ARAÚJO, M., GUÉRIN, P. A., AND BAUMELER, A. Quantum computation with indefinite causal structures. *Physical Review A 96* (2017), p. 052315.

- [12] ARNON-FRIEDMAN, R., DUPUIS, F., FAWZI, O., RENNER, R., AND VIDICK, T. Practical device-independent quantum cryptography via entropy accumulation. *Nature Communications* 9, 1 (2018), p. 459.
- [13] AUDENAERT, K. Subadditivity of q-entropies for q > 1. Journal of Mathematical Physics 48 (2007), p. 083507.
- [14] AVIS, D., IMAI, H., ITO, T., AND SASAKI, Y. Deriving tight Bell inequalities for 2 parties with many 2-valued observables from facets of cut polytopes. arxiv:quantph/0404014 (2004).
- [15] BANCAL, J.-D., GISIN, N., AND PIRONIO, S. Looking for symmetric Bell inequalities. Journal of Physics A: Mathematical and Theoretical 43, 38 (2010), p. 385303.
- [16] BARRETT, J. Information processing in generalized probabilistic theories. *Physical Review A* 75 (2007), p. 032304.
- [17] BARRETT, J., HARDY, L., AND KENT, A. No signaling and quantum key distribution. *Physical Review Letters 95* (2005), p. 010503.
- [18] BARRETT, J., LORENZ, R., AND ORESHKOV, O. Cyclic quantum causal models. arxiv:2002.1215731 (2020).
- [19] BAUMANN, V., AND WOLF, S. On Formalisms and Interpretations. Quantum 2 (2018), p. 99.
- [20] BAUMELER, A., DEGORRE, J., AND WOLF, S. Bell correlations and the common future. Quantum Foundations, Probability and Information (2018), pp. 255–268.
- [21] BAUMELER, A., AND WOLF, S. The space of logically consistent classical processes without causal order. *New Journal of Physics*, 1 (2016), p. 013036.
- [22] BELL, J. S. On the Einstein Podolsky Rosen paradox. Physics Physique Fizika 1 (1964), pp. 195–200.
- [23] BEMPORAD, A., FUKUDA, K., AND TORRISI, F. D. Convexity recognition of the union of polyhedra. *Computational Geometry* 18, 3 (2001), pp. 141–154.
- [24] BENNETT, C., BRASSARD, G., CREPEAU, C., AND MAURER, U. Generalized privacy amplification. *IEEE Transactions on Information Theory* 41 (1995), pp. 1915–1923.
- [25] BENNETT, C., AND SCHUMACHER, B. Talk at QUPON Wien, (2005).
- [26] BENNETT, C. H., BRASSARD, G., CRÉPEAU, C., JOZSA, R., PERES, A., AND WOOT-TERS, W. K. Teleporting an unknown quantum state via dual classical and Einstein-Podolsky-Rosen channels. *Physical Review Letters* 70 (1993), pp. 1895–1899.
- [27] BOHM, D. A suggested interpretation of the quantum theory in terms of "hidden" variables. I. *Physical Review* 85 (1952), pp. 166–179.

- [28] BOHR, N. Discussions with Einstein on epistemological problems in atomic physics. Albert Einstein: Philosopher-Scientist, The Library of Living Philosophers, Ed. Schilpp, P. A. (1949), pp. 200–241.
- [29] BOLTZMANN, L. Über die mechanische bedeutung des zweiten hauptsatzes der wärmetheorie. Wiener Berichte 53 (1866), pp. 195–220.
- [30] BONG, K.-W., UTRERAS-ALARCÓN, A., GHAFARI, F., LIANG, Y.-C., TISCHLER, N., CAVALCANTI, E. G., PRYDE, G. J., AND WISEMAN, H. M. A strong no-go theorem on the Wigner's friend paradox. *Nature Physics* (2020), https://doi.org/10.1038/s41567-020-0990-x.
- [31] BOYD, S., AND VANDENBERGHE, L. Convex optimization. *Cambridge University Press* (2004).
- [32] BRANCIARD, C., ARAÚJO, M., FEIX, A., COSTA, F., AND BRUKNER, Č. The simplest causal inequalities and their violation. New Journal of Physics 18, 1 (2016), p. 13008.
- [33] BRANCIARD, C., ROSSET, D., GISIN, N., AND PIRONIO, S. Bilocal versus nonbilocal correlations in entanglement-swapping experiments. *Physical Review A* 85 (2012), p. 032119.
- [34] BRAUNSTEIN, S. L., AND CAVES, C. M. Information-theoretic Bell inequalities. *Physical Review Letters 61* (1988), pp. 662–665.
- [35] BRIEGEL, H.-J., DÜR, W., CIRAC, J. I., AND ZOLLER, P. Quantum repeaters: The role of imperfect local operations in quantum communication. *Physical Review Letters 81* (1998), pp. 5932–5935.
- [36] BRUKNER, Č. A no-go theorem for observer-independent facts. Entropy 20, 350 (2018).
- [37] BRUKNER, Č., TAYLOR, S., CHEUNG, S., AND VEDRAL, V. Quantum entanglement in time. arXiv:quant-ph/0402127 (2004).
- [38] BRULÉ, J. A causation coefficient and taxonomy of correlation/causation relationships. arxiv:1708.05069 (2017).
- [39] BRUNNER, N., CAVALCANTI, D., PIRONIO, S., SCARANI, V., AND WEHNER, S. Bell nonlocality. *Reviews of Modern Physics 86* (2014), pp. 419–478.
- [40] BUHRMAN, H., CLEVE, R., MASSAR, S., AND DE WOLF, R. Nonlocality and communication complexity. *Reviews of Modern Physics* 82 (2010), pp. 665–698.
- [41] CARNAP, R. Meaning and necessity: a study in semantics and modal logic. University of Chicago Press (1988).
- [42] CASTRO-RUIZ, E., GIACOMINI, F., BELENCHIA, A., AND BRUKNER, C. Quantum clocks and the temporal localisability of events in the presence of gravitating quantum systems. *Nature Communications* 11, 1 (2020), p. 2672.

- [43] CERF, N. J., AND ADAMI, C. Negative entropy and information in quantum mechanics. *Physical Review Letters 79* (1997), pp. 5194–5197.
- [44] CHAVES, R. Entropic inequalities as a necessary and sufficient condition to noncontextuality and locality. *Physical Review A* 87 (2013), p. 022102.
- [45] CHAVES, R., CARVACHO, G., AGRESTI, I., DI GIULIO, V., AOLITA, L., GIACOMINI, S., AND SCIARRINO, F. Quantum violation of an instrumental test. *Nature Physics* 14, 3 (2018), pp. 291–296.
- [46] CHAVES, R., AND FRITZ, T. Entropic approach to local realism and noncontextuality. *Physical Review A* 85 (2012), pp. 032113.
- [47] CHAVES, R., LUFT, L., AND GROSS, D. Causal structures from entropic information: geometry and novel scenarios. New Journal of Physics 16, 4 (2014), p. 043001.
- [48] CHAVES, R., MAJENZ, C., AND GROSS, D. Information-theoretic implications of quantum causal structures. *Nature Communications* 6, 1 (2015), p. 5766.
- [49] CHERNIKOV, S. N. Contraction of systems of linear inequalities. Doklady Akademii Nauk SSSR 131, 3 (1960), pp. 518–521.
- [50] CHERNIKOV, S. N. The convolution of finite systems of linear inequalities. USSR Computational Mathe- matics and Mathematical Physics 5 (1965), pp. 1–24.
- [51] CHIRIBELLA, G., BANIK, M., BHATTACHARYA, S. S., GUHA, T., ALIMUDDIN, M., ROY, A., SAHA, S., AGRAWAL, S., AND KAR, G. Indefinite causal order enables perfect quantum communication with zero capacity channel. arXiv:1810.10457 (2018).
- [52] CHIRIBELLA, G., D'ARIANO, G. M., AND PERINOTTI, P. Probabilistic theories with purification. *Physical Review A 81* (2010), p. 062348.
- [53] CHIRIBELLA, G., D'ARIANO, G. M., PERINOTTI, P., AND VALIRON, B. Quantum computations without definite causal structure. *Physical Review A* 88, 2 (2013), p. 022318.
- [54] CHOI, M.-D. Completely positive linear maps on complex matrices. Linear Algebra and its Applications 10, 3 (1975), pp. 285–290.
- [55] CHRISTOF, T., AND LOEBEL, A. POlyhedron Representation Transformation Algorithm. (PORTA) (1997).
- [56] CLAUSER, J. F., AND HORNE, M. A. Experimental consequences of objective local theories. *Physical Review D 10* (1974), pp. 526–535.
- [57] CLAUSER, J. F., HORNE, M. A., SHIMONY, A., AND HOLT, R. A. Proposed experiment to test local hidden-variable theories. *Physical Review Letters 23* (1969), pp. 880–884.
- [58] CLAUSIUS, R. Ueber die bewegende kraft der wärme und die gesetze, welche sich daraus für die wärmelehre selbst ableiten lassen. Annalen der Physik 155, 3 (1850), pp. 368–397.

- [59] COLBECK, R. Quantum and relativistic protocols for secure multi-party computation. PhD Dissertation, University of Cambridge. arxiv:0911.3814 (2007).
- [60] COLBECK, R., AND KENT, A. Private randomness expansion with untrusted devices. Journal of Physics A 44, 9 (2011), p. 095305.
- [61] COLBECK, R., AND RENNER, R. A short note on the concept of free choice. arxiv:1302.4446 (2013).
- [62] COLBECK, R., AND VILASINI, V. LPAssumptions (Mathematica package). https://github.com/rogercolbeck/LPAssumptions.
- [63] COLLINS, D., AND GISIN, N. A relevant two qubit Bell inequality inequivalent to the CHSH inequality. Journal of Physics A: Mathematical and General 37, 5 (2004), p. 1775.
- [64] COLLINS, D., GISIN, N., LINDEN, N., MASSAR, S., AND POPESCU, S. Bell inequalities for arbitrarily high-dimensional systems. *Physical Review Letters* 88 (2002), p. 040404.
- [65] СООК, R. T. Patterns of paradox. The Journal of Symbolic Logic 69, 3 (2004), pp. 767–774.
- [66] COPE, T., AND COLBECK, R. Bell inequalities from no-signaling distributions. *Physical Review A 100* (2019), p. 022114.
- [67] COSTA, F., AND SHRAPNEL, S. Quantum causal modelling. New Journal of Physics 18, 6 (2016), p. 63032.
- [68] CURADO, E. M. F., AND TSALLIS, C. Generalized statistical mechanics: connection with thermodynamics. *Journal of Physics A* 24 (1991), pp. L69–L72.
- [69] DANTZIG, G. B. Linear programming and extensions. *Princeton University Press* (1963).
- [70] DANTZIG, G. B. Origins of the simplex method. A History of Scientific Computing (1990), pp. 141–151.
- [71] DARÓCZY, Z. Generalized information functions. Information and Control 16, 1 (1970), pp. 36–51.
- [72] DE BROGLIE, L. La mécanique ondulatoire et la structure atomique de la matiére et du rayonnement. *Journal of Physics Radium 8*, 5 (1952), pp. 225–241.
- [73] DEL RIO, L., KRÄMER, L., AND RENNER, R. Resource theories of knowledge. arXiv:1511.08818 (2015).
- [74] DEUTSCH, D. Quantum theory as a universal physical theory. International Journal of Theoretical Physics 24, 1 (1985), pp. 1–41.
- [75] DEUTSCH, D. Quantum mechanics near closed timelike lines. Physical Review D 44 (1991), pp. 3197–3217.
- [76] DEWITT, B. S. Quantum mechanics and reality. *Physical Today* 23 (1970), pp. 155–165.

- [77] DOURDENT, H. A quantum Gödelian hunch. arXiv:2005.04274 (2020).
- [78] DÜRR, D., AND TEUFEL, S. Bohmian mechanics: The physics and mathematics of quantum theory. *Springer* (2009).
- [79] EKERT, A. K. Quantum cryptography based on Bell's theorem. *Physical Review Letters* 67 (1991), pp. 661–663.
- [80] ELGA, A. Self-locating belief and the sleeping beauty problem. Analysis 60, 2 (2000), pp. 143–147.
- [81] EVERETT, H. "Relative state" formulation of quantum mechanics. Reviews of Modern Physics 29 (1957), pp. 454–462.
- [82] FAGIN, R., HALPERN, J. Y., MOSES, Y., AND VARDI, M. Reasoning about knowledge. MIT press (2004).
- [83] FEHR, S., AND BERENS, S. On the conditional Rényi entropy. IEEE Transactions on Information Theory 60 (2014), pp. 6801–6810.
- [84] FORRÉ, P., AND MOOIJ, J. M. Markov properties for graphical models with cycles and latent variables. arxiv:1710.08775 (2017).
- [85] FRAUCHIGER, D., AND RENNER, R. Quantum theory cannot consistently describe the use of itself. *Nature Communications* 9, 1 (2018), p. 3711.
- [86] FREEDMAN, S. J., AND CLAUSER, J. F. Experimental test of local hidden-variable theories. *Physical Review Letters 28* (1972), pp. 938–941.
- [87] FRIEDMAN, M. The fed's thermostat. The Wall Street Journal (2003).
- [88] FRITZ, T. Beyond Bell's theorem: correlation scenarios. New Journal of Physics 14, 10 (2012), p. 103001.
- [89] FRITZ, T., AND CHAVES, R. Entropic inequalities and marginal problems. *IEEE Trans*actions on Information Theory 59, 2 (2013), pp. 803–817.
- [90] FRITZ, T., AND CHAVES, R. Entropic inequalities and marginal problems. *IEEE Trans*actions on Information Theory 59 (2013), pp. 803–817.
- [91] FUCHS, C. A., AND SCHACK, R. Quantum-bayesian coherence. Reviews of Modern Physics 85 (2013), pp. 1693–1715.
- [92] FUKUDA, K. Polyhedral computation (lecture notes), Spring 2016. http://wwwoldurls.inf.ethz.ch/personal/fukudak/lect/pclect/notes2016/.
- [93] FURUICHI, S. Information theoretical properties of Tsallis entropies. Journal of Mathematical Physics 47 (2004).
- [94] FURUICHI, S., YANAGI, K., AND KURIYAMA, K. Fundamental properties of Tsallis relative entropy. *Journal of Mathematical Physics* 45 (2004).

- [95] GARSON, J. Modal logic. The Stanford Encyclopedia of Philosophy (2016), Ed. Edward N. Zalta.
- [96] GEIGER, D. Towards the formalization of informational dependencies. Tech. rep. 880053. UCLA Computer Science (1987).
- [97] GIBBS, J. W. Elementary principles in statistical mechanics. *Charles Scribner's Sons* (1902).
- [98] GIUSTINA, M., VERSTEEGH, M. A., WENGEROWSKY, S., HANDSTEINER, J., HOCHRAINER, A., PHELAN, K., STEINLECHNER, F., KOFLER, J., LARSSON, J.-Å., ABELLÁN, C., AMAYA, W., PRUNERI, V., MITCHELL, M. W., BEYER, J., GERRITS, T., LITA, A. E., SHALM, L. K., NAM, S. W., SCHEIDL, T., URSIN, R., WITTMANN, B., AND ZEILINGER, A. Significant-loophole-free test of Bell's theorem with entangled photons. *Physical Review Letters* 115, 25 (2015), p. 250401.
- [99] GLÄSSLE, T., GROSS, D., AND CHAVES, R. Computational tools for solving a marginal problem with applications in Bell non-locality and causal modeling. *Journal of Physics* A: Mathematical and Theoretical 51, 48 (2018), p. 484002.
- [100] GÖDEL, K. An example of a new type of cosmological solutions of einstein's field equations of gravitation. *Reviews of Modern Physics 21* (1949), pp. 447–450.
- [101] GOLDBLATT, R. Mathematical modal logic: A view of its evolution. Handbook of the History of Logic 7 (2006), pp. 1–98.
- [102] GREENBERGER D.M., HORNE M.A., Z. A. Going beyond Bell's theorem. Quantum Theory and Conceptions of the Universe. Fundamental Theories of Physics 37 (1989).
- [103] GROSS, D., MÜLLER, M., COLBECK, R., AND DAHLSTEN, O. C. O. All reversible dynamics in maximally nonlocal theories are trivial. *Physical Review Letters 104* (2010), p. 080402.
- [104] GRÖTSCHEL, M., LOVASZ, L., AND SHRIJVER, A. Geometric algorithms and combinatorial optimization. Springer Science and Business Media (2012).
- [105] GRUNHAUS, J., POPESCU, S., AND ROHRLICH, D. Jamming nonlocal quantum correlations. *Physical Review A 53* (1996), pp. 3781–3784.
- [106] GUÉRIN, P. A., AND BRUKNER, Č. Observer-dependent locality of quantum events. New Journal of Physics 20, 10 (2018), p. 103031.
- [107] GUÉRIN, P. A., FEIX, A., ARAÚJO, M., AND BRUKNER, Č. Exponential communication complexity advantage from quantum superposition of the direction of communication. *Physical Review Letters* 117 (2016), p. 100502.
- [108] HARDY, L. Quantum mechanics, local realistic theories, and Lorentz-invariant realistic theories. *Physical Review Letters* 68 (1992), pp. 2981–2984.
- [109] HARDY, L. Nonlocality for two particles without inequalities for almost all entangled states. *Physical Review Letters* 71 (1993), pp. 1665–1668.

- [110] HARDY, L. Quantum theory from five reasonable axioms. arXiv:quant-ph/0101012 (2001).
- [111] HARDY, L. Probability theories with dynamic causal structure: A new framework for quantum gravity. arxiv:gr-qc/0509120 (2005).
- [112] HARDY, L. Operational general relativity: Possibilistic, probabilistic, and quantum. arxiv:1608.06940 (2016).
- [113] HEISENBERG, W. Ist eine deterministische ergänzung der quantenmechanik möglich? Wolfgang Pauli. Wissenschaftlicher Briefwechsel mit Bohr, Einstein, Heisenberg, Springer 2 (1935), Ed. Hermann, A. and von Meyenn, K. and Weisskopf, V. F. pp. 409–418.
- [114] HENSEN, B., BERNIEN, H., DRÉAU, A. E., REISERER, A., KALB, N., BLOK, M. S., RUITENBERG, J., VERMEULEN, R. F. L., SCHOUTEN, R. N., ABELLÁN, C., AMAYA, W., PRUNERI, V., MITCHELL, M. W., MARKHAM, M., TWITCHEN, D. J., ELKOUSS, D., WEHNER, S., TAMINIAU, T. H., AND HANSON, R. Loophole-free Bell inequality violation using electron spins separated by 1.3 kilometres. *Nature 526*, 7575 (2015), pp. 682–686.
- [115] HENSON, J., LAL, R., AND PUSEY, M. F. Theory-independent limits on correlations from generalized bayesian networks. New Journal of Physics 16, 11 (2014), p. 113043.
- [116] HORODECKI, P., AND RAMANATHAN, R. The relativistic causality versus no-signaling paradigm for multi-party correlations. *Nature Communications* 10, 1 (2019), p. 1701.
- [117] HORSMAN, D., HEUNEN, C., PUSEY, M. F., BARRETT, J., AND SPEKKENS, R. W. Can a quantum state over time resemble a quantum state at a single time? *Proceedings* of the Royal Society A 473 (2017), p. 20170395.
- [118] HORST, R., AND THOAI, N. V. DC Programming: Overview. Journal of Optimization Theory and Applications 103, 1 (1999), pp. 1–43.
- [119] HU, X., AND YE, Z. Generalized quantum entropy. Journal of Mathematical Physics 47, 2 (2006), p. 023502.
- [120] IWAMOTO, M., AND SHIKATA, J. Revisiting conditional Rényi entropies and generalizing Shannon's bounds in information theoretically secure encryption? *Technical report*, *Cryptology ePrint Archive 440* (2013).
- [121] JAMIOŁKOWSKI, A. Linear transformations which preserve trace and positive semidefiniteness of operators. *Reports on Mathematical Physics* 3, 4 (1972), pp. 275–278.
- [122] KASZLIKOWSKI, D., KWEK, L. C., CHEN, J.-L., ZUKOWSKI, M., AND OH, C. H. Clauser-horne inequality for three-state systems. *Physical Review A* 65 (2002), p. 032118.
- [123] KATTA, M. Linear Programming. John Wiley and Sons (1983), p. 482.
- [124] KHACHIYAN, L., BOROS, E., BORYS, K., ELBASSIONI, K., AND GURVICH, V. Generating all vertices of a polyhedron is hard. *Discrete and Computational Geometry*, 1 (2008), pp. 174–190.

- [125] KIM, J. S. Tsallis entropy and general polygamy of multi-party quantum entanglement in arbitrary dimensions. *Physical Review A 94* (2016).
- [126] KOCHEN, S., AND SPECKER, E. Logical structures arising in quantum theory. in Addison, J., L. Henkin, and A. Tarski (eds.), The theory of models, North-Holland, Amsterdam (1967), pp. 177–189.
- [127] KOCHEN, S., AND SPECKER, E. The problem of hidden variables in quantum mechanics. Indiana University Mathematics Journal 17 (1968), pp. 59–87.
- [128] KÖNIG, R., AND RENNER, R. A de Finetti representation for finite symmetric quantum states. *Journal of Mathematical Physics 46* (2005), p. 122108.
- [129] KRAFT, T., DESIGNOLLE, S., RITZ, C., BRUNNER, N., GÜHNE, O., AND HUBER, M. Quantum entanglement in the triangle network. arXiv:2002.03970 (2020).
- [130] KRÄMER, L., AND DEL RIO, L. Operational locality in global theories. Philosophical Transactions of the Royal Society of London A: Mathematical, Physical and Engineering Sciences 376, 2123 (2018).
- [131] KRIPKE, S. A. Semantical considerations on modal logic. Universal Logic: An Anthology (2012), pp. 197–208.
- [132] LEIFER, M. S., AND SPEKKENS, R. W. Towards a formulation of quantum theory as a causally neutral theory of Bayesian inference. *Physical Review A* 88 (2013), p. 052130.
- [133] LEWIS, C. I. A survey of symbolic logic. University of California Press (1918).
- [134] LEWIS, C. I., LANGFORD, C. H., AND LAMPRECHT, P. Symbolic logic. Dover Publications New York (1959).
- [135] LIEB, E., AND RUSKAI, M. Proof of the strong subadditivity of quantum-mechanical entropy. *Journal of Mathematical Physics* 14 (1973), pp. 1938–1941.
- [136] LINDEN, N., MOSONYI, M., AND WINTER, A. The structure of rényi entropic inequalities. Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences 469, 2158 (2013), p. 20120737.
- [137] LLOYD, S., MACCONE, L., GARCIA-PATRON, R., GIOVANNETTI, V., AND SHIKANO, Y. Quantum mechanics of time travel through post-selected teleportation. *Physical Review D* 84 (2011), p. 025007.
- [138] LLOYD, S., MACCONE, L., GARCIA-PATRON, R., GIOVANNETTI, V., SHIKANO, Y., PIRANDOLA, S., ROZEMA, L. A., DARABI, A., SOUDAGAR, Y., SHALM, L. K., AND STEINBERG, A. M. Closed timelike curves via postselection: Theory and experimental test of consistency. *Physical Review Letters 106* (2011), p. 040403.
- [139] LUDWIG, G. Versuch einer axiomatischen grundlegung der quantenmechanik und allgemeinerer physikalischer theorien. Zeitschrift für Physik 181 (1964), pp. 233–260.

- [140] LUDWIG, G. Attempt of an axiomatic foundation of quantum mechanics and more general theories II. Communications in Mathematical Physics 4, 5 (1967), pp. 331–348.
- [141] LUDWIG, G. Attempt of an axiomatic foundation of quantum mechanics and more general theories III. Communications in Mathematical Physics 9 (1968), pp. 1–12.
- [142] MARTIN, R. K. Large scale linear and integer optimization: A unified approach. Springer Science and Business Media (1999).
- [143] MASANES, L. Tight Bell inequality for d-outcome measurements correlations. Quantum information and computation 3 (2002), p. 345.
- [144] MASANES, L., ACIN, A., AND GISIN, N. General properties of nonsignaling theories. *Physical Review A* 73 (2006), p. 012112.
- [145] MAYERS, D., AND YAO, A. Quantum cryptography with imperfect apparatus. Proceedings of the 39th Annual Symposium on Foundations of Computer Science (FOCS-98) (1998), pp. 503–509.
- [146] MEGIDISH, E., HALEVY, A., SHACHAM, T., DVIR, T., DOVRAT, L., AND EISENBERG, H. S. Entanglement swapping between photons that have never coexisted. *Physical Review Letters* 110 (2013), p. 210403.
- [147] MERMIN, N. D. Simple unified form for the major no-hidden-variables theorems. *Physical Review Letters* 65 (1990), pp. 3373–3376.
- [148] MÜLLER-LENNERT, M., DUPUIS, F., SZEHR, O., FEHR, S., AND TOMAMICHEL, M. On quantum rényi entropies: A new generalization and some properties. *Journal of Mathematical Physics 54*, 12 (2013), p. 122203.
- [149] NAVASCUES, M., AND WOLFE, E. The inflation technique completely solves the classical inference problem. arxiv:1707.06476 (2017).
- [150] NEAL, R. M. On deducing conditional independence from d-separation in causal graphs with feedback (research note). *Journal of Artificial Intelligence Research 12* (2000), pp. 87–91.
- [151] NIELSEN, M. A., AND CHUANG, I. L. Quantum Computation and Quantum Information: 10th Anniversary Edition. *Cambridge University Press* (2011).
- [152] NURGALIEVA, N., AND DEL RIO, L. Inadequacy of modal logic in quantum settings. In Proceedings QPL 2018, EPTCS 287 (2019), pp. 267–297.
- [153] ORESHKOV, O. Time-delocalized quantum subsystems and operations: on the existence of processes with indefinite causal structure in quantum mechanics. *Quantum 3* (2019), p. 206.
- [154] ORESHKOV, O., COSTA, F., AND BRUKNER, Č. Quantum correlations with no causal order. Nature Communications 3 (2012), p. 1092.

- [155] ORESHKOV, O., AND GIARMATZI, C. Causal and causally separable processes. New Journal of Physics 18, 9 (2016), p. 93020.
- [156] PAUNKOVIĆ, N., AND VOJINOVIĆ, M. Causal orders, quantum circuits and spacetime: distinguishing between definite and superposed causal orders. *Quantum* 4 (2020), p. 275.
- [157] PAWŁOWSKI, M., PATEREK, T., KASZLIKOWSKI, D., SCARANI, V., WINTER, A., AND ZUKOWSKI, M. Information causality as a physical principle. *Nature* 461, 7267 (2009), pp. 1101–1104.
- [158] PEARL, J. On the testability of causal models with latent and instrumental variables. UAI'95: Proceedings of the Eleventh conference on Uncertainty in artificial intelligence (1995).
- [159] PEARL, J. Causality: Models, reasoning, and inference. Second edition, Cambridge University Press (2009).
- [160] PEARL, J., AND DECHTER, R. Identifying independencies in causal graphs with feedback. UAI'96: Proceedings of the Twelfth international conference on Uncertainty in artificial intelligence (2013).
- [161] PERES, A. Incompatible results of quantum measurements. Physics Letters A 151, 3 (1990), pp. 107–108.
- [162] PETZ, D., AND VIROSZTEK, D. Some inequalities for quantum Tsallis entropy related to the strong subadditivity. *Mathematical Inequalities and Applications* 18 (2014), p. 555.
- [163] PIENAAR, J. A time-reversible quantum causal model. arxiv:1902.00129 (2019).
- [164] PIENAAR, J. Quantum causal models via quantum bayesianism. Physical Review A 101 (2020), p. 012104.
- [165] PIENAAR, J., AND BRUKNER, Č. A graph-separation theorem for quantum causal models. New Journal of Physics 17, 7 (2015), p. 073020.
- [166] PIRANDOLA, S., ANDERSEN, U. L., BANCHI, L., BERTA, M., BUNANDAR, D., COL-BECK, R., ENGLUND, D., GEHRING, T., LUPO, C., OTTAVIANI, C., PEREIRA, J., RAZAVI, M., SHAARI, J. S., TOMAMICHEL, M., USENKO, V. C., VALLONE, G., VIL-LORESI, P., AND WALLDEN, P. Advances in quantum cryptography. arxiv:1906.01645 (2019).
- [167] PIRONIO, S. Lifting Bell inequalities. Journal of Mathematical Physics 46 (2005), p. 062112.
- [168] PIRONIO, S., ACÍN, A., MASSAR, S., BOYER DE LA GIRODAY, A., MATSUKEVICH, D., MAUNZ, P., OLMSCHENK, S., HAYES, D., LUO, L., A MANNING, T., AND MONROE, C. Random numbers certified by Bell's theorem. *Nature* 464 (2010), pp. 1021–4.
- [169] PITOWSKI, I. Quantum probability quantum logic. Springer-Verlag Berlin Heidelberg 321 (1989).

- [170] POPESCU, S., AND ROHRLICH, D. Quantum nonlocality as an axiom. Foundations of Physics 24, 3 (1994), pp. 379–385.
- [171] PORTMANN, C., MATT, C., MAURER, U., RENNER, R., AND TACKMANN, B. Causal Boxes: Quantum Information-Processing Systems Closed under Composition. *IEEE Transactions on Information Theory* 63, 5 (2017), pp. 3277–3305.
- [172] PRESKILL, J. Quantum shannon theory. arXiv:1604.07450 (2016).
- [173] PROCOPIO, L. M., MOQANAKI, A., ARAÚJO, M., COSTA, F., ALONSO CALAFELL, I., DOWD, E. G., HAMEL, D. R., ROZEMA, L. A., BRUKNER, Č., AND WALTHER, P. Experimental superposition of orders of quantum gates. *Nature Communications 6* (2015), p. 7913.
- [174] PUSEY, M. F. An inconsistent friend. *Nature Physics* 14, 10 (2018), pp. 977–978.
- [175] PUSEY, M. F., AND LEIFER, M. S. Logical pre- and post-selection paradoxes are proofs of contextuality. In Proceedings QPL 2015, EPTCS 195 (2015), pp. 295–306.
- [176] REICHENBACH, H. The direction of time. *Dover Publications* (1956).
- [177] RENNER, R. Security of quantum key distribution. *PhD Dissertation, ETH Zürich.* arxiv:quant-ph/0512258 (2006).
- [178] RENOU, M.-O., BÄUMER, E., BOREIRI, S., BRUNNER, N., GISIN, N., AND BEIGI,
  S. Genuine quantum nonlocality in the triangle network. *Physical Review Letters 123* (2019), p. 140401.
- [179] RÉNYI, A. On measures of information and entropy. Proceedings of the 4th Berkeley Symposium on Mathematics, Statistics and Probability, University of California Press (1961), pp. 547–561.
- [180] RICHARDSON, T. A discovery algorithm for directed cyclic graphs. Proceedings of the Twelfth International Conference on Uncertainty in Artificial Intelligence (1996), pp. 454–461.
- [181] RIED, K., AGNEW, M., VERMEYDEN, L., JANZING, D., SPEKKENS, R. W., AND RESCH, K. J. A quantum advantage for inferring causal structure. *Nature Physics 11*, 5 (2015), pp. 414–420.
- [182] ROSSET, D., GISIN, N., AND WOLFE, E. Universal bound on the cardinality of local hidden variables in networks. *Quantum Information and Computation 18* (2018), pp. 0910–0926.
- [183] ROWE, N. Why there's so little good evidence that fiscal (or monetary) policy works (online), (2009). https://worthwhile.typepad.com/worthwhile\_canadian\_initi/2009/ 01/why-theres-so-little-good-evidence-that-fiscal-or-monetary-policyworks.html.

- [184] RUBINO, G., ROZEMA, L. A., FEIX, A., ARAÚJO, M., ZEUNER, J. M., PROCOPIO, L. M., BRUKNER, Č., AND WALTHER, P. Experimental verification of an indefinite causal order. *Science Advances* 3, 3 (2017).
- [185] SALAZAR, R., KAMON, M., GOYENECHE, D., HORODECKI, K., SAHA, D., RA-MANATHAN, R., AND HORODECKI, P. A no-go theorem for device-independent security in relativistic causal theories. arxiv:1712.01030 (2020).
- [186] SCHEINES, R. An introduction to causal inference. In McKim and Turner (eds.) (1997), pp. 185–99.
- [187] SCHRIJVER, A. Theory of linear and integer programming. John Wiley and Sons, Inc. (1986).
- [188] SEGAL, I. E. Postulates for general quantum mechanics. Annals of Mathematics 48, 4 (1947), pp. 930–948.
- [189] SHALM, L. K., MEYER-SCOTT, E., CHRISTENSEN, B. G., BIERHORST, P., WAYNE, M. A., STEVENS, M. J., GERRITS, T., GLANCY, S., HAMEL, D. R., ALLMAN, M. S., COAKLEY, K. J., DYER, S. D., HODGE, C., LITA, A. E., VERMA, V. B., LAMBROCCO, C., TORTORICI, E., MIGDALL, A. L., ZHANG, Y., KUMOR, D. R., FARR, W. H., MARSILI, F., SHAW, M. D., STERN, J. A., ABELLÁN, C., AMAYA, W., PRUNERI, V., JENNEWEIN, T., MITCHELL, M. W., KWIAT, P. G., BIENFANG, J. C., MIRIN, R. P., KNILL, E., AND NAM, S. W. Strong Loophole-Free Test of Local Realism. *Physical Review Letters 115*, 25 (2015), p. 250402.
- [190] SHANNON, C. E. A mathematical theory of communication. The Bell System Technical Journal 27, 3 (1948), pp. 379–423.
- [191] SHOR, P. W. Algorithms for quantum computation: discrete logarithms and factoring. Proceedings 35th Annual Symposium on Foundations of Computer Science (1994), pp. 124–134.
- [192] SHORT, A. J., POPESCU, S., AND GISIN, N. Entanglement swapping for generalized nonlocal correlations. *Physical Review A* 73, 1 (2006), p. 012101.
- [193] SKRZYPCZYK, P., BRUNNER, N., AND POPESCU, S. Emergence of quantum correlations from non-locality swapping. *Physical Review Letters* 102 (2009), p. 110402.
- [194] SPEKKENS, R. W. Contextuality for preparations, transformations, and unsharp measurements. *Physical Review A* 71, 5 (2005), p. 052108.
- [195] SPEKKENS, R. W. Evidence for the epistemic view of quantum states: A toy theory. *Physical Review A* 75 (2007), p. 032110.
- [196] SUDBERY, A. Single-world theory of the extended wigner's friend experiment. Foundations of Physics 47, 5 (2017), pp. 658–669.
- [197] SUDBERY, A. The hidden assumptions of Frauchiger and Renner. arxiv:1905.13248 (2019).

- [198] SVETLICHNY, G. Time Travel: Deutsch vs. Teleportation. International Journal of Theoretical Physics 50, 12 (2011), pp. 3903–3914.
- [199] TOMAMICHEL, M., COLBECK, R., AND RENNER, R. A fully quantum asymptotic equipartition property. *IEEE Transactions on Information Theory* 55, 12 (2009), pp. 5840–5847.
- [200] TSALLIS, C. Possible generalization of Boltzmann-Gibbs statistics. Journal of Statistical Physics 52, 1 (1988), pp. 479–487.
- [201] TSIRELSON, B. Some results and problems on quantum Bell-type inequalities. *Hadronic Journal Supplement* (1993), pp. 329–345.
- [202] TSIRELSON, B. Some results and problems on quantum Bell-type inequalities. *Hadronic Journal Supplement 8* (1993), pp. 329–345.
- [203] VAN HIMBEECK, T., BOHR BRASK, J., PIRONIO, S., RAMANATHAN, R., SAINZ, A. B., AND WOLFE, E. Quantum violations in the Instrumental scenario and their relations to the Bell scenario. *Quantum 3* (2019), p. 186.
- [204] VAN STOCKUM, W. J. The gravitational feild of a distribution of particles rotating about an axis of symmetry. *Proceedings of the Royal Society of Edinburgh* 57 (1937), pp. 135–154.
- [205] VERMA, T., AND PEARL, J. Causal networks: Semantics and expressiveness. Proceedings of the Fourth Annual Conference on Uncertainty in Artificial Intelligence (UAI '88) (1990), pp. 69–78.
- [206] VILASINI, V., AND COLBECK, R. Analyzing causal structures using Tsallis entropies. *Physical Review A 100* (2019), p. 062108.
- [207] VILASINI, V., AND COLBECK, R. Limitations of entropic inequalities for detecting nonclassicality in the postselected Bell causal structure. *Physical Review Research 2* (2020), p. 033096.
- [208] VILASINI, V., NURGALIEVA, N., AND DEL RIO, L. Multi-agent paradoxes beyond quantum theory. New Journal of Physics 21, 11 (2019), p. 113028.
- [209] VILASINI, V., PORTMANN, C., AND DEL RIO, L. Composable security in relativistic quantum cryptography. New Journal of Physics 21, 4 (2019), p. 043057.
- [210] VITTER, J. S., LARMORE, L., LEIGHTON, T., AND RAZ, R. Exponential separation of quantum and classical communication complexity. *Proceedings of the Thirty-First Annual* ACM Symposium on Theory of Computing (1999), pp. 358–367.
- [211] VON NEUMANN, J. Mathematical foundations of quantum mechanics. *Princeton uni*versity press (1955).
- [212] VON NEUMANN, J., WHEELER, N. A., AND BEYER, R. T. Mathematical foundations of quantum mechanics. *Princeton University Press* (1932).

- [213] WAJS, M., KURZYNSKI, P., AND KASZLIKOWSKI, D. Information-theoretic Bell inequalities based on Tsallis entropy. *Physical Review A 91* (2015), p. 012114.
- [214] WEILENMANN, M., AND COLBECK, R. Inability of the entropy vector method to certify nonclassicality in linelike causal structures. *Physical Review A* 94 (2016), p. 042112.
- [215] WEILENMANN, M., AND COLBECK, R. Analysing causal structures with entropy. Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences 473, 2207 (2017).
- [216] WEILENMANN, M., AND COLBECK, R. Non-Shannon inequalities in the entropy vector approach to causal structures. *Quantum 2* (2018), p. 57.
- [217] WEILENMANN, M., AND COLBECK, R. Analysing causal structures in generalised probabilistic theories. *Quantum* 4 (2020), p. 236.
- [218] WEILENMANN, M., AND COLBECK, R. Self-testing of physical theories, or, is quantum theory optimal with respect to some information-processing task? *Physical Review Letters* 125 (2020), p. 060406.
- [219] WHEELER, J. A. Assessment of Everett's "relative state" formulation of quantum theory. *Reviews of Modern Physics 29* (1957), pp. 463–465.
- [220] WIGNER, E. P. Remarks on the mind-body question. The Scientist Speculates, Heineman (1961), Ed. I. J. Good.
- [221] WILLIAMS, H. P. Fourier's method of linear programming and its dual. The American Mathematical Monthly 93, 9 (1986), pp. 681–695.
- [222] WOLFE, E., SCHMID, D., SAINZ, A. B., KUNJWAL, R., AND SPEKKENS, R. W. Quantifying Bell: the resource theory of nonclassicality of common-cause boxes. *Quantum* 4 (2020), p. 280.
- [223] WOLFE, E., SPEKKENS, R. W., AND FRITZ, T. The inflation technique for causal inference with latent variables. *Journal of Causal Inference* 7 (2019).
- [224] WOOD, C. J., AND SPEKKENS, R. W. The lesson of causal discovery algorithms for quantum correlations: causal explanations of Bell-inequality violations require finetuning. New Journal of Physics 17, 3 (2015), p. 33002.
- [225] YEUNG, R. W. A framework for linear information inequalities. IEEE Transactions on Information Theory 43, 6 (1997), pp. 1924–1934.
- [226] ZHALAMA, ZHANG, J., AND MAYER, W. Weakening faithfulness: some heuristic causal discovery algorithms. *International Journal of Data Science and Analytics* 3, 2 (2017), pp. 93–104.
- [227] ZHANG, Z., AND YEUNG, R. W. A non-Shannon-type conditional inequality of information quantities. *IEEE Trans. Information Theory* 43 (1997), pp. 1982–1986.

- [228] ZUKOWSKI, M., KASZLIKOWSKI, D., BATURO, A., AND LARSSON, J.-Å. Strengthening the Bell theorem: conditions to falsify local realism in an experiment. arXiv:quantph/9910058 (1999).
- [229] ZUKOWSKI, M., ZEILINGER, A., HORNE, M. A., AND EKERT, A. K. "Event-readydetectors" Bell experiment via entanglement swapping. *Physical Review Letters* 71 (1993), pp. 4287–4290.
- [230] ZYCH, M., COSTA, F., PIKOVSKI, I., AND BRUKNER, Č. Bell's theorem for temporal order. *Nature Communications 10*, 1 (2019), p. 3772.